

ICES/PICES/PAME WORKING GROUP ON INTEGRATED ECOSYSTEM ASSESSMENT (IEA) FOR THE CENTRAL ARCTIC OCEAN (WGICA)

VOLUME 2 | ISSUE 79

ICES SCIENTIFIC REPORTS

RAPPORTS SCIENTIFIQUES DU CIEM



ICES INTERNATIONAL COUNCIL FOR THE EXPLORATION OF THE SEA CIEM CONSEIL INTERNATIONAL POUR L'EXPLORATION DE LA MER

International Council for the Exploration of the Sea Conseil International pour l'Exploration de la Mer

H.C. Andersens Boulevard 44-46 DK-1553 Copenhagen V Denmark Telephone (+45) 33 38 67 00 Telefax (+45) 33 93 42 15 www.ices.dk info@ices.dk

The material in this report may be reused for non-commercial purposes using the recommended citation. ICES may only grant usage rights of information, data, images, graphs, etc. of which it has ownership. For other third-party material cited in this report, you must contact the original copyright holder for permission. For citation of datasets or use of data to be included in other databases, please refer to the latest ICES data policy on ICES website. All extracts must be acknowledged. For other reproduction requests please contact the General Secretary.

This document is the product of an expert group under the auspices of the International Council for the Exploration of the Sea and does not necessarily represent the view of the Council.

ISSN number: 2618-1371 I © 2020 International Council for the Exploration of the Sea

ICES Scientific Reports

Volume 2 | Issue 79

ICES/PICES/PAME WORKING GROUP ON INTEGRATED ECOSYSTEM AS-SESSMENT (IEA) FOR THE CENTRAL ARCTIC OCEAN (WGICA)

Recommended format for purpose of citation:

ICES. 2020. ICES/PICES/PAME Working Group on Integrated Ecosystem Assessment (IEA) for the Central Arctic Ocean (WGICA). ICES Scientific Reports. 2:79. 144 pp. http://doi.org/10.17895/ices.pub.7454

Editors

Lis Lindal Jørgensen• Sei-ichi Saitoh

Authors

John Bengtson • Bodil Bluhm • Tom Christensen • Karen Edelvang • Hauke Flores • Anne Kirstine Frie Yasushi Fukamachi • Maria Gavrilo • Harald Gjøsæter • Jacqueline Grebmeier • Grøsvik, Bjørn Einar Naomi Harada • Kevin Hedges • Martine van den Heuvel-Greve • Taka Hirata • Toru Hirawake • Alf Håkon Hoel • Randi Ingvaldsen • Vladimir Ivanov • Lis Lindal Jørgensen • Kathy Kuletz • Jan René Larsen • Pauline Snoejis Leonmalm • Igor Melnikov • Anders Mosbech • Ingeborg Mulder • Barbara Niehoff • Shigeto Nishino • Fugio Ohnishi • Natsuhiko Otsuka • Thomas Van Pelt • Cecilie von Quillfeldt • Sei-ichi Saitoh • Hyoung Chul Shin • Hein Rune Skjoldal • Lisa Speer • Senjie Yang



Contents

i	Executive summaryiv		iv		
ii	Expert group information				
iii	Summ	Summary of the main conclusions			
1	Introduction				
	1.1	Terms of Reference	10		
	1.2	Summary of work plan	10		
	1.3	Summary of achievements of the WG during 3-year term	11		
2	Status	for WGICA IEA report in the CAO (CRR) (Report 1)	12		
3	ToR b. Review and report on on-going and recent changes and events in the CAO				
	ecosystem associated with changes such as in sea-ice, oceanographic circulation, and				
	hydrog	graphic properties	13		
	3.1	Introduction	13		
	3.2	Changes in Arctic Sea ice	13		
	3.2.1	Central Arctic Ocean	14		
	3.2.2	Arctic seas	15		
	3.3	Changes in the gateways	16		
	3.3.1	Atlantic gateways (Barents Sea, Fram Strait and Western Nansen Basin)	16		
	3.3.2	Pacific gateway (Bering and Chukchi seas)	17		
	3.4	Changes in CAO hydrographic properties	18		
	3.4.1	Upper ocean layer	18		
	3.4.2	Halocline layer	19		
	3.4.3	Intermediate and deep layers	20		
	3.5	Summary	20		
	3.6	Possible ecosystem implications	22		
	3.7	References	22		
4	ToR c.	Continue to examine effects of climate change on the CAO ecosystem	29		
	4.1	Arctic microalgae	29		
	4.1.1	General introduction	29		
	4.1.2	Possible effects of climate change	29		
	Refere	nces	32		
	4.2	Invertebrate ice fauna	35		
	4.2.1	Sea ice meiofauna	35		
	4.2.2	Sea ice macrofauna	37		
	4.2.3	Needs for research and monitoring	38		
	Refere	nces	40		
	4.3	Zooplankton in the Central Arctic Ocean, a linchpin to keep the system in place	44		
	Refere	nces	47		
	4.4	Benthos in the CAO	48		
	4.4.1	The seabed	48		
	4.4.2	Bottom temperatures of the Central Arctic Ocean	49		
	4.4.3	Benthos	50		
	4.4.4	Climate impact on benthos	51		
	4.4.5	Possible human activities impacting benthos	51		
	Refere	nces	53		
	4.5	Fish in the Central Arctic Ocean having to cope with unprecedented warming	57		
	References				
	4.6	Marine mammals in the CAO	62		
	4.6.1	Ecological Effects of Climate Change	62		
	4.6.2	Effects of anthropogenic noise	73		
	4.6.3	Knowledge gaps and monitoring needs for marine mammals	74		

	Refere	nces	76
	4.7	Marine birds in the CAO	88
	4.7.1	Recent information on marine birds in the CAO	88
	4.7.2	Climate and anthropogenic impacts on marine birds in the future CAO	89
	4.7.3	Marine bird research, management, and monitoring resources in adjacent seas	89
	Refere	nces	90
5	ToR d.	Assess the consequences of recent and on-going climatic and oceanographic	
	change	s on transport pathways and potential effects of contaminants	91
	5.1	Introduction	91
	5.2	Pollution pressures in the present-day CAO	91
	5.2.1	Contaminants	91
	5.2.2	Chemical pollution	91
	5.2.3	Plastic pollution	92
	5.3	Human activities in the CAO related to pollution: present and future	92
	5.3.1	Ship traffic	93
	5.3.2	Fisheries in the CAO	94
	5.3.3	Oil and gas activities and related shipping	95
	5.3.4	Offshore mining	95
	5.3.5	Tourism	96
	Refere	nces	96
6	ToR e.	Review and report on new studies on fishes in the CAO ecosystem	102
-	6.1	Introduction	102
	6.2	An analysis of literature published in the past three years (2017-2020)	103
	6.2.1	Shelf area	103
	622	Gateways	105
	6.2.3	CAO	105
	624	Experimental studies	106
	625	Policy	106
	626	Research tonics	106
	Refere	nces	107
7		Continue to identify priority research needs and monitor how identifies	
,	knowle	dge gans are being addressed and filled	116
	7 1	Introduction	116
	7.2	Examples of Case studies form ongoing research manning and monitoring in	
	7.2	the CAO	116
	721	Distributed Biological Observatory (DBO)	116
	722	Pacific Arctic Climate Ecosystem Observatory (PACEO)	117
	722	Ioint Ocean Ice Studies (IOIS)	118
	724	Canadian Beaufort Sea Marine Ecosystem Assessment (CRS-MEA)	119
	725	Canadian Beaufort Sea Marine Mammal research	120
	7.2.5	Nansen Legary	120
	7.2.0	Multidiciplinary Drifting Observatory for the Study of Arctic Change (MOSAiC)	121
	1.2.1	Expedition	122
	728	Expedition	122
	7.2.0	Synontic Arctic Survey (SAS)	172
	7.2.5	North Dala 2020 avaidition	123
	7.2.10	North Pole 2020 expedition	123
	7.3 7.3 1	New tools, methods, and sampling ged for studies of the CAU	124
	7.3.1 7.2.2	The SUIT in the Arctic Ocean - Decent compliant of poles and under set in the	124
	1.3.2	The soft in the Arctic Ocean – Recent sampling of polar cod under sea ice in the	174
	7 2 2	Chukchi-Beautort region	124
	/.3.3	The SULL In the Arctic Ocean – Results from the Go-West project	124
	7.3.4	A new, non-invasive tool, environmental DNA	126
	/.4	Priority research and monitoring needs per ToR	127

	7.4.1	(ToR b): On-going and recent changes and events in the CAO	127
	7.4.2	(ToR c): Continue to examine effects of climate change on the ecosystem and	
		assess projected future changes	127
	7.4.3	(ToR d): Assess the consequences of climatic and oceanographic changes on	
		transport pathways and potential effects of contaminants	131
	7.4.4	(ToR e): Review and report on new studies on fish as well as other biological	
		components	131
	7.5	Summary and Conclusions	132
	Refere	References	
8	ToR g.	ToR g. Prepare an Ecosystem Overview for the CAO ecosystem	
	8.1	Introduction	134
	8.2	Pressure assessment	135
	8.3	Linking main pressures with human activities and ecosystem components	135
	8.3.1	Linking main pressures with human activities	136
	8.3.2	Linking main pressures with ecosystem components	136
9	Outline	e of Report 2	137
Annex 1:		List of participants	139
Annex 2:		Resolutions	141

I

i Executive summary

The Working Group on the Integrated Assessment of the Central Arctic Ocean (WGICA) aims to provide a holistic analysis of the present and future status of the ecosystem and human activities therein. Data collection in the Central Arctic Ocean (CAO) has been inconsistent both spatially and temporally, which can limit the ability to conduct comprehensive analyses of trends and warning signals. However, coverage of data collection has been improving over the past few years. WGICA collates and analyses this regional information, which will be used to help guide the production of an Ecosystem Overview (EO) that relates the main regional pressures in the CAO with the human activities and the ecosystem components that are most impacted by these pressures.

Climate change reduces sea ice, increases light penetration, causes regionally variable trends in stratification and mixing of the water column, increases inflow in both the Atlantic and Pacific sectors, and heating of waters at the surface and extending deeper and deeper. These changes in turn affect primary production and cascade through the food web to ice-associated fauna, zoo-plankton, fish, benthos, and sea mammals.

These changes may be exacerbated by increasing human activities in and around the CAO, including increasing pollution from ship traffic and from the transport of contaminants to the ecoregion by rivers and ocean currents. The number of ships and distances traveled are increasing and it is anticipated that both commercial and tourist traffic by sea and air will continue to rise. The CAO has become an important sink for many pollutants such as microplastics, which have been found in sea ice and wildlife. Current and future threats to the ecoregion from these activities also include increased risk of oil spills and biodiversity loss if ocean mining expands into the Arctic.

While an agreement has been made to ban commercial fishing in the high seas of the Central Arctic Ocean; fish populations continue to be impacted by the effects of a warming ocean, retreating ice-cover and acidification. These threats have important ecological and policy implications for the entire food web and the Arctic community. For example, negative impacts on the polar cod population will negatively impact ringed seals and beluga whales and therefore will also affect subsistence harvests in the future.

In upcoming years, WGICA plans to further define and describe human activities and resulting pressures and related management organizations, and develop a climate and vulnerability assessment of the CAO.

ii Expert group information

Expert group name	ICES/PICES/PAME Working Group on Integrated Ecosystem Assessment (IEA) for the Central Arctic Ocean (WGICA)
Expert group cycle	Multiannual fixed-term
Year cycle started	2019
Reporting year in cycle	2/3
Chair(s)	Lis Lindal Jørgensen, Norway
	Sei-ichi Saitoh, Japan
Meeting venue(s) and dates	9-10 May 2019, Sapporo, Japan(26 participants)
	27-19 April, 2020, online meeting (36 participants + 5 ICES Secretariat and observers)

I

iii Summary of the main conclusions

Oceanography:

Summer sea ice extent in the past decade or so has remained fairly stable at -22 ± 8 per cent relative to the mean climatic norm for the available range of satellite observations (1979-2019). Ice thickness declined across the central Arctic by 65%, from 3.59 to 1.25 m, between 1975 and 2012.

The Atlantic gateways to the Arctic Ocean are currently is experiencing greater inflows, manifested in a warmer ocean and atmosphere, northward and eastwards spread of Atlantic Water in the Barents Sea and the Nansen Basin, and reduced stratification and increased mixing in the upper ocean in the Eurasian Basin.

The Bering Sea recorded unprecedented high sea surface temperatures in 2014 and the warm condition has persisted into 2019. The Canadian Basin showed strong freshening and a deepening of the nutricline and deep chlorophyll maximum.

A more energetic state of the intermediate water layer in the CAO is projected in future years. This new state will be presumably supported by stronger currents and shear, leading to increased turbulent mixing and larger upward oceanic heat fluxes.

Human activities:

The CAO and adjoining waters remain relatively unpolluted, but the Arctic is a sink for pollutants transported from lower latitudes, and pollution from local sources is increasing. Emissions of chemical compounds (e.g. mercury, POPs) from outside the CAO are currently the main source of air pollution. Other pollutants, including flame retardants, pesticides, and phthalates, are an emerging concern. Macro-, micro- and nanoplastics, transported by rivers and ocean currents, have been found in sea ice and wildlife. Sea ice is an important sink for microplastics. The number of ships and the distances traveled are increasing in the Arctic; currently up to 45% of traffic is associated with fishing vessels, but larger vessel traffic is increasing. We also consider future activities that may impact the CAO as sea-ice cover decreases. For example, oil spills from activities on the continental shelves may affect the CAO ecosystem. Ocean mining may expand into the Arctic, resulting in biodiversity loss even as most species in the CAO remain undiscovered or unidentified. An agreement has been made to ban commercial fishing in the high seas of the Central Arctic Ocean; still, the problem of Abandoned, Lost or otherwise Discarded Fishing Gear (ALDFG) is a factor to consider in the CAO area even if commercial fishing is not allowed. Tourism is generally associated with ships or, in smaller volumes, as flights to the North Pole.

Ecosystem components:

Microalgae, Sea Ice Biota and Zooplankton: Changes in the amount, type, timing and location of sea ice in the Arctic, along with related changes in light availability, temperature, salinity and nutrient concentrations in surface layers, are affecting the timing and abundance of primary production and the biomass and species composition of ice biota and the major zooplankton species, with likely important consequences for food web functioning.

Benthos: Diverse deep-sea habitats, including undersea mountains, ridges, glacial deposits and other features support increased benthic biodiversity, but little is known about changes to these and other deep-sea communities. On the shelves, there is evidence of declining benthic biomass in the northern Bering Sea and the southern Chukchi Sea along with a northward shift in dominant macrofaunal biomass, which has also been observed in the European Arctic. The long lifetimes, slow growth rates and low fecundity of deep-sea organisms make them more vulnerable to human activities such as mining, oil exploitation, bottom fisheries, climate change as well as light, noise, plastic and chemical pollution.

Fish: Diminishing sea ice, earlier melt, higher ocean temperatures and resulting changes in salinity, nutrient availability and prey availability are affecting the biomass, abundance and distribution of different fish populations, including that of polar cod (*Boreogadus saida*), a keystone Arctic species. Increased human presence in and near the CAO is exposing fish populations to plastic and other pollutants and increasing ocean noise, which has been shown to affect polar cod (*B. saida*) behavior. Should commercial fishing, mining or oil development commence in the CAO, a variety of impacts on fish can be expected.

Marine mammals: Based on current knowledge, the CAO appears to be relatively scarcely populated by marine mammals; with continued ice retreat, however the importance of the CAO may increase over time, especially for ice dependent pelagic-feeding or generalist species like ringed, ribbon, harp and hooded seals, belugas,(walrus, polar bear,) narwhals and bowhead whales, some of which are already experiencing population declines. Some polar bears currently also use sea ice in the CAO as a summer hunting habitat but increasing distance to denning habitats on land may compromise this strategy in the future. Many marine mammals rely on acoustics for key life functions; increasing human presence in the region will bring increasing noise from seismic airguns, ship engines, military operations, fisheries, research sonars and possible mining, which can interfere with vital behaviors.

Seabirds: Large numbers of breeding, non-breeding, and migratory seabird individuals use open water habitats in marginal shelf waters of the Central Artic region during summer and fall, foraging on zooplankton, invertebrates and forage fish. However, few seabirds occupy the mostly ice-covered CAO region itself. Predicted ice-free summers in the CAO are likely to impact seabird populations, in part through impacts on prey species (e.g. *B. saida*). Other impacts of diminished sea ice range from smaller scales, e.g. reduction in very ice-dependent species like ivory gulls and Ross's gulls, to possible extremely large-scale changes in longstanding migration and distribution patterns of northern hemisphere marine birds. Also, post-breeding and migrating marine birds in newly ice-free CAO waters would overlap with increased vessel traffic (birdship collisions), and pollution or plastic debris can also be detrimental to marine birds.

Review and report on new studies on fish in the CAO ecosystem:

Research on marine fishes in the CAO and adjacent waters published during 2017-2020 is summarized, building on earlier summaries by other groups (e.g. related to the Agreement to Prevent Unregulated High Seas Fisheries in the Central Arctic Ocean). Polar cod (*Boreogadus saida*), a keystone species in Arctic food webs, continues to be research focus, with publications detailing more about the species ecology, distribution, genetics, links to other species and impacts from climate change. Other research has examined fish community structure and anticipated impacts of climate change (e.g. potential for species to expand northward into the CAO). Policy papers have also developed recommendations regarding commercial fishery development. L

Priority research needs and monitoring:

Dato collection of physics, contaminants, primary and secondary producers and sea ice biota must be standardized and also obtained i.e. during winter. Mapping and baseline studies were identified as priorities for benthos, sea mammals and fishes including the development/use of new technology. Marine mammals need studies on life history, health and ecology.

Regarding possible future fisheries development of new management tools can safeguard sustainable development of ecosystems and human stakeholders in the face of a rapidly changing environment. Studies of arctic mammal sensitivity to low frequency anthropogenic noise, were identified as priority research.

For monitoring are a coordinated array of regional transect lines recommended with key parameters to evaluate the regional to pan-Arctic scales. Such international, coordinated observation can help identify focal areas for monitoring on an interannual and decadal basis. This involves also CAFF and its Circumpolar Seabird Expert Group (Cbird), the IUCN Polar Bear Specialist Group, and other relevant working groups.

Ecosystem overviews:

Development of an Ecosystem Overview commenced in 2020 with the identification and prioritization of ecosystem pressures. Links between the five main pressures (sea-ice loss, non-indigenous species, contaminants, marine litter and noise), human activities and ecosystem components were initially discussed; these links will be further examined by WGICA during 2020-2021.

1 Introduction

The international governance framework for the Central Arctic Ocean

The central Arctic Ocean (CAO) – as every ocean in the world – is governed by Law of the Sea Convention (UNCLOS). UNCLOS (in force 1994) is the "Constitution of the oceans", providing the rules for how countries are to establish maritime zones, principles for the management of natural resources, conduct marine science and protection of the marine environment.

Coastal states are entitled to establish 200 nautical mile exclusive economic zones (EEZs) where they have sovereign rights to the natural resources. In the high seas beyond national jurisdiction, flag states are responsible for the vessels flying their flag, ensuring that they comply with international obligations in UNCLOS and other international treaties. Also, coastal states can submit claims to the continental shelf extending beyond the 200 nautical miles limit. UNCLOS therefore has implications for managing all human activities in the oceans such as shipping, tourism, seabed mining, oil and gas activities, marine scientific research, and fisheries.

UNCLOS has 169 parties and is ratified by four of the five CAO coastal states. USA is not party to the Convention but considers it customary international law.

UNCLOS provides rules for the management of living marine resources which are to be managed sustainably and utilized in an optimal manner. These provisions are reinforced by the 1995 UN Fish Stocks Agreement which is an implementing agreement to the Convention. With respect to environmental protection UNCLOS stipulates a general obligation to protect the environment. A number of international treaties, such as the Convention on Biodiversity, pertain to the marine environment, and currently an implementing agreement for the conservation and use of biodiversity in areas beyond national jurisdiction is under negotiation ("BBNJ"). As regards marine scientific research, the main thrust of UNCLOS is that scientific activity is to be promoted.

There are a number of scientific initiatives and bodies in the Arctic. Under the auspices of the Arctic Council an agreement on international scientific cooperation (2016) has been negotiated. Additionally, the working groups in the Arctic Council produce scientific assessments. Since 2016, Arctic Science Ministerial addresses issues of concern in the Arctic. The International Arctic Science Committee (IASC) is a bottom-up scientific cooperation involving scientists from more than 20 countries. In the North Atlantic part of the Arctic, the International Council for the Exploration of the Sea plays an important role, providing scientific advice for the management of fisheries and the marine environment in that region.

The CAO governance regime has been strengthened considerably over the last decade through numerous international agreements. In addition, there is many international agreements besides those mentioned above that pertain to the CAO (Arctic Ocean Review 2011). As regards science in particular, the International Arctic Science Committee represents the scientific community, whereas the Arctic Council science cooperation agreement, the recent Arctic science ministerial, as well as the International Council for the Exploration of the Sea (ICES) and the North Pacific Marine Science Organization (PICES) have important roles in these areas.

Arctic Ocean Review 2011: <u>https://www.pame.is/index.php/document-library/shipping-docu-ments/arctic-ocean-review-documents/347-aor-phase-i-report-to-ministers-2011-nov-2013/file</u>

L

WGICA

The International Council for the Exploration of the Sea (ICES) is committed to building a scientific foundation for integrated ecosystem understanding. Part of this integrated approach is ecosystem-based management (EBM). EBM requires consideration of the impacts of different marine sectors on marine ecosystems and trade-offs between such activities, in the context of climate change and other forms of environmental change.

The Arctic Council (AC) also promotes the use of EBM. A core component of EBM is Integrated Ecosystem Assessment (IEA), the topic of this report.

The joint ICES/PICES/PAME Working Group for Integrated Ecosystem Assessment of the Central Arctic Ocean – WGICA is undertaking an IEA for the Central Arctic Ocean (CAO), integrating assessment of the changing ecosystem with current and future human activities.

The WGICA was established in 2015 to 1) provide a holistic and integrated view of the status, trends and pressures of key components of the (CAO) ecosystem, and 2) to contribute to the implementation of EBM in the CAO.

The CAO (see black lines in figure 1) is defined by the slopes down to the deep Eurasian and Amerasian basins. The fluxes and properties of water through the Atlantic and Pacific gateways, connecting the Arctic to other oceans, are part of the physical and biological variability of the basins of the CAO.



Figure 1. The study area of WGICA is the Central Arctic Ocean (CAO) Large Marine Ecosystem (LME), bounded in red in the center of the map, which includes areas under national jurisdiction (EEZs boundaries are in green) as well as the High Seas beyond national jurisdiction (source: ICES WKICA REPORT 2015 ACOM/SCICOM Steering Group On Integrated Ecosystem Assessments, ICES CM 2015/SSGIEA:11, REF. SCICOM & ACOM).

The WGICA CAO area includes international waters, the High Seas beyond national jurisdiction (see red lines in figure 1,) as well as areas within the exclusive economic zones (EEZs) of coastal states.

Infobox

What is the ARCTIC COUNCIL (AC)? https://arctic-council.org/en/about/

The Arctic Council is the leading intergovernmental forum promoting cooperation, coordination and interaction among the Arctic States, Arctic indigenous communities and other Arctic inhabitants on common Arctic issues, in particular on issues of sustainable development and environmental protection in the Arctic.

The Arctic Council does not and cannot implement or enforce its guidelines, assessments or recommendations. That responsibility belongs to individual Arctic States or international bodies.

<u>The Arctic states</u>: Canada, The Kingdom of Denmark, Finland, Iceland, Norway, The Russian Federation, Sweden and The united States are the members of the AC, while <u>the permanent participants</u> are the Aleut International Association (AIA), the Arctic Athabaskan Council (AAC), the Gwich'in Council International (GCI), the Inuit Circumpolar Council (ICC), the Russian Association of Indigenous Peoples of the North (RAIPON) and the Saami Council.

<u>The working groups</u> in the AC are: Protection of the Arctic Marine Environment (**PAME**), Arctic Monitoring and Assessment Programme (**AMAP**), Conservation of Arctic Flora and Fauna (**CAFF**), Arctic Contaminants Action Program (**ACAP**), the Emergency Prevention, Preparedness and Response (**EPPR**), and the Sustainable Development Working Group (**SDWG**).

Conduct of the meeting in 2020

The 5th annual meeting of the WGICA was planned as a physical meeting at the ICES HQ in Copenhagen, Denmark, but was conducted as a Webex meeting 27-29 April 2020 due to the COVID- 19 travel restrictions.

This meant that the Regional Arctic Ocean Decade Workshop, planned back to back with the WGICA meeting in Copenhagen, was canceled and postponed to an online meeting in October 2020 (<u>http://www.oceandecade.dk/</u>).

The meeting focused on six of the seven action items in the 2018 Terms of Reference for the WGICA, and on finishing Report 1 and starting the work on an Ecosystem Overview (EO) and Report 2.

Before the meeting, one Atlantic and one Pacific lead for each of the action items was tasked to reach out to experts in the WGICA group.

The meeting was built up as two duplicate meetings per 24 hours, with 12 hours in between, allowing participants from around the world to participate.

Norway, Denmark, Finland, Netherlands, USA, Japan, Korea and China, Germany, Russia, Canada, and Sweden are WGICA members and 36 persons participated in the meeting in addition to four participants from the ICES secretariat and one observer.


Top photo: the Asian Pacific meeting, Bottom photo: the European and North American meeting

The first two days were dedicated to presentations of the work to address the annual ToRs (see appendix 2), while the final day was used for each of the ToR groups to define the work ahead and discuss the content of the ToR.

During the meeting it was clarified that in 2021 new ToRs must be defined for the next WGICA term (2022-2024). This work includes the writing of the Ecosystem Overview and WGICA Report 2.

WGICA co-lead, Senior Scientist Lis Lindal Jørgensen (IMR), presented the 2020-2024 workplan for the Arctic Monitoring and Assessment Programme (AMAP) Heads of Delegation meeting 18 June 2020. There is interest in co-operation between WGICA and the AMAP/CAFF for the upcoming CAO Climate- and Vulnerability Assessments. The format of such cooperation remains to be decided on.

Information given as "notes" during the webex meeting:

AMAP will publish "Climate Issues of concern report 2021" in 2021. Draft table of contents:

- 1) Recent developments in major climate indicators
- 2) Extreme events and thresholds in the Arctic
- 3) Future of The Arctic Seen from CMIP6
- 4) Arctic/Mid-latitude Weather Connectivity
- 5) Climate change impacts on Arctic ecosystems and ecosystem feedbacks
- 6) Trends, effects and societal implications of Arctic climate change.

Point 5) above is intended to serve as a basis for a planned AMAP/CAFF assessment of climate change impacts on Arctic ecosystems and ecosystem feedbacks, as part of the 2021-2023 AMAP Work Plan. The scope of this work is tentatively planned also to cover the marine environment

IPCC will release the 6th assessment report in 2021; including both the physical basis and impacts, adaptation and vulnerability. There will be a separate chapter on Polar Oceans

The State of the Arctic Ocean report has been released April 2020: <u>https://www.dfo-mpo.gc.ca/oceans/soto-rceo/arctic-arctique/index-eng.html</u>

"Plastic in a bottle" project is planning five launching sites, and one site is planned from the Eastern coast of U.S.A. And provided report in PAME shows an idea of launching from the Arctic coast of Alaska. The capsule launched in Iceland recently stranded in the northwest of Scotland: https://www.pame.is/projects/arctic-marine-pollution/plastic-in-a-bottle-live-map

ICES Working Group on Shipping Impacts in the Marine Environment (EGSHIP) <u>https://www.ices.dk/community/groups/Pages/WGSHIP.aspx</u>

WGICA work, deliveries and meetings in 2021-24:

WGICA shall produce and deliver annual reports as well as three final reports before year 2024:

• Annual Reports (Interim Reports)

The annual report of this year is *this* report. New reports will follow each year answering the ToRs and give the last updates on scientific work from the CAO

• <u>Final Reports</u>

Three final reports are planned.

• <u>Report 1 - Ecosystem description of the CAO</u>

This report (approximately 350 pages) is in production and details are given in chapter 2 of this annual report.

• <u>Report 2 – part I: Human activities, pressures and management bodies; part 2: climate-and vulnerability assessment of the Central Arctic Ocean (CAO)</u>

This report will build on Report 1 and will start in winter 2020 with a deadline October 2024.

• Ecosystem Overview (EO) – CAO

Ecosystem overviews provide a description of the ecosystems, identify the main human pressures, and explain how these affect key ecosystem components. Such overviews are approximately 15 pages and updated each third year. We started the EO for the CAO in June 2020 and a first draft will be delivered in November. See Chapter 8 of this annual report for more details.

Planned meetings:

Web-based meeting in subgroups scattered throughout the year, arranged by ToR/sub-groups leaders

2021: Webex-based meeting April early May 2021 to discuss assessment models to be built for data-poor areas, with the Ecosystem Overview (EO) as a first step to develop more methods.

2021: to write part 1 (Human activities and management bodies) by following the structure of the EO

2021: In fall 2021, to have a physical meeting in ICES HQ to discuss the way forward to:

- Finalize the Part 1
- Draft the ToRs for 2022-2024

L

L

- Develop Ecological Overview
- Writing of Report 2 part 2.

2022-2024: to write part 2 (Climate- and Vulnerability- Assessment).

Acknowledgement: Thank you to Thomas Van Pelt, Kevin Hedges, Henry P. Huntington and Lisa Speer to language edit this report.

1.1 Terms of Reference

ToR a: Review and consider approaches and methodologies for conducting an IEA of the CAO ecosystem (year 1 - 2019).

ToR b: Review and report on ongoing and recent changes and events in the CAO ecosystem associated with changes such as in sea ice, oceanographic circulation, and hydrographic properties.

ToR c: Continue to examine effects of climate change on the CAO ecosystem by compiling and reviewing information on changes in response to the ongoing 'Great melt', and assess likely consequences to the CAO ecosystem of projected future changes associated with further loss of sea ice and other climate-related changes (i.e. a climate impact assessment).

ToR d: Assess the consequences of recent and ongoing climatic and oceanographic changes on transport pathways (physical and biological) and potential effects of contaminants in the CAO ecosystem.

ToR e: Review and report on new studies on fish as well as other biological components of the CAO ecosystem ("other biological components" not included here, but in ToR c)

ToR f: Continue to identify priority research needs and monitor how identified knowledge gaps (needed to improve IEA and management effectiveness) are being addressed and filled.

ToR g: Prepare an Ecosystem Overview (EO) for the CAO ecosystem as a draft version in interim report in 2020, and final version in 2021 (draft version of EO will be available October-November 2020).

1.2 Summary of work plan

Year 1 2019	Review IEA methodologies for IEA of the CAO. Review and report new information and changes in the CAO ecosystem.
Year 2 2020	Review and report new information and changes in the CAO ecosystem. Address pathways and effects of contaminants, make an initial list of research needs, and prepare draft Ecosystem Overview.
Year 3 2021	Prepare a second version IEA eport for the CAO with information on status and trends, including impacts of climate change, pollution, and other relevant human pressures. Report on research needs and prepare final draft of Ecosystem Overview.

1.3 Summary of achievements of the WG during 3-year term

For 2019:

- The CAO differs from other LMEs in having few biological and oceanographic time series available to describe status and changes of the ecosystem.
- Methodological development: The 'scientific review' method will continue to be a main avenue for compiling information and assessing the changing state of the CAO ecosystem
- Datasets: Satellite-based monitoring of sea ice and some oceanographic parameters provides a very important source of information on the CAO.

2020: (In bullet form, and as specific as possible -e.g. publications, advisory products, datasets, modelling outputs, methodological developments, etc. These will be used by ICES structures to demonstrate delivery. Typically 1-2 pages).

Final report on ToRs, workplan and Science Implementation Plan (=to be made in 2021)

(A summary of work done on the ToRs and in fulfilment of the work plan and the ICES (Strategic) Science Implementation Plan; Science Highlights, etc.

Format to include:

- Progress and fulfilment by ToR
- Science highlights

Τ

2 Status for WGICA IEA report in the CAO (CRR) (Report 1)

Report 1: Integrated Ecosystem Assessment of the Central Arctic Ocean: Ecosystem description and vulnerability characterization.

The first final report from the WGICA will be published by ICES in the Cooperative Research Report (CRR) series. This includes peer review and technical editing by ICES prior to publication.

The title of the report is: *Integrated Ecosystem Assessment of the Central Arctic Ocean: ecosystem description*. The plan was to include also a section on *vulnerability characterization*. This part is now moved to the second report where emphasis will be on human activities and their impacts on the CAO ecosystem.

The first report is made up of 9 chapters:

- Chapter 1 Introduction
- Chapter 2 Topography, oceanography and sea ice
- Chapter 3 Algae and primary production
- Chapter 4 Zooplankton and invertebrate ice fauna
- Chapter 5 Sympagic and pelagic bacterial communities
- Chapter 6 Arctic benthos
- Chapter 7 Fishes in the Central Arctic Ocean
- Chapter 8 Marine birds: species occurrence and habitat use
- Chapter 9 Marine mammals of and near the central Arctic Ocean

Draft versions of the chapters have been sent out to the full list of WGICA participants for their review and comments. Chapters 7 and 8 were sent 13 February, chapters 2 and 3 on 1 April, chapters 4 and 9 on 17 April, and chapters 1, 5 and 6 on 22 April. The texts are currently revised and updated based on comments received and graphics are being produced. The aim is to have a complete draft of the report ready to be sent to ICES by the end of 2020 for their further review, editing and publication of the report.

The chapters together provide a description of the ecosystem by ecosystem components following a traditional breakdown into oceanography, plankton, fishes, birds, etc. Each chapter is a review and summary of published information with an emphasis on spatial aspects (where in the CAO) and trophic interactions (prey and predators). This type of information provides some linkages between the chapters. In addition, the text on *Key features of the CAO ecosystem* will be included as a synthesis section of the report. The text that was included in the WGICA 2017 report, will be updated and constitute the main part of the executive summary of the report.

3 ToR b. Review and report on on-going and recent changes and events in the CAO ecosystem associated with changes such as in sea-ice, oceanographic circulation, and hydrographic properties

Contributors: Vladimir Ivanov, Shigeto Nishino and Randi Ingvaldsen.

3.1 Introduction

The purpose of this chapter is to provide up-to-date information on the recent changes in sea ice and oceanographic conditions in the Central Arctic Ocean (CAO) on the basis of open-access sources. The chapter consists of six sections, including the introduction.

Section 2 describes recent changes in sea ice conditions in context of general trends observed in the CAO within the last 20 years. Section 3 examines changes in the Atlantic and Pacific gateways to the Arctic. Section 4 looks at changes in the hydrographic properties of the CAO. Section 5 provides a summary, and Section 6 considers possible ecosystem implications of oceanographic change. Suggested priority research tasks are given at the end of the overall report, in Chapter 7.

A strong seasonal cycle, low temperatures, extensive permanent and seasonal ice cover, a large supply of freshwater from melting ice and rivers, and heat/salt input through the gateways are key physical factors shaping the CAO hydrographic regime. According to the latest report of the International Panel on Climate Change (IPCC, 2019: Chapter 3) over the first twenty years of this century the Arctic surface air temperature has increased more than twice as fast as the global mean. Although attribution studies confirm that greenhouse gases are the primary drivers of global temperature increase, climate change in the Arctic is progressing much faster than in the lower latitudes due to the so-called "Arctic amplification". New mechanisms of Arctic amplification are still being discovered, but the basic ones are well known, with ice-albedo feedback likely to be among the key ones. Rapid reduction of Arctic sea ice area and volume may thus be considered as the ultimate trigger for other physical changes currently taking place in the Arctic domain. Delayed onset of freezing in 'marginal' Arctic seas (Barents, Kara and Chukchi) provides a positive feedback ("atlantification"/"pacification") allowing warmer water to progress further toward the deep interior of the CAO. Besides unprecedented Arctic sea ice loss in summer in the early 21st century, it should be stressed that the last decade was exceptional in terms of record low ice extent at the peak of seasonal minimum relative to 1979-2018 (http://nsidc.org/arcticseaicenews/).

3.2 Changes in Arctic Sea ice

Independent studies confirm that Arctic sea ice extent in summer has substantially decreased in in 1990-2010s (e.g., Serreze et al., 2007; Stroeve et al., 2012; Comiso et al., 2017). According to the IPCC (2019: Chapter 3) Arctic sea ice extent has progressively declined throughout the year with maximum reduction in September (-12.8 \pm 2.3% per decade; 1979-2018). However, this average decay is not uniform in time. The strongest decrease happened between 1996 and 2007 (see Fig.

L

1). After 2007, the summer minimum of sea ice extent has remained fairly stable at -22 ± 8 per cent relative to the mean climatic norm for the available range of satellite observations (1979-2019). Climate models project further decay of the Arctic sea ice with a wide range in expected timing of a seasonally ice-free Arctic Ocean, and in the duration of the ice-free season as a result of the interplay among various affecting factors as well as uncertainties in models and the IPCC's representative concentration pathways (RCPs). This range varies from a 1% chance of brief ice-free years in September by the end of this century (in case of 1.5°C stable global warming) to a 10-35% chance of ice-free summers after 10 years of steady 2°C warming (IPCC, 2019: Chapter 3).

3.2.1 Central Arctic Ocean

The sharp rise of air temperature in the 1990s led to an "instant" (on a climatic time scale) change in the ice regime of the Arctic Ocean - a shift to a reduced area of ice cover at the seasonal minimum (in September) by an average of $22 \pm 8\%$ relative to the mean climatic norm for the available range of satellite observations (Fig. 3.1).



Figure 3.1: Anomaly in the extent of Arctic sea ice in September (%) relative to the mean climatic norm for the period of satellite observations (1979-2019) (based on data from https://nsidc.org/).

The fraction of multiyear ice (5 years and older) decreased from 30% to 2% between 1979 and 2018; while the fraction of first-year sea ice increased from 40% to 70% (Stroeve and Notz, 2018). Average ice thickness declined across the central Arctic by 65%, from 3.59 to 1.25 m between 1975 and 2012 (Lindsay and Schweiger, 2015). Under conditions of thinner and less consolidated sea ice, the speed of ice drift has increased compared with the 20th century. A vivid example of recent changes in ice dynamics seen by comparing drifting ships in the Arctic Ocean. Both the Tara in 2006-2007 (Gascard et al., 2008) and the Polarstern hosting the MOSAiC expedition (https://mosaic-expedition.org/)currently underway approximately followed the drift of the Fram in 1983-1896. The drift that took the Fram three years was completed by Tara in half that time. The Polarstern started about 600 nautical miles closer to Fram Strait due to ice conditions and had nearly reached Svalbard after only nine months.

I

3.2.2 Arctic seas

The strongest decay of ice cover in summer has occurred in the Eurasian Arctic seas. In the 2000s, the gradual expansion of the ice-free season progressed in two "waves", propagating towards the Arctic interior from the Barents Sea (in the west) and from the Chukchi Sea (in the east). In 2010s this "signal" reached the Laptev Sea (Fig. 3.2), which may be considered as the most remote location along the Arctic coast, far away from the warming influence of either the Atlantic or the Pacific.



Figure 3. 2: Duration of the ice-free days, defined by spatial mean ice concentration lower than 0.15, in the Laptev Sea. (based on data from https://nsidc.org/).

According to Egorov (2020), in 2004–2018, the area of winter ice thinner than 70 cm increased by 12 % compared to 1997–2003, while the area of thicker ice reduced by 18 %. The area of multiyear ice decreased by about 10 times. On average, in 2004–2018 mean-weighted ice thickness in the Eurasian Arctic seas by the end of cold season decreased by about 40 cm (from 153 to 117 cm) compared with 1997–2003.

Ice cover shrinking and ice thinning leads to enhanced ice-albedo feedback (Stroeve et al., 2014b; Liu et al., 2015a), and to excessive heat uptake by the surface water layer. In the next cold season this extra heat is mostly released to the air, providing a reverse warming influence on the lower atmosphere and delaying onset of freezing. In the 2010s, the thermal impact of the ocean on the ice cover and the Arctic atmosphere had increased (Ivanov et al., 2016; Tetzlaff et al., 2014). This impact is realized either directly, through upward heat flux from the intermediate Atlantic-origin water layer - the so-called "atlantification" (Polyakov et al., 2017; Ivanov and Repina, 2019) - or indirectly, through accumulation of extra heat in the upper water layer in summer, which may affect ice properties during the next cold season (Ivanov et al., 2019) and contribute to earlier break-up the following spring (Stroeve et al., 2012b; Serreze et al., 2016). Another emerging effect of increased duration and extended area of open water in the summer season is thinning of ice in mid-winter, with positive consequences for polynya formation. Arctic polynyas (areas of open water or thin ice inside consolidated ice pack) are the zones of intensive ice production and dense water formation on shallow shelves, where vertical haline convection reaches the seabed in winter (Ivanov et al., 2004; Luneva et al., 2020). While polynyas in the western Arctic show a pronounced variability with varying trends, positive trends in polynya area and ice production are I

detected for several regions of the Eurasian Arctic seas, most significantly in the Laptev Sea region as shown in Fig. 3.3 (Preußer et al., 2016).



Figure 3.3: Time series of polynya area and ice production of Laptev Sea polynyas from MODIS satellite data showing significant trends in sea-ice production and polynya area (Fig 6 in Preußer et al., 2016).

3.3 Changes in the gateways

3.3.1 Atlantic gateways (Barents Sea, Fram Strait and Western Nansen Basin)

The Atlantic gateways to the Arctic Ocean are strongly influenced by vigorous flow of Atlantic Water (e.g. Schauer et al., 2002), and currently the region is experiencing a greater role for the Atlantic inflows (Polyakov et al., 2017). This "atlantification" is manifested in a warmer ocean (Skagseth et al., 2020; Polyakov et al., 2020a) and atmosphere (Isaksen et al., 2016), loss of sea ice (Onarheim et al., 2018; Ivanov et al., 2018), and the northward and eastwards spread of Atlantic Water in the eastern (Oziel et al., 2016) and northern (Lind et al., 2018) Barents Sea, as well as in the Nansen Basin (Polyakov et al., 2017; 2020a). Changes in stratification and mixing in the upper ocean are also involved (Polyakov et al., 2020b).

Over the last four decades the temperatures of the inflowing Atlantic Water to the Barents Sea have increased by about 1.5oC (Smedsrud et al., 2013; Skagseth et al., 2020). The associated changes include substantial reductions in sea ice cover during winter (Arthun et al., 2012), increases in heat content of the region (Sandø et al., 2010), and a northeastwards shift of the location of the Polar Front (Oziel et al., 2016; Barton et al., 2018). Recent studies have also suggested that this might partly be due to increased velocities in the Atlantic Water surface currents (Oziel et al., 2020). Reductions of about 50% of the March sea-ice area between 1979 and 2018 have been observed, and have been linked to an increase in Atlantic Ocean heat transport, especially through the western part of the Barents Sea (Docquier et al., 2020). When the Atlantic Water passes through the shallow Barents Sea, it is cooled and mixed with the surrounding waters, and the resulting denser water masses that are exported to the Arctic Ocean are an important part of the Arctic thermohaline system. A recent study shows that Barents Sea cooling has become less efficient with reduced ocean heat loss over the southern Barents Sea (Skagseth et al., 2020). According to the study, the outflows from the Barents Sea to the Arctic Ocean have become warmer, although the outflow densities have remained relatively unperturbed as increasing salinity has compensated for the warming. If the trend continues, decreasing salinities in the upstream

Atlantic Water might cause the export of water masses of record-low density from the Barents Sea within a few years (Skagseth et al., 2020).

Despite substantial long-term trends, the inter-annual variability in the Atlantic gateways to the Arctic is substantial, and since 2015/2016 the temperature in the inflowing Atlantic Water has decreased (González-Pola et al., 2019). For the southwestern Barents Sea, the temperature of the inflowing Atlantic Water in 2019 was about at the same level as in the early 2000s (Figure 1). Observations from a recently established standard section (the Hinlopen section) north of Svalbard imply substantial inter-annual variations in Atlantic Water core temperature in this region (Figure 3.4). Maximum temperatures in the period 2014-2019 were observed in 2016. Associated with the decreasing temperatures, both regions have experienced increased sea ice cover since 2016.



Figure 3.4. Temperature (left) and salinity (right) anomalies in the Atlantic Water flow into the Arctic gateways. The time series sampled in August-September at the Fugløya-Bear Island section at the western entrance to the Barents Sea, and at the Hinlopen section near 200E north of Svalbard.

Also evident from the time series is a substantial freshening of the Atlantic Water in the Arctic gateways, which started around 2010 in the western Barents Sea (Fig.1). An accelerated freshening is evident from 2016, and similar changes were observed in the eastern subpolar North Atlantic, in the Nordic seas influencing the Greenland Sea, the northern Norwegian Sea to Fram Strait, and the southern reaches of the Barents Sea (González-Pola et al., 2016). Throughout the subpolar region, freshening is accompanied by moderate cooling at just a few sites, indicating that the large changes in salinity are decoupled from changes in temperature (González-Pola et al., 2019). It is likely that the Atlantic Water flowing into the Arctic Ocean has become substantially fresher during recent years, and possibly will do so for some years to come.

3.3.2 Pacific gateway (Bering and Chukchi seas)

The Bering and Chukchi shelves are an important region for fisheries (Van Vorhees & Lowther, 2010) and Indigenous subsistence resources (Suydam et al., 2007). The region has several biological hotspots with high benthic biomasses (Grebmeier et al., 1988, 2006, 2015; Grebmeier, 2012) and congregations of large benthic feeders at high trophic levels, such as grey whales and walruses (Feder et al., 2005). The marine ecosystem in this region is likely influenced by the recent ocean warming and reduction of sea ice.

The Bering Sea recorded unprecedented high sea surface temperatures in 2014 (Stabeno et al., 2017) and the warm condition has persisted into 2019 (Cornwall, 2019; Stabeno & Bell, 2019). Sea water in the winter of 2017-2018 was exceedingly warm and late storms affected sea ice duration and extent, resulting in an unprecedented period of low Bering Sea winter ice (Stabeno and Bell, 2019). Ecosystem surveys there showed that the spring bloom was delayed, zooplankton abundances were low, catches of pelagic forage fishes were low, and seabird die-offs were extensive (Duffy-Anderson et al., 2019).

The Bering and Chukchi seas are connected by the narrow (85 km) and shallow (50 m) Bering Strait. Through the strait, Pacific water carrying heat, freshwater, and nutrients is transported into the Arctic Ocean (Coachman and Aagaard, 1966; Stigebrandt, 1984; Walsh et al., 1989). The Bering Strait throughflow has been monitored by mooring observations (e.g., Woodgate et al., 2012, 2015), and it increased 50 % from 2001 (0.7 Sv) to 2011 (1.1 Sv), driving heat and freshwater flux increases. Enhanced Bering Strait throughflow likely transports more nutrients (Woodgate, 2018) that may help increase the primary production in the Chukchi Sea where the ice-free season has extended in recent decades (e.g., Arrigo et al., 2008). On the other hand, Yun et al. (2016) hypothesized that significant decreases of 30-50% in nutrient concentrations occurred over recent decades in the Bering Strait and the Chukchi Sea, resulting in a decrease of primary productivity that was actually estimated from in situ measurements. As a source of heat, freshwater, and nutrients, the Pacific water that flows through the Bering Strait plays important roles in the sea ice and water mass distributions and ecosystem in the Chukchi Sea and the Canada Basin (e.g., Coachman & Barnes, 1961; Lin et al., 2019; Lowry et al., 2015; Nishino et al., 2011; Shimada et al., 2006; Spall et al., 2018; Steele et al., 2004, 2008; Timmermans et al., 2014, 2018; Woodgate et al., 2010).

In the Chukchi Sea, Danielson et al. (2020) found a significant summer and fall warming of 1.4 °C from climatological data over 1922-2018 (0.14 ± 0.07 °C decade-1), and over 1990-2018 the warming rate tripled to 0.43 ± 0.35 °C decade-1. They proposed a conceptual model that summarized the controlling feedback loop for the Pacific Arctic changes of heat content, sea ice, fresh water distributions, surface heat fluxes, and advective fluxes. Interannual variation in solar heating in the Chukchi Sea over 1999-2015 was examined by satellite-derived sea ice concentration and reanalysis of shortwave radiation (Tsukada et al., 2018). The result showed that solar heating during summer varied from 3.6 to 6.7 ×1020 J and was up to twice the northward heat flux through the Bering Strait. In Barrow Canyon, at the northeastern end of the Chukchi Sea, mooring observations have been carried out since the late 1990s (Itoh et al., 2013; Williams et al., 2014), and the data indicate a significant increase in Alaskan Coastal Water (ACW) temperature in the 2000s (~4 °C to 8°C). The ACW further spreads into the Canada Basin via Barrow Canyon. In addition to the Barrow Canyon throughflow, westward jets along the Chukchi shelf slope are identified with a subsurface warm water transport during winter toward the Chukchi Borderland (Watanabe et al., 2017). Anomalously warm water was found over the Chukchi shelf slope in winter 2016/2017 (Kimura et al., 2019). This warm water was likely derived from the Bering Sea via the Bering Strait and Barrow Canyon.

3.4 Changes in CAO hydrographic properties

3.4.1 Upper ocean layer

ontrasting regional responses to atlantification (related to progression of anomalies from the Atlantic sector) and pacification (associated with influx of anomalous Pacific waters). The analyses revealed strong salinification of the upper Eurasian Basin since 2000, with attendant reductions in stratification, and potentially altered nutrient fluxes and primary production. This was in contrast to the situation in the Amerasian Basin, which showed strong freshening, a thickening of the fresh surface layer, and a deepening of the nutricline and deep chlorophyll maximum. Thus, a divergent (Eurasian Basin) gyre responds altogether differently than does a convergent (Amerasian Basin) gyre to climate forcing (Polyakov et al., 2020a).

The decrease in sea ice contributes strongly to the freshening of the Arctic Ocean, especially in the Beaufort Gyre of the Canada Basin, as a result of freshwater accumulation (Proshutinsky et al., 2009; Wang et al., 2018). Proshutinsky et al. (2019) estimated an increase of more than 6,400 km3 of liquid freshwater content in the Beaufort Gyre from 2003 to 2018, a 40% increase relative

to the 1970s. If the freshwater is released to lower latitudes, it may impact sea ice conditions, ocean circulation, and ecosystems of the Sub-Arctic similar to the influence of the Great Salinity Anomaly observed in the 1970s.

Joint analysis of historical observations in the CAO (from 1950 to 2010) allowed Ivanov and Golovin (2019) to reveal multidirectional changes in the thermohaline parameters of the upper layer in the western and eastern parts of the Arctic basin in different seasons. In the Eurasian Arctic in the winter season, salinification and cooling of surface waters are happening, whereas in the eastern Arctic, to the contrary, there is strong desalination and a corresponding increase of water temperature. One of the main reasons for the desalination of surface waters in the eastern Arctic is a reduction in the area of perennial ice and its replacement by seasonal ice at the same time that riverine water discharge is increasing and there are changes in atmospheric circulation over the Arctic. The salinification of surface waters in the western Arctic is associated with an increase in the supply of saline Atlantic waters from the Fram Strait and their rise to the surface under conditions of increased seasonality of Arctic sea ice.

Significant 'seasonal memory', characterized by consistent change of the ice cover parameters in sequential seasons, was revealed in the Eurasian segment of the CAO (60-120°E, 65-80°N) between 2007 and 2017 (Ivanov et al., 2019). This interconnectivity was hypothetically explained by increased accumulation of heat in the upper layer under present conditions of depleted ice cover.

3.4.2 Halocline layer

The observed reductions in stratification in the Nansen Basin have also caused a weakening of the halocline in recent years (Polyakov et al., 2020b). This, in combination with shoaling of the Atlantic Water, has resulted in a substantial increase of upward oceanic heat flux from Atlantic Water during the winter season, from an average of 3-4 W/m2 in 2007-2008 to >10 W/m2 in 2016-2018 (Polyakov et al., 2020b).

Timmermans et al. (2014) explained the maintenance mechanism of a permanent halocline in the Canada Basin in terms of ventilation by Pacific summer water (PSW) from the Chukchi Sea. Thus, the recent warming and freshening of the Chukchi Sea (e.g., Danielson et al., 2020) could result in increasing trends in integrated heat and freshwater content in the halocline of the Canada Basin. Timmermans et al. (2018) estimated a near doubling of ocean heat content relative to the freezing temperature in the Canada Basin halocline over the past three decades (1987–2017). Pacific winter water (PWW) also spreads into the Canada Basin and forms a cold halocline (Jones and Anderson, 1986). Zhong et al. (2019) suggested that the thickening, deepening, and expansion of the PWW layer in the Canada Basin in recent years is associated with the enhanced and westward-shifted Beaufort Gyre.

The permanent halocline in the Canada Basin generally corresponds to the nutricline, the layer where nutrient concentrations increase rapidly with depth. Freshwater accumulation in the Canada Basin causes deepening of the nutricline (McLaughlin & Carmack, 2010) and can change phytoplankton community structure (Li et al., 2009), decrease phytoplankton production (Coupel et al., 2015), and weaken the biological pump (Nishino et al., 2011; Zhuang et al., 2018). On the other hand, the delay in fall freezing of the eastern part of the East Siberian Sea during the late 2000s compared with the early 2000s might have resulted in the formation of a large water mass through cooling and convection. The spread of this water into the southern Makarov Basin may have caused shoaling of the nutricline (Nishino et al., 2013). Shelf water in the western part of the East Siberian Sea, containing high nutrient concentrations, also spreads into the central Arctic Ocean, forming a shallower nutricline than that in the Canada Basin (Alkire et al., 2019).

These shallow nutriclines could create an advantage for phytoplankton production under decreasing sea ice conditions.

3.4.3 Intermediate and deep layers

Intermediate Atlantic water layer in the CAO experienced two warming episodes in 1990s and 2000s (Polyakov et al., 2005). After peak warming in 2005-2007 (about 1.5°C over the climate norm in the 'core' of this water mass), the temperature decreased, but has not returned to mean climate values. During the most recent 10 years the temperature in the warm 'core' has remained about 0.7-1°C over the climate norm, with seasonal and interannual variations (Baumann et al., 2018). Heat content in the Atlantic water layer (150-900 m) in the CAO continues to increase due to shoaling of the upper boundary of zero-degree isotherm (Polyakov et al., 2017), thus affecting upper layers and, hypothetically, ice cover as a result of weakened vertical stratification in the halocline (Polyakov et al., 2020b). Recent estimates show 2-4 times larger heat fluxes in 2014-2015 compared with 2007-2008. In the Canadian Basin, the maximum temperature of the Pacific Water layer increased by ~0.5°C between 2009 and 2013 (Timmermans et al., 2014), resulting in doubling of integrated heat content over 1987-2017 (Timmermans et al., 2018).

In the Nansen Basin, the Atlantic Water flow is manifested as a narrow slope current. Evaluating its impact and recent changes on the Arctic slopes, Bluhm et al. (in review) show that at the upper slope, climate change is currently manifested in sea-ice retreat, increased heat and mass transport by sub-Arctic inflows, surface warming, and altered vertical stratification, while the lower slope has yet to display evidence of change. Model projections suggest that ongoing physical changes will enhance primary production at the upper slope, with suspected enhancing effects for consumers. A Pan-Arctic monitoring effort of slopes is recommended given that many signals of climate change appear there first and are then transmitted along the slope domain (Bluhm et al., in review).

On the basis of long-term mooring observations in the eastern Arctic Ocean, Polyakov et al. (2020b) have proposed an emergence of new, more energetic state dominated by stronger currents and shear, leading to increased turbulent mixing and larger upward oceanic heat fluxes from intermediate Atlantic waters.

To assess interannual variability of the Atlantic water flow along the Eurasian continental slope Pnyushkov et al. (2020) analyzed observations at long term moorings in the Laptev Sea at five moored stations in 2013-2018. These observations revealed a lack of significant changes in the intensity of heat and water volume transports in this part of the Eurasian Basin, suggesting a steady regime of the slope current in the CAO despite substantial changes in ice cover.

No significant changes have been reported in the deep CAO waters in recent years.

Climate model projections suggest that throughout the 21st century the Arctic Ocean will gradually warm up from surface to deep layers (IPCC, 2019: Chapter 3). The basic mechanisms of this warming are presently acting processes and feedbacks, including: excess summer heating of open water areas, leading to additional ice melt, accelerated due to positive ice-albedo feedback. Rapid transformation of vertical hydrographic structure in the Western Nansen Basin (Ivanov et al., 2018) and in the Barents Sea (Lind et al., 2018) in the 2010s may indicate that IPCC scenarios underestimate the pace of this transformation.

3.5 Summary

After rapid decline between 1996 and 2007, summer sea ice extent remains fairly stable at -22 ± 8 per cent relative to the mean climatic norm for the available range of satellite observations (1979-

2019). The fraction of multiyear ice in CAO and Arctic seas decreased from 30% to 2% between 1979 and 2018; while the fraction of first-year sea ice increased from 40% to 70%. Ice thickness declined across the central Arctic by 65%, from 3.59 to 1.25 m, between 1975 and 2012. The speed of ice motion in the Transpolar drift (towards Fram Strait) roughly doubled between the end of the 19th century and the 2000-2010s.

The seasonal duration of open water has substantially increased in the Eurasian Arctic seas and on the margins of the deep ocean interior (continental slopes). In the Laptev Sea the ice-free season in individual years in the late 2010s reached almost three months. Excessive accumulation of heat in the upper mixed layer in summer started to noticeably affect first-year ice thickness, thus increasing the frequency of polynya events in mid-winter and leading to earlier ice breaking in spring. As a result, the area of winter ice thinner than 70 cm increased by 12 % compared with 1997–2003, while the area of thicker ice reduced by 18 %. On average, mean-weighted ice thickness in the Eurasian Arctic seas at the end of cold season decreased by about 40 cm (from 153 to 117 cm) from 1997–2003 to 2004-2018.

The Atlantic gateways to the Arctic Ocean are strongly influenced by vigorous flow of Atlantic Water, and currently the region is experiencing a greater role for the Atlantic inflows: the so-called "atlantification", which is manifested in a warmer ocean and atmosphere, loss of sea ice, northward and eastwards spread of Atlantic Water in the Barents Sea and the Nansen Basin, and changes in stratification and mixing in the upper ocean. Despite substantial long-term trends, the inter-annual variability in the Atlantic gateways to the Arctic is substantial, and since 2015/2016 the temperature in the inflowing Atlantic Water has decreased, while in the southwestern Barents Sea it remains similar to the early 2000s.

As a source of heat, freshwater, and nutrients, the Pacific water that flows through the Bering Strait plays important roles in sea ice and water mass distributions and ecosystem in the Chukchi Sea and the Canada Basin. The Bering Sea recorded unprecedented high sea surface temperatures in 2014 and the warm condition has persisted into 2019. Sea water in winter of 2017-2018 was exceedingly warm and late storms affected sea ice duration and extent, resulting in an unprecedented period of low Bering Sea winter ice. Anomalously warm water was found over the Chukchi shelf slope in winter 2016/2017. This warm water was likely derived from the Bering Sea via the Bering Strait and Barrow Canyon.

Opposing patterns of salinification/freshening in the upper water layer in different regions of the CAO due to ice melt, freshwater transport and saltwater input through the gateways is detected. Salinification generally happens in the Atlantic sector and freshening generally happens in the Pacific sector. Strong salinification is observed in the upper Eurasian Basin since 2000, with attendant reductions in stratification and potentially altered nutrient fluxes and primary production. In contrast, the Canadian Basin showed strong freshening, a thickening of the fresh surface layer, and a deepening of the nutricline and deep chlorophyll maximum.

Weakening of stratification in the halocline layer in the Nansen Basin in combination with shoaling of the Atlantic Water, has resulted in a substantial increase of upward oceanic heat flux from Atlantic Water during the winter season, from an average of 3-4 W/m2 in 2007-2008 to >10 W/m2 in 2016-2018. Recent warming and freshening of the Chukchi Sea could result in increasing trends in integrated heat and freshwater content in the halocline of the Canada Basin.

After a substantial increase (by about 1.5 deg C) in the early 2000s, intermediate Atlantic water in the CAO has remained at a higher (by 0.7-1°C) temperature than the climate norm of the second half of the 20th century. Interannual oscillations of Atlantic Water temperature around this elevated level are observed. Heat content in the Atlantic Water layer in the CAO continues to increase due to shoaling of the upper boundary of the zero-degree isotherm. A more energetic state of the intermediate water layer in the CAO is projected in future years. This new state will be presumably supported by stronger currents and shear, leading to increased turbulent mixing and larger upward oceanic heat fluxes. In the Canadian Basin, the maximum temperature of the Pacific Water layer increased by about 0.5°C between 2009 and 2013, resulting in a doubling of integrated heat content over 1987-2017.

Climate model projections suggest that throughout the 21st century the Arctic Ocean will gradually warm up from surface to deep layers. The basic mechanisms of this warming are presently acting processes and feedbacks, including: excess summer heating of open water areas, leading to additional ice melt, and accelerated due to positive ice-albedo feedback.

3.6 Possible ecosystem implications

Currently observed and projected changes in sea ice properties (concentration, thickness, seasonality and drift) may affect primary production with possible impacts on ecosystems through changes in marine species composition, spatial distribution, and abundance.

The general trend of poleward retreat of sea ice cover provides better conditions for penetration of solar radiation in the water column. This will lead to better light conditions in the upper water column for longer duration, higher water temperatures and more intensive vertical mixing – all these factors positively influence primary production.

Expected expansion of winter polynyas may be significant for the early spring phytoplankton bloom. Downslope cascading of oxygenated dense water, originating in polynya zones, frequently triggers compensatory upwelling of nutrient-rich deep waters, thus making continental slope areas favorable for early blooms.

Ongoing "atlantification" in the Atlantic sector of the CAO may have several consequences for biological and ecological conditions in the affected areas (Barents-Kara seas and western Nansen Basin). Direct effects may be linked with northwards and eastwards expansion of the areal extent of Atlantic species. Indirect effect may be caused by a gradual shift of hydrographic conditions to sub-polar type, characterized by intensive vertical mixing in winter and an ice-free regime throughout the year with corresponding impacts on the entire ecological system.

Progressive freshwater accumulation in the Canada Basin causes a deepening of the nutricline and can decrease phytoplankton production. However, spreading of shelf-origin waters with high nutrient concentrations into the CAO may contribute to formation of a shallower nutricline in the adjacent deep basin than that in the Canada Basin, thus allowing relatively high phytoplankton production under decreasing sea ice conditions.

Contrasting changes in surface salinity in the Eurasian Basin (increase) and Canadian Basin (decrease) might have opposite effects on salinity-dependent species. In a general perspective, such changes might lead to opposite trends in vertical mixing in these basins, resulting in more favorable conditions for primary production (in terms of nutrient supply) in the upper water layer of the Eurasian Basin.

3.7 References

- Alkire, M. B., Rember, R., & Polyakov, I. (2019). Discrepancy in the identification of the Atlantic/Pacific front in the central Arctic Ocean: NO versus nutrient relationships. Geophysical Research Letters, 46, 3843– 3852. https://doi.org/10.1029/2018GL081837
- Arrigo, K. R., van Dijken, G., & Pabi, S. (2008). Impact of a shrinking Arctic ice cover on marine primary production. Geophysical Research Letters, 35, L19603. https://doi.org/10.1029/2008GL035028
- Årthun, M., Eldevik, T., Smedsrud, L.H., Skagseth, Ø., Ingvaldsen, R.B., 2012. Quantifying the influence of Atlantic heat on the Barents Sea ice variability and retreat. Journal of Climate, Vol. 25, No. 13, 4736-4743.

- Barton, BI, Lenn, Y-D, Lique, C., 2018. Observed Atlantification of the Barents Sea Causes the Polar Front to Limit the Expansion of Winter Sea Ice. J. Phys. Oceanogr. (2018) 48 (8): 1849–1866. https://doi.org/10.1175/JPO-D-18-0003.1
- Baumann T.M., I. V. Polyakov, A. V. Pnyushkov, R. Rember, V. V. Ivanov, M. B. Alkire, I. Goszczko, E. C. Carmack, 2018, On the Seasonal Cycles Observed at the Continental Slope of the Eastern Eurasian Basin of the Arctic Ocean, Journ. Phys. Oceanogr., 48, 1451-1470; DOI: 10.1175/JPO-D-17-0163.1
- Bluhm et al., in review. Frontiers in Marine Science: Towards a Unifying Pan-Arctic Perspective of the Contemporary and Future Arctic.
- Coachman, L. K. & Barnes, C. A. (1961). The contribution of Bering Sea water to the Arctic Ocean. Arctic, 14, 147–161.
- Coachman, L. K., & Aagaard, K. (1966). On the water exchange through Bering Strait. Limnology and Oceanography, 11 (1), 44-59. https://doi.org/10.4319/lo.1966.11.1.0044
- Comiso, J. C., Meier, W. N., & Gersten, R. (2017). Variability and trends in the Arctic Sea ice cover: Results from different techniques. Journal of Geophysical Research: Oceans, 122, 6883–6900. https://doi.org/10.1002/2017JC012768
- Cornwall, W. (2019). Vanishing Bering Sea ice threatens one of the richest U.S. seafood sources. Science. https://doi.org/10.1126/science.aay0513
- Coupel, P., Ruiz-Pino, D., Sicre, M. A., Chen, J. F., Lee, S. H., Schiffrine, N., Li, H. L., & Gascard, J. C. (2015). The impact of freshening on phytoplankton production in the Pacific Arctic Ocean. Progress in Oceanography, 131, 113–125. https://doi.org/10.1016/j.pocean.2014.12.003
- Danielson, S. L., Ahkinga, O., Ashjian, C., Basyuk, E., Cooper, L. W., Eisner, L., Farley, E., Iken, K. B., Grebmeier, J. M., Juranek, L., Khen, G., Jayne, S. R., Kikuchi, T., Ladd, C., Lu, K., McCabe, R. M., Moore, G. W. K., Nishino, S., Ozenna, F., Pickart, R. S., Polyakov, I., Stabeno, P. J., Thoman, R., Williams, W. J., Wood, K., & Weingartner, T. J. (2020). Manifestation and consequences of warming and altered heat fluxes over the Bering and Chukchi Sea continental shelves. Deep-Sea Research Part II, in press. https://doi.org/10.1016/j.dsr2.2020.104781
- Docquier, D, Fuentes-Franco, R, Koenigk, T, Fichefet, T, 2020. Sea Ice—Ocean Interactions in the Barents Sea Modeled at Different Resolutions. Frontiers in Earth Science, 8. DOI=10.3389/feart.2020.00172.
- Duffy-Anderson, J. T., Stabeno, P., Andrews, A. G., III, Cieciel, K., Deary, A., Farley, E., et al. (2019). Responses of the northern Bering Sea and southeastern Bering Sea pelagic ecosystems following recordbreaking low winter sea ice. Geophysical Research Letters, 46, 9833–9842. https://doi.org/10.1029/2019GL083396
- Egorov A.G. The Russian Arctic seas ice age composition and thickness variation in winter periods at the beginning of the 21st century. Problemy Arktiki i Antarktiki. Arctic and Antarctic Research. 2020, 66 (2): 124–143. [In Russian]. https://doi.org/10.30758/0555-2648-2020-66-2-124-143.
- Feder, H., Jewett, S., & Blanchard, A. (2005). Southeastern Chukchi Sea (Alaska) epibenthos. Polar Biology, 28, 402–421.
- Gascard J.-C. et al. Exploring Arctic Transpolar Drift During Dramatic Sea Ice Retreat, Eos, Vol. 89, No. 3, 15 January 2008
- González-Pola, C., Larsen, K. M. H., Fratantoni, P., and Beszczynska-Möller, A. (Eds.) 2019. ICES Report on Ocean Climate 2018. ICES Cooperative Research Report No. 349. 122 pp. https://doi.org/10.17895/ices.pub.5461
- Grebmeier, J. M. (2012). Shifting patterns of life in the Pacific Arctic and sub-Arctic seas. Annual Review of Marine Science, 4, 63–78. https://doi.org/10.1146/annurev-marine-120710-100926
- Grebmeier, J. M., Bluhm, B. A., Cooper, L. W., Danielson, S. L., Arrigo, K. R., Blanchard, A. L., Clarke, J. T., Day, R. H., Frey, K. E., Gradinger, R. R., Kedra, M., Konar, B., Kuletz, K. J., Lee, S. H., Lovvorn, J. R., Norcross, B. L., & Okkonen, S. R. (2015). Ecosystem characteristics and processes facilitating persistent macrobenthic biomass hotspots and associated benthivory in the Pacific Arctic. Progress in Oceanography, 136, 92–114. https://doi.org/10.1016/j.pocean.2015.05.006

L

- Grebmeier, J. M., Cooper, L. W., Feder, H. M., & Sirenko, B. I. (2006). Ecosystem dynamics of the Pacificinfluenced Northern Bering and Chukchi Seas in the Amerasian Arctic. Progress in Oceanography, 71, 331–361.
- Grebmeier, J. M., McRoy, C. P., & Feder, H. M. (1988). Pelagic-benthic coupling on the shelf of the northern Bering and Chukchi Seas. I. Food supply source and benthic biomass. Marine Ecology-Progress Series, 48, 57–67.
- IPCC, 2019: IPCC Special Report on the Ocean and Cryosphere in a Changing Climate [H.-O. Pörtner, D.C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, K. Mintenbeck, A. Alegría, M. Nicolai, A. Okem, J. Petzold, B. Rama, N.M. Weyer (eds.)]. <u>https://www.ipcc.ch/site/assets/up-loads/sites/3/2019/12/SROCC_FullReport_FINAL.pdf</u>
- Isaksen, K, Nordli, Ø, Førland, EJ, Łupikasza, E, Eastwood, S, and Niedźwiedź, T 2016 Recent warming on Spitsbergen—Influence of atmospheric circulation and sea ice cover. J. Geophys. Res. Atmos. 121: 11,913–11,931, doi:10.1002/2016JD025606.
- Itoh, M., Nishino, S., Kawaguchi, Y., & Kikuchi, T. (2013). Barrow Canyon volume, heat, and freshwater fluxes revealed by long-term mooring observations between 2000 and 2008. Journal of Geophysical Research, 118, 4363-4379. https://doi.org/10.1002/jgrc.20290
- Ivanov V., V. Alexeev, N.V. Koldunov, I.A. Repina, A.B. Sandoe, L.H. Smedsrud and A. Smirnov, 2016, Arctic Ocean Heat Impact on Regional Ice Decay: A Suggested Positive Feedback, Journ. Phys. Oceanogr., 46, 1437-1456, DOI: 10.1175/JPO-D-15-0144.1
- Ivanov V.V., G.I.Shapiro, J.M. Huthnance, D.M. Aleynik, P.N. Golovin, Cascades of dense water around the World Ocean, 2004, Progress in Oceanography, 60, 47-98
- Ivanov V.V., Golovin P.N. Interannual variability of thermohaline parameters in the deep Arctic Ocean. Fundamentalnaya i Prikladnaya Gidrofizika. 2019, 12, 3, 13–25 DOI: 10.7868/S2073667319030031Polyakov et al., 2005
- Ivanov V.V., Repina I.A., 2019, Mid-winter anomaly of sea ice in the Western Nansen Basin in 2010s. IOP Conf. Ser.: Earth Environ. Sci. 231 012024; doi:10.1088/1755-1315/231/1/012024
- Ivanov, V, Smirnov, A, Alexeev, V, Koldunov, NV, Repina, I, Semenov, V., 2018. Contribution of Convection-Induced Heat Flux to Winter Ice Decay in the Western Nansen Basin. Journal of Geophysical Research: Oceans. Volume 123, Issue 9, 6581-659. https://doi.org/10.1029/2018JC013995.
- Ivanov, V.; Varentsov, M.; Matveeva, T.; Repina, I.; Artamonov, A.; Khavina, E. Arctic Sea Ice Decline in the 2010s: The Increasing Role of the Ocean-Air Heat Exchange in the Late Summer. Atmosphere 2019, 10, 184. http://dx.doi.org/10.3390/atmos10040184
- Jones, E. P., & Anderson, L. G. (1986). On the origin of the chemical properties of the Arctic Ocean halocline. Journal of Geophysical Research, 91, 10,759–10,767.
- Kimura, S., Onodera, J., Itoh, M., Kikuchi, T., Nishino, S., Kawaguchi, Y., Watanabe, E., & Harada, N. (2019). The warming of the Chukchi slope through the Barrow Canyon outflow in the 2016-2017 winter. Journal of Geophysical Research: Oceans, 124, 7437-7456. https://doi.org/10.1029/2019JC015093
- Li, W. K. W., McLaughlin, F. A., Lovejoy, C., & Carmack, E. C. (2009). Smallest algae thrive as the Arctic Ocean freshens. Science, 326 (5952), pp. 539. https://doi.org/10.1126/science.1179798
- Lin, P., Pickart, R. S., McRaven, L. T., Arrigo, K. R., Bahr, F., Lowry, K. E., et al (2019). Water mass evolution and circulation of the northeastern Chukchi Sea in summer: Implications for nutrient distributions. Journal of Geophysical Research: Oceans, 124,4416–4432. https://doi.org/10.1029/2019JC015185
- Lind, S, Ingvaldsen, RB, and Furevik, T 2018 Arctic warming hotspot in the northern Barents Sea linked to declining sea-ice import, Nature Climate Change, 8, 634–639, https://doi.org/10.1038/s41558-018-0205y.
- Lindsay, R. and A. Schweiger. (2015). Arctic sea ice thickness loss determined using subsurface, aircraft, and satellite observations. The Cryosphere, 9 (1), 269-283, doi:10.5194/tc-9-269-2015.

- Liu, J., M. Song, M. H. Radley and Y. Hu. (2015). Revisiting the potential of melt pond fraction as a predictor for the seasonal Arctic sea ice extent minimum. Environmental Research Letters, 10 (5), 054017, doi:10.1088/1748-9326/10/5/054017/meta.
- Lowry, K. E., Pickart, R. S., Mills, M. M., Brown, Z. W., van Dijken, G. L., Bates, N. R., & Arrigo, K. R. (2015). The influence of winter water on phytoplankton blooms in the Chukchi Sea. Deep Sea Research, Part II, 118, 53–72. https://doi.org/10.1016/j.dsr2.2015.06.006
- Luneva M., V V. Ivanov, F. Tuzov, Ye. Aksenov, J. D. Harle, S. Kelly, J. T. Holt (2020), Hotspots of dense water cascading in the Arctic Ocean: implications for the Pacific water pathways, Journal of Geophysical Research: Oceans (in press).
- McLaughlin, F. A., & Carmack, E. C. (2010). Deepening of the nutricline and chlorophyll maximum in the Canada Basin interior, 2003-2009. Geophysical Research Letters, 37, L24602. https://doi.org/10.1029/2010GL045459
- Nishino, S., Itoh, M., Williams, W. J., & Semiletov, I. (2013). Shoaling of the nutricline with an increase in near-freezing temperature water in the Makarov Basin. Journal of Geophysical Research: Oceans, 118, 635-649. https://doi.org/10.1029/2012JC008234
- Nishino, S., Kikuchi, T., Yamamoto-Kawai, M., Kawaguchi, Y., Hirawake, T., & Itoh, M. (2011). Enhancement/reduction of biological pump depends on ocean circulation in the sea-ice reduction regions of the Arctic Ocean. Journal of Oceanography., 67, 305–314. http://doi.org/10.1007/s10872-011-0030-7
- Onarheim, IH, Eldevik, T, Smedsrud, LH, Stroeve, JC, 2018. Seasonal and Regional Manifestation of Arctic Sea Ice Loss. J. Climate (2018) 31 (12): 4917–4932. https://doi.org/10.1175/JCLI-D-17-0427.1.
- Oziel, L, Sirven, J, Gascard, J-C, 2016. The Barents Sea frontal zones and water masses variability (1980–2011). Ocean Sci., 12, 169–184. doi:10.5194/os-12-169-2016.
- Oziel, L., Baudena, A., Ardyna, M., Massicotte, P., Randelhoff, A., et al., 2020. Faster Atlantic currents drive poleward expansion of temperate phytoplankton in the Arctic Ocean Nature Climate Change, https://doi.org/10.1038/s41467-020-15485-5.
- Pnyushkov et al. A steady regime of volume and heat transports in the eastern Arctic Ocean in the early 21st century. (2020). Geophys.Res. Lett. (submitted)
- Polyakov, I., Beszczynska, A., Carmack, E.C., Dmitrenko, I.A., Fahrbach, E., Frolov, I.E., Gerdes, R., Hansen, E., Holfort, J., Ivanov, V.V., Johnson, M.A., Karcher, M., Kauker, F., Morison, J., Orvik, K.A., Schauer, U., Simmons, H.L., Skagseth, Ø, Sokolov, V.T., Steele, M., Timokhov, L.A., Walsh, D., Walsh, J.E One more step towards a warmer Arctic, 2005, Geophys. Res. Lett., 32, L17605, doi: 10.1029/2005GL023740
- Polyakov I.V., A. V. Pnyushkov, M. B. Alkire, I. M. Ashik, T. M. Baumann, E. C. Carmack, I. Goszczko, J. Guthrie, V. V. Ivanov, T. Kanzow, R. Krishfield, R. Kwok, A. Sundfjord, J. Morison, R. Rember, A. Yulin, Greater role for Atlantic inflows on sea-ice loss in the Eurasian Basin of the Arctic Ocean, Science 356, 285–291, 2017; http://science.sciencemag.org/.
- Polyakov, IV, Alkire, MB, Bluhm, BA, Brown, K, Carmack, EC, et al., 2020a. Borealization of the Arctic Ocean in response to anomalous advection from sub-arctic seas. Frontiers in Marine Science, in press.
- Polyakov, IV, Rippeth, TR, Fer, I, Alkire, MB, Baumann, TM, et al., 2020b. Weakening of cold halocline layer exposes sea ice to oceanic heat in the eastern Arctic Ocean. Journal of Climate, in press.
- Preußer, A., Heinemann, G., Willmes, S., Paul, S., 2016: Circumpolar polynya regions and ice production in the Arctic: Results from MODIS thermal infrared imagery for 2002/2003 to 2014/2015 with a regional focus on the Laptev Sea. The Cryosphere 10, 3021-3042. doi:10.5194/tc-10-3021-2016.
- Proshutinsky, A., Krishfield, R., Timmermans, M.-L., Toole, J., Carmack, E., McLaughlin, F., Williams, W. J., Zimmermann, S., Itoh, M., & Shimada, K. (2009). Beaufort Gyre freshwater reservoir: State and variability from observations. Journal of Geophysical Research, 114, C00A10. https://doi.org/10.1029/2008JC005104

- Proshutinsky, A., Krishfield, R., Toole, J. M., Timmermans, M.-L., Williams, W., Zimmermann, S., et al. (2019). Analysis of the Beaufort Gyre freshwater content in 2003–2018. Journal of Geophysical Research: Oceans, 124, 9658–9689. https://doi.org/10.1029/2019JC015281
- Sandø, AB, Nilsen, JE, Gao, Y, Lohmann K, 2010. Importance of heat transport and local air-sea heat fluxes for Barents Sea climate variability. Journal of Geophysical Research: Oceans 115 (C7).
- Schauer U, Loeng H, Rudels B, Ozhigin VK, Dieck W, 2002. Atlantic Water flow through the Barents and Kara Seas, Deep-Sea Res. I, 49 (2002), 2281-2298.
- Serreze, M. C., Holland, M. M., & Stroeve, J. (2007). Perspectives on the Arctic's shrinking sea-ice cover. Science, 315, 1533–1536. https://doi.org/10.1126/science.1139426
- Serreze, M. C., J. Stroeve, A. P. Barrett and L. N. Boisvert, (2016). Summer atmospheric circulation anomalies over the Arctic Ocean and their influences on September sea ice extent: A cautionary tale. Journal of Geophysical Research: Atmospheres, 121 (19), 11,463-11,485, 48, doi:10.1002/2016jd025161.
- Shimada, K., Kamoshida, T., Itoh, M., Nishino, S., Carmack, E., McLaughlin, F., et al. (2006). Pacific Ocean inflow: Influence on catastrophic reduction of sea ice cover in the Arctic Ocean. Geophysical Research Letters, 33, L08605. https://doi.org/10.1029/2005GL025624
- Skagseth, Ø., Eldevik, T., Arthun, M., Asbjørnsen, H, Lien, VS, et al., 2020. Reduced efficiency of the Barents Sea cooling machine. Nature Climate Change. https://doi.org/10.1038/s41558-020-0772-6.
- Smedsrud, L.H., Esau, I., Ingvaldsen, R.B., Eldevik, T., Haugan, P.M., Li, C., et al., 2013. The role of the Barents Sea in the Arctic climate system. Reviews in Geophysics, 51 (7), 1-35, doi:10.1002/rog.20017.
- Spall, M. A., Pickart, R. S., Li, M., Itoh, M., Lin, P., Kikuchi, T., & Qi, Y. (2018). Transport of Pacific water into the Canada Basin and the formation of the Chukchi Slope Current. Journal of Geophysical Research: Oceans, 123, 7453–7471. https://doi.org/10.1029/2018JC013825
- Stabeno, P. J., & Bell, S. W. (2019). Extreme conditions in the Bering Sea (2017-2018): Record-breaking low sea-ice extent. Geophysical Research Letters, 46 (15), 8952-8959. https://doi.org/10.1029/2019GL083816
- Stabeno, P. J., Duffy-Anderson, J. T., Eisner, L. B., Farley, E. V., Heintz, R. A., & Mordy, C. W. (2017). Return of warm conditions in the southeastern Bering Sea: physics to fluorescence. PloS One, 12 (9), e0185464. https://doi.org/10.1371/journal.pone.0185464
- Steele, M., Ermold, W., & Zhang, J. (2008). Arctic Ocean surface warming trends over the past 100 years. Geophysical Research Letters, 35 (2), L02614. https://doi.org/10.1029/2007GL031651
- Steele, M., Morison, J., Ermold, W., Rigor, I., Ortmeyer, M., & Shimada, K. (2004). Circulation of summer Pacific halocline water in the Arctic Ocean. Journal of Geophysical Research, 109, C02027. https://doi.org/10.1029/2003JC002009
- Stigebrandt, A. (1984). The North Pacific: a global-scale estuary. Journal of Physical Oceanography, 14 (2), 464-470. https://doi.org/10.1175/1520-0485(1984)014<0464:TNPAGS>2.0.CO;2
- Stroeve, J. and D. Notz, (2018). Changing state of Arctic sea ice across all seasons. Environmental Research 28 Letters, 13 (10), 103001, doi:10.1088/1748-9326/aade56. 29
- Stroeve, J. C. et al., (2012). The Arctic's rapidly shrinking sea ice cover: a research synthesis. Climatic Change, 110 (3), 1005-1027, doi:10.1007/s10584-011-0101-1.
- Stroeve, J. C. et al., (2014). Changes in Arctic melt season and implications for sea ice loss. Geophysical Research Letters, 41 (4), 1216-1225, doi:10.1002/2013gl058951
- Stroeve, J., Kattsov, V., Barrett, A., Serreze, M., Pavlova, T., Holland, M., & Meier, W. N. (2012). Trends in Arctic sea ice extent from CMIP5, CMIP3 and observations. Geophysical Research Letters, 39, L16502. https://doi.org/10.1029/2012GL052676
- Suydam, R., George, J. C., Rosa, C., Person, B., Hanns, C., Sheffield, G., & Bacon, J. (2007). Subsistence harvest of bowhead whales (Balaena mysticetus) by Alaskan Eskimos during 2006. SC/59/BRG4. Paper presented at 59th International Whaling Commission meeting, Department of Wildlife Management, North Slope Borough, Anchorage, AK.

- Tetzlaff, A., Lüpkes, C., Birnbaum, G., Hartmann, J., Nygård, T., and Vihma, T.: Brief Communication: Trends in sea ice extent north of Svalbard and its impact on cold air outbreaks as observed in spring 2013, The Cryosphere, 8, 1757–1762, https://doi.org/10.5194/tc-8-1757-2014, 2014
- Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Platov, G., Toole, J., Williams, W., Kikuchi, T., & Nishino, S. (2014). Mechanisms of Pacific summer water variability in the Arctic's Central Canada Basin. Journal of Geophysical Research: Oceans, 119 (11), 7523– 7548. https://doi.org/10.1002/2014JC010273
- Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Science Advances, 4 (8), 6773. https://doi.org/10.1126/sciadv.aat6773
- Tsukada, Y., Ueno, H., Ohta, N., Itoh, M., Watanabe, E., Kikuchi, T., Nishino, S., & Mizobata, K. (2018). Interannual variation in solar heating in the Chukchi Sea, Arctic Ocean. Polar Science, 17, 33–39. https://doi.org/10.1016/j.polar.2018.06.003
- Van Vorhees, D., & Lowther, A. (2010). Fisheries of the United States 2009: Current Fishery Statistics No. 2009. Silver Spring, MD: National Marine Fisheries Service. Retrieved from. https://www.st.nmfs.noaa.gov/st1/fus/fus09/fus_2009.pdf
- Walsh, J. J., McRoy, C. P., Coachman, L. K., Goering, J. J., Nihoul, J. J., Whitledge, T. E., Blackburn, T. H., Parker, P. L., Wirick, C. D., Shuert, P. G., Grebmeier, J. M., Springer, A. M., Tripp, R. D., Hansell, D. A., Djenidi, S., Deleersnijder, E., Henriksen, K., Lund, B. A., Andersen, P., Müller-Karger, F. E., & Dean, K. (1989). Carbon and nitrogen cycling within the Bering/Chukchi Seas: Source regions for organic matter effecting AOU demands of the Arctic Ocean. Progress in Oceanography, 22 (4), 277-359. https://doi.org/10.1016/0079-6611(89)90006-2
- Wang, Q., Wekerle, C., Danilov, S., Koldunov, N., Sidorenko, D., Sein, D., Rabe, B., & Jung, T. (2018). Arctic sea ice decline significantly contributed to the unprecedented liquid freshwater accumulation in the Beaufort Gyre of the Arctic Ocean. Geophysical Research Letters, 45, 4956–4964. https://doi.org/10.1029/2018GL077901
- Watanabe, E., Onodera, J., Itoh, M., Nishino, S., & Kikuchi, T. (2017). Winter transport of subsurface warm water toward the Arctic Chukchi Borderland. Deep-Sea Research, 128, 115-130. https://doi.org/10.1016/j.dsr.2017.08.009
- Williams, W. J., Shroyer, E., Kinney, J. C., Itoh, M., & Maslowski, W. (2014). Shelf-break exchange in the Bering, Chukchi and Beaufort Seas. In J. M. Grebmeier & W. Maslowski (Eds.), The Pacific Arctic Region, Ecosystem status and trends in a rapidly changing environment (pp.133-165). Dordrecht, The Netherlands: Springer Science+Business Media. https://doi.org/10.1007/978-94-017-8863-2_1
- Woodgate, R. A. (2018). Increases in the Pacific inflow to the Arctic from 1990 to 2015, and insights into seasonal trends and driving mechanisms from year-round Bering Strait mooring data. Progress in Oceanography, 160, 124-154. https://doi.org/10.1016/j.pocean.2017.12.007
- Woodgate, R. A., Stafford, K. M., & Prahl, F. G. (2015). A synthesis of year-round interdisciplinary mooring measurements in the Bering Strait (1990–2014) and the RUSALCA years (2004–2011). Oceanography, 28(3), 46–67. https://doi.org/10.5670/oceanog.2015.57
- Woodgate, R. A., Weingartner, T. J., & Lindsay, R. (2012). Observed increases in Bering Strait oceanic fluxes from the Pacific to the Arctic from 2001 to 2011 and their impacts on the Arctic Ocean water column. Geophysical Research Letters, 39, L24603. https://doi.org/10.1029/2012GL054092
- Woodgate, R. A., Weingartner, T., & Lindsay, R. (2010). The 2007 Bering Strait oceanic heat flux and anomalous Arctic sea-ice retreat. Geophysical Research Letters, 37, L01602. https://doi.org/10.1029/2009GL041621
- Yun, M. S., Whitledge, T. E., Stockwell, D., Son, S. H., Lee, J. H., Park, J. W., Lee, D. B., Park, J., & Lee, S. H. (2016). Primary production in the Chukchi Sea with potential effects of freshwater content. Biogeosciences, 13, 737–749. https://doi.org/10.5194/bg-13-737-2016
- Zhong, W., Steele, M., Zhang, J., & Cole, S. T. (2019). Circulation of Pacific Winter Water in the western Arctic Ocean. Journal of Geophysical Research: Oceans, 124, 863-881. https://doi.org/10.1029/2018JC014604

Zhuang, Y., Jin, H., Chen, J., Li, H., Ji, Z., Bai, Y., & Zhang, T. (2018). Nutrient and phytoplankton dynamics driven by the Beaufort Gyre in the western Arctic Ocean during the period 2008–2014. Deep Sea Research Part I: Oceanographic Research Papers, 137, 30–37. <u>https://doi.org/10.1016/j.dsr.2018.05.002</u>

4 ToR c. Continue to examine effects of climate change on the CAO ecosystem

4.1 Arctic microalgae

Contributers: Cecilie von Quillfeldt, Haakon Hop

4.1.1 General introduction

Arctic microalgae can be divided by their habitat; phytoplankton live suspended in the upper layer of the water column, while ice algae live attached to ice crystals, in the interstitial water between crystals, or associated with the underside of the ice. These two types of algae are composed of the same taxonomic groups with considerable overlap in species composition, depending on the area in question (Poulin et al. 2011, Bluhm et al. 2017, Lovejoy et al. 2017, Hop et al. 2020).

The high variability in the number of single-celled algae documented across the Arctic can be related to sampling effort in time and space. On the other hand, knowledge of algal biodiversity is increasing as a result of improved sampling techniques, advanced microscopic and molecular methods, etc. (Poulin et al. 2011, Daniëls et al. 2013), as well as increased sampling in the central basins (e.g. Melnikov 1997, Katsuki et al. 2009, Joo et al. 2012, Tonkes 2012). Close to 1276 sympagic algae and other protists associated with sea ice have been recorded (Bluhm et al. 2017) along with 2241 phytoplankton taxa (Lovejoy et al. 2017), dominated by large diatom and dino-flagellate cells that are relatively easy to identify through light microscopy.

While productivity estimates are critically important for understanding food web dynamics, knowledge of which species contribute to production and their size, function, phenology etc., is equally important (Bluhm et al. 2015).

A suite of environmental variables (e.g., sea ice, nutrients, light, water stratification, salinity, temperature) determine abundance, biomass, primary production and taxonomic composition through time (Poulin et al. 2011), but there appear to be ongoing climate-mediated changes in these parameters (von Quillfeldt and Smith 2016) which may have interacting effects on the food web of the CAO. The major changes identified for the ice-associated ecosystem are with regard to production timing and abundance, and biomass of ice flora and fauna, both of which are related to regional changes in sea-ice conditions (Barber et al. 2015). Some examples of reported possible changes in the CAO and adjacent seas are described below. How likely these changes are, and whether they are already ongoing in the CAO, along with possible consequences for the ecosystem, should be further investigated in the second report from WGICA.

4.1.2 Possible effects of climate change

Sea ice

Changes in sea ice (increased amounts of annual sea ice relative to multi-year ice, changes in extent and distribution, etc.) as a result climate change may have substantial effects on primary producers. Sea-ice can be an important driver of the taxonomic structure of protist communities in under-ice and epipelagic habitats of the central Arctic Ocean (Flores et al. 2019).

L
In general, increased amounts of **first year ice (FYI)** relative to **multi-year ice (MYI)** will influence species composition in an area (Poulin et al. 2011, Bluhm et al. 2017, Hop et al. 2020). MYI is a unique habitat for sea ice protists, particularly diatoms (Poulin et al. 2011, Hardge et al. 2017). The decline in MYI in recent decades has already resulted in a decrease in sea-ice protist diversity in the Arctic Ocean (Hop et al. 2020). Furthermore, changing freeze-up scenarios, as currently witnessed in the Central Arctic, may result in long-term changes in the biodiversity of sea-ice protists, as cryopelagic species and pelagic species of both Atlantic/Pacific and Arctic origin are likely to increase in relative importance under the new Arctic sea-ice regime (Hop et al. 2020). Further,, Hardge et al. (2017) suggest that continued reduction of sea ice extent, and particularly of MYI, will likely lead to diminished protist exchange (e.g. between melt pond water, sea ice and under-ice water and deep-chlorophyll maximum water), which could reduce species diversity in all habitats of the Central Arctic Ocean.

A greater proportion of FYI may offer better growth conditions for ice algae during the shorter ice-covered period due to higher initial nutrient load and brine volume (Leu et al. 2015), but longer seasons of open water, particularly on Arctic shelves, have shifted production from ice algae to phytoplankton in water masses (Wassmann and Reigstad 2011, Barber et al. 2015) and moved the open-water production northwards (Renaut et al. 2018).

Decreasing sea ice extent and thickness may also influence timing of both ice-associated and pelagic algal blooms important for successful recruitment of pelagic grazer species (Søreide et al. 2010, Wold et al. 2011, Daase et al. 2013, Leu et al. 2015). Ecologically important species of the central Arctic Ocean rely significantly on carbon synthesized by ice algae (Brown et al. 2017). Due to the close connectivity between sea ice and the pelagic food web, changes in sea ice coverage and ice algal production will likely have important consequences for food web functioning and carbon dynamics of the pelagic system (Kohlbach et al. 2016; Flores et al. 2019).

Light

Altered light conditions (due to changes in thickness and structure of sea ice, the amount of snow on top of the ice, thawing/freezing, and the increasing length of the open water season), will affect primary producers in the ice and the water column. For example, thinner ice with relatively more snow results in a distinct algal community at the snow-ice interface when the ice becomes flooded (McMinn and Hegseth 2004, Fernández-Méndez et al. 2018). This algal community had previously been only rarely observed in the Arctic (Horner et al. 1992, von Quillfeldt et al. 2009). On the other hand, the sub-ice community, which used to have the highest biomass in several areas, is less frequent (Eamer et al. 2013, Hop et al. 2020) but can still be observed within the Arctic Basin (Boetius et al. 2013, Hop et al. 2020). However, Boetius et al. (2013) found widespread deposition of ice algae on the ocean floor of the Amundsen Basin at depths of about 4 km in the record low-ice summer of 2012.

The impact of declining sea ice on Arctic sympagic production is large and complex, with consequent trophic and phenological cascades expected in the rest of the food web (Tedesco et al. 2019). Thinning sea ice is a factor affecting sympagic algae, and bottom communities develop earlier in the season because light penetration increases with decreasing ice thickness (Van

L

Leeuwe et al. 2018). Enhanced light availability may allow bottom sea-ice communities to reach higher biomass, though over a shorter period, as nutrients will be consumed more rapidly (Leu et al. 2015).

Overall, sea-ice communities show a high capacity for photoacclimation, but low maximum productivity compared to pelagic phytoplankton (Van Leeuwe et al. 2018). Species-specific mechanisms in coping with high light and high *p*CO₂ (which may reflect ecological niches) could therefore potentially alter the balance between sympagic and pelagic primary production in a future Arctic (Kvernvik et al. 2020). Furthermore, thinner ice and more leads increase light transmission leading to under-ice blooms of phytoplankton (Arrigo et al. 2012, Leu et al. 2015, Assmy et al. 2017). Earlier thawing and later freeze-up also results in a longer growth season for phytoplankton and may result in increased total primary production assuming other requirements (e.g. nutrients and stability) are in place (Arrigo and van Dijken 2015), and at the same time change the relative importance of ice algae and phytoplankton for total primary production in an area (Melnikov 2018). There are also likely to be regional differences in the extent to which primary production in the CAO is affected by climate change (Barber et al 2015). On a pan-Arctic scale, modeling studies indicate both decrease (Dupont 2012, Duarte et al. 2017) and increase in ice algal production (Matrai and Apollonio 2013).

Temperature and salinity

Warming and freshening of the ocean's surface layer due to increased temperature, freshwater runoff (and possibly more precipitation as rain in the ice-free season) may lead to increased stratification of the upper water column, which is likely to favour small species, (e.g. flagellates over diatoms), because they are more effective in acquiring nutrients and less susceptible to gravitational settling than larger species (Lie et al. 2009, Tremblay et al. 2012). Lower and less frequent downward export of biogenic particulate silicate (bPSi) has been observed during a warm anomaly period, also reflecting a shift in phytoplankton community composition towards dominance of small-sized phytoplankton under warmer conditions, but sea ice is a key factor influencing the frequency of export (Lalande et al. 2013). Small species have already become increasingly important in some areas of the Arctic (Sagen and Dalpadado 2004, Tremblay et al. 2012). On the other hand, some small species are specifically adapted to cold water and are likely to be vulnerable to surface-layer warming (Terrado et al. 2012). It has also been predicted that northward movement of Atlantic waters may replace cold-water phytoplankton with more temperate species (Hegseth and Sundfjord 2008, Harrison et al. 2013, Oziel et al. 2020). The effect of this for the CAO is uncertain, but a shift in phytoplankton community structure will have major consequences for carbon cycling and trophic transfer (Wassmann and Reigstad 2011).

Nutrients

Changes in nutrient supply (through mixing/importation/regeneration) will affect primary producers in the CAO. Nutrient concentrations in the upper polar mixed layer are relatively low due to the strong salinity stratification, which limits vertical mixing and upward transport of nutrients from deeper layers with higher nutrient concentrations (Tremblay et al. 2015). In general, the nutrient concentrations in the upper layer are higher in the Eurasian Basin than in the Canadian Basin (Codispoti et al. 2013, Bluhm et al. 2015), and generally higher on shelves than in the CAO (Randelhoff et al. 2020). Increased freshening and warming of the surface ocean might amplify the permanent halocline and favor a regenerating community (Li et al. 2009).

The winter silicate concentration has declined during the last 20 years in the Atlantic water of the Norwegian and Barents Seas (Rey 2012). Such a decline can have significant consequences for ecosystems (e.g., changes in the spring bloom phytoplankton community due to a decrease in diatom biomass). How a reduced supply of silicate by ocean currents could affect the CAO is unknown, but there are indications that ocean currents are more important for nutrient supply than supply of nutrients from rivers (Holmes et al. 2012).

Concentrations of nutrients have implications for community structure and related primary production. Flores et al. (2019) showed that low-nutrient concentrations in the epipelagic and under-ice habitat of the Arctic Ocean were associated with increasing heterotrophy.

Deeper mixing of the upper ocean, due to less sea ice and an increase in atmospheric low-pressure systems that generate stronger winds, may result in extra nutrient supply supporting primary production if there still is a satisfactory light regime. However, also deeper mixing is likely to favor smaller species (Sakshaug 2004).

References

- Arrigo, K.R., Perovich, D.K., Pickart, R.S., Brown, Z.W., Van Dijken, G.L., Lowry, K.E., et al. 2012. Massive phytoplankton blooms under Arctic sea ice. Science 336, 1408.
- Assmy, P., Fernández-Méndez, M., Duarte, P., Meyer, A., Randelhoff, A., Mundy, C. J., et al. 2017. Leads in Arctic pack ice enable early phytoplankton blooms below snow-covered sea ice. Sci. Rep. 7, 40850. doi:10.1038/srep40850.
- Barber, D.G., Hop, H., Mundy, C.J., Else, B., Dmitrenko, I.A, Tremblay, J.-É., et al., 2015. Selected physical, biological and biogeochemical implications of a rapidly changing Arctic Marginal Ice Zone. Progr. Oceanogr. 139, 122-150.
- Bluhm, B.A., Hop, H., Melnikov, I.A., Poulin, M., Vihtakari, M., Collins, E. Gradinger, R., Juul-Pedersen, T., von Quillfeldt, C., 2017. Sea-ice biota. Pp. 33-62 in:. State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland. 978-9935-431-63-9.
- Bluhm, B.A., Kosobokova, K.N., Carmack, E.C., 2015. A tale of two basins: An integrated physical and biological perspective of the deep Arctic Ocean. Progr. Oceanogr. 139, 89-121.
- Boetius, A., Albrecht, S., Bakker, K., Bienhold, C., Felden, J., Fernández-Méndez, M., et al., 2013. Export of algal biomass from the melting sea ice. Science 339, 1430-1432.
- Brown, T.A., Assmy, P., Hop, H., Wold, A, Belt, S.T. 2017. Transfer of ice algae carbon to ice-associated amphipods in the high-Arctic pack ice environment. J. Plankton Res. 39, 664-674.
- CAFF 2017. State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland. 978-9935-431-63-9.
- Codispoti, L.A., Kelly, V., Thessen, A., Matrai, P., Suttles, S., Hill, V., Steele, M., Light, B., 2013. Synthesis of primary production in the Arctic Ocean: III. Nitrate and phosphate based estimates of net community production. Progr. Oceanogr. 110, 126–150.
- Daase, M., Falk-Petersen S., Varpe, Ø., Darnis, G., Søreide, J.E., Wold, A., Leu, E., Berge J., Philippe, B., Fortier, L., 2013. Timing of reproductive events in the marine copepod *Calanus glacialis*: a pan-Arctic perspective. Can. J. Fish. Aquat. Sci. 70, 871-884.

- Daniëls, F.J.A., Gillespie, L.J., Poulin, M., Afonina, O.M., Alsos, I.G., Bültmann, H., Ickert-Bond, S., Konstantinova, N.A., Lovejoy, C., Väre, H., Westergaard, K.B., 2013. Chapter 9. Plants, in: Meltofte, H. (ed.) Arctic Biodiversity Assessment. Status and trends in Arctic biodiversity. Conservation of Arctic Flora and Fauna, 310-353.
- Duarte, P., Meyer, A., Olsen, L.M., Kauko, H. M., Assmy, P., Rösel, A., et al., 2017. Sea ice thermohaline dynamics and biogeochemistry in the Arctic Ocean: Empirical and model results. J. Geophys. Res. Biogeosci. 7, 1632-1654.
- Dupont, F., 2012. Impact of sea-ice biology on overall primary production in a biophysical model of the pan-Arctic. J. Geophys. Res. Oceans 117, C00D17. doi: 10.1029/2011JC006983.
- Eamer, J., Donaldson, G.M., Gaston, A.J., Kosobokova, K.N., Larusson, K.F., Melnikov, I.A., Reist, J.D., Richardson, E., Staples, L., von Quillfeldt, C.H. 2013. Life Linked to Ice: A guide to sea-ice associated biodiversity in this time of rapid change. CAFF Assessment Series No. 10. Conservation of Arctic Flora and Fauna, Iceland.
- Fernández-Méndez, M., Olsen, L.M., Kauko, H.M., Meyer, A., Rösel, A., Merkouriadi, I., et al. 2018. Algal hot spots in a changing Arctic Ocean: Sea-ice ridges and the snow-ice interface. Front. Mar. Sci. 5:75. doi: 10.3389/fmars.2018.00075.
- Flores, H., David, C., Ehrlich, J., Hardge, K., Kohlbach, D., Lange, B.A., et al., 2019. Sea-ice properties and nutrient concentration as drivers of the taxonomic and trophic structure of high-Arctic protist and metazoan communities. Polar Biol. 42, 1377-1395.
- Hardge, K., Peeken, I., Neuhaus, S., Lange, B.A., Stock, A., Stoeck, T., et al., 2017. The importance of sea ice for exchange of habitat-specific protist communities in the Central Arctic Ocean. J. Mar. Syst. 165, 124-138. doi: 10.1016/j.jmarsys.2016.10.004.
- Harrison, W.G., Børsheim K.Y., Li, W.K.W., Maillet, G.L., Pepin, P., Sakshaug, E., Skogen, M., Yeats, P.A., 2013. Phytoplankton production and growth regulation in the Subarctic North Atlantic: A Comparative study of the Labrador Sea-Labrador/Newfoundland shelves and Barents/Norwegian/Greenland seas and shelves. Progr. Oceanogr. 114, 26-45.
- Hegseth, E.N., Sundfjord A., 2008. Intrusion and blooming of Atlantic phytoplankton species in the high Arctic. J. Mar. Syst. 74, 108-119.
- Hop, H., Vihtakari, M., Bluhm, B.A., Assmy, P., Poulin, M., Gradinger, R., Peeken, I., von Quillfeldt, C.H., Olsen, L.M., Zhitina, L., Melnikov I.A., 2020. Changes in sea-ice protist diversity with declining sea ice in the Arctic Ocean from the 1980s to 2010s. Front. Mar. Sci. 7, 243. doi: 10.3389/mars2020.00243.
- Horner, R., Ackley, S.F., Dieckmann, G.S., Gulliksen, B., Hoshiai, T., Legendre, L., Melnikov, I.A., Reeburgh, W.S., Spindler, M. and Sullivan, C.W., 1992. Ecology of sea ice biota. 1. Habitat, terminology, and methodology. Polar Biol. 12, 417-427.
- Joo, H.M., Lee, S.H., Jung, S.W., Dahms, H.-U., Lee, J.H., 2012. Latitudinal variation of phytoplankton communities in the western Arctic Ocean. Deep-Sea Res. II 81-84, 3-17.
- Katsuki, K., Takahashi, K., Onodera, J., Jordan, R.W., Suto, I., 2009. Living diatoms in the vicinity of the North Pole, summer 2004. Micropalentology 55, 137-170.
- Kohlbach, D., Graeve, M., Lange, B.A., David, C., Peeken, I., Flores, H., 2016. The importance of ice algaeproduced carbon in the central Arctic Ocean ecosystem: food web relationships revealed by lipid and stable isotope analyses. Limiol. Oceanogr. 61, 2027-2044.
- Kvernvik, A.C., Rokitta, S.D., Leu, E., Harms, L., Gabrielsen, T.M., Rost, B., Hoppe, C.J.M., 2020. Higher sensitivity towards light stress and ocean acidification in an Arctic sea-ice-associated diatom compared to a pelagic diatom. New Phycologist, doi: 10.1111/nph.16501.
- Lalande, C., Bauerfeind, E., Nöthig, E.-M., Beszczynska-Möller, A., 2013. Impact of a warm anomaly on export fluxes of biogenic matter in the eastern Fram Strait. Progr. Oceanogr. 109, 70-77.
- Leu, E., Mundy, C.J., Assmy, P., Campbell, K., Gabrielsen, T.M., Gosselin, M., et al., 2015. Arctic spring awakening – Steering principles behind the phenology of vernal ice algal blooms. Progr. Oceanogr. 139, 151-170.

- Li, W.K.W., McLaughlin, F.A., Lovejoy, C., Carmack, E.C., 2009. Smallest algae thrive as the Arctic Ocean freshens. Science 326, 539-539.
- Lovejoy, C., von Quillfeldt, C.H., Hopcroft, R., Poulin, M., Thaler, M., Arendt, K., Debes, H., Gíslason, Á., Kosobokova K., 2017. Plankton Pp. 63-84 in CAFF. 2017. State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland. 978-9935-431-63-9.
- Matrai, P., Apollonio, S., 2013. New estimates of microalgae production based upon nitrate reductions under sea ice in Canadian shelf seas and the Canada Basin of the Arctic Ocean. Mar. Biol. 160, 1297-1309.
- McMinn, A., Hegseth, E. N. 2004. Quantum yield and photosynthetic parameters of marine microalgae from the southern Arctic Ocean, Svalbard. J. Mar. Biol. Assoc. UK 84, 865-871.
- Melnikov, I.A., 1997. The Arctic sea ice ecosystem. Gordon and Branch Science Publisher, Amsterdam.
- Melnikov, I.A., 2018. Characterization of the biodiversity of modern sea ice in the North Pole region. Doklady Akad. Nauk 480, 596-600. Translated from Russian. doi: 10.1134/S1028334X18060132
- Oziel, L., Baudena, A., Ardyna, M., Massicotte, P, Randelhoff, A., Sallée, J. -B., Ingvaldsen, R. B., Devred, E., Babin, M., 2020. Faster Atlantic currents drive poleward expansion of temperate phytoplankton in the Arctic Ocean. Nat. Comm. 11, 1705. doi: 10.1038/s41467-020-15485-5.
- Poulin, M., Daugbjerg, N., Gradinger, R., Ilyash, L., Ratkova, T., von Quillfeldt, C.H., 2011. The pan-Arctic biodiversity of marine pelagic and sea-ice unicellular eukaryotes: A first-attempt assessment. Mar. Biodiv. 41, 13–28.
- Randelhoff A., Holding J., Janout M., Sejr M.K., Babin M., Tremblay J.-É., Alkire M.B., 2020. Pan-Arctic Ocean primary production constrained by turbulent nitrate fluxes. Front. Mar. Sci. 7:150. doi: 10.3389/fmars.2020.00150.
- Renaut, S., Devred, E., Babin, M., 2018. Northward expansion and intensification of phytoplankton growth during the early ice-free season in Arctic. Geophys. Res. Lett. 45, 10590-10598.
- Rey, F., 2012. Declining silicate concentrations in the Norwegian and the Barents Seas. ICES J.Mar. Sci. 69, 208-212.
- Sagen H., Dalpadado P., 2004. *Emiliania huxleyi*-oppblomstringen i Barentshavet sommeren 2003 observert ved hjelp av satellitt. Fisken og havet, Særnummer 2, 96-97.
- Sakshaug E., 2004. Primary and secondary production in the Arctic Sea, in Stein, R., Macdonald, R.W. (Eds): The organic carbon cycle in the Arctic Ocean. Springer, Berlin, pp. 57-81.
- Skjoldal et al. 2020. WGICA report no 1. ICES.
- Søreide, J.E., Leu, E., Berge, J., Graeve, M., and Falk-Petersen, S., 2010. Timing in blooms, algal food quality and *Calanus glacialis* reproduction and growth in a changing Arctic. Glob. Change Biol. 16, 3154-3163.
- Tedesco, L., Vichi, M., Scoccimarro, E., 2019. Sea-ice algal phenology in a warmer Arctic. Sci. Adv. 5, eaav4830. doi: 10.1126/sciadv.aav 4830.
- Terrado, R., Scarcella, K., Thaler, M., Vincent, W.F., Lovejoy, C., 2012. Small phytoplankton in Arctic seas: vulnerability to climate change. Biodiversity doi:10.1080/1488386.2012.704839.
- Tonkes, H., 2012. Phytoplankton composition of central Arctic Ocean in summer 2011: with special emphasis on pico- and nanoplankton. Major thesis, Wageningen University and Research.
- Tremblay, J.-E., Robert, D., Varela, D., Lovejoy, C., Darnis, G., Nelson, R.J., Sastri, A., 2012. Current state and trends in Canadian Arctic marine ecosystems: I Primary production. Clim. Change, doi: 10.1007/s10584-012-0496-3.
- Tremblay, J.-E., Anderson, L. G., Matrai, P. M., Couple, P., Bélanger, S., Michel, C., Reigstad, M., 2015. Global and regional drivers of nutrient supply, primary production and CO₂ drawdown in the changing Arctic Ocean. Progr. Oceanogr. 139, 171-196, doi.org/10.1016/j.pocean.2015.08.009.

- Van Leeuwe, M. A., Tedesco, L., Arrigo, K. R., Assmy, P., Campbell, K., Meiners, K. M., et al., 2018. Microalgal community structure and primary production in Arctic and Antarctic sea ice: A synthesis. Elem. Sci. Anth. 6, 4. doi: 10.1525/elamenta.267.
- von Quillfeldt, C.H., Hegseth, E.N., Johnsen G., Sakshaug, E. & Syvertsen, E.E. 2009. Ice algae. Pp. 285-302 in Sakshaug, E., Johnsen, G. & Kovacs K. (eds) Ecosystem Barents Sea. Tapir Academic Press, Trondheim, Norway.
- von Quillfeldt, C.H. and Smith, W. 2016. Primary producers. In: Jørgensen, L.L, Archambault, P., Armstrong, C., Dolgov, A., Edinger, E., Gaston, T., Hildebrand, J., Piepenburg, D., Smith, W., von Quillfeldt, C., Vecchione, M., Rice, J. (2016) Chapter 36 G. Arctic Ocean. In Inniss, L., Simcock, A. (Joint Coordinators) The First Global Integrated Marine Assessment. World Ocean Assessment I. United nations. www.un.org/Depts/los/woa
- Wassmann, P., Reigstad, M. 2011. Future Arctic Ocean seasonal ice zones and implications for pelagic-benthic coupling. Oceanogr. 24, 220-231.
- Wold, A., Darnis, G., Søreide, J.E., Leu, E., Philippe, B., Fortier, L., Poulin, M., Kattner, G., Graeve, M., Falk-Petersen, S., 2011. Life strategy and diet of *Calanus glacialis* during the winter-spring transition in Amundsen Gulf, south-eastern Beaufort Sea. Polar Biol. 34, 1929–1946. http://dx.doi.org/10.1007/s00300-011-1062-6.

4.2 Invertebrate ice fauna

Contributor; Haakon Hop, Norwegian Polar Institute.

Arctic sea ice is habitat for sympagic or ice-associated organisms (Horner et al. 1992; Melnikov, 1997). The diversity of these organisms is largely dependent on the type and age of the ice (Hop et al. 2000; 2011). Habitat complexity ranges from a flat and uniform under-surface in newly formed landfast ice, (via relatively flat areas with brine channels in FYI), to three-dimensional and often very complex habitat in older MYI and pressure ridges of Arctic pack-ice (Hop et al., 2000). Multiyear sea ice provides a more stable and complex environment and therefore often harbors a more diverse community with higher abundance and biomasses (Arndt and Lønne, 2002). In FYI, ridges represent the most important structural habitat for ice-associated fauna, particularly during advanced ice melt (Gradinger et al. 2010).

4.2.1 Sea ice meiofauna

Sea-ice fauna assemblages live within or attached to the ice matrix and consist of bacteria and microbes, the meiofauna community (heterotrophs >20 μ m) and ice macrofauna community (macroinvertebrates >0.5 mm) (Melnikov 1997). Ice sympagic meiofauna comprises multicellular organisms such as nematodes, harpacticoid copepods and rotifers, typically ranging from 20 to 500 μ m in size (Gradinger and Ikävalko 1998; Gradinger et al. 1999; 2005). Single-celled ciliates are also included as meiofauna in some studies, but in others these are referred to as microfauna (<62 μ m; Carey 1985; Bluhm et al. 2017b). In addition to ice-endemic species, both pelagic and benthic meiofaunal species occur in sea ice, often as larvae or juvenile stages. Meiofaunal species or stages settle in sea ice through active migration, are picked up from the water-column during ice formation, disperse from multiyear ice, or are recruited from resting stages (Carey and Montagna 1982).

Besides adding to the biodiversity of the Arctic, many ice meiofauna taxa graze on the seasonally abundant and highly concentrated ice algae (Grainger and Hsiao 1990) that allow for high meiofaunal growth rates early in the season before the phytoplankton bloom develops (McConnell et al. 2012).

Meiofauna grazing does not appear to limit ice algal growth despite seasonally high meiofaunal abundances, (often with >200,000 meiofaunal individuals m-2 in landfast ice at peak times (Gradinger et al. 2009; Bluhm et al. 2017a)). The ingestion rates by multicellular meiofauna are estimated at <10 % of ice algal biomass (Gradinger et al. 1999; Michel et al. 2002). Such low grazing pressure may in part be related to the size of the brine channel system, mostly <1 mm in a given channel of solid ice, which may restrict some meiofauna from exploiting niches with high ice algal growth (Krembs et al. 2000). This restriction, however, rapidly changes during the onset of melt when the porosity increases above the percolation threshold of about 5 % brine volume fraction (Golden et al. 1998), and the brine channels connect so that organisms in the ice can move about.

Direct absorption of ice-produced dissolved organic matter has been suggested as an alternative feeding mode for sympagic nematodes (Tchesunov and Riemann 1995). Meiofaunal predators, in turn, appear to be rare (Bluhm et al. 2017b), perhaps due to space limitations within the brine channels and prey patchiness. Exploiting the sea ice as nursery ground can, therefore, be a winning life strategy for those organisms capable of living in this extreme environment due to the high concentrations of food and low predation pressure. During melt, ice-derived organic matter including ice meiofauna is released from the sea ice and provides food for pelagic and benthic biota (Moran et al. 2012).

Sea ice extent in the Arctic Ocean has declined by over 30% since the satellite record began in 1979 (Vaughan et al., 2013; Meier et al. 2014) and the reduction in sea ice has occurred during all months of the year (Barber et al. 2015). The average age of individual Arctic ice floes has decreased, changing from multiyear to mainly first- and second-year sea ice, and the thickness of Arctic sea ice has decreased by about 0.5 m from the 1980s to 2010s (Perovich et al. 2015; Bi et al. 2016). Multiyear sea ice, which covered about 75% of the Arctic in 1983, is currently limited to the areas north of Greenland, the central Arctic, and parts of the Canadian Arctic Archipelago, with as little as 62% of the summer ice cover remaining compared to 1978-1988 values (Maslanik et al. 2007; Stroeve et al. 2012; Meier et al. 2014). First-year ice has concomitantly increased in relative proportion and importance, although dates of its freeze-up and break-up have also shifted substantially (Markus et al. 2009), leaving large parts of the Arctic without sea ice for an increasing amount of time (Arrigo et al. 2011). Consequences for sea ice biota – from bacteria to polar bears – seem inevitable but are largely undocumented (CAFF 2017). Documentation of ice biota composition, abundance and natural variability is critical for evaluating responses to the decline in Arctic sea-ice.

Sea ice meiofauna has been known to occur for over a century (Nansen 1906), and its biodiversity has been documented at various levels (Poulin et al. 2011; Bluhm et al. 2017a, b). Distribution and composition of sea ice meiofauna have been reported from some locations in the Arctic, for example the Canada Basin (Gradinger et al. 2010), Beaufort Sea (Marquardt et al. 2011), Frobisher Bay in Arctic Canada (Grainger et al. 1985; Nozais et al. 2001), Hudson Bay (Grainger, 1988), and Fram Strait (Schünemann and Werner 2005). These studies documented both similar community patterns of ice meiofauna, such as likeness of taxa found at the class and phylum levels and concentration at near-bottom layers of the sea ice, as well as different levels of abundance and dominant taxa. The identified patterns and variations have locally primarily been related to different ice types (landfast, first-year and multiyear ice), seasons, hydrographic conditions, snow cover and sediment load (see Bluhm et al. 2017a for review). Pan-Arctic studies are rare, but a recent study by Bluhm et al. (2018) addressed taxonomic (phylum-level) composition and abundance of sea ice meiofauna at multiple spatial scales, ranging from the vertical distribution within a single ice core at a given location – to the pan-Arctic scale, and at a seasonal scale under landfast ice and pack ice conditions.

Landfast ice regions may be more important for meiofauna than the pack-ice regions. Meiofauna abundance was found to be 1-2 orders of magnitude higher in near-shore fast ice compared to MYI and FYI of both the Beaufort Gyre (median of 990 ind. m-2) and Transpolar Drift system pack-ice (11 180 ind. m-2) (Cross 1982; Kern and Carey 1983; Grainger and Hsiao 1990; Nozais et al. 2001; Gradinger et al. 2005). Meiofauna species are also largely allochthonous and may therefore be less affected by the loss of MYI, although Melnikov et al. (2002) observed a complete lack of sea-ice meiofauna, and also species poor under-ice fauna, in FYI and MYI in the Canadian Basin in October 1997- March 1998. Melnikov blamed the ongoing reduction in sea-ice extent and thickness for this striking difference from earlier studies (Melnikov 1997; Gradinger 1999). Gradinger et al. (2005), however, found a typical sea-ice meiofauna community in the same area in summer 2002. High spatial heterogeneity in sea-ice biota distribution and abundance, methodical differences and seasonal factors may explain this discrepancy.

Meiofauna abundance and ice algal biomass are positively correlated (Gradinger 1999), but the grazing rates of meiofauna are on the order of 1-6% of the ice algal production (Nozais et al. 2001; Michel et al. 2002; Gradinger et al. 2005). Similarly, low grazing rates on ice algal standing stock by ice amphipods in the Laptev and Greenland seas (3% d -1) suggest that food is not a limiting factor for ice-associated fauna (Werner 1997). Thus, changes in the onset and productivity of ice algae blooms may have little effect on ice fauna productivity.

4.2.2 Sea ice macrofauna

Sea-ice macrofauna is mainly comprised of ice amphipods, but also some polychaetes and the copepod Jaschnovia brevis (Gulliksen and Lønne 1991; Lønne and Gulliksen 1991b; Scott et al. 2002). Ice amphipods feed on ice algae, sympagic and pelagic fauna and detritus (Werner 1997; Poltermann 2001). The small gadid, polar cod (Boreogadus saida) is often associated with sea ice, where it feeds on ice-amphipods as well as pelagic amphipods and copepods (Lønne and Gulliksen 1989; Gradinger and Bluhm 2004; Hop and Gjøsæter 2013). Ice amphipods and polar cod (particularly the younger year-classes of polar cod, which can be found in drifting pack-ice) are important prey for marine mammals and seabirds feeding in the MIZ (Bradstreet and Cross 1982; Lønne and Gabrielsen 1992; Welch et al. 1992; Bluhm and Gradinger 2008).

Sea-ice fauna is comprised of authochthonous species that spend their entire life cycle associated with the sea ice, and allochthonous species that spend only part of their life in the ice and other parts in benthic or pelagic environments (Melnikov and Kulikov 1980; Gulliksen and Lønne 1991). The gammarid amphipods Apherusa glacialis, Gammarus wilkitzkii, Onisimus nanseni, O. glacialis and Gammarocanthus loricatus are regarded as true autochthonous sympagic species (Gulliksen 1984; Gulliksen and Lønne 1991; Lønne and Gulliksen 1991b). Some of these species are long-lived with life cycles up to 5-6 years and they feed on algae, fauna, and detritus derived from the ice community (Arndt and Swadling 2006; Werner et al. 2007). Gammarus wilkitzkii in particular, with life cycle duration of up to 6 years, has been associated with MYI, whereas Apherusa glacialis has a shorter life cycle (1-2 years), is more mobile and often associated with FYI (Lønne and Gulliksen 1991b; Beuchel and Lønne 2002). Ice amphipods are able to leave the ice for short excursions into the water column to relocate on neighbouring ice floes (Arndt and Lønne, 2002), but are otherwise largely dependent on the sea ice as a habitat.

The loss of sea ice for parts of the year will have a more detrimental effect on autochthonous species than on allochthonous species. Since long-lived amphipod species depend on persistent sea ice to complete their life cycles, regions where MYI is disappearing may be susceptible to loss of their populations. Observations from north of Svalbard, where MYI ice has decreased over recentdecades, indicate that there are indeed drastic changes in abundance with regard to the large G. wilkitzkii, and less for the smaller A. glacialis. Ice amphipod abundance has been variable, occasionally high (<60 ind. m-2) until the mid-1990's and then lower but variable (10-50 ind.

m-2) until 2005 (Bluhm et al. 2018). Abundance has been very low in later years (from 2010 onwards) and difficult to sample quantitatively by scuba divers (methods in Hop and Pavlova, 2008). Correspondingly, the biomass has been variable, with high values (> 6 g m-2) until 1996, and then with 1-2 g m-2 until 2005, after which it dropped to very low levels, similar to the pattern in abundance. These results indicate that the sympagic community has declined. A , similar pattern was observed in the Pacific sector by Gradinger et al. (2010). They postulated that pressure ridges in FYI might be crucial for survival during periods of enhanced summer ice melt.

If summer pack-ice disappears in the Arctic Ocean, the sympagic macrofauna will likely vanish from Arctic drift ice, although there may be mechanisms for retaining ice amphipods in the pelagic system (Berge et al. 2012) or shallow water benthic system (Poltermann 1998).Increasing open water areas will likely cause reduction in the ice-associated autochthonous macrofauna as well as their main predators (Bradstreet and Cross 1982), with the exception of local regions in the western Arctic and shallow regions of the Arctic coast where perennial ice remains.

Reduction in ice will also decrease the southward transport of sympagic biogenic material on the European side of the Arctic. The annual biomass transport of ice fauna into Fram Strait and the Barents Sea is in the million-ton range (Wassmann et al. 2006; Hop and Pavlova 2008). There they are released to the pelagic system as the ice melts (Werner et al., 1999; Arndt and Pavlova, 2005). Hop and Pavlova (2008) calculated that 80×103 t C yr-1 is released to the pelagic and benthic ecosystem in the marginal seas, with 70% through Fram Strait and 30% into the Barents Sea. The reduction in this carbon export may have repercussions for the pelagic and benthic food webs in these regions. However, in shelf seas, the carbon import by ice amphipods is very low (0.3 %) compared to primary production (Wassmann et al. 2006).

4.2.3 Needs for research and monitoring

Sea ice is a species-rich habitat, and many sea ice taxa are widespread across the Arctic. Sea ice supports some species endemic to the Arctic, some species endemic to sea ice and other taxa that are more widespread. Ice biota generally copes with extreme conditions, although little is known about the tolerance limits and preferences of individual species in this habitat.

Ice amphipod abundance and biomass have declined in the Svalbard area since the 1980s. Amphipods appear to have been more abundant in the late 1970s to mid-1990s than afterwards.

The occurrence and distribution of ice biota is highly variable in time and space related to a suite of environmental conditions. Consequently, monitoring the biota of this variable habitat is challenging.

While no monitoring program currently exists for sea ice biota, various research groups around the Arctic have regular field activities involving select sampling as part of short-term funded projects. The accumulated knowledge base can serve as a baseline for monitoring of sea ice biota. However, monitoring requires coordinated plans for sampling at set locations, with consistent sampling and analyses to ensure comparability (Gill et al. 2011). For sea ice, standardized sampling techniques have, for example, been summarized by Eicken et al. (2009, 2014) and Miller et al. (2015). Consistent methodology is required for monitoring of ice biota. Available protocols need to be more widely implemented for monitoring. Monitoring should be standardized with regard to gear, collections, timing, sample preservation and processing, storage, and data management. A central receiving place as well as long-term funding for monitoring should be considered. Data should be deposited in existing databases and made available to researchers and beyond. Proper monitoring requires seasonal and annual field campaigns as well as sufficient, long-term financial support.

Most (but not all) sea ice sampling for ice biota living within the brine channels has been done using ice cores (methods described for example in Gradinger and Bluhm 2009). The thickness of the ice sections studied typically show high variation, depending on the goals of each study, time constraints and ambient ice thickness to name a few, resulting in different thickness horizons sampled (Hop et al. 2020). Often the bottom 0-4 cm or 0-10 cm were sampled (with 0 cm representing the ice-water interface), sometimes the bottom 20-30 cm and occasionally whole cores. As minimum, the bottom 10 cm of ice cores (which can be split into the lowermost 3-4 cm bottom ice and the next 6-7 cm of the core) should be sampled, because the bottom 10 cm often includes the majority of the biomass and abundance of ice biota (~65 % of ice meiofauna, up to 95% for algae based on the here compiled data sets). The CBMP Sea Ice Biota Expert Network also recommends the 10-30 cm section be included (separately) where feasible, as an average 12% of the meiofauna have been observed in this ice section. Microbes, particularly bacteria, are distributed ubiquitously throughout the ice column and are generally sampled in 10 -cm increments using sterile procedures to avoid contaminating the ice cores (e.g., Collins et al. 2010). Ice algal communities and biomass (chlorophyll a) are concentrated in the bottom few centimeters but have in some studies been determined in 20 cm sections up to the surface of the ice (e.g., Mundy et al. 2011). The need for sampling the entire ice column will depend on project goals, which may also include modelling aspects of biota in sea ice (Duarte et al. 2015).

Regarding the choice of taxa, communities or habitats to monitor, the studies reviewed have usually focused either on a sub-habitat (e.g., ice biota inside brine channels) or on a taxonomic group (e.g., ice meiofauna), a particular method (e.g., morphological taxonomy or genetic analysis), or a combination of the three. Ice biota has rarely been studied as a whole at a given location and time period. The Russian ice-drift studies have likely been the most extensive and comprehensive studies (Melnikov 1997), and there has also been coordinated sampling of ice biota and sea ice physics at some locations in Arctic Canada and Alaska (e.g., Ehn et al. 2011, Hop et al. 2011, Mundy et al. 2011), and more recently during the N-ICE2015 campaign by the Norwegian Polar Institute to the Arctic Ocean. The under-ice amphipod Gammarus wilkitzkii due to its association with MYI may be gradually replaced by the more pelagic, but also ice-associated amphipod Eusirus holmi, which is of similar large size as adult. Both amphipod species should therefore be monitored, since changes in their relative abundance may reflect changes in sea ice conditions.

Under-ice sampling has been more variable in approach. Regularly used tools have included (1) SCUBA-operated suction pumps (Lønne 1988), (2) under-ice in situ or surface-operated still photos (Mundy et al. 2007), (3) video in a fixed location or video transects (SCUBA diver or ROV operated) (Gradinger and Bluhm 2004), (4) under-ice traps (Nygård et al. 2012) and (5) under-ice trawl nets (David et al. 2016; approaches summarized by Gradinger and Bluhm 2009). The most quantitative samples appear to be based on SCUBA-operated sampling of squares with electrical suction pump (Hop et al. 2000) and the under-ice trawl net SUIT (van Franeker et al. 2009), as well as imagery with sufficient resolution.

Themain challenges for monitoring include the relatively low number of people working on sea ice biota and the loss of taxonomic expertise for groups such as ice algae and other protists, but also ice-associated fauna (e.g., meiofauna and ice-associated zooplankton). Another great challenge in attempting to monitor sea ice biota is the large variability of the dynamic sea ice habitat. Ice biota composition and abundance are, as detailed above, highly dependent upon light availability (modulated by day length, ice and snow thickness, sediments in ice, etc.), nutrient availability, temperature, salinity and location. These factors modify the habitat seasonally, even in the same place or the same ice floe. This variability, in combination with historic variability in sampling approach and timing, obscure potential temporal trends. Thus, actual changes in ice biota diversity and community structure need to be substantial enough to rise beyond the existing variability in order to be detected.

References

- Arrigo, K.R., Matrai, P.A., & van Dijken, G.L. (2011) Primary productivity in the Arctic Ocean: Impacts of complex optical properties and subsurface chlorophyll maxima on large-scale estimates. Journal of Geophysical Research, 116, C11022.
- Arndt, C., Lønne, O.J., 2002. Transport of bioenergy by large scale Arctic ice drift. Proceedings of the 16th IAHR International Symposium on Ice, Dunedin, New Zealand. International Association of Hydraulic Engeneering and Research. pp. 103-111.
- Arndt, C.E., Pavlova, O., 2005. Origin and fate of ice fauna in the Fram Strait and Svalbard area. Marine Ecology Progress Series 301, 55-66.
- Arndt, C.E., Swadling, K.M., 2006. Crustacea in Arctic and Antarctic sea ice: Distribution, diet and life history strategies. Advances in Marine Biology 51, 197-315.
- Barber, D.G., Hop, H., Mundy, C.J., Else, B., Dmitrenko, I.A., Tremblay, J.E., Ehn, J.K., Assmy, P., Daase, M., Candlish, L.M., & Rysgaard, S. (2015) Selected physical, biological and biogeochemical implications of a rapidly changing Arctic Marginal Ice Zone. Progress in Oceanography, 139, 122–150.'
- Berge, J., Varpe, Ø., Moline, M.A., Wold, A., Renaud, P.E., Daase, M., Falk-Petersen, S., 2012. Retention of ice-associated amphipods: possible consequences for an ice-free Arctic Ocean. Biology Letters 8, 1012-1015. doi:10.1098/rsbl.2012.0517.
- Beuchel, F., Lønne, O.J., 2002. Population dynamics of the sympagic amphipods Gammarus wilkitzkii and *Apherusa glacialis* in sea-ice north of Svalbard. Polar Biology 25, 241-250.
- Bi, H., Fu, M., Sun, K., Liu, Y., Xu, X., & Huang, H. (2016) Arctic sea ice thickness changes in terms of sea ice age. Acta Oceanologica Sinica, 35, 1–10.
- Bluhm, B.A., Gradinger, R., 2008. Regional variability in food availability for arctic marine mammals. Ecological Applications 18 (2), S77-S96.
- Bluhm, B.A., Hop, H., Melnikov, I.A., Poulin, M., Vihtakari, M., Collins, E., Gradinger, R., Juul-Pedersen, T., & von Quillfeldt, C. (2017a) State of the Arctic Marine Biodiversity Report pp. 33–61. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland.
- Bluhm, B.A., Swadling, K.M., Gradinger, R. (2017b) Sea Ice (ed. by D.N. Thomas), pp. 394–414. John Wiley & Sons, Ltd, Chichester, UK.
- Bluhm, BA, Hop, H, Vihtakari, M, Gradinger, R, Iken, K, et al. 2018 Sea ice meiofauna distribution on local to pan-Arctic scales. Ecol Evol 8: 2350-2364.
- Bradstreet, M.S.W., Cross, W.E., 1982. Trophic relationships at high Arctic ice edges. Arctic 35 (1), 1-12.
- CAFF (2017) State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland.
- Carey, A.G. (1985) Sea ice biota pp. 173-190. Boca Raton, FL, USA.
- Carey, A.G, Montagna, P.A. (1982) Arctic sea ice fauna assemblage: first approach to description and source of the underice meiofauna. Marine Ecology Progress Series, 8, 1–8.
- Collins, R.E., Rocap, G. and Deming, J.W. 2010. Persistence of bacterial and archaeal communities in sea ice through an Arctic winter. Environmental Microbiology, 12(7): 1828-1841.
- Cross, W.E., 1982. Under-ice biota at the Pond Inlet ice edge and in adjacent fast ice areas during spring. Arctic 35 (1), 13-27.
- David, C., Lange, B., Krumpen, T., Schaafsma, F., van Franeker, J.A. and Flores, H. 2016. Under-ice distribution of polar cod *Boreogadus saida* in the central Arctic Ocean and their association with sea-ice habitat properties. Polar Biology, 39(6): 919-994
- Duarte, P., Assmy, P., Hop, H., Spreen, G., Gerland, S. and Hudson, S.R. 2015. The importance of vertical resolution in sea ice algae production models. Journal of Marine Systems, 145: 69-90

- Eicken, H., Bluhm, B.A., Collins, R.E., Gradinger, R.R., Haas, C., Ingham, M., Mahoney, A., Nicolaus, M. and Perovich, D. 2014. Field techniques in sea ice research. In: H. Shen (ed.). Cold Regions Science and Marine Technology in Encyclopedia of Life Support Systems (EOLSS), UNESCO, Paris, France.
- Eicken, H., Gradinger, R., Salganek, M., Shirasawa, K., Perovich, D. and Leppäranta M. (eds.). 2009. Handbook on field techniques in sea ice research (a sea ice system services approach). University of Alaska Press, Fairbanks.
- Ehn, J.K., Mundy, C.J., Barber, D.G., Hop, H., Rossnagel, A. and Stewart, J. 2011. Impact of horizontal spreading on light propagation in melt pond covered seasonal sea ice in the Canadian Arctic. Journal of Geophysical Research, 116(C9): C00G02.
- Golden, K.M., Ackley, S.F., & Lytle, V.I. (1998) The percolation phase transition in sea ice. Science, 282, 2238–2241.
- Gill, M.J., Crane, K., Hindrum, R., Arneberg, P., Bysveen, I., Denisenko, N.V., Gofman, V., Grant-Friedman, A., Guðmundsson, G., Hopcroft, R.R., Iken, K., Labansen, A., Liubina, O.S., Melnikov, I.A., Moore, S.E., Reist, J.D., Sirenko, B.I., Stow, J., Ugarte, F., Vongraven, D. and Watkins, J. 2011. Arctic Marine Biodiversity Monitoring Plan. CAFF Monitoring Series Report No. 3, Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland.
- Grainger, E.H. (1988) The influence of a river plume on the sea-ice meiofauna in South-Eastern Hudson Bay. Estuarine, Coastal and Shelf Science, 27, 131–141.
- Gradinger, R.R., 1999. Integrated abundance and biomass of sympagic meiofauna in Arctic and Antarctic pack ice. Polar Biology, 22, 169-177.
- Gradinger, R. and Bluhm, B.A. 2009. Assessment of the abundance and diversity of sea ice biota. In: H. Eicken, R. Gradinger, M. Salganek, K. Shirasawa, D. Perovich and M. Leppäranta (eds.). Handbook on field techniques in sea ice research (a sea ice system services approach). University of Alaska Press, Fairbanks: 283-300
- Gradinger, R.R., Bluhm, B.A., 2004. In-situ observations on the distribution and behavior of amphipods and Arctic cod (*Boreogadus saida*) under the sea ice of the High Arctic Canada Basin. Polar Biology 27 (10), 595-603.
- Gradinger, R., Bluhm, B., Iken, K., 2010. Arctic sea-ice ridges--Safe heavens for sea-ice fauna during periods of extreme ice melt? Deep Sea Research Part II: Topical Studies in Oceanography 57 (1-2), 86-95.
- Gradinger, R., Friedrich, C., Spindler, M., 1999. Abundance, biomass and composition of the sea-ice biota of the Greenland Sea pack ice. Deep Sea Research Part II: Topical Studies in Oceanography 46 (6-7), 1457-1472.
- Gradinger, R., Ikävalko, J., 1998. Organism incorporation into newly forming Arctic sea ice in the Greenland Sea. Journal of Plankton Research 20 (5), 871-886.
- Gradinger, R., Meiners, K., Plumley, G., Zhang, Q., Bluhm, B.A., 2005. Abundance and composition of the sea-ice meiofauna in off-shore pack ice of the Beaufort Gyre in summer 2002 and 2003. Polar Biology 28 (3), 171-181.
- Grainger, E.H., Mohammed, A.A., & Lovrity, J.E. (1985) The sea ice fauna of Frobisher Bay, Arctic Canada. Arctic, 38, 23–30.
- Grainger, E.H. & Hsiao, S.I.C. (1990) Trophic relationships of the sea ice meiofauna in Frobisher Bay, Arctic Canada. Polar Biology, 10, 283–292.
- Gulliksen, B., 1984. Under-ice fauna from Svalbard waters. Sarsia 69 (1), 17-23.
- Gulliksen, B., Lønne, O.J., 1991. Sea-ice macrofauna in the Antarctic and the Arctic. Journal of Marine Systems 2, 53-61.
- Hop, H., Gjøsæter, H., 2013. Polar cod (*Boreogadus saida*) and capelin (*Mallotus villosus*) as key species in marine food webs of the Arctic and the Barents Sea. Marine Biology Research, 9 (9), 878-894. doi:10.1080/17451000.2013.775458.

- Hop, H., Pavlova, O., 2008. Distribution and biomass transport of ice amphipods in drifting sea ice around Svalbard. Deep-Sea Research Part II-Topical Studies in Oceanography 55 (20-21), 2292-2307. doi:10.1016/j.dsr2.2008.05.023.
- Hop, H., Poltermann, M., Lønne, O.J., Falk-Petersen, S., Korsnes, R., Budgell, W.P., 2000. Ice amphipod distribution relative to ice density and under-ice topography in the northern Barents Sea. Polar Biology 23 (5), 357-367.
- Hop, H., Mundy, C.J., Gosselin, M., Rossnagel, A.L., Barber, D.G., 2011. Zooplankton boom and ice amphipod bust below melting sea ice in the Amundsen Gulf, Arctic Canada. Polar Biology 34 (12), 1947-1958. doi:10.1007/s00300-011-0991-4.
- Hop, H., Vihtakari, M., Bluhm, B.A., Assmy, P., Poulin, M., Gradinger, R., Peeken, I., von Quillfeldt, C.H., Olsen, L.M., Zhitina, L., Melnikov I.A., 2020. Changes in sea-ice protist diversity with declining sea ice in the Arctic Ocean from the 1980s to 2010s. Front. Mar. Sci. 7, 243. doi: 10.3389/mars2020.00243.
- Horner, R., Ackley, S.F., Dieckmann, G.S., Gulliksen, B., Hoshiai, T., Legendre, L., Melnikov, I.A., Reeburgh, W.S., Spindler, M., Sullivan, C.W., 1992. Ecology of Sea ice Biota .1. Habitat, Terminology, and Methodology. Polar Biology 12 (3-4), 417-427.
- Kern, J.C., Carey, A.G., 1983. The faunal assemblage inhabiting seasonal sea ice in the nearshore Arctic ocean with emphasis on copepods. Marine Ecology Progress Series 10 (2), 159-167.
- Krembs, C., Gradinger, R., & Spindler, M. (2000) Implications of brine channel geometry and surface area for the interaction of sympagic organisms in Arctic sea ice. Journal of Experimental Marine Biology and Ecology, 243, 55–80.
- Lønne, O.J. 1988. A diver-operated electric suction sampler for sympagic (=under-ice) invertebrates. Polar Research, 6: 135-136.
- Lønne, O.J., Gabrielsen, G.W., 1992. Summer diet of seabirds feeding in sea-ice covered waters near Svalbard. Polar Biology 12, 685-692.
- Lønne, O.J., Gulliksen, B., 1989. Size, age and diet of polar cod, *Boreogadus saida* (Lepechin 1773) in ice covered waters. Polar Biology 9, 187-191.
- Lønne, O.J., Gulliksen, B., 1991b. Sympagic macro-fauna from multiyear sea ice near Svalbard. Polar Biology 11, 471-477.
- Markus, T., Stroeve, J.C., & Miller, J. (2009) Recent changes in Arctic sea ice melt onset, freezeup, and melt season length. Journal of Geophysical Research: Oceans, 114, 1–14.
- Marquardt, M., Kramer, M., Carnat, G., Werner, I. (2011) Vertical distribution of sympagic meiofauna in sea ice in the Canadian Beaufort Sea. Polar Biology, 34, 1887–1900.
- Maslanik, J.A., Fowler, C., Stroeve, J., Drobot, S., Zwally, J., Yi, D., & Emery, W. (2007) A younger, thinner Arctic ice cover: Increased potential for rapid, extensive sea-ice loss. Geophysical Research Letters, 34, L24501.
- McConnell, B., Gradinger, R., Iken, K., & Bluhm, B.A. (2012) Growth rates of arctic juvenile *Scolelepis squamata* (Polychaeta: Spionidae) isolated from Chukchi Sea fast ice. Polar Biology, 35, 1487–1494.
- Meier, W.N., Hovelsrud, G.K., van Oort, B.E.H., Key, J.R., Kovacs, K.M., Michel, C., Haas, C., Granskog, M.A., Gerland, S., Perovich, D.K., Makshtas, A., & Reist, J.D. (2014) Arctic sea ice in transformation: A review of recent observed changes and impacts on biology and human activity. Reviews of Geophysics, 52, 185–217.
- Melnikov, I.A., 1997. The Arctic sea-ice ecosystem. Gordon and Breach Science Publishers, Amsterdam.
- Melnikov, I.A., Kolosova, E.G., Welch, H.E., Zhitina, L.S., 2002. Sea-ice biological communities and nutrient dynamics in the Canada Basin of the Arctic Ocean. Deep Sea Research Part I: Oceanographic Research Papers 49, 1623-1649.
- Melnikov, I.A., Kulikov, A.S., 1980. The Cryoplegaic Fauna of the Central Arctic Basin. Biology of the Central Arctic Basin. Nauka, Science Press, Moskow, pp. 97-111.

- Michel, C., Nielsen, T.G., Nozais, C., Gosselin, M. (2002) Significance of sedimentation and grazing by ice micro- and meiofauna for carbon cycling in annual sea ice (northern Baffin Bay). Aquatic Microbial Ecology, 30, 57–68.
- Miller, L.A., Fripiat, F., Else, B.G.T., Bowman, J.S., Brown, K.A., Collins, R.E., Ewert, M., Fransson, A., Gosselin, M., Lannuzel, D., Meiners, K.M., Michel, C., Nishioka, J., Nomura, D., Papadimitriou, S., Russell, L.M., Sørensen, L.L., Thomas, D.N., Tison, J.-L., van Leeuwe, M.A., Vancoppenolle, M., Wolff, E.W. and Zhou, J. 2015. Methods for biogeochemical studies of sea ice: The state of the art, caveats, and recommendations. Elementa Science of the Anthropocene, 3(38): 000038.
- Moran, S.B., Lomas, M.W., Kelly, R.P., Gradinger, R., Iken, K., Mathis, J.T. (2012) Seasonal succession of net primary productivity, particulate organic carbon export, and autotrophic community composition in the eastern Bering Sea. Deep Sea Research Part II: Topical Studies in Oceanography, 65–70, 84–97.
- Mundy, C.J., Gosselin, M., Ehn, J.K., Belzile, C., Poulin, M., Alou, E., Roy, S., Hop, H., Lessard, S., Papakyriakou, T.N., Barber, D.G. and Stewart, J. 2011. Characteristics of two distinct high-light acclimated microbial communities during advanced stages of sea ice melt. Polar Biology, 34(12): 1869-1886.
- Nozais, C., Gosselin, M., Michel, C., Tita, G. (2001) Abundance, biomass, composition and grazing impact of the sea-ice meiofauna in the North Water, Northern Baffin Bay. Marine Ecology Progress Series, 217, 235–250.
- Nygård, H., Berge, J., Søreide, J.E., Vihtakari, M. and Falk-Petersen, S. 2012. The amphipod scavenging guild in two Arctic fjords: seasonal variations, abundance and trophic interactions. Aquatic Biology, 14: 247–264.
- Perovich, D., Meier, W.N., Tschudi, M., Farrell, S., Gerland, S., Hendricks, S., Krumpen, T., & Haas, C. (2015) Sea Ice. Available at: <u>http://www.arctic.noaa.gov/Report-Card</u>.
- Poltermann, M., 1998. Abundance, biomass and small-scale distribution of cryopelagic amphipods in the Franz Josef Land area (Arctic). Polar Biology 20 (2), 134-138.
- Poltermann, M., 2001. Arctic sea ice as feeding ground for amphipods food sources and strategies. Polar Biology 24 (2), 89-96.
- Poulin, M., Daugbjerg, N., Gradinger, R., Ilyash, L., Ratkova, T., von Quillfeldt, C. (2011) The pan-Arctic biodiversity of marine pelagic and sea-ice unicellular eukaryotes: a first-attempt assessment. Marine Biodiversity, 41, 13–28.
- Schünemann, H. & Werner, I. (2005) Seasonal variations in distribution patterns of sympagic meiofauna in Arctic pack ice. Marine Biology, 146, 1091–1102.
- Scott, C.L., Kwasniewski, S., Falk-Petersen, S., Sargent, J.R., 2002. Lipids and fatty acids in the copepod *Jaschnovia brevis* (Jaschnov) and in particulates from Arctic waters. Polar Biology 25 (1), 65-71.
- Stroeve, J.C., Serreze, M.C., Holland, M.M., Kay, J.E., Malanik, J., & Barrett, A.P. (2012) The Arctic's rapidly shrinking sea ice cover: a research synthesis. Climatic Change, 110, 1005–1027.
- Tchesunov, A. V. & Riemann, F. (1995) Arctic sea ice nematodes (Monhysteroidea), with descriptions of Cryonema crassum Gen. N., Sp. N. and C. Tenue Sp. N. Nematologica, 41, 35–50.
- van Franeker, J.A., Flores, H. and Van Dorssen, M. 2009. The Surface and Under Ice Trawl (SUIT). In: Flores, H. (ed.). Frozen Desert alive—the role of sea-ice for pelagic macrofauna and its predators: implications for the Antarctic pack-ice food web. University of Groningen, Groningen, Netherlands.
- Vaughan, D.G., Comiso, J.C., Allison, I., Carrasco, J., Kaser, G., Kwok, R., Mote, P., Murray, T., Paul, F., Ren, J., Rignot, E., Solomina, O., Steffen, K., & Zhang, T. (2013) Climate Change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (ed. by T.F. Stocker, D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex, and P.M. Midgley), pp. 317–382. Cambridge University Press, Cambridge, UK and New York, NY, USA.
- Wassmann, P., Reigstad, M., Haug, T., Rudels, B., Carroll, M.L., Hop, H., Gabrielsen, G.W., Falk-Petersen, S., Denisenko, S.G., Arashkevich, E., Slagstad, D., Pavlova, O., 2006. Food webs and carbon flux in the Barents Sea. Progress in Oceanography 71 (2-4), 232-287.

- Welch, H.E., Bergmann, M.A., Siferd, T.D., Martin, K.A., Curtis, M.F., Crawford, R.E., Conover, R.J., Hop, H., 1992. Energy flow through the marine ecosystem of the Lancaster Sound region, Arctic Canada. Arctic 45 (4), 343-357.
- Werner, I., 1997. Grazing of Arctic under-ice amphipods on sea-ice algae. Marine Ecology Progress Series 160, 93-99.
- Werner, I., Auel, H., Garrity, C., Hagen, W., 1999. Pelagic occurence of the sympagic amphipod Gammarus wilkitzkii in ice-free waters of the Greenland Sea - dead end or part of life-cycle? Polar Biology 22, 56-60.
- Werner, I., Ikävalko, J., Schünemann, H., 2007. Sea-ice algae in Arctic pack ice during late winter. Polar Biology 30 (11), 1493-1504. doi:10.1007/s00300-007-0310-2.

4.3 Zooplankton in the Central Arctic Ocean, a linchpin to keep the system in place

Contributors: Hyoung Chul Shin.

In a much warmer, continuously warming Arctic Ocean with receding sea ice and stronger stratification, the entire food web is affected. Zooplankton is at the center of these changes. Retreating sea ice, earlier ice break-ups, higher sea surface temperatures are all contributing factors operating in concert. There is already a fair amount of information on zooplankton; nonetheless, the limited access to the region due to the ice cover, and the continually changing ecology warrant a consolidated summary of newly collected information. The focus of this section is on recent and future studies. Below is a list of recent articles describing the zooplankton communities in the higher Arctic.

Wang et al (2019) describes the vertical and geographic distribution of copepod communities inlate summer in the Amerasian Basin. This study investigated species composition and copepod community structures by using geospatial and multiple depth scales, and by using multivariate analyses to evaluate the relationships between sampling stations and layers. During July–August 2010, zooplankton samples were collected with temperature and salinity data from three stations in the Canada Basin and two stations in the Makarov Basin of the Arctic Ocean (maximum distance of approximately 1400 km). A total of 55 copepod species (including 25 species that were solely identified to the generic level) and 7 taxa of copepodites were identified.

Significant differences were detected in copepod community structures among sampling strata and at geospatial scales. At the local scale, copepod compositions responded differently at each of the sampling stations. At the geospatial scale, the distance between stations MS03 and ICE explained variations in the pattern of dominant species and of copepod community richness. This study demonstrated varied spatial distribution which indicates that (1) the abundance of copepods at 0-200 m was significantly higher than at other strata, (2) vertical strata affected the distribution of copepod communities, and (3) the interplay of North Pacific and Atlantic waters is shaping the copepod assemblage structure at geospatial scales in the Arctic Ocean.

The influx and interaction of water masses from the Atlantic and the Pacific are certainly one of the principal factors governing zooplankton changes in the CAO, together with population interactions between shelf and basin. Gluchowskaet al (2017) examined interannual zooplankton variability in the main pathways of the Atlantic water flow into the Arctic Ocean by studying summer zooplankton standing stock and community structure in West Spitsbergen Current (WSC) branch and Barents Sea Opening (BSO) branch. Generally, there was a higher abundance and biomass of *Calanus finmarchicus* in the BSO than in the WSC, where *Oithona* spp. numerically dominated. The observed AW (Atlantic Water) temperature and volume flux modifications measurably affected zooplankton variability, with additional effects likely related to diverse

contributions of advected and local zooplankton production. The high abundances of *Oithona* spp. (predominantly *O. similis*) and *C. finmarchicus* were attributed to warm AW temperature anomalies. These results show the effects of warming on the zooplankton community transported with Atlantic water, before they enter the Arctic Ocean, and this finding will greatly assist the construction and tuning of the plankton components of ecosystem models.

Shelf-basin advection is also essential to the Arctic copepod population in the high basin area, especially for *Calanus hyperboreus*. Xu et al (2018) studied the abundance, population structure and body size of *Calanus hyperboreus* in pelagic layers over a large area of the western Arctic Ocean during summer 2003, in order to evaluate the geographical variation in recruitment pattern. *Calanus hyperboreus* was absent from the shallow areas of the Chukchi Sea and was most abundant in the slope area between the Chukchi Sea and Chukchi Abyssal Plain (CS-slope). These results indicate that *C. hyperboreus* can benefit from primary production increases through accelerated development in the first growth season, and the productive CS-slope area is a potential source for slope-basin replenishment.

There are different ways and mechanisms for warming to generate changes in zooplankton populations. Lower ice cover, early ice retreat and ocean warming can change various facets of the Arctic marine ecosystem, including the biogeographic distribution. Feng et al (2016) coupled a copepod individual-based model with an ice-ocean-ecosystem model to simulate temperatureand food-dependent copepod life history development, using *Calanus glacialis*. Model results agreed with commonly known biogeographic distributions of *C. glacialis*, which is a shelf/slope species that cannot colonize the vast majority of the central Arctic basins. Individuals along the northern boundaries of this species' distribution were most susceptible to changes in reproduction timing and early food availability (released sea ice algae). Early ice retreat and elevated ocean temperatures increased phytoplankton food and prolonged the growth season, creating favorable conditions for *C. glacialis* development, which fueled a remarkable poleward expansion of its distribution.

From a pan-Arctic perspective, despite the heterogeneity in the temperature and food regimes, common biogeographic zones could be identified from model simulations.

Kvile et al (2018) provided a good overview of how living under extreme and fluctuating environmental conditions is made possible by life history strategies such as multiyear life cycles, resting stages and capital breeding, using *Calanus hyperboreus* as an example. This study showed, from statistical analysis of compiled observational data, that abundances are relatively low and later stages and adults dominate in the CAO basins, indicating expatriation. They further demonstrated that the resilience of *C. hyperboreus* to environmental fluctuations enables its life cycle to be completed in the central Arctic basins, while the species can reach higher abundances in areas with greater food availability outside the central Arctic basins. Specifically, the energy level required to reach the first overwintering stage-a prerequisite for successful local production-is likely met in some-but not all-years. This fine balance between success and failure indicates that *C. hyperboreus* functions as a peripheral population in the central Arctic basins and its abundance will likely increase in areas with improved growth conditions in response to climate change. By illustrating a key Arctic species' resilience to extreme and fluctuating environmental conditions, the results of this study have implications for projections of future biogeography and food web dynamics in the Arctic Ocean, a region experiencing rapid warming and sea ice loss.

Lower ice cover in the CAO can affect the major zooplankton species by altering their reproductive ecology. But predation can offset the benefit of improved growth conditions. Darnis et al (2019) studied life cycle and reproduction of *Calanus hyperboreus* during a year of record low ice cover in the southeastern Beaufort Sea. Although vigorous reproduction delivered numerous eggs and nauplii N1-N3 early spring, subsequent CI copepodite recruitment, coincident with the phytoplankton bloom, was only modest in Amundsen Gulf. Consequently, *C. hyperboreus* abundance and biomass stagnated throughout summer in the Gulf. A mismatch between the first-feeding stages and food was unlikely under the favourable feeding conditions. Rather, predation on the egg and larval stages in late winter probably limited subsequent recruitment and population growth. Particularly abundant copepods *Metridia longa* and *C. glacialis* were likely important consumers of *C. hyperboreus* eggs and nauplii. With the ongoing climate-driven lengthening of the ice-free season, intensification of top-down control of *C. hyperboreus* recruitment by thriving populations of mesopelagic omnivores and carnivores like *M. longa* may counteract the potential benefits of increased primary production over the Arctic shelves and margins for this key prey of pelagic fish, seabirds and the bowhead whale.

During the long Arctic winter, *Calanus glacialis* and *Calanus hyperboreus*, large and lipid-rich copepods dominating the zooplankton biomass in the Arctic Ocean, have been known to enter a form of resting stage called diapause, surviving on lipid reserves accumulated during the short growth season. In this way, copepod lipids can fuel the Arctic marine ecosystem, but information on the fine-scale distribution of copepods and lipids is scarce. Schmid et al (2018) showed that lipid load can trigger migration to diapause in Arctic *Calanus* copepods. They investigated the fine-scale (1 m) vertical distribution of the copepods *Calanus hyperboreus*, *Calanus glacialis* and *Metridia longa* during a Lagrangian drift in the North Water Polynya using the Light frame Onsight Key-species Investigation (LOKI) imaging system. Vertical distribution profiles and individual lipid loads suggested that individuals with lower lipid load continued DVM, while others with high lipid load ceased migrating, remaining at depth. A bioenergetics model showed that *Calanus* females had enough lipids to diapause for over a year, highlighting the significant lipid overhead they use for capital breeding.

Nonetheless, diapause at depth seems a facultative strategy, not a must. Kylie et al (2019) showed based on a large set of pan-Arctic observational data that Arctic *Calanus* has a rather broad depth distribution, not necessarily descending deep for diapause in winter. The larger *C. hyperboreus* has a deeper diapause depth than *C. glacialis*, potentially to alleviate predation pressure or as a result of predation loss near the surface. Moreover, the mean depth of *C. hyperboreus* in winter is negatively associated with latitude, indicating a shoaling of the diapause population in the central Arctic Ocean where predation pressure is lower. In the central Arctic Ocean where visual predation pressure is lower, copepods might be relieved from the incentive to descend and can remain closer to the surface in winter.

With ocean warming, borealization of zooplankton communities might be expected as a matter of natural course. With the reduction in sea ice cover, the Arctic species are likely to have already declined. There are now more studies available that have examined community changes along with documented physical changes. Møller and Nielsen (2020) analyzed a subset of 13 year data from Western Greenland, from the period 1992 to 2018, for trends in the key zooplankton genus *Calanus* during May and June in relation to physical conditions. In the 1990s, the small North Atlantic species *Calanus finmarchicus* and the two larger Arctic species *Calanus hyperboreus* and *Calanus glacialis* contributed equally to the copepod biomass. With the Arctic species having declined, *C. finmarchicus* currently dominates the biomass. Because of the species shift, the *Calanus* community is now dominated by smaller individuals and the lipid content of *Calanus* females during spring and summer has decreased by 34%. Moreover, during the last decade the annual variation in population size has been prominent, *Calanus* virtually being absent in some years. Because of the central role of *Calanus* in the Arctic food web, the changes will likely impact higher trophic levels, including fish, sea birds, and marine mammals.

A year-round study of the relationship between copepod phenology and environmental drivers is a challenge, particularly in the CAO where non-summer season sampling is more difficult. Tokuhiro et al (2019) offered the possibility of using a sediment trap to address this, although their work was from the Pacific Sector of the Arctic. Seasonal changes in the flux (abundance of swimmers), population structure, lipid accumulation and gonad maturation of four dominant copepods (*Calanus hyperboreus*, *Metridia longa*, *Paraeuchaeta glacialis* and *Heterorhabdus norvegicus*) were studied. *C. hyperboreus* C6Fs were dominated by lipid-rich specimens year-round, and gonad development was observed from February to April, while the *M. longa* flux showed no clear seasonality. From generalized additive models, *C. hyperboreus*, *M. longa* and *P. glacialis* showed relationships with daytime length and/or sea ice concentrations, but the relationship patterns were different. The findings from this trap study were enough to imply that the response to changing environmental parameters could vary with species and that drastic sea ice reductions may affect the copepod phenology in the area.

The Central Arctic will almost certainly be subject to more human activities. One of the consequences of greater human activity is that natural light sources will in many places be masked by the much stronger illumination from artificial light. Berge et al (2020) showed that artificial light during the polar night can disrupt Arctic fish and zooplankton behavior down to 200 m depth. Normal working-light from a ship may disrupt fish and zooplankton behavior down to at least 200m depth across an area of >0.125 km² around the ship. Both the quantitative and qualitative nature of the disturbance differed between the examined regions. Biological surveys in the dark from illuminated ships may introduce biases on biological sampling, bioacoustic surveys, and possibly stock assessments of commercial and non-commercial species and this has ramifications for increased shipping in the future Arctic.

References

- Berge J, Geoffroy M, Daase M, Cottier Finlo, Priou Pierre, Cohen Jonathan H., Johnsen Geir, David McKee, Kostakis Ina, Renaud Paul E., Vogedes Daniel, Anderson Philip, Last Kim S. & Gauthier Stephane (2020) Artificial light during the polar night disrupts Arctic fish and zooplankton behaviour down to 200 m depth. Commun Biol. 3(1):102. doi:10.1038/s42003-020-0807-6
- Darnis Gérald, Woldb Anette, Falk-Petersen Stig, Graevee Martin, Fortiera Louis (2019) Could offspring predation offset the successful reproduction of the arctic copepod *Calanus hyperboreus* under reduced sea-ice cover conditions? Progress in Oceanography 170. 107–118.
- Feng, Z., R. Ji, R. G. Campbell, C. J. Ashjian, and J. Zhang (2016) Early ice retreat and ocean warming may induce copepod biogeographic boundary shifts in the Arctic Ocean, J. Geophys. Res. Oceans, 121, 6137– 6158, doi:10.1002/2016JC011784.
- Gluchowska, Marta & Dalpadado, Padmini & Beszczynska-Moeller, Agnieszka & Olszewska, Anna & Ingvaldsen, Randi & Kwasniewski, Slawomir. (2017). Interannual zooplankton variability in the main pathways of the Atlantic water flow into the Arctic Ocean (Fram Strait and Barents Sea branches). ICES Journal of Marine Science. 74. 10.1093/icesjms/fsx033.
- Koki Tokuhiroa, Yoshiyuki Abea,b, Kohei Matsunoa, Jonaotaro Onoderac, Amane Fujiwarac, Naomi Haradac, Toru Hirawakea,b, Atsushi Yamaguchia, (2018) Seasonal phenology of four dominant copepods in the Pacific sector of the Arctic Ocean: Insights from statistical analyses of sediment trap data. Polar Science 19 (2019) 94–111.
- Kvile Kristina Øie, Ashjian Carin, Feng Zhixuan, Zhang Jinlun, Ji Rubao (2018) Pushing the limit: Resilience of an Arctic copepod to environmental fluctuations. Glob. Change Biol. 24. 5426-5439.
- Kvile Kristina Øie, Ashjian Carin, Ji Rubao (2019) Pan-Arctic depth distribution of diapausing *Calanus* copepods. Biol. Bull. 237: 76–89.
- Møller Eva Friis, Nielsen Torkel Gissel (2020) Borealization of Arctic zooplankton smaller and less fat zooplankton species in Disko Bay, Western Greenland. Limnol. Oceanogr. 65, 2020, 1175–1188
- Schmid, Moritz & Maps, Frederic & Fortier, Louis. (2018). Lipid load triggers migration to diapause in Arctic *Calanus* copepods - Insights from underwater imaging. Journal of Plankton Research. 40. 10.1093/plankt/fby012.

- Tokuhiro, Koki & Abe, Yoshiyuki & Matsuno, Kohei & Onodera, Jonaotaro & Fujiwara, Amane & Harada, Naomi & Hirawake, Toru & Yamaguchi, Atsushi (2019) Seasonal phenology of four dominant copepods in the Pacific sector of the Arctic Ocean: Insights from statistical analyses of sediment trap data. Polar Science 19. 94–111.
- Wang Y-G, Tseng L-C, Lin M, Hwang J-S (2019) Vertical and geographic distribution of copepod communities at late summer in the Amerasian Basin, Arctic Ocean. PLoS ONE 14(7): e0219319. https://doi.org/10.1371/journal.pone.0219319
- Xu, Zhiqiang & Zhang, Guangtao & Sun, Song. (2018). Accelerated recruitment of copepod Calanus hyperboreus in pelagic slope waters of the western Arctic Ocean. Acta Oceanologica Sinica. 37. 87-95. 10.1007/s13131-018-1166-8.

4.4 Benthos in the CAO

Contributors: Lis Jørgensen, Bodil Bluhm, Jackie Grebmeier, Karen Edelvang, Motoyo Itoh, Shigeto Nishino (JAMSTEC, Japan), William Williams (IOS, Canada).

4.4.1 The seabed

The 3.3 million km2 Central Arctic Ocean (CAO, max. depth of 4,000 m) is divided into two primary basins by a submarine ridge of continental crust (the Lomonosov Ridge, 1,800 km long, 60-200 km wide, min. depth 954m) with steep slopes including canyon structures on each side (figure 4.1). The Amerasian Basin is divided into two unequal basins by the Alpha Ridge, a broad, rugged submarine mountain chain that extends to within 1,402 m of the ocean surface. The Makarov Basin lies between the Alpha Cordillera and the Lomonosov Ridge, and its floor is at a depth of 4,023 m.

The largest sub-basin of the Arctic Ocean is the Canada Basin which has a smooth basin floor interrupted by regions of sea knolls. The average depth of the Canada Basin is 3810 m. In the Canada Basin the sediment magnitude increases from the Chukchi Plateau (see also Mayer et al 2010) toward the Alaskan and Canadian margins with the MacKenzie River delta forming a massive sediment deposit of an estimated 15 km based on geophysical investigations. The Makarov–Podvodnikov Basin forms the second part of the Amarasian Basin and is bordered by the Lomonosov Ridge and the Alpha–Mendeleev Ridge. The Podvodnikov Basin (2,800m) is much shallower than the Makarov Basin (4,000m). They are both covered by a several km thick Cretacious-Cenoozoic sediment layer (Pease et al 2014).

Sediment cores and seismic reflection from the margin of the Lomonosov Ridge in the Lincoln Sea indicates that only a thin sediment cover (less than 50 m) overlies the basement. In the deep passage between the Lomonosov Ridge and the Lincoln Sea continental margin, more than 1 km of sediment is present (Mikkelsen et al., 2005). The rivers also discharge suspended sediment into the seabed. The CAO receives about 10% of the world's fresh-water discharge from the major Arctic rivers (MacKenzie, Yukon, Kolyma, Lena, Yenisey, and. The major part of the suspended sediment discharged from the rivers is deposited on the CAO seabed.

Speculation is ongoing regarding the possibility of exploring for resources in the area, not least because large natural gas reserves are believed to be beneath parts of the CAO's sea floor.



Figure 4.1. The basins of the Central Arctic Ocean (source: Jakobsson, M., Grantz, A., Kristoffersen, Y. and Macnab, R., 2003. Physiographic provinces of the Arctic Ocean seafloor. Geological Society of America Bulletin, 115(12), pp.1443-1455). Permission was granted by Geological Society of America Bulletin for use of the figure.

4.4.2 Bottom temperatures of the Central Arctic Ocean

The bottom temperatures of the Fram Strait Branch of Atlantic Water (T>0oC, 300-500m depth) show large interannual variabilities ranging from 0.4oC to 0.7oC during 2002 to 2018 (Figure 2 left).

The variability of Barents Sea Branch of Atlantic Water (T<0oC, 750-1250m depth) was small until 2016, when the bottom temperature increased up to 0.25 oC (Figure 4.2 middle).

The Deep Water (uniform temperature and salinity, deeper than 2500 m) shows a significant increasing trend of approximately 0.005 oC during the last 16 years in the Canadian Basin (Figure 2 right). This warming contributes significantly to the increasing heat content of the water column.

Further monitoring is needed to evaluate this recent warming trend.



Figure 4.2. Interannual variabilities in bottom temperatures (left) Fram Strait Branch of Atlantic Water at 300-500 m, (middle) Barents Sea Branch of Atlantic Water at 750-1250 m and (right) Deep Water below 2500 m for 16 years (2002 to 2018).

Red open circles show 2-year averages. Red bars indicate standard deviation. (Source: Motoyo Itoh, Shigeto Nishino (JAM-STEC, Japan) and William Williams (IOS, Canada)).

4.4.3 Benthos

Restructuring of the benthic shelf ecosystem in the Arctic region is ongoing and likely to continue in the future (Waga et al 2020), though species expansions into the deep sea are unlikely in the Amerasian Basin given the confinement of Pacific-boreal taxa to the shelf and upper slope (Carey et al., Ravelo et al. 2020). Atlantic-boreal taxa are already common in the Eurasian Basin (Bluhm et al. 2011, Dilman and Mironov 2011).

Half of the Arctic Ocean is deep sea (>1000 m), and this area is currently changing from being permanently ice-covered to being seasonally ice-free. Studies on sympagic, pelagic, and especially benthic organisms are crucially needed to get a better understanding for the interwoven ecosystem structures in the AO basins, including quantifying the intensity of carbon advection, benthic carbon production, as well as the downward carbon flux in the form of ice algae aggregates. This will help in predicting how changes in the sea ice, the primary production, and the grazer community may change the carbon supply to the benthos in the AO basins (Wiedmann et al 2020)– a system, which is today already affected by anthropogenic waste (e.g., microplastic, La Daana et al 2019).

Awareness is increasing that deep-sea benthic communities are not as homogenous as previously thought, and that 'special habitats' increase biodiversity and add heterogeneity; this finding also holds true for the CAO. Special habitats include seamounts housing unexpectedly rich sponge beds and associated benthic biota (Boetius and Purser 2017), fauna associated with the spreading of Gakkel Ridge including chemoautotrophic biota (Edmonds et a.l 2003; Bünz et al. 2020), fauna on other ridges such as the Alpha Ridge (Schewe 2001), glacial drop stones forming biodiversity islands (Mayer and Piepenburg 1996, Zhulay et al. 2019), a thin benthic boundary layer combining hyperbenthic and abysso-pelagic taxa (Zhulay et al. 2019, Raskoff et al. 2010), distinct density and biomass levels associated with the overlying marginal ice zone (Rybakova et al. 2019), and the more 'typical' meio- and macrobenthos in soft sediment (Paul and Menzies 1974, Kröncke 1994, Renaud et al 2006, Kröncke et al. 2000, Bluhm et al. 2011).

Lower slope and basin benthic community structure is distinct from upper slopes and certainly adjacent shelves. This is true for community structure based on taxonomic identity (Vedenin et al. 2019, Bluhm et al. in review); biogeographic affinity being dominated by Arctic-Atlantic and cosmopolitan fauna all across the basins versus more diverse affinities in shallower water (Mironov et al. 2013, Zhulay et al. 2019); and biological traits characterizing Arctic deep-sea fauna as generally (yet with substantial variation), comparatively small (though gigantism also exists), slow-growing, non-sessile (for epifauna), predatory or deposit-feeding (Iken et al. 2005, Wei et al. 2010, Zhulay et al. 2019).

Given lower overall densities of Arctic deep-sea fauna, sediment oxygen demand is on average one to two orders of magnitude lower in the Arctic basins than on adjacent shelves (Bourgeois et al. 2017, Wiedmann et al. 2020). Data on vertical flux, ice cover and incidental observations suggest carbon supply to the Arctic deep-sea floor covering this oxygen demand is highly variable in time, space and quality (and hence unpredictable and challenging to measure). Shortterm events may include quick sinking of large clumps of fresh sea ice algae Melosira arctica (Boetius et al. 2013), increased flux through ballasting (Wollenburg et al. 2018) related to brief blooms (Menge and Nöthig 2020); their contribution to overall carbon flux may be substantial but needs quantifying. In addition, the overlying position of the marginal ice zone influences food availability (Rybakova et al. 2019), enhancing patchiness in flux.

4.4.4 Climate impact on benthos

There are no time series in the CAO. But observations from the HAUSGARTEN observatory in eastern Fram Strait can be used to get a proxy of what may happen in the CAO. Over an 11-year period (2004-2015) epibenthic communities at the seafloor were not static, but instead variable from year to year with no consistent community state (Taylor et al. 2017). Variation was, for example, particularly high for sea cucumber densities. Whether slight increases in Chl a in eastern Fram Strait in summer and in standing stocks of POC in the central AO over the period from 1991-2015 (Noethig et al. 2020) translated to changes in benthic stocks and communities is unclear.

On the Pacific Arctic shelves, there is evidence of declining benthic biomass in the decadal benthic times-series data in both the northern Bering Sea (Grebmeier 2012, Grebmeier and Cooper 2016, Grebmeier et al. 2018) and downstream through Bering Strait in the southern Chukchi Sea (Grebmeier et al. 2015b, 2018). In addition, over the last decade a northward shift in dominant macrofaunal biomass in the persistent northern Bering Sea hotspot south of St. Lawrence Island has been observed related to varying current patterns (Grebmeier and Cooper 2016, Goethel et al. 2018, Grebmeier et al. 2018). This northward shift in the distributional pattern of benthic species, and subsequent changes in community composition has also been recorded at places in the European Arctic (e.g., Svalbard Archipelago and Barents Sea; Kortsch et al. 2015, Jørgensen et al. 2017, Jørgensen et al 2019), including for commercial crab species (Fossheim et al. 2015).

4.4.5 Possible human activities impacting benthos

Characteristics of deep-sea organisms include increased longevity, slow growth rates, reproduction late in life and low fecundity (Carreiro-Silva et al., 2013; Levin et al., 2016; Danovaro et al., 2017b; Montero-Serra et al., 2018). These life history strategies mean that many deep-sea species have an increased sensitivity to human activities such as mining, fisheries and climate change. Much of the Arctic deep-sea floor has until now experienced only a weak human footprint, but a predicted ice-free summer in the Arctic in the near future may change that situation, thus an up-to-date inventory is urgently needed given the last one is a decade old (Bluhm et al. 2011).

4.4.5.1 Pollution and Plastics

Litter in the Arctic (HAUSGARTEN observatory in the Fram Strait, 2500 m depth) has increased from 3635 to 7710 items km2 between 2002 and 2011 and reached densities similar to those reported from a canyon near the Portuguese capital Lisboa (Bergman et al. 2012), with a strong increase in marine litter over the period of 2002–2014 (Tekman et al. 2017). Plastic constituted the majority of litter (47 - 59%) followed by a black fabric (11%) and cardboard/paper (7%). Sixtyseven percent of the litter was entangled or colonized by invertebrates such as sponges (41%) or sea anemones (15%). Litter densities were positively correlated with the counts of ships entering harbor at Longyearbyen, the number of active fishing vessels and the extent of summer sea ice. Sea ice may act as a transport vehicle for entrained litter, releasing it during periods of melting.

There are also indications of a widespread occurrence of high numbers of microplastics (42–6595 microplastics kg–1) at the HAUSGARTEN observatory (Bergmann et a.l 2017). The northernmost stations harbored the highest quantities, indicating sea ice as a possible transport vehicle. The authors postulate that a positive correlation between microplastic abundance and chlorophyll a content suggests vertical export via incorporation in sinking (ice-) algal aggregates. Overall, 18 different polymers were detected. Chlorinated polyethylene accounted for the largest proportion (38%), followed by polyamide (22%) and polypropylene (16%). Almost 80% of the microplastics were \leq 25 µm. The microplastic quantities are among the highest recorded from benthic sediments. This corroborates the deep sea as a major sink for microplastics and the presence of

accumulation areas in this remote part of the world, fed by plastics transported to the North via the Thermohaline Circulation.

4.4.5.2 Bottom fishing

Most of the CAO is too deep for demersal fisheries. Fishable depths are identified on the Chukchi Borderland (Figure 4.3) and impact on the seabed-fauna is therefore currently unlikely and will most likely be restricted to the Chukchi Plateau of the CAO with depths of 480-2000 m. Here, research suggests taxon richness, biomass and density of epifauna decrease with depth and communities markedly differ between basin and plateau/ridge (Zhulay et al 2019). Echinodermata (e.g. *Ophiopleura borealis, Elpidia* sp., and *Pontaster tenuispinus*) and Arthropoda (e.g. *Bythocaris* sp.) dominate the overall taxon richness here. Polynoid annelids and sponges were characteristic taxa of the basin stations. Epifaunal densities were estimated of up to 14,346 individuals and 906 g wet weight /1000 m². The lack of Arctic deep-sea data restricts the evaluation of biological responses to large-scale change in the Arctic environment. In order to provide adequate answers concerning how Arctic deep-sea ecosystems will change, long-term observations are needed also in the Amerasian sector of the Arctic deep sea.



Figure 4.3. Map of the Arctic region showing the High Seas (red line) and the shallower parts of the High Seas (yellow area) that may be more "fishable" than the deep basins. L = Lomonosov Ridge, A = Alpha Ridge, M = Mendeleev Ridge, C = Chukchi Plateau. (Source: Adapted from PEW map based on fishable depths derived from IBCAO v3 bathymetry (www.ngdc.noaa.gov/mgg/bathymetry/arctic).

4.4.5.3 Deep-sea mining

(https://www.iucn.org/resources/issues-briefs/deep-sea-mining) (<u>https://www.isa.org.jm/about-isa</u>)

A concern of many marine scientists is that the mining industry will progress to large-scale extraction without the implementation of adequate measures to avoid significant environmental

L

harm (Levin et al., 2016; Niner et al., 2018). Seabed mining will damage the marine environment for many centuries (Van Dover et al., 2017). At hydrothermal vent systems, chemosynthesis can play a large role in providing nutrients to deep-sea benthic communities (Sweetman et al. 2013). Even if restoration is part of the wider management of seabed mining, it is highly unlikely that biodiversity will fully recover (Jones et al., 2018).

Direct impacts occur through the removal of target material and associated organisms within the mine site and include the destruction of biota as well as habitat loss, fragmentation, and modification through altered mineral and sediment composition, geomorphology, and biogeochemical processes (Ellis, 2001; Van Dover, 2014; Jones et al., 2017). It also includes the smothering of habitat and biota, interference with feeding activities, and the release and spread of nutrient-rich and toxin-laden water from the generation of plumes (Ellis, 2001; Boschen et al., 2013). The scale over which these indirect impacts are likely to occur is largely unknown and most of the effects remain unstudied (Van Dover, 2014; MIDAS, 2016).

This vulnerable nature of deep-sea environments to mining impacts, currently limited technological capacity to minimize harm, significant gaps in ecological knowledge, and uncertainties of recovery potential of deep-sea ecosystems means that the industry cannot at present deliver an acceptable outcome (Niner et al. 2018).

4.4.5.4 Oil exploitation

(https://atlantic-community.org/petroleum-resources-of-the-arctic-ocean-a-broken-promise/)

Around 22% of the world's undiscovered oil and gas resources is in the Arctic, most of it offshore. The areas of the Eurasian Basin, Amerasian Basin and Lomonossov-Makarov are thought to house an estimated 13.5% of undiscovered Arctic oil reserves, 5 % of total gas and close to7% of total barrels of combined oil and oil-equivalent natural gas (Bird et al. 2008). This assessment means the potential for oil and gas exploration in deep areas exists, though is not considered immediate. However, ongoing and northward moving exploration on shelves may indirectly affect CAO areas also. Russia, Norway and the US have ongoing oil and gas extraction in their Arctic territories (<u>https://atlantic-community.org/petroleum-resources-of-the-arctic-ocean-abroken-promise/</u>).

In the south western Barents Sea, drill cuttings released around wells have caused changes in sediment quality and ecosystem structure with partial recovery 15 years after drill cuttings releases (Dijkstra et al. 2020).

Effects of drilling discharges are generally limited to within 1 km from the platform but have been observed to impact up to 1.2 km away and impacts persist for at least 6–8 years. Impacts were undetectable in the areas where cuttings (Henry et al 2017).

After oil spills, marine snow on the sediment affects organisms and can be a vector for introducing oil into the food web. Amphipods shows a dose-dependent decrease in survival while gastropods were only affected by the highest concentration of oil-contaminated marine snow (van Eenennaam et al 2019).

References

- Bergmann, Melanie, and Michael Klages (2012). Increase of litter at the Arctic deep-sea observatory HAUS-GARTEN. Marine Pollution Bulletin 64, 12: 2734-2741.
- Bergmann, M., Wirzberger, V., Krumpen, T., Lorenz, C., Primpke, S., Tekman, M.B. and Gerdts, G., (2017). High quantities of microplastic in Arctic deep-sea sediments from the HAUSGARTEN observatory. *Environmental Science & Technology*, 51(19), pp.11000-11010.

- Bluhm, B.A., W.G. Ambrose Jr., M. Bergmann, L.M. Clough, A.V. Gebruk, C. Hasemann, K. Iken, M. Klages, I.R. MacDonald, P.E. Renaud, I. Schewe, T. Soltwedel, and M. Wlodarska-Kowalczuk. 2011. Diversity of the Arctic deep-sea benthos. Marine Biodiver. 41, 87–107.
- Bluhm B.A., K.N. Kosobokova, and E.C. Carmack. 2015. A tale of two basins: An integrated physical and biological perspective of the deep Arctic Ocean. Prog. Oceangr. 139, 89-121.
- Bluhm BA, Janout M, Danielson S, Ellingsen I, Gavrilo M, Grebmeier JM, Hopcroft RR, Iken KB, Ingvaldsen RB, Jørgensen LL, Kosobokova KN, Kwok R, Polyakov I, Renaud PE, Carmack EC. (*in review*) Arctic continental slopes: sharp gradients of physical processes affect pelagic and benthic ecosystems *Frontiers in Marine Science*
- Bourgeois, S., Archambault, P. and Witte, U., 2017. Organic matter remineralization in marine sediments: A Pan-Arctic synthesis. *Global Biogeochemical Cycles*, 31(1), pp.190-213.
- Boetius, A., Albrecht, S., Bakker, K., Bienhold, C., Felden, J., Fernández-Méndez, M., Hendricks, S., Katlein, C., Lalande, C., Krumpen, T. and Nicolaus, M., 2013. Export of algal biomass from the melting Arctic sea ice. *Science*, 339(6126), pp.1430-1432.
- Boetius, A. and Purser, A. (2017): The Expedition PS101 of the Research Vessel POLARSTERN to the Arctic Ocean in 2016, Berichte zur Polar- und Meeresforschung = Reports on polar and marine research, Bremerhaven, Alfred Wegener Institute for Polar and Marine Research, 706, 230 p. . doi: 10.2312/BzPM_0706_2017
- Bünz, S., Ramirez-Llodra, E., German, C., Ferre, B., Sert, F., Kalenickenko, D., Reeves, E., Hand, K., Dahle, H., Kutti, T., Purser, A., Hilario, A., Ramalho, S., Rapp, H. T., Ribeiro, P., Victorero, L., Hoge, U., Panieri, G., Bowen, A., Jakuba, M., Suman, S., Gomez-Ibanez, D., Judge, C., Curran, M., Nalicki, V., Vagenes, S., Lamar, L., Klesh, A., Dessandier, P. A., Steen, I., Mall, A., Vulcano, F., Meckel, E. M. and Drake, N. (2020): RV Kronprins Håkon (cruise no. 2019708) Longyearbyen – Longyearbyen 19.09. – 16.10.2019, cruise report
- Carreiro-Silva, M., Andrews, A., Braga-Henriques, A., de Matos, V., Porteiro, F. M., and Santos, R. S. (2013). Variability and growth of long-lived black coral Leiopathes sp. from the Azores. Mar. Ecol. Prog. Ser. 473, 189–199. doi: 10.3354/meps10052
- Danovaro, R., Corinaldesi, C., Dell'Anno, A., and Snelgrove, P.V.R. (2017). The deep-sea under global change deep-sea under global change. Curr. Biol. 27, R461-R465. doi: 10.1016/j.cub. 2017.02.046
- Dijkstra, N., Junttila, J. and Aagaard-Sørensen, S., 2020. Impact of drill cutting releases on benthic foraminifera at three exploration wells drilled between 1992 and 2012 in the SW Barents Sea, Norway. *Marine Pollution Bulletin*, *150*, p.110784.
- Edmonds, H.N., Michael, P.J., Baker, E.T., Connelly, D.P., Snow, J.E., Langmuir, C.H., Dick, H.J.B., Mühe, R., German, C.R. and Graham, D.W., 2003. Discovery of abundant hydrothermal venting on the ultraslow-spreading Gakkel ridge in the Arctic Ocean. *Nature*, 421(6920), pp.252-256.
- Ellis. D.V. (2001) A review of some environmental issues affecting marine mining. Mar. Georesour. Geotechnol. 19, 51–63. doi: 10.1080/106411901093 53804
- Fossheim. M-, Primicerio. R-, Johannesen. E-, Ingvaldsen, R.B., Aschan, M.M., Dolgov, A.V. 2015. Recent warming leads to a rapid borealization of fish communities in the Arctic Nature Clim. Change 5, 673-678.
- Goethel, C.L., J.M. Grebmeier, and L.W. Cooper. 2018. Changes in abundance and biomass of the bivalve Macoma calcarea in the northern Bering Sea and the southeastern Chukchi Sea from 1998-2014 tracked through dynamic factor analysis models. Deep-Sea Research II Topical Studies in Oceanography, https://doi.org/10.1016/j.dsr2.2018.10.007 (in press).
- Grebmeier J.M. 2012. Shifting patterns of life in the Pacific Arctic and sub-arctic seas. Annual Review of Marine Science 4, 63-78.
- Grebmeier, J.M, and L.W. Cooper. 2016. The Saint Lawrence Island Polynya: A 25-Year Evaluation of an Analogue for Climate Change in Polar Regions. In: P.M. Glibert, T.M. Kana (eds.), Aquatic Microbial Ecology and Biogeochemistry: A Dual Perspective, Springer; DOI 10.1007/978-3-319-30259-1_14.

- Grebmeier, J.M., B.A. Bluhm, L.W. Cooper, S.G. Denisenko, K. Iken, M. Kedra, and C. Serratos. 2015b. Timeseries benthic community composition and biomass and associated environmental characteristics in the Chukchi Sea during the RUSALCA 2004–2012 program. Oceanogr. 28(3): 3):116–133, http://dx.doi.org/10.5670/ oceanog.2015.61
- Grebmeier, J.M, K.E. Frey, L.W. Cooper, and M. Kędra. 2018. Trends in benthic macrofaunal populations, seasonal sea ice persistence and bottom water temperatures in the Bering Strait region. *Oceanography 31*(2), 136-151.
- Henry, L.A., Harries, D., Kingston, P. and Roberts, J.M., 2017. Historic scale and persistence of drill cuttings impacts on North Sea benthos. *Marine environmental research*, *129*, pp.219-228.
- Iken, K., Bluhm, B., Gradinger, R., 2005. Food web structure in the high Arctic Canada Basin: evidence from δ 13 C and δ 15 N analysis. Polar Biol. 28, 238–249.
- Jørgensen, L.L., P. Archambault, M. Blicher, N. Denisenko, G. Mmundsson, K. Iken, V. Roy, J. Sørensen, N. Anisimova, C. Behe, B.A. Bluhm, S. Denisenko, V. Metcalf, S. Olafsdóttir, T. Schiøtte, O. Tendal, A.M. Ravelo, M. Kędra, and D. Piepenburg. 2017. Chapter 3.3: Benthos. In: Conservation of Arctic Flora and Fauna (CAFF) (eds) State of the Arctic Marine Biodiversity Report. CAFF Secretariat, Akureyri, pp 85-107.
- Jørgensen, L. L., Primicerio, R., Ingvaldsen, R. B., Fossheim, M., Strelkova, N., Thangstad, T. H., ... & Zakharov, D. (2019). Impact of multiple stressors on sea bed fauna in a warming Arctic. Marine Ecology Progress Series, 608, 1-12.
- Jones, H.P., Jones, P.C., Barbier, E.B., Blackburn, R.C., Rey Benayas, J.M., Holl. D., et al. (2018). Restoration and repair of Earth's damaged ecosystems. Proc. R. Soc. B285 :2017257. doi: 10.1098/rspb.2017.2577
- Jones D. O. B, S. Kaiser, A. K. Sweetman, C. R. Smith, L. Menot, A. Vink, D. Trueblood, J. Greinert, D. S. M. Billett, P. M. Arbizu, T. Radziejewska, R. Singh, B. Ingole, T. Stratmann, E. Simon-Lledo, J. M. Durden, M. R. Clark, Biological responses to disturbance from simulated deep-sea polymetallic nodule mining. PLOS ONE 12, e0171750 (2017).
- Kenneth J. Bird, Ronald R. Charpentier, Donald L. Gautier (CARA Project Chief), David W. Houseknecht, Timothy R. Klett, Janet K. Pitman, Thomas E. Moore, Christopher J. Schenk, Marilyn E. Tennyson, and Craig J. Wandrey (2008) Circum-Arctic Resource Appraisal: Estimates of Undiscovered Oil and Gas North of the Arctic Circle U.S. Geological Survey Fact Sheet 2008-3049
- Kröncke, I., 1994. Macrobenthos composition, abundance and biomass in the Arctic Ocean along a transect between Svalbard and the Makarov Basin. *Polar Biology*, *14*(8), pp.519-529.
- Kröncke, I., Vanreusel, A., Vincx, M., Wollenburg, J., Mackensen, A., Liebezeit, G. and Behrends, B., 2000. Different benthic size-compartments and their relationship to sediment chemistry in the deep Eurasian Arctic Ocean. *Marine Ecology Progress Series*, 199, pp.31-41.
- Kortsch S, R. Primicerio, M. Fossheim, A. Dolgov, and M. Aschan. 2015. Climate change alters the structure of arctic marine foodwebs due to poleward shifts of boreal generalists. In Proc. R. Soc. B, 282, No. 1814, p. 20151546. The Royal Society.
- La Daana, K.K., Johansson, C., Frias, J.P.G.L., Gardfeldt, K., Thompson, R.C. and O'Connor, I., 2019. Deep sea sediments of the Arctic Central Basin: A potential sink for microplastics. *Deep Sea Research Part I: Oceanographic Research Papers*, 145, pp.137-142.
- L. A. Levin, K. Mengerink, K. M. Gjerde, A. A. Rowden, C. L. van Dover, M. R. Clark, E. Ramirez-Llodra, B. Currie, C. R. Smith, K. N. Sato, N. Gallo, A. K. Sweetman, H. Lily, C. W. Armstrong, J. Brider (2016) Defining "serious harm" to the marine environment in the context of deep-seabed mining. Mar. Policy 74, 245–259.
- Mayer, M., Piepenburg, D., 1996.Epibenthic community patterns on the continental slope off East Greenland at 75* N. Mar. Ecol. Prog. Ser. 143, 151–164.
- Mayer, L.A., Armstrong, A., Calder, B., Gardner, J., 2010. Sea floor mapping in the Arctic: support for a potential US extended continental shelf. Int. Hydrogr. Rev. 3, 14–23.

- MIDAS (2016). Managing Impacts of Deep-Sea Resource Exploitation. Available at: <u>https://www.eu-midas.net/</u>
- Mikkelsen, N., Nørgaard-Pedersen, N., Kristoffersen, Y., Juul Lassen, S., and Sheldon, E. 2005. Radical past climatic changes in the Arctic Ocean and a geophysical signature of the Lomonosov Ridge north of Greenland. Geological Survey of Denmark and Greenland. Bulletin 10, 61-64
- Mironov, A.N., Dilman, A.B. and Krylova, E.M., 2013. Global distribution patterns of genera occurring in the Arctic Ocean deeper 2000 m. *Invertebrate Zool*, *10*(1), pp.167-194.
- Menge, M. and Nöthig, E.M., 2020. Phytoplankton succession in the eastern Fram Strait (August 2017-July 2018)–Microscopic analysis of autonomously collected water samples.
- Montero-Serra, I., Linares, C., Doak, D. F., Ledoux, J. B., and Garrabou, J. (2018). Strong linkages between depth, longevity and demographic stability across marine sessile species. Proc. R. Soc. B 285:20172688. doi: 10.1098/rspb.2017. 2688
- Niner, H. J., Ardron, J. A., Escobar, E. G., Gianni, M., Jaeckel, A., Jones, D. O. B., et al. (2018). Deep-sea mining with no net loss of biodiversity – an impossible aim. Front. Mar. Sci. 5:53.doi: 10.3389/fmars.2018.00053
- Nöthig, E.M., Ramondenc, S., Haas, A., Hehemann, L., Walter, A., Bracher, A., Lalande, C., Metfies, K., Peeken, I., Bauerfeind, E. and Boetius, A., 2020. Summertime in situ chlorophyll a and particulate organic carbon standing stocks in surface waters of the Fram Strait and the Arctic Ocean (1991–2015). *Frontiers in Marine Science*, 7, p.350.
- Paul, A.Z. and Menzies, R.J., 1974. Benthic ecology of the high Arctic deep sea. Marine Biology, 27(3),
- Renaud, P.E., Ambrose Jr, W.G., Vanreusel, A. and Clough, L.M., 2006. Nematode and macrofaunal diversity in central Arctic Ocean benthos. *Journal of Experimental Marine Biology and Ecology*, 330(1), pp.297-306.
- Pease V, S. Drachev, R. Stephenson, X. Zhang 2014. Arctic lithosphere A review. Tectonophysics 628 (2014) 1–25
- Pemberton, P. and Nilsson, J., 2016. The response of the central Arctic Ocean stratification to freshwater perturbations. JGR Oceans Volume121, Issue1. Pages 792-817
- Raskoff, K., Hopcroft, R., Kosobokova, K., Purcell, J., Youngbluth, M., 2010. Jellies under ice: ROV observations from the Arctic 2005 hidden ocean expedition. Deep Sea Res. Part II Top. Stud. Oceanogr. 57, 111– 126.
- Rybakova, E., Kremenetskaia, A., Vedenin, A., Boetius, A. and Gebruk, A., 2019. Deep-sea megabenthos communities of the Eurasian Central Arctic are influenced by ice-cover and sea-ice algal falls. *PloS* one, 14(7).
- Schewe, I., 2001. Small-Sized Benthic Organisms of the Alpha Ridge, Central Arctic Ocean. International Review of Hydrobiology: A Journal Covering all Aspects of Limnology and Marine Biology, 86(3), pp.317-335.
- Shapovalova, D., 2020. Special rules for the Arctic? The analysis of Arctic-specific safety and environmental regulation of offshore petroleum development in the Arctic Ocean States. In Arctic Marine Sustainability (pp. 275-301). Springer, Cham.
- Snoeijs-Leijonmalm P., Flores H, Volckaert F, Niehoff B, Schaafsma FL, Hjelm J, Hentati-Sundberg J, Niiranen S, Crépin A, Österblom H (2020). Review of the research knowledge and gaps on fish populations, fisheries and linked ecosystems in the Central Arctic Ocean (CAO). European Commission B-1049 Brussels
- Sweetman, A.K., Levin, L.A., Rapp, H.T. and Schander, C., 2013. Faunal trophic structure at hydrothermal vents on the southern Mohn's Ridge, Arctic Ocean. *Marine Ecology Progress Series*, 473, pp.115-131.
- Taylor, J., Krumpen, T., Soltwedel, T., Gutt, J. and Bergmann, M., 2016. Regional-and local-scale variations in benthic megafaunal composition at the Arctic deep-sea observatory HAUSGARTEN. *Deep Sea Research Part I: Oceanographic Research Papers*, 108, pp.58-72.

- Tekman, M.B., Krumpen, T. and Bergmann, M., (2017). Marine litter on deep Arctic seafloor continues to increase and spreads to the North at the HAUSGARTEN observatory. *Deep sea research part I: oceanographic research papers*, *120*, pp.88-99
- Van Dover, C. L., Aronson, J., Pendleton, L., Smith, S., Arnaud-Haond, S., Moreno-Mateos, D., et al. (2014). Ecological restoration in the deep sea: desiderata. Mar. Policy 44, 98–106. doi: 10.1016/j.marpol.2013. 07.006
- VanDover, C.L., Ardron, J.A., Escobar, E., Gianni, M., Gjerde, K.M., Jaeckel, A., et al. (2017). Biodiversity loss from deep-sea mining. Nat. Geosci. 10, 464–465. doi:10.1038/ngeo2983
- van Eenennaam, J.S., Rohal, M., Montagna, P.A., Radović, J.R., Oldenburg, T.B., Romero, I.C., Murk, A.J. and Foekema, E.M., 2019. Ecotoxicological benthic impacts of experimental oil-contaminated marine snow deposition. *Marine pollution bulletin*, 141, pp.164-175.
- Vedenin, A., Gusky, M., Gebruk, A., Kremenetskaia, A., Rybakova, E., Boetius, A. (2018). Spatial distribution of benthic macrofauna in the Central Arctic Ocean. PLoS ONE 13(10): e0200121.
- Wei, C.-L., Rowe, G.T., Escobar-Briones, E., Boetius, A., Soltwedel, T., Caley, M.J., Soliman, Y., Huettmann, F., Qu, F., Yu, Z., 2010. Global patterns and predictions of seafloor biomass using random forests. PLoS One 5, e15323.
- Wiedmann, I., Ershova, E., Bluhm, B.A., Nöthig, E.M., Gradinger, R.R., Kosobokova, K. and Boetius, A., 2020. What Feeds the Benthos in the Arctic Basins? Assembling a Carbon Budget for the Deep Arctic Ocean. *Frontiers in Marine Science*, 7, p.224.
- Wollenburg, J.E., Katlein, C., Nehrke, G., Nöthig, E.M., Matthiessen, J., Wolf-Gladrow, D.A., Nikolopoulos, A., Gázquez-Sanchez, F., Rossmann, L., Assmy, P. and Babin, M., 2018. Ballasting by cryogenic gypsum enhances carbon export in a Phaeocystis under-ice bloom. *Scientific reports*, 8(1), pp.1-9.
- Zhulay, I., Iken, K., Renaud, P.E. and Bluhm, B.A., 2019. Epifaunal communities across marine landscapes of the deep Chukchi Borderland (Pacific Arctic). *Deep Sea Research Part I: Oceanographic Research Papers*, 151, p.103065.

4.5 Fish in the Central Arctic Ocean having to cope with unprecedented warming

Contributors: Hyoung Chul Shin

In a rapidly warming Arctic, fish populations are faced with major challenges, as are other components of the ecosystem. Receding sea ice, earlier ice breakups and higher sea surface temperatures are contributing factors operating in combination. Fish biodiversity throughout the diverse habitats of the Central Arctic Ocean (CAO) has yet to be explored in detail, and future research is needed.

Since the last update, however, relevant studies are steadily accumulating. Polar cod and Arctic cod, (*Boreogadus saida* and *Arctogadus glacialis*) are known to occur and are likely to be the main forage fish for the apex predators of the CAO. How warming of the Arctic Ocean and the loss of sea ice will affect fish communities of the CAO will depend on several factors, for example the effect on recruitment. A number of studies have been conducted not quite inside the CAO but on the same or related species in nearby waters and are thus highly relevant.

Bouchard et al. (2017) examined, in Canadian coastal waters, whether earlier sea ice break-up and warmer sea surface temperatures (SST) in spring-summer led to higher recruitment of juvenile polar cod, *Boreogadus saida* in late summer by enabling early hatchers to survive and grow larger by late summer. The density (number m⁻²) and biomass (mg m⁻²) of age-0 polar cod in August and September, estimated by hydroacoustics over 9 years in 9 areas of the Canadian Arctic, were negatively correlated with ice break-up week and positively correlated to SST. Ice breakup, in some areas, has occurred earlier by as much as 9.3 days per decade since 1979. Bouchard et al. project an increase in polar cod biomass, at least transiently, over the first part of the present century. Whether this will be sustained is not clear, however. It is possible that the ongoing migration of sub-Arctic and boreal forage fish northward in response to the moderation of extreme climatic conditions in the Arctic may cause the replacement of the hyper-specialized cod species. If that were to happen, cascading impacts would be expected on the unique pelagic ecosystems and biodiversity of the CAO.

LeBlanc et al. (2019b) investigated whether, apart from a longer growth season, an earlier ice breakup can also provide superior feeding conditions for young polar cod, *Boreogadus saida* by enhancing microalgal and zooplankton production over the summer months. Earlier breakups resulted in earlier phytoplankton blooms. Zooplankton backscatter in August increased with earlier breakup and bloom and plateaued at chlorophyll a > 1 mg m⁻³. Juvenile cod biomass in August increased with earlier breakup and plateaued at a zooplankton backscatter > 5 m² nmi⁻², supporting the hypothesis that higher food availability promotes the growth and survival of age-0 fish in years of early ice melt. Intense avian predation could explain the lower biomass of juvenile cod in the polynya complex, which is known to be a biological hotspot for energy transfer to higher trophic levels.

There are other aspects to the question of how changing sea ice (with a static photoperiod) will impact planktivorous fish in a future Arctic Ocean. Climate change will drive fish and plankton species toward the poles, but the distribution of fish may also be governed by less acknowledged mechanisms. For example, ice limits access to air for physostomous fish filling their swim bladders at the surface. Kaartvedt et al. (2018) hypothesized that reduction of ice constraints may facilitate northward expansion of physostomes, with implied impacts on their zooplankton prey. On the other hand, even in a changing Arctic, the extreme high-latitude photoperiod will persist.

The abundance of mesopelagic fish is low in the Arctic Ocean. Feeding conditions may be inferior during the darkness of winter and in bright summer nights. If the photoperiod is constraining distribution, biogeographic boundaries of mesopelagic fish may be relatively unaffected by climate change. Alternatively, if low temperatures are the main constraint, it is hypothesized that northward extensions in a warmer ocean may be detrimental to key Arctic copepods if their current success relates to low mortality during overwintering in the absence of mesopelagic fish. It is therefore critical to distinguish the effects of light at high latitudes from those related to temperature in assessing future biogeographic boundaries.

Salinity change associated with sea ice melt can also impact early life stages of Arctic fish species by changing the dispersal potential of marine organisms, salinity change can do so for most fish species, if not any. Spencer et al. (2018) examined whether loss of sea ice and salinity changes associated with warming in Alaskan waters have a strong effect on the distribution and survival of eggs and larvae of key fish species at the Pacific-Arctic interface, such as polar cod (Boreogadus saida) in the Chukchi Sea and walleye pollock (Gadus chalcogrammus) in the northern Bering Sea. Laboratory experiments were conducted to investigate how changes in ontogeny and salinity influenced the buoyancy and survival of eggs and larvae of these and two other gadids (Pacific cod, Gadus macrocephalus and saffron cod, Eleginus gracilis). Polar cod egg density varied among family groups, but eggs were more buoyant than those of walleye pollock overall. After hatching, polar cod and walleye pollock larvae were similar in density and were notably more buoyant than the larvae of species with demersal eggs (Pacific cod and saffron cod). Larvae of the two Arctic species (polar cod and saffron cod) were less sensitive to salinity challenges than the sub-Arctic gadids, but polar cod became more sensitive at 6 weeks post-hatch. These results highlight possible adaptations of polar cod to ensure that their eggs are positioned under or near ice after spawning so that larvae hatch in productive surface waters during ice break-up. The synergistic effects of environmental variables on vertical distribution of eggs and their survival potential

L

will be important biogeographic mechanisms to consider with climatic warming and continued loss of sea ice.

The pan-Arctic population structure of polar cod can provide insights as to how the species may respond to environmental change and allow for establishment of meaningful management units. Nelson et al. (2020) examined the population genetic structure of *Boreogadus saida* by analysing nine microsatellite DNA loci in 2269 fish collected at 19 locations across the species' range. Genetic differentiation was detected (FST = 0.01, p < 0.01), in concordance with geography.

A Canada East group consisted of fish collected from Resolute Bay to the Gulf of St. Lawrence. The Canada West group contained specimens from the Canadian Beaufort Sea and Amundsen Gulf. The Europe group contained specimens from the Greenland Sea, Iceland and the Laptev Sea, and a US group consisted of specimens collected in the North Bering, Chukchi and Western Beaufort seas. Very little genetic differentiation was detected within the identified groups. Physical distance, geophysical structure and oceanography all appeared to have the potential to influence levels of genetic divergence. The population genetic structure is likely to be important for the response of *B. saida* to environmental change and should be considered in management of human activities that may impact this foundational species.

Characterizing the condition of fish in dynamic seasonal environments requires an understanding of their energy allocation strategies. Varying energy sources for the early life stages of fish in relation to prey organisms is also an important aspect associated with Arctic Ocean warming. Copeman et al. (2020) examined the ontogenetic patterns in lipid and fatty acid biomarkers of juvenile polar cod (Boreogadus saida) and saffron cod (Eleginus gracilis) from across the Alaska Arctic, as changes in lipid allocation could have important implications for overwintering survival as well as energetic value for predators. A combination of laboratory and field approaches was employed to describe allometric relationships in lipid storage of polar cod and spatial patterns in field-caught juvenile gadid condition were explored in consecutive years. Lipid density in wild juvenile Arctic gadids increased with size leading into the first overwintering period, but age-1 + fish showed a reduction in lipid density with size prior to the 2nd overwintering period. Fatty acid biomarkers from the total lipid pool indicated that polar cod have a higher reliance on calanoid copepods than saffron cod. Collectively, these data suggest polar cod as well as saffron cod will likely respond differently to regional warming depending upon the shift in zooplankton communities, such that the energetic contribution of these fish to higher trophic levels could be transformed with ocean warming in coming decades.

With the CAO becoming more accessible and hence subject to enhanced human activities, fish populations will also have to deal with a number of new circumstances they have not experienced before. Several recent studies on CAO fish species may be relevant, although not necessarily undertaken inside the CAO.

Increased human presence in and close to the CAO will mean a higher plastic pollution load, and plastic litter is among the recently recognised stressors in Arctic ecosystems. With increasing human activity, plastic ingestion may act as an increasing stressor on polar cod in combination with ocean warming and sea-ice decline in peripheral regions of the Arctic Ocean. Any ingestion at any trophic level will go up the food web with cascading effects.

Kühnet et al. (2018) investigated the plastics in the stomachs of juvenile polar cod (*Boreogadus saida*), one of the key species in the Arctic ecosystem. The stomachs of 72 individuals collected both directly from underneath the sea ice in the Eurasian Basin and in open waters around Svalbard were analysed. Two stomachs contained non-fibrous microplastic particles. According to μ FTIR analysis, the particles consisted of epoxy resin and a mix of Kaolin with polymethylmethacrylate (PMMA). Fibrous objects were excluded from this analysis to avoid bias due to contamination with airborne micro-fibres. Based on the two non-fibrous objects found in polar

cod stomachs, it was shown that ingestion of microplastic particles by this ecologically important fish species is possible. To fully assess the significance of this stressor and its spatial and temporal variability, future studies will need to apply a rigorous approach to avoid secondary contamination.

Increasing Arctic shipping will definitely affect the behaviour of fish and influence the underwater soundscape. Anthropogenic noise associated with shipping has emerged as a major disruptor of aquatic animal behavior worldwide. The Arctic marine realm is no exception, and indeed may be more vulnerable in that it has historically experienced little noise-generating human activity.

Ivanova et al. (2019) used a combination of acoustic telemetry and modeling of ship noise to examine the temporo-spatial habitat use of a key arctic forage fish, polar cod (*Boreogadus saida*), in the presence and absence of vessels in Resolute Bay, Canada, a high latitude Arctic embayment. The presence and movement of vessels induced a horizontal shift in the home ranges of Arctic cod with low core overlap when compared to periods without vessel activity. Results indicate that Arctic cod perceive vessel noise and presence as a threat and react by moving away and decreasing exploratory activities. These changes in fish behavior also coincide with the critical open water feeding period, suggesting an interruption in exploitation of important and seasonally abundant food resources, and carry broader implications for dependent seabirds and marine mammals, and indirectly for all Arctic indigenous peoples' subsistence and long-term cultural traditions.

The soundscape is an important habitat feature for marine animals, and sea ice loss and increased human activity may cause large changes to the Arctic marine soundscape. Halliday et al. (2020) examined the marine soundscape over eight months near Ulukhaktok, Northwest Territories, Canada, and assessed the relative contribution of the geophony (wind and wave sounds), biophony (marine mammal and fish sounds), and anthrophony (noise from vessel traffic). Sound pressure levels (SPL) were significantly higher during the summer than during the autumn and winter, and these differences were caused by increased wind/waves and vessel traffic in the summer. Increased wind speed drove increased SPL, while increased ice concentration resulted in decreased SPL. When vessel traffic was closer, SPL was higher. Overall, the marine soundscape near Ulukhaktok varied greatly through time and may be prone to large changes in the future as the ice-free season continues to lengthen and more vessels travel through the region.

With the disappearance of sea ice that provides refuge for many fish and zooplankton species, the interaction between forage fish and top predators will likely respond. LeBlanc et al. (2019a) tested the hypothesis that thick-billed murre (*Uria lomvia*), northern fulmar (*Fulmarus glacialis*) and black-legged kittiwake (*Rissa tridactyla*) congregate in areas of high polar cod food abundance and low ice concentration at different spatial scales across the marginal ice zone in Baffin Bay. At a large spatial scale (western versus eastern Baffin Bay), thick-billed murre, northern fulmar and their polar cod resource proxy were generally more abundant on the western ice-covered side of Baffin Bay. No clear spatial match was found, however, when comparing seabird abundances and their food-resource proxy in different ice concentrations across the marginal ice zone or at small scale (5 km). A lack of schooling behavior and a successful strategy to avoid predation by hiding under the ice could explain the absence of any strong spatial match between polar cod and its seabird predators at these different scales.

It will be important for the CAO ecosystem and marine living resources management to synthesize results from observations, laboratory experiments and models to showcase how the integration of scientific methods and indigenous knowledge can improve our understanding of (a) past and projected changes in environmental conditions and marine species; (b) their effects on social and ecological systems in the respective communities; and (c) support management and planning tools for climate change adaptation and mitigation. Steiner et al. (2019) used an integrated approach by linking climate-ecosystem-economic (CEE) models for the key forage fish Boreogadus saida and subsistence fisheries in the western Canadian Arctic and discussing uncertainties within those models by changing "this" to "an", it will be a general, beginning statement. Arctic cod can be trophically linked to sea-ice algae and pelagic primary producers and are key vectors for energy transfers from plankton to higher trophic levels (e.g., ringed seals, beluga), which are harvested by Inuit peoples. Fundamental changes in ice and ocean conditions in the region affect the marine ecosystem and fish habitat. Model simulations suggest increasing trends in oceanic phytoplankton and sea-ice algae with high interannual variability. The latter might be linked to interannual variations in Arctic cod abundance and mask trends in observations. CEE simulations incorporating physiological temperature limits data for the distribution of Arctic cod result in an estimated 17% decrease in Arctic cod populations by the end of the century (high emission scenario) but suggest increases in abundance for other Arctic and sub-Arctic species. Uncertainties remain with respect to detailed future changes, but general results are likely correct and in line with results from other approaches. This makes a case for the involvement of Arctic communities both in monitoring and decision making and illustrates the need for a precautionary and adaptive approach.

References

- Bouchard, Caroline & Geoffroy, Maxime & LeBlanc, Mathieu & Majewski, Andrew & Gauthier, Stéphane & Walkusz, Wojciech & Reist, James & Fortier, Louis. (2017). Climate warming enhances polar cod recruitment, at least transiently. Progress in Oceanography. 156. 121-129. 10.1016/j.pocean.2017.06.008.
- Copeman, Louise & Spencer, Mara & Heintz, Ron & Vollenweider, Johanna & Sremba, Angie & Helser, Thomas & Logerwell, Libby & Sousa, Leandra & Danielson, Seth & Pinchuk, Alexei & Laurel, Ben. (2020). Ontogenetic patterns in lipid and fatty acid biomarkers of juvenile polar cod (*Boreogadus saida*) and saffron cod (*Eleginus gracilis*) from across the Alaska Arctic. Polar Biology. 10.1007/s00300-020-02648-9.
- Halliday, William & Pine, Matthew & Mouy, Xavier & Kortsalo, Piia & Hilliard, Casey & Insley, Stephen. (2020). The coastal Arctic marine soundscape near Ulukhaktok, Northwest Territories, Canada. Polar Biology. 10.1007/s00300-020-02665-8.
- Ivanova, Silviya & Kessel, Steven & Espinoza, Mario & McLean, Montana & O'Neill, Caitlin & Landry, Justin & Hussey, Nigel & Williams, Rob & Vagle, Svein & Fisk, Aaron. (2019). Shipping alters the movement and behavior of Arctic cod (*B. saida*), a keystone forage fish in Arctic marine ecosystems. Ecological Applications. 30. 10.1002/eap.2050.
- Kaartvedt, Stein & Titelman, Josefin. (2018). Planktivorous fish in a future Arctic Ocean of changing ice and unchanged photoperiod. ICES Journal of Marine Science. 75. 2312-2318. 10.1093/icesjms/fsx248.
- Kühn, Susanne & Schaafsma, Fokje & van Werven, Bernike & Flores, Hauke & Bergmann, Melanie & Egelkraut-Holtus, Marion & Tekman, Mine B & Van Franeker, Jan. (2018). Plastic ingestion by juvenile polar cod (*Boreogadus saida*) in the Arctic Ocean. Polar Biology. 41. 10.1007/s00300-018-2283-8.
- LeBlanc, Mathieu & Gauthier, Stéphane & Garbus, Svend & Mosbech, Anders & Fortier, Louis. (2019a). The co-distribution of Arctic cod and its seabird predators across the marginal ice zone in Baffin Bay. Elem Sci Anth. 7. 1. 10.1525/elementa.339.
- LeBlanc, Mathieu & Geoffroy, Maxime & Bouchard, Caroline & Gauthier, Stéphane & Majewski, Andrew & Reist, James & Fortier, Louis. (2019b). Pelagic production and the recruitment of juvenile polar cod *Boreogadus saida* in Canadian Arctic seas. Polar Biology. 10.1007/s00300-019-02565-6.
- Nelson, R. & Bouchard, Caroline & Fortier, Louis & Majewski, Andrew & Reist, Jim & Præbel, Kim & Madsen, M. & Rose, G. & Kessel, Steven & Divoky, George. (2020). Circumpolar genetic population structure of polar cod, *Boreogadus saida*. Polar Biology. 10.1007/s00300-020-02660-z.

- Spencer, Mara & Vestfals, Cathleen & Mueter, Franz & Laurel, Ben. (2020). Ontogenetic changes in the buoyancy and salinity tolerance of eggs and larvae of polar cod (*Boreogadus saida*) and other gadids. Polar Biology. 1-18. 10.1007/s00300-020-02620-7.
- Steiner NS, Cheung WWL, Cisneros-Montemayor AM, Drost H, Hayashida H, Hoover C, Lam J, Sou T, Sumaila UR, Suprenand P, Tai TC and VanderZwaag DL (2019) Impacts of the changing ocean-sea ice system on the key forage fish Arctic cod (*Boreogadus saida*) and subsistence fisheries in the Western Canadian Arctic-evaluating linked climate, ecosystem and economic (CEE) Models. Front. Mar. Sci. 6:179. doi: 10.3389/fmars.2019.00179

4.6 Marine mammals in the CAO

11 species of marine mammals in the Northern hemisphere may be considered ice dependent based on the criterion that maximum population size is recorded in areas with at least seasonal sea ice cover. This list comprises seven species of pinnipeds (ringed seals (*Pusa hispida*), bearded seals (*Erignatus barbatus*), harp seals (*Pagohilus groenlandicus*), hooded seals (*Cystophora cristata*), spotted seals (*Phoca largha*), ribbon seals (*Histriophoca fasciata*), and walrus (*Odobenus rosmarus*), three species of whales (bowhead whales (Balaena mysticetus), beluga (*Delphinapterus leucas*), and narwhal (*Monodon monodon*), and polar bears (*Ursus maritimus*). Detailed descriptions of distribution, biology and population history of these species are found in the upcoming report 1 of WGICA.

All of the listed species have been observed in the CAO. In addition, some more temperate large pelagic-feeding cosmopolitan species like blue whales (*Balaenoptera musculus*), fin whales (*Balaenoptera physalus*), humpback whales (*Megaptera novaeangliae*) and minke whales (*Balaenoptera acutorostrata*). are increasingly observed off northern Svalbard during summer (Storrie et al. 2018). The same species are found in the North Pacific, but are rarely north of the Bering Strait (Moore et al. 2019). Pacific endemic gray whales (*Eschritius robustus*) are more commonly found in the Northern parts of the Chukchi Sea and Southern Beaufort Sea, but are not observed or expected to move off the shelf because they are primarily benthic feeders (e.g. Iwahara et al. 2016, Moore et al. 2019).

Based on current knowledge, the slopes and high seas of the CAO appear to be relatively scarcely populated by marine mammals. However, with increasing sea temperatures, the importance of the CAO may increase, especially for ice dependent species in adjacent areas. The importance of the WGICA focus area for any species can therefore only be adequately assessed if information on population status in adjacent areas is also taken into consideration. Updated summaries of known effects of climate change and other anthropogenic pressures on ice dependent marine mammal species observed in the CAO are given below.

4.6.1 Ecological Effects of Climate Change

4.6.1.1 Walruses

The community of ice dependent marine mammals in the Pacific Arctic region is dominated by benthic-feeding species, including a population of around 200000 walruses (Lowry 2016a, Beatty et al. 2019), which has been severely affected by reduced ice cover in their summer feeding habitat in the Chukchi Sea. Female pacific walruses (*Odobenus rosmarus divergens*) depend on the sea ice over shallow waters as a platform for nursing their calves and resting between feeding bouts. Over the past decades, the summer ice cover has retreated faster to deep waters off the shelf, where female walruses cannot successfully combine feeding and nursing (McCracken 2012). This has led to increased calf mortality due to abandonment on the ice or trampling and predation at crowded terrestrial haul-outs (Cooper et al. 2006, McCracken,2012). Panics leading to stampedes may be caused by human disturbance or polar bear attacks (Garlich-Miller et al. 2011).

I

Ovsyannikov (1995) described intensive harassment of walrus by polar bears on Wrangell Island in 1990, which was a year with severely reduced availability of sea ice haulouts. Numerous attacks were made by polar bears, but the direct success rate was low. Such attacks may, nevertheless, increase the mortality rate of walruses due to stampedes or stress/exhaustion and make them easily available for scavenging bears (Ovsyannikov 1995). Cases of killer whale predation on Pacific walruses have also been reported by Russian scientists and Alaska natives (Fay 1982; Kryukova and Kochnev 2014). Of 52 walrus carcasses washed ashore on St. Lawrence Island in 1951, 17 (33%) were estimated to have died from injuries consistent with killer whale predation (Fay 1982). Passive acoustic data collected from 2009 to 2016 in the southern Chukchi Sea showed an increase in the occurrence of killer whale vocalizations over time (Stafford 2018). Killer whale calls were detected every year and with increasing frequency in fall months (September - November). The earliest signals were detected in early June and the latest in November. Incidental sightings of killer whales have been reported since the 1950s, but the majority of reported sightings are from the period 2008-2015 and are primarily made in areas far offshore in the Northern Chukchi Sea. Reduced sea ice cover also weakens the benthopelagic coupling in the Chukchi Sea leading to reductions in the benthic prey base for walruses (Grebmeier et al. 2015). Walruses may diversify their diet and forage partly on pelagic species including seals and birds (Seymour et al. 2014, Clark et al. 2019), but movement studies have so far not indicated significant feeding off the shelf of the CAO (Jay et al. 2012). Surveys in the Eastern Chukchi Sea corroborate this finding (Kuletz et al. 2015). Very little is known about the ecology of the Laptev Sea population of pacific walruses, but they are generally reported to be rather stationary within polynyas on the Laptev Sea shelf (Spiridonov et al. 2011).

Atlantic walruses (Odobenus rosmarus rosmarus) in the Atlantic gateway area only number in the low thousands and appear to be slowly recovering after historical hunting pressures (Lydersen and Kovacs 2014, CAFF 2017). Due to low densities, calving and nursing in terrestrial summer haul-outs are likely not associated with significantly increased mortality in the East Greenland -Franz Josef Land area (Lydersen and Kovacs 2014). Telemetry data are available for males in these areas and show strong association with coastal waters during the summer feeding period and heavy ice during winter (Freitas et al. 2009). None of the available data, however, suggest significant use of waters off the slope of the CAO. Diet data show a predominantly low trophic benthic diet, but some individuals also feed on birds and mammals (Fox et al. 2010, Lydersen and Kovacs, 2014). Feeding on higher trophic levels is a strategy that could allow some walruses to forage from summer haul-outs on the ice edge in the CAO, but so far this does not appear to happen. A possible hint of more pronounced non-benthic feeding in walruses may, however, be suggested by the observed wintering of Atlantic walrus males on thick ice over relatively deep waters in the Barents Sea (Freitas et al. 2009). The males are not feeding during winter, but they are courting females, which are believed to be feeding during winter.

4.6.1.2 Bearded Seals

Bearded seals are generally considered benthic feeders, but they can also adopt more pelagic feeding strategies in areas with high availability of fish (Hindell et al.2012, Hamilton et al. 2018) and/or significant competition with other benthic feeders like walruses (Lowry et al. 1980).

Pacific bearded seals (*Erignatus barbatus nauticus*) tagged along the coast of the Eastern Bering and Chukchi Seas appear to aggregate in the Bering Strait region during winter and adopt a more northern distribution during summer, but few animals appear to use the slope areas of the CAO and Beaufort seas as summer feeding areas (Citta et al. 2018, Cameron et al. 2018). Bearded seal vocalisations have, however, been recorded year-round over the slope of the southern Beaufort Sea (MacIntyre et al. 2014). Call activity increased with the formation of pack ice in the winter and peaked in April-June, coinciding with the mating season. Mating usually occurs at the end

of a 3-4 week lactation period, suggesting that pupping may take place in this pack ice habitat. Studies of bearded seals harvested in the Bering and Chukchi Seas over the recent period of declining sea ice cover have not found clear signs of a shift towards a more pelagic prey composition (Crawford et al.2014). Biological parameters like residual length, blubber thickness and proportions of pups have also not shown any significant relationship with ice cover (Crawford et al. 2014). A decline in age at maturity by 1.6 years has, however, been observed along with an increased proportion of pups in the harvests, indicating increased postweaning reproductive success (Crawford et al. 2014).

In Svalbard, Atlantic bearded seals (*Erignatus barbatus barbatus*) have responded to a dramatic decline in fjord ice by pupping and nursing on small pieces of calved glacier ice. Estimates of pup survival, growth and condition has not changed between periods of contrasting ice conditions (Kovacs et al. 2020a). Bearded seal pups swim with their mothers within hours from birth and are thus much more robust than other arctic seal pups (Hamilton et al. 2019). At few months of age, they may dive down to 300m. Juveniles continue to dive deeper than adult seals during their first year, possibly due to competition with adult seals in the more lucrative shallow waters. Bearded seals may haul out on land, but so far, have not been observed to pup and nurse on land. As fjord ice and glaciers retreat from the coast, ice may therefore become a limiting factor for bearded seals in several of their current pupping areas (Kovacs et al. 2020a, McCracken et al.2012). Based on the general affiliation of bearded seals with shelf waters, it seems unlikely that pack ice over deep waters of the CAO could be used as pupping areas, but little visual survey data are available from these and other pack ice areas during the pupping period.

4.6.1.3 Ringed seals

Ringed seals have a circumpolar distribution with little genetic structure (Lowry 2016b, Martinez-Bakker et al. 2013). Tagging studies have nevertheless suggested pronounced female philopatry in some areas (e.g. Kelly et al. 2010) providing some stability to management units. Few estimates of ringed seal abundance are available, but the total population in the circumpolar Arctic is thought to number at least 1.4 million in the circumpolar Arctic (Lowry 2016b). They are the single most important prey of polar bears in throughout their range (e.g. Wiig et al. 2015). Due to their small size, ringed seal pups require snow cover for insulation for several weeks after birth (Lydersen and Kovacs 1999). Successful breeding therefore requires stable snow -covered ice, where ringed seal females can dig out a cave under the snow with underwater access through a hole in the ice beneath. These caves also offer protection from visual predators like gulls and to some extent polar bears and arctic foxes (Lydersen and Smith 1989). The breeding habitat must, however, also offer feeding opportunities for the female during the 6-7 week long lactation period (Lydersen and Kovacs 1999). The size of the foraging range of lactating females during winter has been found to vary in response to thickness of ice cover, which could be driven by changes in prey distribution (Harwood et al. 2015). Ringed seals are known to be able to maintain breathing holes in ice up to 2 meters thick, but the extra effort needed for maintaining holes in thick ice could also affect the foraging range. During the past decades of increasing temperatures in the Arctic, suitable pupping habitat has become scarce in many areas, including the west coast of Svalbard, where landfast ice has almost ceased to be formed in fjords that were previously prime breeding habitat (Lydersen et al. 2014). The degree to which, the pack ice area of the CAO is used as a breeding substrate is currently unknown. Pack ice breeding is, however, known to occur in the Barents Sea and eastern Canada (Finley et al. 1983, Wiig et al. 1999).

During the summer and fall feeding period, subadult ringed seals tagged in western Svalbard migrate to the ice edge to the north of Svalbard in summer (Hamilton et al. 2015), while adult ringed seals generally remain in Svalbard coastal waters (Hamilton et al. 2016). The increasingly long summer migrations of subadult ringed seals are expected to reduce their energy balance and potentially their growth rates and life-time reproductive output (Hamilton et al. 2015).

I

Glacial fronts are highly productive and appear to be a preferred habitat for ringed seals staying close to land (Lydersen et al. 2014, Weslawski et al. 1994). Ringed seals have, however, also been observed and sampled along the ice edge in the Barents Sea (Wathne et al. 2020). For these samples, occurrence of krill and amphipods in gastrointestinal tracts was almost as high as for polar cod, but a strong preference was shown for the latter (Wathne et al. 2000). Diet studies in western Svalbard in the 1980s also found a higher preference for fish (mainly polar cod but also some redfish (Sebastes spp) than for pelagic crustaceans (mainly Pandalus spp and amphipods) (Weslawski et al. 1994). Mysids were not eaten although they were abundant and of high energy density. No substantial changes in ringed seal diets around Svalbard were found by Labansen et al. (2007) during the period 2002-2004 or Bengtsson et al. (2020) for the period 2014-2017. So far no diet data are available for the pack ice areas of the CAO.

In the Pacific gateway region, the main breeding areas of ringed seals appear to be located close to the Bering Strait on both sides, with some breeding also occurring along the North Slope and in the Amundsen Gulf region bordering on the Southeastern Beaufort Sea (Kelly and Quakenbush 1990, Kelly et al. 2010, Citta et al. 2018). Surveys in the post breeding season for ringed seals in the Chukchi Sea showed highest densities in the southern areas and densities were expected to decline further towards the East, due to much lower overall productivity (Bengtsson et al. 2005).

Long term data series of ringed seal vital rates are available from the eastern Beaufort Sea and Amundsen Gulf area and offer insights to effects of environmental change on growth and reproductive rates. One data series uses a growth index based on relative thickness of cementum growth layers in teeth from ringed seals killed by polar bears (Nguyen et al. 2017. These data showed a general negative trend in body growth over the period 1965-2007. Among candidate ice-related explanatory variables, only timing of ice break-up was retained in the final model (Nguyen et al. 2017). This parameter was, however, only available from 1992. Galley et al. (2016) did not find any trend in timing of sea ice break-up over the period 1983-2014, but a significant linear decline in date of break-up was found by Harwood et al. (2020) based on local data for the eastern Amundsen Gulf. The significant effect of break-up time was, however, due to reductions in growth index in years with late ice break-up. These years also coincided with years of reduced ovulation rates of ringed seals in the Amundsen Gulf area (Harwood et al. 2012). In a later study based on an extended data series, Harwood et al. (2020) found a decline in blubber depth over the period 1991-2019 in both males and females. Ovulation rates were generally around 100 % but showed episodic reductions to 60-70% and even 40% in years of unusually late ice break-up. The ovulation rates correlated well with observed proportions of pups of the year in the summer harvest, which is a more reliable indicator of recruitment. The declines in blubber depth may suggest a negative trend in energy balance, but a similar trend is not seen in asymptotic lengths, which increased slightly for both males and females between the periods 1991-2011 and 2012-2019 (Harwood et al. 2020). Estimated age at first pregnancy was also lower in the latter period by 0.5 year. The effects of ice conditions on the fitness of ringed seals in the eastern Beaufort Sea area therefore appear to be complex.

Ringed seals tagged in the Chukchi Sea and western Beaufort Sea 2011-2016, generally did not go into the eastern Beaufort Sea, but spread out over the Chukchi Sea and western Beaufort Sea (Von Duyke et al. 2020). Ringed seals tagged in 2011, however, showed an unusual affinity for the pack ice off the shelf and were thought to represent a previously unsampled offshore ecotype (Von Duyke et al. 2020). Most of the tagged seals in this study spent winter in the Bering Sea, but appreciable numbers also stayed in the southern Chukchi Sea (Von Duyke et al. 2020). Crawford et al (2014) found significant increases in body growth rate and blubber thickness of harvested ringed seals with reduced sea ice cover in the Bering and Chukchi Seas between the periods 1975-84 and 2003-2012. Ringed seals also matured 2 years earlier in the latter period and there was an increase in the proportions of pups in the harvests (Crawford et al. 2014). Both growth of young
seals and the proportion of pups in the harvest were significantly negatively correlated with ice concentration. Diet studies over the same period showed a considerable increase in occurrence of fish species, particularly rainbow smelt (*Osmerus mordax*) and a decline in the occurrence of invertebrates, including the sympagic *amphipod Gammarus wilkitzkii* (Crawford et al. 2014). Polar cod, however, remained the dominant prey item across the sampling periods.

4.6.1.4 Ribbon seals and spotted seals

Satellite telemetry shows that some ribbon seals summer on the northern slope of the Chukchi Sea, where they may forage on several hundred meters depth (Boveng et al. 2013). Based on available evidence, however, the number of ribbon seals using this area appears to be small. The capacity for deep diving, however, suggests that ribbon seal could potentially increase their use of the waters off the shelf. Spotted seals are found in the Northern Chukchi Sea in summer, but appear to stay on the shelf (Citta et al. 2018). Abundances of these Pacific species in the Bering Sea area have been estimated at about 180000 for ribbon seals (Conn et al. 2014) and about 460000 spotted seals (Conn et al.2014).

4.6.1.5 Harp seals

Harp seals have both been observed in low numbers in the pack ice areas on the slopes of the CAO close to the Atlantic gateway (Haug et al. 2017) and may be potential prey for polar bears in the area. It is assumed that only harp seals from the Northeast Atlantic populations enter the CAO, as the access from to the CAO for the Northwest Atlantic populations is much more restricted.

The Barents Sea/White Sea harp seal population pups on ice in the White Sea and moults in a larger area also including the southern Barents Sea (Sergeant 1991). During spring they migrate through the Barents Sea to the edge of the sea ice (Nordøy et al. 2008). During the past decades, the summer ice edge has retreated from the Barents Sea to areas north of Svalbard (Hamilton et al. 2015). Harp seal diets along the ice edge during summer and autumn is mainly comprised by krill and amphipods but also some polar cod and capelin (e.g. Wathne et al. 2000, Lindstrøm et al. 2013). The increased distance to the ice edge may have played a role in the 50% reduction in pup production of Barents Sea/White Sea harp seals since 2003 (ICES, 2019). No pup production surveys have been conducted since 2013, but projections of previous data estimates current abundance at 1497189 (SD=104209) (ICES, 2019).

Harp seals from the Greenland Sea population breed and moult in pack ice areas off Northeast Greenland during spring and then embark on a feeding migration towards the Northeast along the ice edge into the Fram Strait, Barents Sea and to some extent also the waters north of Svalbard (Folkow et al. 2004, Haug et al. 2017). This population was estimated at 426 808 (SD =58 063) in 2018 (ICES, 2019). Studies within the Greenland Sea have shown diets dominated by amphipods, polar cod and some capelin (Haug et al. 2004, Enoksen et al. 2017). During winter, both harp seal populations typically switch to a more fish dominated diet close to the pupping area (Haug et al. 2004, Nilssen et al. 2000). Pup production of Greenland Sea harp seals also declined by 50% in 2018 compared to previous surveys over the period 2002 to 2012 (ICES 2019), but it is not yet known if this was just a single bad year. Over the past decades, killer whales have been observed to prey on harp seal pups from the Greenland Sea area (Foote et al.2013). The overall extent of this behavior is unknown but currently not thought to be extensive (Jourdain et al. 2019).

4.6.1.6 Hooded seals

Hooded seals are known to occur in low numbers in the pack ice areas on the slopes of the CAO close to the Atlantic gateway (Folkow et al. 1996, Haug et al. 2017). These seals likely belong to the Greenland Sea hooded seal population, which breeds and moults on the pack ice off northeast

I

Greenland. This population is currently on a historically low level of about 76 623 individuals (SD=9348) (ICES, 2019). A dramatic decline by about 80% occurred from the 1940s to the 1980s for this stock, probably due to overexploitation (Øigård et al. 2014). The lack of recovery after full protection since 2007 is, however, likely related to environmental changes. Increased predation from polar bears may be part of this explanation, but likely not all (Øigård et al. 2014).

Hooded seals forage over continental slopes of the northeast Atlantic during most of the year (Folkow et al. 1996, Folkow et al. 2010, Vacquie-Garcia et al. 2017a). A satellite-tracked pup of the year has even been tracked to the North Pole (Vacquie-Garcia et al. 2017a). Hooded seals are deep diving specialists foraging on a wide range of prey including arctic species like polar cod, the squid *Gonatus fabricii*, Greenland halibut and redfish (Folkow and Blix 1999, Enoksen et al. 2016). Three observations of vagrant hooded seals were made in the Beaufort Sea between 1970 and 1975 underlining the dispersal potential of this species (Burns and Gavin, 1980). Reductions in sea ice cover in the CAO should facilitate this type of dispersal, but no cases appear to have been recorded recently. It has, however, been noted that outbreaks of phocine distemper virus among marine mammals in the Pacific Arctic region have been caused by virus strains spread from the North Atlantic, presumably by a live vector (VanWormer et al. 2019).

4.6.1.7 Polar bears

Polar bear neonates only weigh about 0.7 Kg and depend on an insulating maternity den of snow for several months after birth in mid-winter (Blix and Lentfer 1979, Spady et al. 2007). Dens are generally dug into snow drifts on land but denning on sea ice is also observed in the Beaufort and Chukchi Seas, the CAO and off the coast of North Greenland (Laidre et al. 2020). Reproducing polar bear females may spend up to 7 months fasting in the den (Rode et al. 2018). Growth and early survival of the suckling cubs are therefore highly dependent on maternal resources accumulated prior to denning (Robbins et al. 2012, Rode et al. 2014) as well as resource availability immediately after emergence. In all of the world's 19 subpopulations of polar bears, choice of denning habitat and timing of emergence appears to have evolved to ensure access to ringed seal pups for the lactating females. Still, there are important ecological differences between subpopulations and habitats, also among the 8-11 subpopulations bordering on or situated in the CAO (e.g. Regehr et al. 2016). An important habitat characteristic is whether sea ice is divergent or convergent i.e. retreating from land during summer or available close to land all summer (Amstrup et al. 2008). Both of these regimes are represented in the Pacific and Atlantic gateway regions (see Fig.1). These are also among the most intensively studied areas with respect to polar bears and their interactions with physical, biological and anthropogenic elements of the Arctic ecosystem.

The Southern Beaufort Sea polar bear subpopulation is known to den both on land and on sea ice. Some of the recorded den locations are far off the shelf and located within the CAO (see Fig.2). Rode et al. (2018) found that survival was higher among cubs born in dens on land than in dens on sea ice over the period 1985–2013. Part of this difference was explained by longer denning duration, later emergence from dens and thicker snow cover on land-based dens. Some of these factors may, however, also have been influenced by prey availability prior to denning, which is thought to be greater close to shore than at the sea ice edge (Rode et al. 2018). In the southern Beaufort Sea in particular, bowhead blubber and carcasses left on the shore by hunters may imply a nutritional advantage to bears summering on land (e.g. Atwood et al. 2016, Olson et al. 2017). Both behavioural studies and stable isotope analyses confirm that shore-based bears use this resource to a considerable extent (Olson et al. 2017; Ware et al. 2017). The carcasses are shared with grizzly bears, which appear to be dominant when both species are present (Miller et al. 2015). Summer habitat is tightly linked to the choice of denning habitat. Olson et al (2017) found that all tagged females summering along the shore also denned on land and 61% of the females summering on the sea ice denned on the ice.

Between the periods 1985-1995 to 1996-2006, the proportion of Southern Beaufort Sea polar bears denning on sea ice declined by 20 % and then remained stable up to the end of the study in 2013 (Olsen et al. 2017). An average abundance of 907 (95% CI = 548 – 1,270) individuals was estimated for the Southern Beaufort polar bear population over the period 2001-2010 (Bromaghin et al. 2015) The decline in sea ice denning coincided with retreat of the sea ice and a decrease in the availability of multiyear ice preferred for denning (Fischbach et al. 2007, Olson et al. 2017). It also coincided with a decline in pup survival (Bromaghin et al. 2015) and several reports of nutritionally stressed bears in the sea ice habitat (Cherry et al. 2009, Rode et al. 2010). Polar bear survival returned to previous levels after 2006, but the proportion of females denning on sea ice has not changed (Rode et al. 2014, Olson et al. 2017). The females that do den on sea ice tend to do so further north, probably because availability of suitable multiyear ice has decreased in the southern areas (Olson et al. 2017). Significant differences in activity budgets and diets have been found between females summering on land and along the ice edge of the Beaufort Sea (Rogers et al. 2015, Olson et al. 2017, Ware et al. 2017). For example, more than half of the female polar bears summering in coastal areas of the Beaufort Sea visited bowhead carcasses on the shore and likely fed on whale blubber (Ware et al. 2017). Analyses of fatty acid composition of 569 polar bears from 2004-2012 show high seasonal variability in prey composition and better condition among adult males and females that had fed on bowhead carcasses in one or more seasons (McKinney et al. 2017). Coastal polar bears also preyed more on bearded seals during summer and autumn (McKinney et al. 2017). Overall, alternative prey types like bowhead whales, seabirds, bearded seals and beluga whales have made up about 50% of the diets of polar bears in the Southern Beaufort Sea from 2004-2014 (Bourque et al. 2020).

A separate polar bear subpopulation resides in the northeastern Beaufort Sea (=Northern Beaufort Sea subpopulation, PBSG 2019) but there is some overlap in distribution with the Southern Beaufort Sea population and the geographic delineation of these subpopulations is currently being revised (Stirling 2002, PBSG 2019). The habitat of the Northern Beaufort Sea subpopulation is classified as a convergent ice region (PBSG 2019), but the sea ice transport into this area by the Beaufort gyre is highly variable (Galley et al. 2016). In the past decade mainly annual ice has been present and the open water season has increased (Galley et al.2016). Polar bears from the Northern Beaufort Sea subpopulation den both on land and on sea ice (Stirling et al. 2002), but less is known about trends in habitat use for this subpopulation than for the Southern Beaufort Sea subpopulation. Polar Bears from the two subpopulations tagged over the period 1970-1998, partly overlapped in summer distributions in the Amundsen Gulf, but Northern Beaufort Sea polar bears were more prone to using the northern and eastern areas overlapping the harvesting areas of the ringed seals studied by Harwood et al. (2020) mentioned in the previous section. Stirling et al. (2011) estimated the abundance of Northern Beaufort Sea polar bears over the period 2000-2006 at 980 ± 155 (mean and 95% CI) and this estimate has not since been updated. This subpopulation is, however, expected to be in decline (PBSG 2019), due to indications of poor body condition and very few observed polar bears in previous high use areas (PBSG, 2019).

A third Pacific region subpopulation of polar bears is found in the Chukchi Sea. Based on markrecapture data collected 2008-2016, the abundance of this subpopulation has been estimated at 2937 95% CI = 1552-5944 (Regehr et al. 2018) in the American part of their distribution range. A similar number may be found on the Russian side based on surveys of denning areas in the 1980s (Belikov, 1993). In spite of increased rates of sea ice retreat, the body condition of Chukchi Sea polar bears was improving or stable between the periods 1986–1995 and 2008–2013 (Rode et al. 2014). This may be because suitable ice for hunting did not completely disappear from the productive shelf waters during spring and summer. Increased condition of ringed seals reported by Crawford et al. 2014 probably also had a positive effect on the polar bears. Between the periods 1986–1995 and 2008–2013, the proportion of polar bears spending > 7 days on land between August and October increased from 20.0% to 38.9%, and the average duration of their stay on land increased by 30 days (Rode et al. 2015). Most of the bears came to Wrangel and Herald Islands (Russia), which are also important denning areas (Rode et al. 2015). These islands are also increasingly used as summer haul-outs by walruses and polar bear predation on walruses has been reported by Ovsyannikov (1995) and Kochnev (2002). According to the latter, polar bears killed at least 226 walruses between 1989 and 2006, mainly by causing stampedes. About 20% of the Chukchi Sea polar bears summered on mainland Chukotka, where they were observed to feed opportunistically on marine and terrestrial mammal carcasses, such as gray whales, muskox (*Ovibos moschatus*) and reindeer (*Rangifer tarandus*) (Kochnev 2002). In addition, predation on live lemming (*Synaptomys* spp.) has been observed during lemming peak years. The population effect of these supplements to ringed and bearded seals is uncertain, but expected to be small, especially for terrestrial food resources (Rode et al. 2015).

The polar bear subpopulation inhabiting the Barents Sea area has experienced the fastest loss of sea-ice habitat of all the 19 recognized polar bear subpopulations (Laidre et al. 2015). Surveys around Svalbard and in the pack ice to the North of Svalbard have nevertheless indicated a stable population size over the period 2004 to 2015 of ~1000 bears (95% CI = 334 - 1026), possibly because the population was well below carrying capacity, when the sea ice reductions began (Aars et al. 2017). Retreat of the ice edge in the Barents Sea has led to abandonment of previous prime breeding habitat such as the Island of Hopen (Derocher et al. 2011). In addition, there has likely been an eastward shift in the distribution of dens from the east side of Svalbard to the Franz Josef Land archipelago (Aars et al. 2017). No recent counts are available from the Russian components of this subpopulation.

Polar bears around Svalbard are split into a smaller component summering in coastal waters and a larger offshore component summering along the edge of the pack ice, which has recently been situated over the slope of the CAO (Aars et al. 2017). Pregnant females summering offshore appear to return to Svalbard to breed on land, even if this requires swimming long distances (Lone et al. 2018). In spite of this, the offshore life style currently appears to be energetically superior to the coastal resident strategy (Blanchet et al. 2020). There are so far no signs of temporal trends in body condition or pup production in any of the two polar bear ecotypes in the Svalbard area (Aars et al. 2017). The resident bears also swim a great deal between feeding areas along the shores of Svalbard (Lone et al. 2018, Blanchet et al. 2020). Females with cubs of the year swim significantly less than other females, but they do in fact swim considerable distances, possibly carrying the pups on their back (Aars and Plumb 2010, Lone et al. 2018). Even cubs of the year appear to be able to survive being displaced several tens of kilometers through water (Lone et al. 2018). This may be a crucial factor for the ability of mothers to maintain sufficient resources for lactation after emergence from the den. Currently, many females are thus able to pursue an offshore life style even during the cubs first summer. With continued ice retreat, however, this will eventually become impossible and more of these females may have to stay on land, choose denning sites less affected by warm Atlantic water, or den on the drift ice itself.

Telemetry and observational data suggest that some polar bears around Svalbard have developed new foraging strategies suitable for extended periods in ice free coastal waters (e.g. Lone et al. 2018). These strategies rely on alternative hunting techniques like "aquatic stalking" of bearded seals and hunting for harbour seals, which are not ice-associated (Lone et al. 2018). These skills appear to be rare but could become the target of strong selection. Even so, predation on harbour seals likely cannot sustain the current number of polar bears in Svalbard as the Svalbard harbour seal population was only estimated at ~1800 individuals in 2010 (Merkel et al. 2013). Polar bears are also known to feed on harp seals, which are, however, generally only available, when ice is present. Smith and Stirling (2019) noted that harp seals react very slowly or not at all to polar bear attacks when hauling out on the ice in summer. In 2014, several polar bears were found to be scavenging on 9 different carcasses of white-beaked dolphins from spring through early autumn in three different areas in Northern Svalbard (Aars et al. 2015). One bear appeared to be hiding left-overs with snow for later use (Aars et al. 2015). The dolphin carcasses were likely a significant food source for the polar bears in the area that year, but similar incidences have not been reported since. The dolphins are believed to have been entrapped by ice while foraging in early spring in previously ice-covered areas. With increased presence of southern cetacean species in Svalbard, whale and dolphin carcasses could become an increasingly important food supply for polar bears, although they are likely to be only scarcely and sporadically available. Reindeer, seabirds and eggs are also occasionally consumed by polar bears in Svalbard (Stempniewicz et al. 2014, Prop et al. 2015, Aars et al. 2015, Lone et al. 2018). As terrestrial prey generally does not have as high a fat content as marine prey, these resources are likely of less energetic value, especially for adult females depending on fat accumulation for winter gestation and lactation (e.g. Rode et al. 2015). For cubs and subadults they may be more useful supplements, although not outweighing the loss of sea ice habitat.

The East Greenland subpopulation of polar bears inhabits a convergent sea ice region (PBSG, 2019). The southern and eastern extent of the ice has nevertheless decreased over the past decades and changed the hunting habitat of the polar bears (e.g. Strong 2012, AMAP 2017). Polar bears in East Greenland are believed to den primarily on land (Born et al. 2011). Recently, however, denning on sea ice off North and Northeast Greenland has been documented (Laidre et al. 2020). Precipitation is scarce in this area, but sufficient snow for denning is found around calvings of glacier ice stuck in the sea ice. The movement pattern of these northern polar bears is unknown, but areas bordering on the CAO could potentially be part of their habitat. Interviews with local hunters suggest that the occurrence of polar bears near settlements in East Greenland has increased in recent years (Laidre et al. 2018), possibly indicating an increase in abundance or a change in distribution pattern. The latter could be driven by ongoing changes in sea ice and snow cover, also noted by the hunters (Laidre et al. 2018). Changes in polar bear foraging patterns are supported by analyses of fatty acids, stable isotopes and contaminant profiles in blubber samples from East Greenland collected over the period 1984-2011 (McKinney et al. 2013). These analyses show a change from a diet dominated by ringed seals to a diet dominated by pack-ice seals, mainly hooded seals.

Overall, recent studies of polar bears in the gateways to the CAO reveal considerable levels of ecological plasticity and opportunistic behavior both with respect to habitat use, foraging strategies and diets. This could be important for the survival chances of polar bear species in the face of severe expected reductions due to climate change (Lone et al. 2018, Molnar et al. 2020). Understanding potential new foraging strategies and changes in denning habitat as they occur or change in frequency is also important for efforts to mitigate the effects of climate change on polar bears and their prey species (e.g. Prop et al. 2015) and their interactions with humans (e.g. Wilder et al. 2017).

4.6.1.8 Bowhead whales

Bowhead whales feed efficiently on small crustaceans both pelagically and on the sea floor (Simon et al. 2009). Relatively shallow waters may therefore be an optimal habitat for them as also suggested by their successful recovery from overexploitation in the Pacific Arctic region to recent estimates of ~16000 animals (Givens et al. 2016). Bowheads overwintering in the Bering Sea follow the ice edge through the Bering Strait and feed on copepods and krill in the Chukchi and Beaufort seas during summer (Citta et al. 2015). In this area, reduced sea ice cover has been accompanied by improved body condition of bowhead whales, especially juvenile whales (George et al. 2015). There has also been a considerable increase in the number of observed bowhead calves in recent years (Clarke et al., 2017). These responses are likely driven by increased pelagic productivity, partially due to increased coastal upwelling (Moore 2016).

The bowhead whale population around the Atlantic gateway is believed to be slowly recovering from historic overexploitation and is estimated to number in the low hundreds (Vacquie-Garcia

et al. 2017). A survey in 2015 estimated bowhead abundance to the North of Svalbard at 343 (CV = 0.488) (Vacquie-Garcia et al. 2017b). Many of these whales inhabit the pack ice over the slope of the CAO to the north of Svalbard (see Fig. 3). This summering habitat differs somewhat from typical bowhead habitat in the Pacific Arctic region where the whales are more strongly associated with the shelf waters (Citta et al. 2018). Observations of very large groups of bowheads have also been reported in the Fram strait pack ice areas in recent years (de Boer et al. 2019). Although bowhead whales may be responding positively to recent levels of sea ice reduction, they are nevertheless strongly ice associated and prefer waters below 2°C (Kovacs et al. 2020b). Modelling of future availability of preferred habitat for bowhead whales suggest that suitable areas may be reduced by 50% by 2050 (Kovacs et al. 2020b).

4.6.1.9 Beluga whales

The Pacific gateway region is one of the most important habitats for beluga whales globally with an estimated total of at least 60000 individuals during summer and fall. The majority of these belugas belong to the Beaufort Sea population, which was estimated at ~39000 individuals in 1992 (Muto et al. 2018). Subsequent studies of relative abundance have shown an increasing trend within a selected index area, but at least part of this is thought to be due to distribution changes (Harwood and Kingsley 2013). The genetically distinct Chukchi Sea population was estimated at 20752 (CV = 0.70) individuals in 2012 (Lowry et al. 2017).

There are significant differences in spatiotemporal distribution both between populations and between sexes within stocks in the Pacific Gateway region (Hauser et al. 2014). For all groups, however, dive data suggest that the main foraging depth is 200-300 meters (Hauser et al. 2015), which is consistent with a diet primarily composed by polar cod as indicated by fatty acid analyses (Loseto et al. 2009). On the shelf, most dives extend to the bottom and significant amounts of benthic organisms appear to be consumed including shrimp, octopuses, eciurid worms and amphipods (Quakenbush et al. 2015). Stable isotope values from Beaufort Sea and Chukchi Sea belugas suggest important contributions of both pelagic and benthic prey (Dehn et al. 2006, Horstmann-Dehn et al. 2012). This fits the mixed pattern of benthic and pelagic diving in the summer and fall dive data (Hauser et al. 2015). No diet data are available from the wintering areas in the Bering Sea.

Males generally spend most time in waters on and off the slope of the Beaufort Sea and Chukchi Plateau. Chukchi Sea males were particularly prone to explore waters in the central and northern Beaufort Sea bordering on the CAO (Hauser et al.2014). In deep water areas, belugas occasionally dive down to 900m depth, but the function of these dives is unclear (Hauser et al. 2015). Females spent more time than males in nearshore areas around the Mackenzie Estuary and Kasegaluk bay, which are the main moulting and nursing areas for the Beaufort Sea and Chukchi Sea belugas, respectively. Beluga habitat use of belugas in the Chukchi and Beaufort seas appears to be driven mainly by bathymetry and oceanographic features rather than ice cover (Hauser et al. 2018). Hence little change in home ranges and habitat use was observed over a period of retreating ice cover between 1998-2002 and 2007-2012. Longer ice- free season has, however, triggered a month delay in the migration of Chukchi Sea belugas from the western Beaufort Sea in 2007-2012 (Hauser et al. 2018). Chukchi Sea belugas also spent more time at greater depths than during 1998-2002. This could be indirectly driven by ice retreat, which may change the depths of productive zones through effects of increased wind exposure and resulting changes in frontal systems and mixing depths (Stafford et al. 2016).

Harwood et al. (2014) found some signs of decreasing size at age in belugas harvested in the Beaufort Sea from 1989-2008. This was mainly due to a trend over the period 1998 to 2008, which was, however, not separately analysed. Data on blubber thickness also showed a decline over the period 2000-2008, albeit not statistically significant. Moore and Kuletz (2019) noted that the potential decline in beluga blubber thickness co-occurred with negative responses in ringed seals

and black guillemots (*Cepphus grylle*) in the same area and suggested that a common factor may have affected pelagic food availability for these species. Interestingly, the same factor does not appear to have affected bowhead whales, which also feed pelagically albeit on a lower trophic level. The distribution areas of both Chukchi and Beaufort Sea beluga whales overlap with areas, where killer whales have been observed over the past decades (Stafford 2018). Very few cases of killer whale predation on belugas have, however, been reported for the Chukchi Sea from the 1950s to the 1990s (Lowry et al. 1987, George and Suydam 1998) and the present status is unknown.

In the Svalbard area, abundance of beluga whales has been estimated at 549 individuals (95% CI: 436-723) in 2018 (Vacquie-Garcia et al. 2020). They are almost exclusively inhabiting coastal waters and often forage along glacier fronts (Vacquie-Garcia et al. 2018). Fatty acid analyses suggest that polar cod is their primary prey (Dahl et al. 2000). Karlsen et al. (1999) found that Belugas in Svalbard waters were unusually quiet during summer and suggested this was due to a perceived threat from killer whales. This is also suggested to be a possible explanation for their strong preference for waters close to shore (Vacquie-Garcia et al. 2018). No instances of killer whale predation on belugas or other cetaceans have, however, been reported over the past decades from this area.

Beluga whales are also thought to be quite common around Franz Josef Land and even more so in the Kara and Laptev Seas (Belikov and Boltunov 2002, Solovyov et al. 2012). Some observations have also been made in the CAO to the north of these areas (Belikov and Boltunov 2002). Few belugas have been observed in the East Siberian Sea (Belikov and Boltunov 2002).

4.6.1.10 Narwhales

The Atlantic gateway to the CAO is dominated by the deep waters of the Northern Norwegian Sea, the Greenland Sea and the Nansen Basin, which are home to the Arctic-Atlantic endemic narwhals (Ahonen et al., 2019). This deep-diving species inhabits the ice-covered waters of the Fram Strait year-round and 837 (95% CI 314-2233) narwhals were estimated to occur in the western Nansen basin pack ice in summer 2015 (Vacquie-Garcia et al. 2017b, See Fig. 3). This estimate is thought to be negatively biased, as narwhals were still observed at the end of the transects (Vacquie-Garcia et al 2017b). Narwhals have generally been found to summer in coastal habitats and the presence of narwhals in the Nansen basin during this time of year is therefore very atypical for the species (Heide-Jørgensen et al. 2015, Vacquie-Garcia et al. 2017b). Narwhals generally feed most intensively during winter and spring (Watt et al. 2015), but it is not known, if the narwhals summering in the Nansen basin were feeding. Typical prey of narwhals are polar cod, Gonatus fabricii, shrimp, capelin, Greenland halibut and redfish. While narwhals in the Northwest Atlantic feed mainly on Greenland halibut, East Greenland narwhals have a broader niche and feed more on capelin than the Northwest Atlantic narwhals (Watt et al. 2013). The genetic affiliation of the narwhals in the Nansen basin is not known, but narwhals sampled in coastal waters of East Greenland and Svalbard appear to belong to genetically differentiated populations (Louis et al. 2020).

Abundance of narwhals in East Greenland was recently estimated at ~700 individuals, which is substantially less than a previous estimate of ~2700 individuals (NAMMCO 2019) and much less than an estimate of 6,444 (95% CI: 2,505–16,575) reported by Heide-Jørgensen et al. (2010) based on surveys in 2008. It is not clear, if these apparent reductions are due to a genuine decline, low precision in the estimates, or a shift in distribution. It is notable, however, that the recent survey did not detect any narwhals in the southernmost management region in southeast Greenland (NAMMCO 2019). The surveyed area only extended northward to Scoresby Sound fjord. Groups of at least 100 narwhals have been observed in areas further to the North along the coast, but no

abundance estimates exist for this area (Boertman et al. 2009, NAMMCO 2019). There are also no abundance estimates for narwhals in the Russian Arctic, but it is quite clear that the narwhal abundance in the area from eastern Greenland to Franz Josef Land is vastly lower than in the Northwest Atlantic, where total abundance is estimated around 170000 individuals (Lowry et al. 2017).

Overall, narwhals are characterized by very low genetic diversity, which has likely been maintained over long periods of their evolutionary history (Westbury et al.2019). A threefold increase in effective population size is, however, estimated to have occurred at the start of the Holocene in response to deglaciation (Louis et al.2020). According to Polyak et al. (2010) this period was in fact warmer than recent decades in the Arctic. Future climate scenarios are however predicted to reduce suitable narwhal habitat by 25 % (Louis et al. 2020). This result is partly due to the strong preference shown for cold waters in narwhal telemetry studies.

4.6.2 Effects of anthropogenic noise

Cetaceans and pinnipeds rely on passive and active acoustics for prey location, detection of predators, identification of conspecifics, and navigation. Sounds emitted from ship engines, seismic airguns, military, fisheries and research sonars may interfere with these vital behaviours and in some cases inflict physical hearing damages of temporary or permanent nature (Gomez et al. 2016, Southall et al.2019). Adverse effects are mainly expected within the auditory range of the animals although in rare cases very powerful directed sounds may generate other trauma at close range (Gray and Van Waerebaek, 2011)). With the exception of fisheries and research sonars, the listed sources of anthropogenic noise are primarily in the low-mid- frequency range (PAME 2019, Kvadsheim et al., 2020).

For many species, evaluations on sensitive hearing ranges are mainly based on recorded frequencies of sound production and sounds eliciting behavioral responses documented by visual observations, acoustic monitors or telemetric devices (Southall et al. 2019). Controlled experimental studies of total hearing range are only available for pinnipeds and a few toothed whales but no baleen whales. For several species analyses of auditory anatomy have played an important role in sensitivity evaluations. However, significant gaps still remain in our understanding of the biophysical principles of hearing in cetaceans, particularly in baleen whales. Hence, it was only recently discovered, that baleen whales likely receive sound impulses in water partly through a fat organ located on the sides of their heads (Yamato et al. 2012).

Based on a comprehensive review of available data on hearing ranges, Southall et al. (2019) assigned marine mammals to 6 different hearing groups of which 4 are known to be regularly present in the gateways to the CAO. These are 1) Low Frequency cetaceans (all baleen whales), 2) High frequency cetaceans (killer whales, narwhals, belugas and more), 3) Phocid carnivores (all phocid seals) and 4) Other carnivores (walrus, polar bears and more). Somewhat counterintuitively, several high frequency cetaceans are highly sensitive to low frequency anthropogenic sounds. This is illustrated by the strong response to shipping noise reported for belugas (Erbe and Farmer, 2000). High sensitivity to seismic noise for both belugas and narwhals is also suggested by recent studies in the Baffin Bay area (Kyhn et al. 2019). Both species appear to vacate ensonified areas at distances of more than 10 km from the source. Depending on the timing, location and duration of the disturbance, this may significantly affect feeding or other important behaviors. Little is, however, known about the actual behavioral state of the animals during the reported noise exposures. Narwhals are generally known to feed little during summer, which is the main season of anthropogenic noise in their habitats (Ahonen et al. 2017, Kyhn et al. 2019).

Bowhead whales are also more likely to move away from seismic airgun noise when they are migrating (Richardsson et al. 1999) than when they are feeding (Koski et al. 2009). Feeding

bowheads whales exposed to seismic airgun noise have, however, been found to reduce their surfacing and respiration rates (Robertson et al. 2016), which could affect the energetic costs of diving. It could also affect their foraging efficiency by disturbing the pattern of short exploratory dives before longer lasting feeding dives reported by Simon et al. (2009). Exposure to low frequency anthropogenic noise has also been shown to increase the levels of stress hormones in baleen whales even in the absence of clear behavioral responses (e.g. Rolland et al. 2012).

In some cases, observed noise tolerance levels in marine mammals may pose a risk of temporary or permanent hearing disabilities (threshold shifts). This was suggested by Harris (2001) for seals (mainly ringed seals) observed to remain close to a seismic survey vessel during a full range seismic operation in the Beaufort Sea. No differences in dive rates were observed between seals within 150 from the ship and >250 m from the ship.

Phocid seals exhibit the most sensitive low-frequency hearing abilities among marine mammal species for which audiometric data are available (Reichmuth et al., 2013; Erbe et al., 2016). They could therefore be especially vulnerable to seismic airgun noise, which is predominately concentrated below 1 kHz. No signs of temporal threshold shift were, however, recorded by Reichmut et al. (2016) exposing 2 ringed seals and 2 spotted seals to single pulses of seismic airgun noise of 100 Hz at levels predicted to generate TTS by Southall et al. (2007). Unlike the reported studies on seismic exposures, Kvadsheim et al. (2010) found strong evasive reactions to simulated sonar signals between 1-7 kHz at source levels from 134-194dB (RMS, re1uPa@1m). The sound pressure level inside the net cage was 10-27dB below the source level. Initial responses included rapid surface swimming away from the sound source. The horizontal escape reaction soon stopped but the seals generally remained at the surface with their heads out of the water. Heart rates were increased compared to pre-exposure levels during surface time, but not during dives (Kvadsheim et al. 2010). More studies using comparable metrics for exposure levels are, however, needed to understand thresholds for auditory effects in seals exposed to low frequency noise. Depending on temporal overlaps, there is also a clear potential for masking of phocid vocalizations, which has not received much attention.

The calls and mating songs of bowhead whales and other baleen whales are also prone to masking by low frequency anthropogenic noise. At low levels of masking noise, singing bowhead whales respond by increasing the volume, but beyond a certain threshold level, the whales stop vocalizing (Blackwell et al. 2015). Critical levels of this type of disturbance are unknown, but past levels in the Pacific Arctic region have not prevented an observed population increase (Givens et al. 2016). In contrast to the Pacific Arctic soundscape, almost no overlap between bowhead mating displays and anthropogenic noise occurs in the western Fram Strait, which is the preferred mating area of the endangered East-Greenland Spitsbergen bowhead whale population (Ahonen et al. 2017, Hiemer et al. 2020). Distant seismic noise and occasional local shipping noise is, however, heard during summer, concurrently with vocalizations by narwhals and mating calls from bearded seals. This study is an important base line study of a critical habitat for arctic marine mammals prior to a potential increase in anthropogenic activity due to reduced ice occurrence. Similar studies are available for the Pacific Arctic region (Moore et al. 2012, Stafford et al. 2018, Halliday et al. 2020) and would be highly valuable other areas of the CAO close to areas of high biological activity such as the Nansen basin, Laptev Sea and East Siberian Sea.

4.6.3 Knowledge gaps and monitoring needs for marine mammals

Among the research and monitoring needs for marine mammals in the CAO are:

1. Continued and expanded monitoring of marine mammal occurrence in the CAO

Monitoring of geographic and seasonal occurrence patterns is of fundamental importance for the understanding of highly mobile species in a changing environment. Historically, monitoring of

marine mammals in the CAO has mainly relied on opportunistically collected data from ice breakers, aerial ice monitoring programs and various research and tourist expeditions (e.g. Belikov and Boltunov 2002). Collection, collation and analyses of this type of data is highly valuable and should continue and expand with the increase of human activities in the region. Efforts should be made to optimize the quality, synthesis and dissemination of opportunistic observations. In addition, passive acoustic monitoring stations within the CAO would valuable tools for the majority of the arctic marine mammals. Based on prior indications and judgement of ecological habitat suitabilities, reconnaissance surveys should be carried out in selected areas to evaluate the need and feasibility of specialized abundance surveys.

3. Continued and expanded dedicated abundance surveys in the CAO

Continued and geographically expanded dedicated abundance surveys for ice dependent species like narwhals, polar bears, seals and bowheads around the CAO ice edge are needed for responsible management of human activities. In particular, a more complete survey of narwhals in the Nansen basin should be a near future priority as this population component has shown high affinity to conditions prevailing in the CAO. It is furthermore likely to constitute a significant proportion of the total abundance of a distinct grouping of narwhals to the East of Greenland.

4. Telemetry based studies of marine mammal movements in the CAO

Satellite telemetry, preferably with oceanographic tags, provide highly valuable information on movement patterns, habitat selection, diving behavior, potential diets and population affiliation. Currently available telemetry data for marine mammals in the CAO are generally based on animals originally tagged outside the area, which may be biased towards short term visitors. Tagging animals already within the CAO would be more likely to reveal information on potential long-term residents of the CAO. Elucidating the population affiliation of Nansen basin narwhals is of particularly high importance as these animals are likely to represent a significant proportion of the total abundance of narwhals to the East of Greenland.

5.Genetic studies

Genetic assignment of population affiliation of marine mammals observed in the CAO is a high priority for all species. In addition, genetic samples may provide information on sex and some physiological parameters through quantitative analyses of gene expression (Sigsgaard et al. 2019). High quality DNA from individual biopsies provide the most straightforward technical basis for these analyses, but faecal dna or environmental DNA from targeted areas may provide similar information with extended protocols (Sigsgaard et al. 2019). In addition, eDNA may provide community data on suites of species.

6.Studies of life history, health and ecology based on biological material

For marine mammals observed in the CAO access to biological material from live animals for assessment of health, age and reproductive status are generally limited to biopsies, faeces, cells in exhaled air, and occasional live sampling of blood samples, epithelial cells, hair and whiskers. These samples can provide useful data on reproductive status (sex hormones), health (stress hormones, contaminants, pathogens and antibodies), trophic information (stable isotope and fatty acid analyses). With the exception of data collected from live-captured animals for tagging, many of these techniques are, however, highly opportunistic and time consuming and unlikely to be widely operational for realistic levels of field time in the CAO. For abundant species like ringed seals, lethal sampling may be defendable for more detailed analyses of health and reproductive status, contaminant levels, age and diet. Similar data derived from indigenous and commercial hunting have previously proven highly useful for arctic populations (e.g. Nguyen et al. 2017, Beatty et al. 2019). The relevance of available hunted samples for CAO related studies, should be assessed.

7. Analyses of soundscapes relevant for marine mammals within the CAO

In addition to data on occurrence patterns of vocal marine mammals, general analyses of CAO soundscapes are important for the overall understanding of marine mammal acoustic ecology in the area. This is because levels of both anthropogenic noise and sounds from natural sources like wind and wave action are likely to change in response to changes in ice cover (Stafford et al. 2018). Soundscape data can be obtained from passive acoustic monitoring devices. Baseline studies for the pacific and arctic gateways are available (Moore et al. 2012, Ahonen et al. 2017, Stafford et al. 2018).

8. Studies of arctic mammal sensitivity to low frequency anthropogenic noise

Although several studies are available on the sensitivity of bowhead and beluga whales to seismic airguns and vessel noise, there are still considerable gaps in the knowledge base required for vulnerability assessments of marine mammals in the CAO. Very few studies of behavioural responses to noise exist for other arctic marine mammals and no studies appear to exist on effects of military sonars on arctic marine mammals (Gomez et al. 2016). With expected increases in industrial and military activity in the CAO and other parts of the Arctic, these gaps need to be filled.

Since marine mammals in the CAO are most likely closely connected with larger populations outside the core area of the CAO, continued monitoring of these more central populations are important in order to put the information from the CAO into a population and species-wide context. Generally agreed circumpolar research and monitoring recommendations for arctic mammals are summarized in CAFF (2017).

References

- Aars J. and Plumb A. 2010. Polar bear cubs may reduce chilling from icy water by sitting on mother's back. *Polar Biol.* **33**, 557–559
- Aars J. Andersen, M., Brenière A. and Blanc S. 2015. White beaked dolphins trapped in the ice and eaten by polar bears, Polar Research, 34:1, 26612, DOI: 10.3402/polar.v34.26612
- Aars J, Marques T.A., Lone K., Andersen M. and others. 2017. The number and distribution of polar bears in the western Barents Sea. Polar Res 36: 1374125
- Ahonen H., Stafford K.M., De Steur L., Lydersen C., Wiig Ø. and Kovacs K. 2017. The underwater soundscape in western Fram Strait: Breeding ground of Spitsbergen's endangered bowhead whales. Marine Pollution Bulletin 123 (2017) 97–112
- AMAP 2017. Snow, Water, Ice and Permafrost in the Arctic (SWIPA) 2017. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway. xiv + 269 pp
- Amstrup S.C., Marcot B.G. and Douglas D.C.. 2008. A Bayesian Network Modeling Approach to Forecasting the 21st Century Worldwide Status of Polar Bears, in Arctic Sea Ice Decline: Observations, Projections, Mechanisms, and Implications. Geophys. Monogr. Ser., Vol 180, edited by E.T. DeWeaver, C.M. Bitz, and L. B. Tremblay, pp. 213-268, AGU, Washington, D.C., Mechanisms and Implications. Geophysical Monograph 180. American Geophysical Union, Washington, D.C.
- Atwood T. C., et al. 2016. Rapid environmental change drives increased land use by an Arctic marine predator. PLoS One 11e0155932.
- Beatty W.S., Lemons P.R., Sethi S.A., Everett J., Lewis C.J., Olsen J.B., Lynn R.J., Cook G.M., Garlich-Miller J and Wenburg, J.K. 2019. Estimating Pacific Walrus Abundance and Demographic Rates from Genetic Mark-Recapture. Anchorage, Alaska, US Department of the Interior, Bureau of Ocean Energy Management. OCS Study BOEM 2019-059. 18 p.
- Belikov S.E. 1993. The polar bear. Pp. 420–478 In Vaisfeld, M.A., and Chestin, I.E. (eds.). *Bears: Distribution, Ecology, Use and Protection*. Moscow, Nauka. [In Russian with English summary].

- Belikov S. E. and Boltunov A. N. 2002. Distribution and migrations of cetaceans in the Russian Arctic according to observations from aerial ice reconnaissance. In: M. P. Heide-Jørgensen and O. Wiig (eds), *Belugas in the North Atlantic and Russian Arctic*, pp. 69-86. NAMMCO Scientific Publications 4.
- Bengtson J L., Hiruki-Raring L M.; Simpkins M. A. and Boveng P. L.2005. Ringed and bearded seal densities in the eastern Chukchi Sea, 1999–2000. *Publications, Agencies and Staff of the U.S. Department of Commerce.* Paper 143. <u>http://digitalcommons.unl.edu/usdeptcommercepub/143</u>
- Bengtsson O., Lydersen C., Kovacs K.M. and Lindström U.2020. Ringed seal (*Pusa hispida*) diet on the west coast of Spitsbergen, Svalbard, Norway: during a time of ecosystem change. Polar Biology (2020) 43:773–788 https://doi.org/10.1007/s00300-020-02684-5
- Blackwell SB, Nations CS, McDonald TL, Thode AM, Mathias D, Kim KH, et al. (2015) Effects of Airgun Sounds on Bowhead Whale Calling Rates: Evidence for Two Behavioral Thresholds. PLoS ONE 10(6): e0125720. <u>https://doi.org/10.1371/journal.pone.0125720</u>
- Blackwell SB, Tervo OM, ConradAS, Sinding MHS, Hansen RG, Ditlevsen S, et al. (2018) Spatial and temporal patterns of sound productionin East Greenland narwhals. PLoS ONE 13(6): <u>e0198295.https://doi.org/10.1371/journal.pone.0198295</u>
- Blanchet M-A., Aars J., Andersen M., Routti H.2020.Space-use strategy affects energy requirements in Barents Sea polar bears . Marine Ecology Progress Series. Vol. 639: 1–19, <u>https://doi.org/10.3354/meps1329</u>
- Blix A.S. andLentfer J.W.1979. Modes of thermal protection in polar bears cubs at birth and on emergence from the den. Am J Physiol 263: 67–74
- Boertmann D., Olsen K. and Nielsen R.D. 2009. Seabirds and marine mammals in Northeast Greenland. Aerial surveys in spring and summer 2008. NERI Technical Report no. 721. National Environmental Research Institute, Aarhus University, Denmark.
- Born E.W., Dietz R., Wiig Ø., Aars J., and Andersen M. 2009. Polar bear Ursus maritimus. Pp. 93–100 In Boertmann, D., Mosbech, A., Schiedek, D., and Johansen, K. (eds.). The Western Greenland Sea. A Preliminary Strategic Environmental Impact Assessment of Hydrocarbon Activities in the KANUMAS East Area. NERI Technical Report No. 719, National Environmental Research Institute, Aarhus University, Denmark
- Born E.W., Laidre K.L., Dietz R., Wiig Ø., Andersen M., Rosing-Asvid A., Ugarte F., Aars J., Rasmussen LM 2011 In: Boertmann D, Mosbech A (eds) The western Greenland Sea: A strategic environmental impact assessment of hydrocarbon activities. Aarhus University, DCE – Danish Center for Environment and Energy. Scientific Report from DCE – Danish Centre for Environment and Energy no. 22: 268, pp 100– 148
- Bourque J., Atwood, T.A., Divoky, J.G., Stewart C., McKinney, M.A.2020. Fatty acid-based diet estimates suggest ringed seal remain the main prey of southern Beaufort Sea polar bears despite recent use of onshore food resources. Ecology and Evolution 10: 2093-2103. doi.org/10.1002/ece3.6043
- Boveng P. L., Bengtson J.L., Cameron M.F., Dahle S.P., Logerwell E.A, London J.M., Overland J.E., Sterling, J.T., Stevenson D.E., Taylor B.L., and Ziel J.L. 2013. Status review of the ribbon seal. U.S. Dep. Commer., NOAA Tech. Memo. NMFSAFSC-255, 174 p.
- Bromaghin J. F., et al. 2015. Polar bear population dynamics in the southern Beaufort Sea during a period of sea-ice decline. Ecological Applications 25:634–651.
- Burns J.J. and Gavin A. 1980. Recent records of hooded seals, Cystophora cristata Erxleben, from the western Beaufort Sea. Arctic 33: 326-329.
- CAFF 2017. State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland. 978-9935-431-63-9
- Calvert W. and Stirling, I. 1990. Interactions between Polar Bears and Overwintering Walruses in the Central Canadian High Arctic. Bears: Their Biology and Management, Vol. 8, A Selection of Papers from the Eighth International Conference on Bear Research and Management, Victoria, British Columbia, Canada, February 1989 (1990), pp. 351-356 Published by: International Association of Bear Research and Management Stable URL: <u>http://www.jstor.org/stable/3872939</u>

- Cameron M.F., Frost K.J., Ver Hoef J.M., Breed G.A., Whiting A.V., Goodwin J., et al. 2018. Habitat selection and seasonal movements of young bearded seals (Erignathus barbatus) in the Bering Sea. PLoS ONE 13(2): e0192743.https:// doi.org/10.1371/journal.pone.0192743
- Heide-Jorgensen, M.P., 2015. Ecological characteristics of core-use areas used by Bering-Chukchi-Beaufort (BCB) bowhead whales, 2006–2012. Prog. Oceanogr.136, 201–222.
- Cherry S.G., Derocher A.E., Stirling I, and Richardson E.S.2009. Fasting physiology of polar bears in relation to environmental change and breeding behavior in the Beaufort Sea. Polar Biol 32: 383–391
- Citta J.J., Lowry L.F., Quakenbush L.T., Kelly B.P., Fischbach A.S., London J.M., Jay C.W., Frost K., O'Corry-Crowe G., Crawford J., Boveng P.L., Cameron M., Von Duyke A.L., Nelson M., Harwood L.A., Richard P., Suydam R., Heide-Jørgensen M.P., Hobbs R.C., Litovka D.I., Marcoux M., Whiting A., Kennedy, A.S., George, J.C., Orr, J. and Gray, T. 2018. A multi-species synthesis of satellite telemetry data in the Pacific Arctic (1987–2015): Overlap of marine mammal distributions and core use areas. Deep-Sea Research Part II 152 132–153
- Clarke J.T., Brower, A.A., Ferguson, M.C., Willoughby, A.L., 2017. Distribution and Relative Abundance of Marine Mammals in the Eastern Chukchi and Western Beaufort Seas, 2016. Annual Report, OCS Study BOEM 2017-078.
- Clark C. T., Horstmann, L., de Vernal A., Jensen A. M., Misarti N. 2019. Pacific walrus diet across 4000 years of changing sea ice conditions. Quaternary Research 1–17.
- Conn P.B., Ver Hoef J.M., McClintock B.T., Moreland E.E., London J.M., Cameron M.F., Dahle S.P., Boveng P.L. 2014. Estimating multispecies abundance using automated detection systems: ice-associated seals in the Bering Sea. *Methods in Ecology and Evolution* 5: 1280-1293.
- Cooper L.W., C.J. Ashjian, S.L. Smith, L.A. Codispoti, J.M. Grebmeier, R.G. Campbell and E.B. Sherr. 2006. Rapid seasonal sea-ice retreat in the Arctic could be affecting Pacific walrus (Odobenus rosmarus divergens) recruitment. Aquatic Mammals 32:98-102.
- Craighead J., Suydam Gand Suydam R. 1998. Observations of killer whale (Orcinus orca) predation in the northeastern Chukchi and western Beaufort Seas. Marine Mammal Science 14 : 330-332, DOI: 10.1111/j.1748-7692.1998.tb00722.x
- Crawford J.A., Quakenbush L.T. and Citta J.J.2014. A comparison of ringed and bearded seal diet, condition and productivity between historical (1975-1984) and recent (2003-2012) periods in the Alaskan Bering and Chukchi seas. Progress in Oceanography. 2015; 136:133±50. https://doi.org/10.1016/j.pocean.2015.05.011 PubMed PMID: WOS:000358626900009
- Crawford J.A., Frost K.J., Quakenbush L.T. and Whiting A. 2019. Seasonal and diel differences in dive and haul-out behavior of adult and subadult ringed seals (Pusa hispida) in the Bering and Chukchi seas. Polar Biology (2019) 42:65–80 <u>https://doi.org/10.1007/s00300-018-2399-x</u>
- Dahl T.M., Lydersen C., Kovacs K.M., Falk-Petersen S., Sargent J., Gjertz I. and Gulliksen, B., 2000. Fatty acid composition of the blubber in white whales (Delphinapterus leucas). Polar Biol. 23, 401–409.
- De Boer M.N., Janinhoff N., Nijs G., Verdaat H. 2019. Encouraging encounters: unusual aggregations of bowhead whales Balaena mysticetus in the western Fram Strait. Endang Species Res 39:51-62. <u>https://doi.org/10.3354/esr00948</u>
- Dehn L.A., Follmann E.H., Rosa C., Duffy L.K. and others.2006. Stable isotope and trace element status of subsistence hunted bowhead and beluga whales in Alaska and gray whales in Chukotka. Mar Pollut Bull 52: 301–319
- Derocher A.E., Wiig Ø. and Andersen, M. 2002. Diet composition of polar bears in Svalbard and the western Barents Sea. Polar Biol 25: 448–452: DOI 10.1007/s00300-002-0364-0
- Derocher A. *et al.* Sea ice and polar bear den ecology at Hopen Island, Svalbard. 2011. *Mar. Ecol. Prog. Ser.* **441**, 273–279.
- Erbe C. and FarmerD.M. 2000. Zones of impact around icebreakers affecting beluga whales in the Beaufort Sea. The Journal of the Acoustical Society of America 108, 1332 (2000); doi: 10.1121/1.1288938

- Erbe C., Reichmuth C., Cunningham K., Lucke K., and Dooling R. 2016. Communication masking in marine mammals: A review and research strategy. Mar. Pollut. Bull. 103, 15–38.
- Enoksen S., Haug T., Lindstrøm U. and Nilssen K.T.2016. Recent summer diet of hooded Cystophora cristata and harp Pagophilus groenlandicus seals in the drift ice of the Greenland Sea Polar Biol DOI 10.1007/s00300-016-2002-2
- Fay F.H. 1982. Ecology and biology of the Pacific walrus, *Odobenus rosmarus divergens* illiger. U.S. Fish and Wildlife Service, North American Fauna, Washington, D. C., 279 pp.
- Finley K. J., G. W. Miller, R. A. Davis, and W. R. Koski. 1983. A distinctive large breeding population of ringed seals (Phoca hispida) inhabiting the Baffin Bay pack ice. Arctic 36:162-173
- Fischbach A.S., Amstrup S.C. and Douglas D.C.2007. Landward and eastward shift of Alaskan polar bear denning associated with recent sea ice changes. Polar Biol (2007) 30:1395–1405. DOI 10.1007/s00300-007-0300-4
- Fox A.D., Fox G.F., Liaklev A. and Gerhardsson N.2010. Predation of flightless pink-footed geese (Anser brachyrhyncus) by Atlantic walruses (Odobenus rosmarus rosmarus) in southern Edgeøya, Svalbard. Polar Res. 29(3):455–457. doi: http://dx.doi.org/10.1111/j.1751-8369.2010.00180.
- Folkow, L.P., Mårtensson, P.E. and Blix, A.S., 1996. Annual distribution of hooded seals (Cystophora cristata) in the Greenland and Norwegian Seas. *Polar Biology*, *16*(3), pp.179-189.
- Folkow L.P. and Blix, A.S. 1999. Diving behaviour of hooded seals (Cystophora cristata) in the Greenland and Norwegian Seas. Polar Biol 22: 61-74
- Folkow L.P., Nordøy E.S. and Blix A.S.2004. Distribution and diving behaviour of harp seals
- (Pagophilus groenlandicus) from the Greenland Sea stock.Polar Biol (2004) 27: 281–298 DOI 10.1007/s00300-004-0591-7
- Folkow L. P., Nordøy E. S. and Blix A. S. 2010. Remarkable development of diving performance and migrations of hooded seals (Cystophora cristata) during their first year of life. Polar Biol. 33, 433-441.
- Foote A.D., Newton J., Avila-Arcos M.C., Kampmann M.-L., Samaniego J.A., Post K., Rosing-Asvid A., Sinding M.-H.S., Gilbert M.T.P., 2013. Tracking niche variation over millennial time scales in sympatric killer whale lineages. Proc. Roy. Soc. B 280, 20131481, 9 pp.
- Freitas C., Kovacs K.M., Ims R.A., Fedak M.A. and Lydersen C.2009.Deep into the ice. Over-wintering
- and habitat selection in male Atlantic walruses. Mar. Ecol. Progr. Ser. 375:247–261. doi: http://dx.doi.org/10.3354/meps07725
- George J.C., Druckenmiller M.L., Laidre K.L., Suydam R.and Person B., 2015. Bowhead whale body condition and links to summer sea ice and upwelling in the Beaufort Sea. Prog. Oceanogr. 136, 250–262.
- Givens G.H., Edmondson S.L., George J.C., Suydam R., Charif R.A., Rahaman A., Hawthorne D., Tudor B., DeLong R.A. and Clark C.W., 2016. Horvitz-Thompson whale abundance estimation adjusting for uncertain recapture, temporal availability variation, and intermittent effort. Environmetrics. https://doi.org/10.1002/env.2379.
- Galicia M. P., Thiemann G.W., Dyck M.G., Ferguson S.H., and Higdon J.W.. 2016. Dietary habits of polar bears in Foxe Basin, Canada: possible evidence of a trophic regime shift mediated by a new top predator. Ecology and Evolution 6:6005–6018.
- Galley R. J., Babb D., Ogi M. Else, B.G.T., Geilfus N.-X., Crabeck O.,Barber D.G., and Rysgaard S. 2016. Replacement of multiyear sea ice and changes in the open water season duration in the Beaufort Sea since 2004, J. Geophys. Res. Oceans, 121, 1806–1823, doi:10.1002/2015JC011583.
- Garlich-Miller J., MacCracken J.G., Snyder J., Meehan R., Myers M., Wilder J.M., Lance E. and Matz A. 2011. Status Review of the Pacific Walrus (Odobenus rosmarus divergens)
- Garlich-Miller S., Wilder J. and Wilson.R.R. 2015. Polar bear-grizzly bear interactions during the autumn open-water period in Alaska. Journal of Mammalogy, 96(6):1317–1325 DOI:10.1093/jmammal/gyv140

- George J.C. and R. Suydam.R. 1998. Observations of killer whale (Orcinus orca) predation in the northeastern Chukchi and western Beaufort Seas. Marine Mammal Science 14(2): 330–332.
- George J.C., Druckenmiller M.L., Laidre K.L., Suydam, R. and Person, B. 2015. Bowhead whale body condition and links to summer sea ice and upwelling in the Beaufort Sea. Progress in Oceanography 136 (2015) 250–262
- Gomez C., Lawson J.W., Wright A.J., Buren A.D., Tollit D. and LesageV. 2016. A systematic review on the behavioural responses of wild marine mammals to noise: the disparity between science and policy. Can. J. Zool. 94: 801–819 (2016) dx.doi.org/10.1139/cjz-2016-0098
- Gray H, and Van Waerebeek K.2011. Postural instability and akinesia in a pantropical spotted dolphin, Stenella attenuata, in proximity to operating airguns of a geophysical seismic vessel. Journal for Nature Conservation 19 (2011) 363–367
- Grebmeier J.M., Bluhm B.A., Cooper L.W., Danielson S.L., Arrigo K.R., Blanchard A.L., Clarke, J.T., Day R.H., Frey K.E., Gradinger R.R., Kedra M., Konar B., Kuletz K.J., Lee S.H., Lovvorn J.R., Norcross B.L., Okkonen S.R., 2015. Ecosystem characteristics and processes facilitating persistent macrobenthic biomass hotspots and associated benthivory in the Pacific Arctic. Prog. Oceanogr. 136, 92–114
- Halliday W.D., Scharffenberg K., Whalen D., MacPhee S.A., Loseto L.L., Insley S.J.2020. The summer soundscape of a shallow-water estuary used by beluga whales in the western Canadian Arctic. Arctic Science, Published on the web 27 March 2020, https://doi.org/10.1139/as-2019-0022
- Hamilton C.D., Lydersen C., Ims R.A., Kovacs K.M. 2015. Predictions replaced by facts: a keystone species' behavioural responses to declining arctic sea-ice. Biol Lett 11: 20150803
- Hamilton C.D., Lydersen C., Ims R.A., Kovacs K.M..2016 Coastal habitat use by ringed seals Pusa hispida following a regional sea-ice collapse: importance of glacial refugia in a changing Arctic. Mar Ecol Prog Ser 545: 261–277
- Hamilton, C. D., Kovacs, K. M., & Lydersen, C. (2018). Individual variability in diving, movement and activity patterns in adult bearded seals in Svalbard, Norway. Scientific Reports, 8, 16988
- Hamilton C.D., Lydersen C., Fedak M.A., Freitas C, Hindell M.A., Kovacs K.M. 2019. Behavioural ontogeny of bearded seals *Erignathus barbatus* through the first year of life. Mar Ecol Prog Ser 627:179-194. https://doi.org/10.3354/meps13072
- Harris R.E., Miller G.W. and Richardson, J. 2001. Seal responses to airgun sounds during summer seismic surveys in the Alaskan Beaufort Sea. Marine Mammal Science 17(4):795-812
- Harwood L.A. and Kingsley M.C.S.K.2013. Trends in the Offshore Distribution and Relative Abundance of Beaufort Sea Belugas, 1982 85 vs 2007 09. ARCTIC 3:247-256
- Harwood L.A., Smith T.G., Auld J.C., Melling H. and Yurkowski D.J.2015a. Seasonal Movements and Diving of Ringed Seals, Pusa hispida, in the Western Canadian Arctic, 1999 – 2001 and 2010 – 11. ARC-TIC 68: 193 – 209 <u>http://dx.doi.org/10.14430/arctic4479</u>
- Harwood L.A., Kingsley M.C.S.K. and Smith T.G.2015b. An Emerging Pattern of Declining Growth Rates in Belugas of the Beaufort Sea: 1989 – 2008. **ARCTIC** 67, NO. 4: 483 – 492 <u>http://dx.doi.org/10.14430/arctic4423</u>
- Harwood L.A., Smith T.G., Alikamik J., Alikamik E., Lea E.V., Stirling I., Wright, H., Melling, H. and Zhu, X.2020. Long-term, Harvest-based Monitoring of Ringed Seal Body Condition and Reproduction in Canada's Western Arctic: An Update through 2019. ARCTIC VOL. 73, NO. 2 (JUNE 2020) P. 206 – 220 <u>https://doi.org/10.14430/arctic70428</u>
- Haug T., Nilssen K.T., Lindblom L.2004. Feeding habits of harp and hooded seals in drift ice waters along the east coast of Greenland in summer and winter. Polar Res 23:35–42
- Haug T.,, Bogstad B., Chierici M., Gjøsæter H., Hallfredsson E.H., Høines Å.S., Hoel A.H., Ingvaldsen R.B., Jørgensen L.L., Knutsen T., Loeng H., Naustvoll L_J., Røttingen I.and Sunnanå K. 2017. Future harvest of living resources in the Arctic Ocean north of the Nordic and Barents Seas: A review of possibilities and constraints. Fisheries Research 188: 38–57

- Hauser D.W., Laidre, K.L., Suydam, R.S. and Richard, P.R. 2014. Population-specific home ranges and migration timing of Pacific Arctic beluga whales (Delphinapterus leucas). Polar Biol 37:1171–1183DOI 10.1007/s00300-014-1510-1
- Hauser D.W., Laidre K.L. and Parker-Stetter S.L.2015. Regional diving behavior of Pacific Arctic beluga whales Delphinapterus leucas and possible associations with prey. Marine Ecology Progress Series 541:245-264doi: 10.3354/
- Heide-Jørgensen M.P., Laidre K.L., Burt M.L., Borchers, D.L., Marquez, T.A., Hansen, R.G., Rasmussen, M. and Fossette, S. 2010. Abundance of narwhals (Monodon monoceros) on the hunting grounds in Greenland. Journal of Mammalogy, 91(5):1135–1151
- Heide-Jørgensen M.P., Nielsen N.H., Hansen R.G., Schmidt H.C., Blackwell S.B., Jørgensen O.A. 2015 The predictable narwhal: satellite tracking shows behavioural similarities between isolated subpopulations. J. Zool. 297, 54–65. (doi:10.1111/jzo.12257)
- Hiemer K., Van Opzeeland I., Boebel O., Burkhardt, E., Spiesecke S., Thomisch K. 2020 Vocal behaviour of bowhead whales (Balaena mysticetus) in eastern Fram Strait. SC/68/NH/03 Available from: https://iwc.int/sc-documents
- Higdon J.W., Byers T., Brown, L. and Ferguson, S.H. 2012. Observations of killer whales (Orcinus orca) in the Canadian Beaufort Sea. Polar Record 49 (250): 307–314 (2013). c Cambridge University Press 2012. doi:10.1017/S0032247412000356 307
- Hindell M. A., Lydersen C., Hop H. and Kovacs K. M. 2012. Pre-partum diet of adult female bearded seals in years of contrasting ice conditions. *PLoS ONE* 7, e38307, https://doi.org/10.1371/journal.pone.0038307.
- Horstmann-Dehn L, Follmann EH, Rosa C, Zelensky G, George JC (2012) Stable carbon and nitrogen isotope ratios in muscle and epidermis of arctic whales. Mar Mamm Sci 28: E173–E190
- ICES. 2019. ICES/NAFO/NAMMCO Working Group on Harp and Hooded Seals (WGHARP). ICES Scientific Reports. 1:72. 193 pp. <u>http://doi.org/10.17895/ices.pub.5617</u>
- Jay C.V., Fischbach A.S. and Kochnev A.A. 2012. Walrus areas of use in the Chukchi Sea during sparse sea ice cover. Mar Ecol Prog Ser 468: 1–13, 2012
- Jones J.M., Thayre B.J., Roth E.H., Mahoney M., Sia I., Merculief K., Jackson C., Zeller C., Clare M., Bacon A., Weaver S., Gentes Z., Small R.J., Stirling I., Wiggon S.M. and Hildebrand J.A.2014. Ringed, Bearded, and Ribbon Seal Vocalizations North of Barrow, Alaska: Seasonal Presence and Relationship with Sea Ice ARCTIC: 67, 203 222 http://dx.doi.org/10.14430/arctic4388
- Jourdain E., Ugarte F., Vikingsson G., Samarra F., Ferguson S., Lawson J., Vongraven D. and Desportes G. 2019. North Atlantic killer whale Orcinus orca populations: a review of current knowledge and threats to conservation. Mammal Review 49 (2019) 384–400 © 2019 The Mammal Society and John Wiley & Sons Ltd
- Karlsen J. D., Bisther A., Lydersen C., Haug T., and Kovacs K. M. (2002). Summer vocalisations of adult male white whales (Delphinapterus leucas) in Svalbard, Norway. Polar Biol. 25, 808–817.
- Kelly B. P., and Quakenbush L. T. 1990. Spatiotemporal use of lairs by ringed seals (*Phoca hispida*). Can. J. Zool. 68: 2503-2512.
- Kelly B.P., Badajos O.H., Kunnasranta M., Moran J.R., Martinez-Bakker M.M., Wartzok D. and Boveng, P. 2010. Seasonal home ranges and fidelity to breeding sites among ringed seals. Polar Biol (2010) 33:1095– 1109 DOI 10.1007/s00300-010-0796-x
- Kochnev A. A. 2002. Autumn aggregations of polar bears on the Wrangel Island and their importance or the population.Proceedings of the Marine Mammals of the Holarctic meeting, Sept10–15,2002,Baikal,Russia.
- Koski W.R., Funk D.W., Ireland D.S., Lyons C., Christie K., Macrander A.M., Blackwell, S.B., 2009. An Update on Feedieng by Bowhead Whales near an Offshore Seismic Survey in the Central Beaufort Sea. IWC SC, pp. 61.

- Kovacs K.M. 2016. Erignathus barbatus. The IUCN Red List of Threatened Species 2016: e.T8010A45225428. https://dx.doi.org/10.2305/IUCN.UK.2016-1.RLTS.T8010A45225428.en. Downloaded on 18 July 2020.
- Kovacs K.M., Krafft B.A.and Lydersen C. Bearded seal (Erignathus barbatus) birth mass and pup growth in periods with contrasting ice conditions in Svalbard, Norway. Mar Mam Sci. 2020a;36: 276–284. https://doi.org/10.1111/mms.12647
- Kovacs K.M., Lydersen C., Vacquiè-Garcia J., Shpak O., Glazov D., Heide-Jørgensen M.P. 2020b. The endangered Spitsbergen bowhead whales' secrets revealed after hundreds of years in hiding. Biol. Lett. 16: 20200148. http://dx.doi.org/10.1098/rsbl.2020.0148
- Kryukova N.V. and Konchev, A.A. 2014. Marine mammals in Vankarem Cape region (The Chukchi Sea, August-November, 2010-2011). Zoologicheskii Zhurnal 93: 274–283
- Kuletz K.J., Ferguson M.C., Hurley B., Gall A.E., Labunski E.A., Morgan T.C. 2015. Seasonal spatial patterns in seabird and marine mammal distribution in the eastern Chukchi and western Beaufort seas: Identifying biologically important pelagic areas. Progress in Oceanography 136 (2015) 175–200
- Kvadsheim P. H., Sevaldsen E. M., Folkow L. & Blix, A. S. 2010. Behavioural and Physiological Responses of Hooded Seals (Cystophora cristata) to 1 to 7 kHz Sonar Signals. Aquatic Mammals 36:239-247. ISSN 0167-5427. doi: 10.1578/AM.36.3.2010.239.
- Kvadsheim P.H., Forland T.N., De Jong K., Nyqwist D., Grimsbø E. and Sivle, L.D.2020. Impact of anthropogenic noise on the marine environment – status of knowledge and Management. In Norwegian. FFI report 2020/01015 (<u>http://rapporter.ffi.no/rapporter/2020/01015.pdf</u>)
- Kyhn L.A.Wisniewska D.M., Beedholm K., Tougaard J., Simon M., Mosbech A and Madsen P.T.2019. Basinwide contributions to the underwater soundscape by multiple seismic surveys with implications for marine mammals in Baffin Bay, Greenland. Marine Pollution Bulletin 138: 474–490
- Labansen A.L., Lydersen C., Haug T., Kovacs K.M.. 2007. Spring diet of ringed seals (*Pusa hispida*) from north-western Spitsbergen, Norway. ICES J Mar Sci 64:1246–1256
- Laidre K.L., Born E.W., Gurarie E., Wiig, Ø., Dietz R., and Stern H. 2013. Females roam while males patrol: divergence in breeding season movements of pack-ice polar bears (Ursus maritimus). Proceedings of the Royal Society B 280:20122371, doi:10.1098/rspb.2012.2371.
- Laidre K., Stern H., Kovacs K.M., Lowry L., Moore, S.E., Regehr, E.V., Ferguson, S.H., Wiig, Ø., Boveng P., Angliss, R.P., Born, E.W., Litovka, D., Quakenbush L., Lydersen C., Vongraven, D. and Ugarte, F. 2015. Arctic marine mammal population status, sea ice habitat loss, and conservation recommendations for the 21st century. *Conservation Biology*, Volume 00, No. 0, 1–14. DOI: 10.1111/cobi.12474
- Laidre K.L., Northey A.D. and Ugarte F. (2018) Traditional Knowledge About Polar Bears (Ursus maritimus) in East Greenland: Changes in the Catch and Climate Over Two Decades.Front. Mar. Sci. 5:135.doi: 10.3389/fmars.2018.00135
- Laidre K. and Stirling I. 2020. Grounded icebergs as maternity denning habitat for polar bears (Ursus maritimus) in North and Northeast Greenland. Polar Biology (2020) 43:937–943 <u>https://doi.org/10.1007/s00300-020-02695-2</u>
- Lindstrøm U., Nilssen, K.T., Pettersen, L.M.S. and Haug, T. (2013). Harp seal foraging behaviour during summer around Svalbard in the northern Barents Sea: diet composition and the selection of prey. *Polar Biology*, 36, 305-320.
- Lone K., Kovacs K.M., Lydersen C., Fedak M., Andersen M., Lovell P., Aars J. 2018. Aquatic behaviour of polar bears (*Ursus maritimus*) in an increasingly ice-free Arctic. Sci Rep 8: 9677
- Loseto L.L., Stern G.A., Connelly T.L., Deibel D. and others.2009. Summer diet of beluga whales inferred by fatty acid analysis of the eastern Beaufort Sea food web. J Exp Mar Biol Ecol 374: 12–18
- Louis M et al. 2020 Influence of past climate change on phylogeography and demographic history of narwhals, Monodon monoceros. Proc. R. Soc. B 287: 20192964. http://dx.doi.org/10.1098/rspb.2019.2964
- Lowry F.L., Frost K.J. and Burns J.J.1980. Feeding of Bearded Seals in the Bering and Chukchi Seas and Trophic Interaction with Pacific Walruses. ARCTIC VOL. 33, NO. 2. P. 330-342

- Lowry L.F., Nelson R.R., and Frost K.J.1987. Observations of killer whales, *Orcinus orca*, in western Alaska: sighting, strandings and predation on other marine mammals. *Canadian Field-Naturalist* 101: 6–12.
- Lowry L. 2016a. Odobenus rosmarus. The IUCN Red List of Threatened Species 2016: e.T15106A45228501. https://dx.doi.org/10.2305/IUCN.UK.2016-1.RLTS.T15106A45228501.en. Downloaded on 20 July 2020.
- Lowry L. 2016b. Histriophoca fasciata. The IUCN Red List of Threatened Species 2016: e.T41670A45230946. https://dx.doi.org/10.2305/IUCN.UK.2016-1.RLTS.T41670A45230946.en. Downloaded on 06 August 2020.
- Lowry L. 2016b. Pusa hispida. The IUCN Red List of Threatened Species 2016: .T41672A45231341. https://dx.doi.org/10.2305/IUCN.UK.2016-1.RLTS.T41672A45231341.en. Downloaded on 05 August 2020
- Lowry L., Laidre K. & Reeves R. 2017. Monodon monoceros. The IUCN Red List of Threatened Species 2017: e.T13704A50367651. https://dx.doi.org/10.2305/IUCN.UK.2017-3.RLTS.T13704A50367651.en. Downloaded on 06 August 2020.
- Lowry L.F., Kingsley M.C.S. Hauser D.D.W., Clarke J. and Suydam R. 2017. Abundance of the eastern Chukchi stock of Beluga Whales (*Delphinapterus leucas*) based on aerial surveys conducted in the Beaufort Sea in 2012. *Arctic* 70(3): 273-286.
- Lydersen C. and Smith T.G. 1989. Avian predation on ringed seal *Phoca hispida* pups. *Polar Biology* 9: 489-490.
- Lydersen C. and Kovacs K. M. 1999. Behaviour and energetics of ice-breeding, North Atlantic phocid seals during the lactation period. Marine Ecology Progress Series 187: 265-281.
- Lydersen C. and Kovacs K.M.2014. Walrus Odobenus rosmarus research in Svalbard, Norway, 2000-2010. NAMMCO Scientific Publications, Volume 9:175-190
- Lydersen C., Assmy P., Falk-Petersen S., Kohler J., Kovacs K.M., Reigstad M., Steen H., Strøm H., Sundfjord A., Varpe Ø., Walczowski W., Weslawski J.M. and Zajaczkowski M. 2014. The importance of tidewater glaciers for marine mammals and seabirds in Svalbard, Norway. Journal of Marine Systems 129, 452–471.
- MacIntyre K.Q., Stafford K.M., Berchok C.L. and Boveng P.L.2013. Year-round acoustic detection of bearded seals (Erignathus barbatus) in the Beaufort Sea relative to changing environmental conditions, 2008–2010. Polar Biol 36:1161–1173 DOI 10.1007/s00300-013-1337-1
- Martinez-Bakker M.E., Sell S.K., Swanson B.J., Kelly B.P., Tallmon D.A. 2013. Combined Genetic and Telemetry Data Reveal High Rates of Gene Flow, Migration, and Long-Distance Dispersal Potential in Arctic Ringed Seals (Pusa hispida). PLoS ONE 8(10): e77125. doi:10.1371/journal.pone.0077125
- McCracken J.G., 2012. Pacific walrus and climate change:observations and predictions. Ecol. Evol. 2, 2072–2090.
- McKinney M., Iverson S., Fisk A., Sonne C., Riget F., Letcher R., Arts M., Born E., Rosing-Asvid A. and Dietz R. 2013. Global change effects on the long-term feeding ecology and contaminant exposures of East Greenland polar bears. Glob Chang Biol 19:2360–2372. doi:10.1111/gcb.12241
- McKinney M. A., Atwood T.C., Iverson S.J., and Peacock E.. 2017. Temporal complexity of southern Beaufort Sea polar bear diets during a period of increasing land use. Ecosphere 8(1):e01633. 10.1002/ecs2.1633
- Merkel B., Lydersen C., Yoccoz N.G., Kovacs K.M. 2013. The World's Northernmost Harbour Seal Population–How Many Are There? PLoS ONE 8(7): e67576. doi:10.1371/journal.pone.0067576
- Miller G.W. and Richardson W.J.2001. Seal responses to airgun sounds during seismic surveys in the Alaskan Beaufort Sea. MARINE MAMMAL SCIENCE, 17(4):795-812 (October 2001)
- Miller S., Wilder J. and Wilson R.R. 2015. Polar bear-grizzly bear interactions during the autumn openwater period in Alaska. Journal of Mammalogy, 96(6):1317–1325
- Molnár P.K., Bitz C.M., Holland M.M., Kay J.E., Penk S.R. and Amstrup S.C..2020. Fasting season length sets temporal limits for global polar bear persistenceNature Climate Change 10: 732–738

- Moore S.E., Stafford K., Melling H., Berchok C., Wiig Ø., Kovacs K., Lydersen C. and Menge J.R. 2012. Comparing marine mammal acoustic habitats in Atlantic and Pacific sectors of the High Arctic: yearlong records from Fram Strait and the Chukchi Plateau. Polar Biol (2012) 35:475–480 DOI 10.1007/s00300-011-1086-y
- Moore S.E., 2016. Is it 'boom times' for baleen whales in the Pacific Arctic region? Biol.Lett. 12. https://doi.org/10.1098/rsbl.2016.0251
- Moore S. E., Haug T., Vikingsson G. A. and Stenson G. B.2019. Baleen whale ecology in arctic and subarctic seas in an era of rapid habitat alteration. Progress in Oceanography, 176: 102118.
- Muto M. M., Helker V.T., Angliss R.P., Boveng P.L.,Breiwick J.M., Cameron M.F., Clapham P.J., Dahle S.P.,Dahlheim M.E., Fadely B.S., Ferguson M.C., Fritz L.W., Hobbs R.C.,Ivashchenko Y.V., Kennedy A.S., JLondon J.M., Mizroch S.A.,Ream R.R., Richmond E.L., Shelden K.E.W., Sweeney K.L., Towell R.G., Wade P.R., Waite J.M., and Zerbini A.N.. 2019. Alaska marine mammal stock assessments, 2018. U.S. Dep. Commer., NOAATech. Memo. NMFS-AFSC-393, 390 p.Document available: http://www.afsc.noaa.gov/Publications/AFSC-TM/NOAA-TM-AFSC-393.pdf
- NAMMCO-North Atlantic Marine Mammal Commission.2019.. Report of the Ad hoc Working Group on Narwhal in East Greenland. September 2019, Copenhagen, Denmark. Available at <u>https://nammco.no/topics/sc-wg-reports/</u>
- Nilssen K.T., Pedersen O.-P., Folkow L.P. and Haug T. 2000. Food consumption estimates of Barents Sea harp seals. *NAMMCD Sci. Publ.* 2:9-27.
- Nguyen L., Pilfold N.W., Derocher A.E., Stirling I., Bohart A.M. and Richardson E. 2017. Ringed seal (Pusa hispida) tooth annuli as an index of reproduction in the Beaufort Sea. Ecological Indicators 77 (2017) 286–292
- Nordøy E.S., Folkow L.P., Potelov, V., Prischemikhin, V. and Blix, A.S. 2008. Seasonal distribution and dive behaviour of harp seals (*Pagophilus groenlandicus*) of the White Sea–Barents Sea stock Polar Biol2008) 31:1119–1135 DOI 10.1007/s00300-008-0453-9
- Øigård T.A., Haug T., Nilssen K.T. 2014.Current status of hooded seals in the Greenland Sea.Victims of climate change and predation? Biol Conserv 172: 29–36
- Olson J. W., et al. 2017. Collar temperature sensor data reveal long-term patterns in southern Beaufort Sea polar bear den distribution on pack ice and land. Marine Ecology Progress Series 564:211–224.
- Ovsyanikov N.G.1995. Polar bear predation of walruses on Wrangell Island. Bulletin of the Moscow association of Natural Scientists, Section of Biology. VOL. 100. Issue #4.
- PAME 2019. Underwater Noise in the Arctic: A State of Knowledge Report, Roveniemi, May 2019. Protection of the Arctic Marine Environment (PAME) Secretariat, Akureyri (2019)
- PBSG 2019. Status Report on the World's Polar Bear Subpopulations. IUCN/SSC Polar Bear Specialist Group. http://pbsg.npolar.no/export/sites/pbsg/en/docs/2019-StatusReport.pdf
- Polyak L. et al. History of sea ice in the Arctic. Quaternary Sci. Rev. 29, 1757–1778 (2010).
- Prop J., Aars J., Bårdsen B-J., Hanssen S.A., Bech C., Bourgeon S., de Fouw J., Gabrielsen G.W., Lang J., Noreen E., Oudman T., Sittler B., Stempniewicz L., Tombre I., Wolters E. and Moe B. (2015) Climate change and the increasing impact of polar bears on bird populations. Front. Ecol. Evol. 3:33. doi: 10.3389/fevo.2015.00033
- Quakenbush L.T., Suydam R.S., Bryan A.L., Lowry L.F., Frost K.J., and Mahoney B.A. 2015. Diet of beluga whales, (Delphinapterus leucas), in Alaska from stomach contents, March-November. Marine Fisheries Review 77(1): 70-84.
- Reichmut C., Holt M.M., Mulsow J.M., Sills J.M. and Southall B.L. 2013. Comparative assessment of amphibious hearing in pinnipeds. Journal of Comparative Physiology A. Neuroethology, Sensory, Neural, and Behavioural Physiology, 199(6), 491-507. https://doi.org/10.1007/s00359-013-0813-v

- Reichmuth C., Ghoul A., Sills J.M., Rouse A. and Southall B.L.2016.Low-frequency temporary threshold shift not observed in spotted or ringed seals exposed to single air gun impulses). J. Acoust. Soc. Am. 140 (4): 2646-2658
- Regehr E.V. et al. 2016.Conservation status of polar bears (Ursus maritimus) in relation to projected sea ice declines. Biol. Lett. 12: 20160556. <u>http://dx.doi.org/10.1098/rsbl.2016.0556</u>
- Regehr E.V., Hostetter N.J., Wilson R.R. *et al.* Integrated Population Modeling Provides the First Empirical Estimates of Vital Rates and Abundance for Polar Bears in the Chukchi Sea. *Sci Rep* **8**, 16780 (2018). https://doi.org/10.1038/s41598-018-34824-7
- Richardson W.J., Miller G.W., Greene C.R., 1999. Displacement of migrating bowhead whales by sounds from seismic surveys in shallow waters of the Beaufort Sea. J. Acoust. Soc. Am. 106, 2281.
- Robbins C.T., Ben-David M., Fortin J.K. and Nelson O.L. 2012. Maternal condition determines birth date and growth of newborn bear cubs. Journal of Mammalogy, 93(2):540–546
- Robertson F., Koski W.R. and Trites A.W. 2016. Behavioral responses affect distribution analyses of bowhead whales in the vicinity of seismic operations. Marine Ecology Progress Series Volume: 549 Pages: 243-262.DOI: 10.3354/meps11665
- Rode K.D., Amstrup S.C., Regehr E.V.2010Reduced body size and cub recruitment in polar bears associated with sea ice decline. Ecol Appl 20:768–782. PMID: 20437962
- Rode K.D., Regehr E.V., Douglas D.C., Durner G., Derocher A.E., Thiemann G.W., and Budge S.M. 2014. Variation in the response of an Arctic top predator experiencing habitat loss: feeding and reproductive ecology of two polar bear populations. Global Change Biology 20:76–88.
- Rode K.D., Wilson R.R., Regehr E.V., St. Martin M., Douglas D.C., Olson J. 2015. Increased Land Use by Chukchi Sea Polar Bears in Relation to Changing Sea Ice Conditions. PLoS ONE 10(11): e0142213. doi:10.1371/journal.pone.0142213
- Rode K.D., Olson J., Eggett D., Douglas D.C., Durner G.M., Atwood T.C., Regehr E.V., Wilson R.R., Smith T., and St Martin M.2018. Den phenology and reproductive success of polar bears in a changing climate Journal of Mammology 99:16-26.DOI: 10.1093
- Rogers M.C., Peacock E., Simac K., O'Dell M.B., Welker J.M.2015. Diet of female polar bears in the southern Beaufort Sea of Alaska: evidence for an emerging alternative foraging strategy in response to environmental change. Polar Biol 38: 1035–1047
- Rolland R. Parks S. Hunt K. et al. 2012.Evidence that ship noise increases stress in right whales. <u>Proc Biol</u> <u>Sci.</u> 279(1737): 2363–2368.
- Sigsgaard E.E., Jensen M.R., Winkelmann I.E., Møller P.R., Hansen M.M., Thomsen P.F. Population-level inferences from environmental DNA-Current status and future perspectives. Evol Appl. 2019;13(2):245-262. Published 2019 Nov 18. doi:10.1111/eva.12882
- Simon M., Johnson M., Tyack P. and Madsen P.T. Behaviour and kinematics of continuous ram filtration in bowhead whales (Balaena mysticetus). 2009. Proc. R. Soc. B (2009) 276, 3819–3828 doi:10.1098/rspb.2009.1135
- Seymour J., Horstmann-Dehn L. and Wooler M.J. 2014. Proportion of higher trophic-level prey in the diet of Pacific walruses (Odobenus rosmarus divergens). Polar Biology 37: 941-952.
- Smith T.G. and Stirling I. 2019. Predation of Harp Seals, Pagophilus groenlandicus, by Polar Bears, Ursus maritimus, in Svalbard. Arctic 72: 197-202
- Solovyev B.A., Platonov N.G., Glazov D.M., Shpak O.V. and Rozhnov V.V.2012. Distribution of Beluga Whales (Delphinapterus leucas) in the Russian Arctic Seas According to the Results of Expedition aboard RV Mikhail Somov, September–November 2010. BIOLOGY BULLETIN 39: 654-658
- Southall B. L., Bowles A. E., Ellison W. T., Finneran J. J., Gentry R. L., Greene C. R. J., Kastak D., Ketten D. R., Miller J. H., Nachtigall P. E., Richardson J. W., Thomas J. A., and Tyack P. L. (2007). "Marine mammal noise exposure criteria: Initial scientific recommendations," Aquat. Mamm. 33, 411–521.

- Spady T.J., Lindburg D.G. and Durrant B. 2007. Evolution of reproductive seasonality in bears. *Mammal Rev.* 2007, Volume 37, No. 1, 21–53.
- Spiridonov A.V., Gavrilo M.V., Krasnova E.D. and Nikolaeva N.G.(Eds). 2011. Atlas of marine and coastal biological diversity of the Russian Arctic. – Moscow: WWF Russia, 2011. – 64 pp. ISBN 978_5_9902786_2_2
- Stafford K.M., Citta J.J., Okkonen S.R. and Suydam, R.S. 2016. Wind-dependent beluga whale dive behavior in Barrow Canyon, Alaska. Deep–Sea Research I 118 (2016) 57–65
- Stafford K.2018. Increasing detections of killer whales (Orcinus orca), in the Pacific Arctic. MARINE MAM-MAL SCIENCE, 35(2): 696–706
- Stafford K.M., Castellote M., Guerra M., and Berchok C.L. 2018. Seasonal acoustic environments of beluga and bowhead whale core-use regions in the Pacific Arctic. Deep Sea Res., Part II: Top. Stud. Oceanogr. 152: 108–120. doi: 10.1016/j.dsr2.2017.08.003
- Stempniewicz L., Kidawa D., Barcikowski M., & Iliszko, L. (2014). Unusual hunting and feeding behaviour of polar bears on Spitsbergen. Polar Record, 50(2), 216-219. doi:10.1017/S0032247413000053
- Stirling I. 2002. Polar Bears and Seals in the Eastern Beaufort Sea and Amundsen Gulf: A Synthesis of Population Trends and Ecological Relationships over Three Decades. ARCTIC: 55: 59–76
- Stirling I., Richardson E., Thiemann G.W. and Derocher A.E. 2008. Unusual Predation Attempts of Polar Bears on Ringed Seals in the Southern Beaufort Sea: Possible Significance of Changing Spring Ice Conditions. ARCTIC 61:14–22
- Storrie L., Lydersen C., Andersen M., Wynn R.B. and KovacsK.M.2018. Determining the species assemblage and habitat use of cetaceans in the Svalbard Archipelago, based on observations from 2002 to 2014. Polar Research 37.doi.org/10.1080/17518369.2018.1463065
- Strong C. Atmospheric influence on Arctic marginal ice zone position and width in the Atlantic sector, February–April 1979–2010. Clim Dyn 39, 3091–3102 (2012). <u>https://doi.org/10.1007/s00382-012-1356-6</u>
- Vacquie-Garcia J., Lydersen C., Biuw M., Haug T., Fedak M.A., Kovacs K.M. 2017a. Hooded seal Cystophora cristata foraging areas in the Northeast Atlantic Ocean DInvestigated using three complementary methods. PLoS ONE 12(12): e0187889. <u>https://doi.org/10.1371/journal.pone.0187889</u>
- Vacquié-Garcia J., Lydersen C., Marques T.A., Aars J., Ahonen H., Skern-Mauritzen M., Øien N. and Kovacs K.M. 2017b. Late summer distribution and abundance of ice-associated whales in the Norwegian High Arctic. Endangered Species Research 32: 59-70.
- Vacquié-Garcia J., Lydersen C., Ims R.A., Kovacs K.M. 2018. Habitats and movement patterns of white whales Delphinapterus leucas in Svalbard, Norway. Mov Ecol 6: 21
- Vacquié-Garcia, J., Lydersen, C, Marques, T.A., Andersen, M., and Kovacs, K.M.2020. First abundance estimate for white whales Delphinapterus leucas in Svalbard, Norway. Endangered species Research 41: 253–263, 2020
- VanWormer E., Mazet J. A. K., Hall A., Gill V.A., Boveng P.L., London J.M., Gelatt T., Fadely B.S., Lander M.E., Sterling J., Burkanov V.N., Ream R.R., Brock P. M., Rea L.D., Smith B.R., Jeffers A., Henstock M., Rehberg M.J., Burek-Huntington K.A., Cosby S.L. 10, Hammond J.A. and Goldstein T. 2019. Viral emergence in marine mammals in the North Pacific may be linked to Arctic sea ice reduction Scientific Reports 9:15569 | https://doi.org/10.1038/s41598-019-51699-4
- Von Duyke A.L., Douglas D.C., Herreman J.K., Crawford J.A. Ringed seal (Pusa hispida)seasonal movements, diving, and haul-out behavior in the Beaufort, Chukchi, and Bering Seas (2011–2017). Ecol Evol.2020;10:5595–5616. https://doi.org/10.1002/ece3.6302
- Ware J. V., et al. 2017. Habitat degradation affects the summer activity of polar bears. Oecologia 184:87–99.

- Wathne, J. A., Haug, T., and Lydersen, C. 2000. Prey preference and niche overlap of ringed seals Phoca hispida and harp seals *P. groenlandica* in the Barents Sea. Marine Ecology Progress Series, 194: 233–239.
- Watt C. A., Heide-Jørgensen M.P., and Ferguson S.H. 2013. How adaptable are narwhal: a comparison of foraging patterns among the world's three narwhal populations. Ecosphere 4(6):71. <u>http://dx.doi.org/10.1890/</u> ES13-00137.1
- Watt, C.A.,Orr, J.R.,Heide-Jørgensen, M.P., Nielsen, N.H. and Ferguson, S.H. 2015 Differences in dive behaviour among the world's three narwhal Monodon monoceros populations correspond with dietary differences.Marine Ecology Progress Series **525**: **273–285** doi: 10.3354/meps11202
- Weslawski, J.M., Ryg, M., Smith, T.G., Øritsland, N.A., 1994. Diet of ringed seals (Phoca hispida) in a fjord of West Svalbard. Arctic 47, 109–114.
- Westbury M.V., Petersen B., Garde E., Heide-Jørgensen M.P., Lorenzen E.D. 2019 Narwhal genome reveals long-term low genetic diversity despite current large abundance size. iScience 15, 592–599. (doi:10.1016/j.isci.2019.03.023)
- Wilder J. M., Vongraven D., Atwood T., Hansen B , Jessen A. Kochnev A. York G., Vallender R., Hedman D. and Gibbons M. 2017. Polar bear attacks on humans: implications of a changing climate. Wildlife Society Bulletin 41:537–547.
- Wiig Ø. 1995. Distribution of polar bears (Ursus maritimus) in the Svalbard area. Journal of Zoology, London 237:515–529.
- Wiig Ø., A. E. Derocher, and S. E. Belikov. 1999. Ringed seal (Phoca hispida) breeding in the drifting pack ice of the Barents Sea. Marine Mammal Science 15:595-598.
- Wiig Ø., Born, E.W., and Pedersen, L.T. 2003. Movements of female polar bears (Ursus maritimus) in the East Greenland pack ice. Polar Biology 26:509–516.
- Yamato M., Ketten D.R., Arruda J., Cramer S.and Moore, K.2012. The Auditory Anatomy of the Minke Whale (Balaenoptera acutorostrata): A Potential Fatty Sound Reception Pathway in a Baleen Whale. The Anatomical Record 295:991–998

4.7 Marine birds in the CAO

Contributor: Thomas Van Pelt and Kathy Kuletz

Marine birds occur in relatively low densities throughout the Central Arctic Ocean (CAO), but are substantial ecosystem components along the continental edges of the CAO, particularly in the 'gateway' ecosystems; as such they are sensitive to a wide range of current and future climate and anthropogenic impacts. Marine birds breed on coasts and islands, so terrestrial aspects of their life cycles are not be directly impacted by change in the open-ocean Central Arctic. But as highly mobile predators, their foraging range in the breeding season can include open water areas hundreds of kilometers from land, and large numbers of non-breeding or migratory individuals use open water habitats in marginal waters of the Central Artic region during summer and fall. Marine bird species also play a role in linking marine and terrestrial habitats by returning nutrients from offshore waters to the coast. Marine birds thus span a large breadth of ecological and management considerations.

The *Circumpolar Seabird Monitoring Plan* (CSMP; Irons et al., 2015) recognizes 64 marine bird species in the Arctic ecosystem; about half of these breed in the Arctic while others visit in summer and fall to forage on zooplankton, invertebrates and forage fish. About a dozen seabird species comprise over 90% of total individual birds in the broader "Arctic" region (following the Arctic Council geographic definition of Arctic); of those, only a few species have been regularly recorded in the CAO, and about two dozen species recorded overall. In the Pacific sector of the CAO, murres *Uria spp*], black-legged kittiwakes (*Rissa tridactyla*), and jaegars (*Stercorarius* spp) are most frequently encountered, while Ross's gull (*Rhodostethia rosea*) and ivory gull (*Pagophila eburnean*) predominate in the Atlantic sector. Throughout the CAO, numbers of birds are extremely low, defined largely by lack of open water, but perhaps also related to water depth and prey availability. In contrast, the adjacent continental shelves, and in particular the gateways linking the CAO with adjacent seas, have high densities of marine birds and high species richness. A comprehensive resource for further marine bird information in the Arctic Ocean and surrounding areas can be found in the seabird chapter of the 2017 SAMBR report (CAFF 2017).

For this WGICA 2020 annual report, we first provide a very brief summary of recent information relevant to marine birds in the context of climate and human impacts in the basin waters of the Central Arctic region that are the WGICA core area. Then we briefly refer to related ICES and other programs that support marine bird information in marginal seas and coastlines and gateway regions.

4.7.1 Recent information on marine birds in the CAO

Historically, there have been few formal at-sea surveys for upper-trophic organisms (fish, marine birds and mammals, etc.) in the CAO, largely due to near-total ice coverage year-round. Now, with the combination of recent Arctic warming and consequent sea ice reductions allowing greater access to the Arctic Ocean, and increased scientific, commercial, and political interest motivating greater research presence, some new information is emerging from deskwork and fieldwork. A European Commission-funded review of "fish populations and related ecosystems" published in 2020 (Snoeijs-Leijonmalm et al., 2020) provides an up-to-date review of information on prey species important for marine birds in the CAO, with select examples of potential impacts connected to climate change and to human impacts. New fieldwork with some marine bird focus includes surveys from transarctic icebreaker cruises, documenting expected low densities of marine birds and mammals (e.g. Joiris et al., 2016).

4.7.2 Climate and anthropogenic impacts on marine birds in the future CAO

As reviewed in the SAMBR report and in other sections of this WGICA annual report, an array of climate and human impacts are predictable for the future CAO region. The most prominent of these is the projection of ice-free summer and fall conditions in the CAO within the next ~20-30 years (Notz & Stroeve 2018), and related changes in freshwater inputs, currents, water column structures, and primary and secondary productivity. These changes will impact marine bird prey species such as polar cod (*Boreogadus saida*; e.g. Huserbråten et al. 2019). Considering that marine birds reproduce on land, an ice-free CAO is unlikely to result in rapid changes in marine bird habitat use in the region. However, unexpected effects may result across a range of scales, from reductions in certain very ice-dependent species like ivory gulls and Ross's gulls (Joiris 2017) to extreme changes in longstanding migration and distribution patterns of northern hemisphere marine birds (Clairbeaux et al. 2019).

Decreasing sea ice is likely to be followed by increased shipping traffic, seabed exploration, and tourism, with consequential impacts on marine birds and the prey resources upon which they depend. Increased use of the CAO by post-breeding and migrating marine birds in newly ice-free waters would overlap with increased vessel traffic, which could lead to more bird-ship collisions, such as occurs in sub-Arctic and Arctic seas during poor weather and fog, when birds are attracted to ship's lights. Pollution and oily or plastic debris dumped illegally from vessels can also be detrimental to marine birds. In summer 2020, dead seabirds have washed up with trash and non-petroleum oil on beaches in the northern Bering Sea, coincident with increased fishing activity in the region (Kuletz, pers. comm.); similar risks could increase in the CAO.

4.7.3 Marine bird research, management, and monitoring resources in adjacent seas

The circumpolar northern continental shelf regions support enormous populations of marine birds, with high ecological, economic, social, and cultural values - the marine bird chapter of the SAMBR report (CAFF 2017) is a current and authoritative resource for marine bird information in the areas adjacent to the CAO. Shelf and Atlantic 'gateway' ecoregions within the ICES framework provide assessments and information on marine birds as components of the Barents, Norwegian, and Greenland Seas (http://www.ices.dk/community/groups/Pages/IEASG.aspx). In the Pacific gateway region, the North Pacific Marine Science Organization (PICES) and its working groups and programs, in partnership with ICES, are building an integrated ecosystem assessment for the Bering and Chukchi Sea Large Marine Ecosystem that encompasses marine birds and their prey (https://meetings.pices.int/members/working-groups/wg44). The Arctic Council's Circumpolar Seabird Expert Group (CBird; https://www.caff.is/seabirds-cbird) support an array of circumpolar initiatives and programs including the Circumpolar Biodiversity Monitoring Program — which generated the recent SAMBR report (CAFF 2017) — and the Circumpolar Seabird Monitoring Plan. Other practitioner-based research programs and networks include marine bird studies and surveys in conjunction with physical and biological studies, e.g.: the Distributed Biological Observatory program in the Pacific gateway region (https://arcticdata.io/catalog/projects/DBO); the Synoptic Arctic Survey (https://synopticarcticsurvey.w.uib.no/); the Pacific Arctic Group (https://pag.arcticportal.org/); and the Arctic Marine Biodiversity Observation Network in the Chukchi Sea (AMBON; https://www.uaf.edu/cfos/research/projects/arctic-marine-biodiversit/).

References

- CAFF 2017. State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland. 978-9935-431-63-9
- Clairbaux, M., Fort, J., Mathewson, P. *et al.* (2019). Climate change could overturn bird migration: Transarctic flights and high-latitude residency in a sea ice free Arctic. *Sci Rep* **9**, 17767. https://doi.org/10.1038/s41598-019-54228-5
- Huserbråten, M.B.O., Eriksen, E., Gjøsæter, H. *et al.* Polar cod in jeopardy under the retreating Arctic sea ice. *Commun Biol* **2**, 407 (2019). https://doi.org/10.1038/s42003-019-0649-2
- Jakubas D, Wojczulanis-Jakubas K, Iliszko LM, et al. (2020). Flexibility of little auks foraging in various oceanographic features in a changing Arctic. *Sci Rep.* 10:8283. doi:10.1038/s41598-020-65210-x
- Joiris, C. R., Boos, K., D'Hert, D., & Nachtsheim, D. A. (2016). Low density of top predators (seabirds and marine mammals) in the high Arctic pack ice. *Scientifica*. http://dx.doi.org/10.1155/2016/1982534
- Joiris, C.R. (2017). Drastic decrease in high Arctic gulls—ivory *Pagophila eburnea* and Ross's *Rhodostethia rosea*—density in the northern Greenland Sea and Fram Strait between 1988 and 2014. *Polar Biol* **40**, 1029–1034. https://doi.org/10.1007/s00300-016-2027-6
- Notz, D., Stroeve, J. (2018). The trajectory towards a seasonally ice-free Arctic Ocean. *Curr Clim Change Rep* **4**, 407–416. https://doi.org/10.1007/s40641-018-0113-2
- Snoeijs-Leijonmalm, P., Flores, H., Volckaert, F., Niehoff, B., Schaafsma, F., Hjelm, J., Hentati-Sundberg, J., Niiranen, S., Crépin, A.S. and Österblom, H., 2020. Review of the research knowledge and gaps on fish populations, fisheries and linked ecosystems in the Central Arctic Ocean (CAO). European Commission. DOI: 10.2826/387890
- Irons, D.B., Petersen, A., Anker-Nilssen, T., Arthukin, Y., Barrett, R., Boertman, D., Gavrilo, M., Gilchrist, H.G., Hansen, E.S., Hario, M., Kuletz, K., Mallory, M.L., Merkel, F., Mosbech, A., Labansen, A.L., Olsen, B., Österblom, H., Reid, J., Robertson, G., Rönkä, M. and Strøm, H. 2015. <u>Circumpolar Seabird Monitoring Plan</u>. CAFF Monitoring Report No. 17, CAFF International Secretariat, Akureyri, Iceland

5 ToR d. Assess the consequences of recent and ongoing climatic and oceanographic changes on transport pathways and potential effects of contaminants

Contributing: Natsuhiko Otsuka, Martine van den Heuvel-Greve, Bjørn Einar Grøsvik, Anders Mosbech, Lisa Speer, Jan Rene Larsen, Lis Lindal Jørgensen

5.1 Introduction

In the Central Arctic Ocean (CAO), rapid climate-driven sea ice loss is putting pressure on the highly sensitive ecosystems, while also opening the area for an unprecedented expansion of human activities (De Lucia et al 2018).

For the ecosystem of the CAO, five main pressures have been identified, such as sea-ice decline and pollution. In this chapter, we focus on pollution, including both chemical pollution and marine litter. Pollution in the Central Arctic Ocean is primarily a consequence of activities in other regions of the northern hemisphere, with long-range transport by air or sea bringing pollution into the CAO. Additionally, some human activities that generate pollution occur within the Arctic. Both types of sources, and both current and predicted future pollution, are reported here.

5.2 Pollution pressures in the present-day CAO

5.2.1 Contaminants

The CAO is relatively unpolluted compared to other marine regions of the world. However, the Arctic is a sink for global pollutants transported from lower latitudes via the atmosphere, oceans and rivers, and local sources of pollution are also increasing. Winds provide a relatively fast route to the CAO for volatile contaminants and for substances that adhere to small particles. With rain and snow, contaminants are washed from the air and deposit on the sea ice, open water, or on the ground. Rivers carry contaminants to the coasts bordering the Arctic Ocean, where processes of sedimentation and re-suspension of particles occur. Both ocean currents and sea-ice drift also carry contaminants (partly dissolved in water or in ice) to and from the CAO.

It is indicated that ongoing climate change in the Arctic will increase the potential of long-range atmospheric transport of contaminants such as Hg and POPs (AMAP, 2011; Armitage et al., 2011; Ma et al., 2011; Wöhrnschimmel et al., 2012a, b, 2013). At the same time, POPs in the Arctic may be influenced by warming, resulting a shorter lifetime of the POPs in the environment (AMAP 2016). Changes in ocean currents may affect ocean transport (AMAP, 2011; Kallenborn et al., 2012; Grannas et al., 2013), as well as transport caused by sea-ice drift.

5.2.2 Chemical pollution

Emissions of chemical compounds from the Northern Hemisphere mid-latitude (from Europe, Asia, and North America) are currently the main source of Arctic air pollution. These include toxic contaminants (e.g. mercury) and certain persistent organic pollutants (POPs), along with particulates and emerging pollutants (AMAP 2017).

While levels of many POPs are declining in the Arctic due to tighter global regulation, other pollutants have been listed as 'chemicals of emerging concern" for the Arctic (AMAP 2017). Examples include pharmaceuticals and personal care products, a large number of per- and polyfluoroalkyl substances (PFASs), new brominated, organophosphorus-based and chlorinated flame retardants (BFRs, PFRs, CFRs), hexachlorobutadiene (HCBD), siloxanes, byproduct polychlorinated biphenyls (PCBs, e.g. PCB11), phthalates, halogenated natural products (HNPs), and pesticides (AMAP 2016).

In addition, increasing industrial activity in the Arctic region itself, such as oil and gas development, mining, shipping, and industrial runoff from onshore facilities, will likely result in higher levels of pollution in the Arctic Ocean.

Since 1991, AMAP has been investigating the transport of pollutants to the Arctic, as well as their fate and effect. AMAP (2016) provided maps of modeled estimates of changes in concentrations of γ -hexachlorocyclohexanes (γ -HCH), PCB52, and PCB153 in the atmosphere and oceans between 1995–1999 and 2095–2099. In the CAO, future γ -HCH concentration is estimated to be stable or to slightly increase, whereas concentrations of PCB52 are predicted to decrease.

5.2.3 Plastic pollution

Plastic debris, including microplastics, have been documented in Arctic marine habitats, sea ice, and wildlife (Obbard et al 2014). Debris are transported by rivers and ocean currents from regions south of the CAO. Additionally, debris can result from human activities onsite, such as shipping activities (see figure 5.1).

Plastic microparticles and nanoparticles are entering food webs in poorly understood ways (IP-BES 2019). Plastic debris can be divided into three categories: macro-, micro- and nanoplastic. Particle analyses suggests that microplastics in the Arctic may either result from the breakdown of larger items (transported over large distances by prevailing currents, or derived from local vessel activity), or from direct input of microplastics in sewage and wastewater of coastal areas (Lusher et al 2015). Recently, transport and deposit of microplastics from air has received more attention as microplastic by Arctic marine species has been recorded in organisms ranging from blue mussels (Mytilus, Lusher et al 2017; Kühn et al 2018), polar cod (Kühn et al 2018), marine mammals (Martin and Clarke 1986), to seabirds (Lydersen et al 1989; Mallory 2008; Provencher et al 2010; Trevail et al 2015).

The effects of climate change are expected to alter the amount of plastic debris present in the Arctic (AMAP 2016). In the future, the incidence of plastic ingestion in Arctic food webs is likely to increase as plastic pollution rises in the Arctic (Tekman et al 2017). The likely increased influx of Atlantic water and accelerated ice drift may enhance the advection of microplastic particles into the CAO (AMAP 2017), and possible toxic contaminants adhered to microplastics (Karlsson et al. 2018). In addition, recent analyses of ice cores from across the Arctic found levels of microplastic pollution in sea ice up to three orders of magnitude greater than previously thought, suggesting that Arctic sea ice is an important sink for microplastic pollution (Peeken et al 2018).

5.3 Human activities in the CAO related to pollution: present and future

Two main activities that are currently occurring in the CAO and adjoining regions are ship traffic and fisheries. Human activities that may increase in the future — when the CAO becomes ice free in summer — are oil and gas activities, offshore mining and tourism.

93

5.3.1 Ship traffic

See also: PAME's Shipping Expert Group (SEG) Arctic Ship Traffic Data (ASTD) System. Data from this system could be used to conduct an assessment of ship traffic in the CAO.

Smith and Stephenson (2013) modeled shipping routes under several climate change scenarios and concluded that September sea ice will have retreated in the Northwest Passage by midcentury, allowing approximately 30% geographic distance savings in shipping between western North America and Europe.

Potential vessel traffic growth in the Arctic could result in higher rates of noise and air pollution, vessel discharges (sewage, graywater, etc.), non-native species introductions, small vessel interference, oiling, and other concerns related to the wellbeing of humans and environment (Huntington et al 2015).

Cooperation among the Arctic states, the non-Arctic shipping states, and the global maritime enterprise will be critical to effective protection of Arctic people and the marine environment and developing sustainable strategies for the region (Safety 2018).

PAME has been monitoring ship activities in the Polar Code governed area under the Arctic Ship Traffic Data Project. The latest report (PAME 2020) indicates that the number of ships and total distance sailed has been increasing continuously since 2013. Fishing vessels are the dominant group, accounting for 45% of total sailed distance by vessels in the Polar Code area (Figure 5.1). At the same time, large cargo vessel activity is taking place as LNG transport from the Yamal Peninsula and crude oil transport from the Kara Sea occurs via the Northern Sea Route along the Russian Arctic coast. However, most of these activities are limited to nearshore and shelf areas. In the CAO itself, a limited amount of ship activity is currently taking place, mainly involving tourism and scientific research.

The Arctic Marine Shipping Assessment (AMSA 2009) identified oil spills and disturbance of wildlife as the two main potential effects of shipping. Shipping as a local source of pollution was not considered to be significant, because discharges to water and emissions to air are governed by IMO regulations and are negligible. On the other hand, if vessel traffic in the Arctic Ocean increases, oil spills from maritime accidents was thought to become a major threat to the Arctic marine ecosystems.

Oil and gas activities in the Arctic were assessed by AMAP in 2007-2010 (e.g. AMAP Assessment 2009). In that assessment, oil spills were regarded as the major environmental threat, particularly in ice-covered waters. The likelihood of a major spill in the Arctic is increasing as vessel traffic rises and offshore oil development accelerates (Eckle et al., 2012; Nevalainin et al 2016). The effects of a major spill depend on the physical and chemical composition of the oil or other chemical spilled; spill size, location, and season; and the vulnerability and sensitivity of affected species and ecosystems. Oil will last a long time when spilled in ice (or transported into ice-covered waters). Therefore an oil spill will have a large total geographical range as it drifts with the ice, although its spatial extent at any one time may be limited, even for a large spill.

IMO (2020), MARPOL Convention Annex IV, gives a set of regulations for the discharge of sewage into the sea from ships. And its Annex V prohibits the discharge of all garbage into the sea except as provided in regulations that are related to food waste, cargo residues, cleaning agents and additives, and animal carcasses. The IMO Polar Code applies these regulations into polar waters.



Figure 5.1. Ship tracks (#75) along the NSR (larger than 5,000DWT, Jun.~Nov.2014). Navigable speed related to ice condition along the Northern Sea Route, The 5th International Symposium on Arctic Research (ISAR5) (Source: Otsuka, N., Izumiyama, K. and Tateyama, K. (2018)).

From Annual Report 2017:

The following species or groups of organisms are identified as the key **ecosystem components to consider in a shipping vulnerability** context:

• Polar bear with relevant subpopulations: Barents Sea, Kara Sea, Laptev Sea, Chukchi Sea, southern, and northern Beaufort Sea, and CAO subpopulations;

• Ringed seal, as the anticipated food base for polar bears;

• Bowhead of the Critically Endangered Spitsbergen stock, and the large migratory Bering-Chukchi-Beaufort stock; • Beluga whale of several stocks in the Atlantic and Pacific gateway areas;

• Narwhal of the stock (or stock complex) found in the Atlantic gateway area;

• Ivory gull, which uses the peripheral pack ice of the CAO as feeding habitat in summer and autumn; • Ross's gull, similar habitat use as ivory gull;

• Polar cod *Boreogadus saida*, presumably with large migratory populations surrounding the CAO and spawning under ice;

• Arctic cod *Arctogadus glacialis*, possibly occurring with a migratory population in the Canada Basin;

• Sea ice amphipods, living on the underside of ice and being an important part of the CAO food-webs.

5.3.2 Fisheries in the CAO

The "Agreement to Prevent Unregulated High Seas Fisheries in the Central Arctic Ocean" bans commercial fishing in the CAO until there is scientific evidence that fishing can be done in a sustainable manner. Of the nine nations (USA, Canada, Russia, Norway, Denmark, Japan, China, Korea, Iceland) and one party (EU), eight have currently ratified the agreement as of May 2019. Ratification by China and Iceland remains to be completed.

Capture volume of the Arctic Ocean (FAO major fishing area 18) yields only up to 589 tons annually in 2000-2018 (FAO 2018). This survey however does not mention the CAO specifically, and might also include adjacent shelf seas. By another estimation, the fish catch of Alaskan Arctic coast yields 89 thousand tons, Canadian commercial and small-scale local fishery yields 94 thousand tons, and Russian Arctic waters yield 770 thousand tons from 1950-2006 (Zeller et., al. 2011). In all cases, fishery activity in the Arctic Ocean is small-scale and limited to coastal waters and brackish nearshore waters.

Today, the problem of Abandoned, Lost or otherwise Discarded Fishing Gear (ALDFG) is causing great concern for marine environment. It is estimated that ALDFG accounts for less than 10 percent of global marine litter by volume (Macfadyen et al 2009). ALDFG has a number of negative environmental impacts, including ghost fishing, entanglement, and being a vector for invasive species. ALDFG should be taken into account when assessing effects of high seas fisheries in the CAO. In the Barents Sea, where extensive fisheries occur, fishery-related macro litter encompasses approximately 40-60 % of litter as bycatch in trawls (Prokhorova and Grøsvik, 2019).

5.3.3 Oil and gas activities and related shipping

Oil- and gas-related shipping activity in the Northern Sea Route (NSR) is increasing year-round mainly in the Kara Sea area, with increasing oil and Liquefied Natural Gas (LNG) production planned from coastal areas. By 2024, Russia plans to quadruple cargo shipment volume compared to 2018, mainly due to natural resources shipments.

From the coast off the Kara Sea and Pechora Sea of the Russian Arctic, crude oil and LNG is extracted and transported by tankers to the non-Arctic regions. Additionally, the new LNG development project in the Ob Bay (Arctic LNG 2) is about to start. For Arctic LNG 2, most of the LNG tankers are powered by LNG, without carrying heavy fuel oil. However, the overall increase of large vessel traffic along the NSR is becoming a reality.

There are currently no crude oil developments in the CAO area, but oil spills from activities on the shelves (e.g. in the Russian Arctic) may affect the CAO ecosystem and should be taken into account in an assessment of potential impacts of oil and gas activities on the CAO ecosystem.

5.3.4 Offshore mining

Ocean mining, while relatively small, has expanded and likely will expand into the Arctic and Antarctic regions as sea ice melts (IPBES 2019).

There is increasing interest worldwide in the potential for deep-sea mining to serve as an engine for "Blue Growth" and to drive sustainable economic development (European Commission, 2012; Wedding et al., 2015). There is also an increasing concern that the direct and indirect impacts of mineral extraction in the deep sea will result in a significant loss of biological diversity (CBD, 1992; Wedding et al., 2015).

Direct impacts occur through the removal of target material and associated organisms within the mine site, and include the destruction of biota as well as habitat loss, fragmentation, and modification through altered mineral and sediment composition, geomorphology, and biogeochemical processes (Ellis, 2001; Van Dover, 2014; Jones et al., 2017).

Potential indirect impacts on the seabed and water column, both within and outside of the directly mined area, include the smothering of habitat and biota, interference with feeding activities, and the release and spread of nutrient-rich and toxin-laden water from the generation of plumes (Ellis, 2001; Boschen et al., 2013). Additional potentially harmful diffuse effects include those from light, noise, and electromagnetic disturbance (Van Dover, 2014; MIDAS, 2016). The scale over which these indirect impacts are likely to occur is largely unknown, and most of the effects remain unstudied.

While biodiversity loss is recognized as a major global environmental problem (Weikard, 2002), the importance of biodiversity in the deep ocean merits clarification, particularly given that most species in the CAO remain undiscovered or unidentified (Bluhm et al 2011).

The seafloor of the CAO is not fully defined in geographical terms, pending decisions on the claims by coastal states for extended jurisdiction of mineral resources on their continental shelves. The "Area", also in the CAO, is governed by the International Seabed Authority (ISA), which is a UN body based in Jamaica. ISA has the duty to ensure the effective protection of the marine environment from harmful effects that may arise from deep-seabed related activities.

5.3.5 Tourism

In the 21 century, international tourists in the Arctic region has been increasing to reach 6 million including 1.2 million cruise tourists in 2015 (National Ocean Economics Program) (Tanaka et., al. 2018). Some Arctic tourism cruise routes are beginning to enter the CAO. Tourism in the CAO is likely to generally be associated with ships, although tourists are flown in to the Russian Barneo station established at the North Pole in spring.

References

- AMAP Assessment (2009) Persistent Organic Pollutants (POPs) in the Arctic <u>https://www.amap.no/doc-uments/doc/amap-assessment-2009-persistent-organic-pollutants-pops-in-the-arctic/45</u>
- AMAP Assessment (2011) Mercury in the Arctic <u>https://www.amap.no/documents/doc/amap-assess-</u> ment-2011-mercury-in-the-arctic/90
- AMAP Assessment (2016) Chemicals of Emerging Arctic Concern. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway. xvi+353pp. Available at: https://www.amap.no/documents/doc/AMAP-Assessment-2016-Chemicals-of-Emerging-Arctic-Concern/1624.
- AMAP (2017). AMAP Assessment 2016: Chemicals of Emerging Arctic Concern. Arctic Monitoring and Assessment Programme (AMAP), Oslo, Norway. xvi+353pp. <u>file:///C:/Users/bjoerneg/Desk-top/Amap%202016%20Chemicals%20of%20emerging%20concern.pdf</u>.
- AMSA (2009). Arctic Marine Shipping Assessment (AMSA). Arctic Council, second printing.
- Armitage, J.M., C.L. Quinn and F. Wania, 2011. Global climate change and contaminants an overview of opportunities and priorities for modelling the potential implications for long-term human exposure to organic compounds in the Arctic. Journal of Environmental Monitoring, 13:1532-1546.
- Bergmann M, B. Lutz, M.B. Tekman, L. Gutow, 2017. Citizen scientists reveal: marine litter pollutes arctic beaches and affects wildlife. Mar Pollut Bull 125:535–540. <u>https://doi.org/10.1016/j.marpolbul.2017.09.055</u>;
- Bluhm, B.A., Ambrose, W.G., Bergmann, M., Clough, L.M., Gebruk, A.V., Hasemann, C., Iken, K., Klages, M., MacDonald, I.R., Renaud, P.E. and Schewe, I., 2011. Diversity of the arctic deep-sea benthos. *Marine Biodiversity*, 41(1), pp.87-107.
- Boschen, R. E., Rowden, A. A., Clark, M. R., and Gardner, J. P. A. (2013). Mining of deep-sea seafloor massive sulfides: a review of the deposits, their benthic communities, impacts from mining, regulatory frameworks and management strategies. Ocean Coast. Manag. 84, 54–67. doi: 10.1016/j.ocecoaman.2013.07.005

- De Lucia, V., Prip, C., Kraabel, K.D. and Primicerio, R., 2018. Arctic Marine biodiversity in the high seas between regional and global governance. Arctic Review, 9, pp.264-266.
- Ellis, D. V. (2001). A review of some environmental issues affecting marine mining. Mar. Georesour. Geotechnol. 19, 51–63. doi: 10.1080/106411901093 53804
- European Commission (2012). Blue Growth—Opportunities for Marine and Maritime Sustainable Growth. Luxembourg: European Commission.

FAO Fishstat, FAO SOFIA 2018

- Grannas, A.M., C. Bogdal, K.J. Hageman, C. Halsall, T. Harner, H. Hung, R. Kallenborn, P. Klán, J. Klánová, R.W. Macdonald, T. Meyer and F. Wania, 2013. The role of the global cryosphere in the fate of organic contaminants. Atmospheric Chemistry and Physics, 13:3271-3305.
- Huntington H.P., R. Daniel, A. Hartsig, K. Harun, M. Heiman, R. Meehan, et al. 2015 Vessels, risks, and rules: planning for safe shipping in Bering Strait Mar. Policy, 51 (2015), pp. 119-127
- IMO 2020. <u>http://www.imo.org/en/About/Conventions/ListOfConventions/Pages/International-</u> Convention-for-the-Prevention-of-Pollution-from-Ships-(MARPOL).aspx
- IPBES (2019). Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. eds. Sandra Díaz and others. Paris, France: IPBES secretariat.
- Jones, D. O., Kaiser, S., Sweetman, A. K., Smith, C. R., Menot, L., Vink, A., et al. (2017). Biological responses to disturbance from simulated deep-sea polymetallic nodule mining. PLoS ONE 12:e0171750. doi: 10.1371/journal.pone.0171750
- Karlsson T.M., Arneborg, L., Broström, G., Almroth BC, Gipperth L, Hassellöv, M. 2018. The unaccountability case of plastic pellet pollution. <u>Marine Pollution Bulletin</u>, 129: 52-60. CBD (1992). Convention on Biological Diversity. 1760 UNTS 79.
- Kallenborn, R., C. Halsall, M. Dellong and P. Carlsson, 2012. The influence of climate change on the global distribution and fate processes of anthropogenic persistent organic pollutants. Journal of Environmental Monitoring, 14:2854-2869.
- Kühn, S., Schaafsma, F.L., van Werven, B. et al., 2018. Plastic ingestion by juvenile polar cod (*Boreogadus saida*) in the Arctic Ocean, Polar Biol 41: 1269-1278. https://doi.org/10.1007/s00300-018-2283-8
- Kühn, S., Schaafsma, F.L., van Werven, B., Flores, H., Bergmann, M., Egelkraut-Holtus, M., Tekman, M.B. and van Franeker, J.A., 2018. Plastic ingestion by juvenile polar cod (*Boreogadus saida*) in the Arctic Ocean. *Polar biology*, 41(6), pp.1269-1278.
- Lusher A.L., V. Tirelli, I. O'Connor, R. Officer, 2015. Microplastics in Arctic polar waters: the first reported values of particles in surface and sub-surface samples. Sci Rep 5:14947. https://doi.org/10.1038/srep14947;
- Lusher, A., Bråte, I.L.N., Hurley, R., Iversen, K. and Olsen, M., 2017. Testing of methodology for measuring microplastics in blue mussels (Mytilus spp) and sediments, and recommendations for future monitoring of microplastics (R & D-project).
- Lydersen, C., Gjertz, I., Weslawski, J.M. 1989. Stomach contents of autumn-feeding marine vertebrates from Hornsund, Svalbard. Polar Rec 25:107–114. https://doi.org/10.1017/S0032247400010408;
 Mallory, M.L. 2008. Marine plastic debris in Northern Fulmars from the Canadian high Arctic. Mar Pollut Bull 56:1501–1504. https://doi.org/10.1016/j.marpolbul.2008.04.017;
- Ma, J., H. Hung, C. Tian and R. Kallenborn, 2011. Revolatilization of persistent organic pollutants in the Arctic induced by climate change. Nature Climate Change, 1:255-260.
- Martin, A.R., Clarke, M.R.,1986. The diet of sperm whales (Physeter macrocephalus) captured between Iceland and Greenland. J Mar Biol Assoc UK 66:779–790. https://doi.org/10.1017/S0025315400048426; and Finley, K.J. 2001. Natural history and conservation of the Greenland whale or bowhead, in the northwest Atlantic. Arctic 54:55–76. DOI: <u>https://doi.org/10.14430/arctic764</u>

- Macfadyen, G., Huntington, T. And Cappell, R. 2009: "Abandoned, lost or otherwise discarded fishing gear", FAO FISHERIES AND AQUACULTURE TECHNICAL PAPER No. 523, UNEP REGIONAL SEAS REPORTS AND STUDIES No. 185: <u>http://www.fao.org/3/i0620e/i0620e00.htm</u>
- MIDAS (2016). Implications of MIDAS Results for Policy Makers: Recommendations for Future Regualtions. Available online at: http://www.eu-midas.net/sites/ default/files/downloads/MIDAS_recommendations_for_policy_low-res.pdf
- Nevalainin M., I. Helle, and J. Vanhatalo, 2016. Preparing for the unprecedented Toward quantitative oil risk assessment in the Arctic marine areas. Marine Pollution Bulletin 114 (2017) 90–101.
- Nevalainin M., I. Helle, and J. Vanhatalo, 2018. Estimating the acute impacts of Arctic marine oil spills using expert elicitation. Marine Pollution Bulletin 131 (2018) 782-792.
- Obbard R.W., S. Sadri, Y.Q. Wong, A.A. Khitun, I. Baker, R.C. Thompson, 2014. Global warming releases microplastic legacy frozen in Arctic sea ice. Earth's Future 2:315–320. https://doi.org/10.1002/2014EF000240.;
- PAME, 2020: THE INCREASE IN ARCTIC SHIPPING 2013-2019, ARCTIC SHIPPING STATUS REPORT (ASSR) #1 March 31, 2020, Available at: <u>https://www.pame.is/projects/arctic-marine-shipping/arctic-shipping-status-reports/723-arctic-shipping-report-1-the-increase-in-arctic-shipping-2013-2019-pdf-version/file</u>
- Peeken I, S. Primpke, B. Beyer, J. Gutermann, C. Katlein, T. Krumpen, M. Bergmann, Laura Hehemann and G. Gerdts, 2018. "Arctic sea ice is an important temporal sink and means of transport for microplastic." Nature Communications 9, 24 April 2018.
- Prokhorova T. and Grøsvik B.E. 2019. Marine litter. In: Survey report from the joint Norwegian/Russian ecosystem survey in the Barents Sea and adjacent waters, August-October 2018 (eds by van der Meeren, G.I and Prozorkevich D.). IMR/PINRO Joint Report Series, No. 2/2019
- Provencher, J.F., Gaston, A.J., Mallory, M.L., O'Hara, P.D., Gilchrist, H.G. 2010. Ingested plastic in a diving seabird, the Thick-Billed Murre (Uria Iomvia), in the eastern Canadian Arctic. Mar Pollut Bull 60:1406–1411. https://doi.org/10.1016/j.marpolbul.2010.05.017
- Safety, E.A.M., 2018. Arctic Marine Shipping Assessment Recommendations. Sustainable Shipping in a Changing Arctic, p.447.
- Smith, L.C. and Stephenson, S.R., 2013. New Trans-Arctic shipping routes navigable by midcentury. Proceedings of the National Academy of Sciences, 110(13), pp.E1191-E1195.
- Stevenson, T.C., Davies, J., Huntington, H.P. and Sheard, W., 2019. An examination of trans-Arctic vessel routing in the Central Arctic Ocean. *Marine Policy*, *100*, pp.83-89.
- Tanaka, M. and Osabe, T. 2018 Current Activities and Perspectives of the Arctic Tourism, The 33rd Intl. Smp. on Okhotsk Sea & Polar Oceans, pp.109-112
- Tekman, M.B., Krumpen, T., Bergmann, M., 2017. Marine litter on deep Arctic seafloor continues to increase and spreads to the north at the Hausgarten observatory. Deep Sea Resl 120:88–99. <u>https://doi.org/10.1016/j.dsr.2016.12.011</u>
- Trevail, A.M., Gabrielsen, G.W., Kühn, S., Van Franeker, J.A. 2015. Elevated levels of ingested plastic in a high Arctic seabird, the Northern Fulmar (Fulmarus glacialis). Polar Biol 38:975–981. <u>https://doi.org/10.1007/s00300-015-1657-4</u>Van Dover, C. L. (2014). Impacts of anthropogenic disturbances at deep-sea hydrothermal vent ecosystems: a review. Mar. Environ. Res. 102, 59–72. doi: 10.1016/j.marenvres. 2014.03.008
- Wedding, L. M., Reiter, S. M., Smith, C. R., Gjerde, K. M., Kittinger, J. N., Friedlander, A. M., et al. (2015). Managing mining of the deep seabed. Science 349, 144–145. doi: 10.1126/science.aac6647
- Weikard, H.-P. (2002). Diversity Functions and the Value of Biodiversity. Land Econ. 78, 20–27. doi: 10.2307/3146920
- Wöhrnschimmel, H., M. MacLeod and K. Hungerbühler, 2012a. Global multimedia source–receptor relationships for persistent organic pollutants during use and after phase-out. Atmospheric Pollution Research, 3:392-398.
- Wöhrnschimmel, H., P. Tay, H. von Waldow, H. Hung, Y.F. Li, M. MacLeod and K. Hungerbuhler, 2012b. Comparative assessment of the global fate of *α*- and β-hexachlorocyclohexane before and after phaseout. Environmental Science and Technology, 46:2047-2054.

Zeller, Dirk & Booth, Shawn & Pakhomov, Evgeny & Swartz, W. & Pauly, D.. (2011). Arctic fisheries catches in Russia, USA, and Canada: Baselines for neglected ecosystems. Polar Biology. 34. 955-973. 10.1007/s00300-010-0952-3.

More literature on plastic

Bergmann M, Mützel S, Primpke S, Tekman MB, Trachsel J, Gerdts G. 2019. White and wonderful? Microplastics prevail in snow from the Alps to the Arctic. Sci. Adv. 5: eaax1157 14 August 2019.

- Cózar A., et. al. 2017. The Arctic Ocean as a dead end for floating plastics in the north Atlantic branch of the thermohaline circulation. Sci Adv 3: e1600582. DOI: 10.1126/sciadv.1600582.
- Lusher A.L.,.V. Tirelli, I. O'Connor, R. Officer, 2015. Microplastics in Arctic polar waters: the first reported values of particles in surface and sub-surface samples. Sci Rep 5:14947. https://doi.org/10.1038/srep14947;
- Mallory, M.L. 2008. Marine plastic debris in Northern Fulmars from the Canadian high Arctic. Mar Pollut Bull 56:1501–1504. https://doi.org/10.1016/j.marpolbul.2008.04.017;
- Trevail AM, S. Kühn, G.W. Gabrielsen, 2015. The state of marine microplastic pollution in the Arctic. Norwegian Polar Institute, Tromso, p 23;

More literature on Garbage

Buhl-Mortensen L, P. Buhl-Mortensen, 2017. Marine litter in the nordic seas: distribution composition and abundance. Mar Pollut Bull 125:260–270. https://doi.org/10.1016/j.marpolbul.2017.08.048;

More AMAP REPORTS

POPs-climate interactions

Proposed assessment products currently include:

- AMAP Assessment Report on "POPs climate interactions" (possible targets for release include scientific conferences in 2021/2022, e.g. SETAC meetings)
- Summary for Policy Makers (target for release/delivery: Arctic Council Ministerial meeting 2021) The scope of this work is not just 'central arctic ocean' and not just marine.
- A scoping document is posted on the AMAP website on the following link: https://www.amap.no/workarea/document/2918. A table of contents is found in Annex 1. Timeline is found in Annex 2.

Mercury Assessment 2021

Proposed assessment products currently include:

- AMAP Assessment Report on "AMAP Assessment 2021: Mercury in the Arctic" (possible targets for release include Minamata COP (fall? 2021) and International Conference on Mercury as a Global Pollutant (South Africa, July 2021))
- Summary for Policy Makers (target for release/delivery: Arctic Council Ministerial meeting 2021) The scope of this work is not just 'central arctic ocean' and not just marine.
- A scoping document is posted on the AMAP website on the following link: https://www.amap.no/workarea/document/2917. A table of contents is found in Annex 1. Timeline is found in Annex 2.

POPs-climate interactions: https://www.amap.no/work-area/document/2918 (upcoming)

Mercury Assessment 2021: https://www.amap.no/work-area/document/2917 (upcoming)

- AMAP Assessment 2015: Temporal Trends in Persistent Organic Pollutants in the Arctic <u>https://www.amap.no/documents/doc/amap-assessment-2015-temporal-trends-in-persis-</u> <u>tent-organic-pollutants-in-the-arctic/1521</u>
- AMAP, 2015. Summary for Policy-makers: Arctic Pollution Issues 2015 <u>https://www.amap.no/docu-ments/doc/summary-for-policy-makers-arctic-pollution-issues-2015/1195</u>
- AMAP Assessment 2018: Arctic Ocean Acidification <u>https://www.amap.no/documents/doc/amap-as-sessment-2018-arctic-ocean-acidification/1659</u>
- AMAP, 2019: Arctic Ocean Acidification Assessment 2018: Summary for Policy-Makers https://www.amap.no/documents/doc/arctic-ocean-acidification-assessment-2018-summary-for-policy-makers/1762
- AMAP, 2017. Adaptation Actions for a Changing Arctic: Perspectives from the Bering-Chukchi-Beaufort Region (applicable to the region mentioned) <u>https://www.amap.no/documents/doc/adaptation-actions-for-a-changing-arctic-perspectives-from-the-bering-chukchi-beaufort-region/1615</u>
- AMAP, 2017. Adaptation Actions for a Changing Arctic (AACA) Bering/Chukchi/Beaufort Region Overview report (applicable to the region mentioned) <u>https://www.amap.no/documents/doc/adaptation-actions-for-a-changing-arctic-aaca-beringchukchibeaufort-region-overview-report/1531</u>
- AMAP, 2017. Adaptation Actions for a Changing Arctic: Perspectives from the Baffin Bay/Davis Strait Region (applicable to the region mentioned) <u>https://www.amap.no/documents/doc/adaptation-actions-for-a-changing-arctic-perspectives-from-the-baffin-baydavis-strait-region/1630</u>
- AMAP, 2017. Adaptation Actions for a Changing Arctic (AACA) Baffin Bay / Davis Strait Region Overview report (applicable to the region mentioned) <u>https://www.amap.no/documents/doc/adaptation-actions-for-a-changing-arctic-aaca-baffin-bay-davis-strait-region-overview-report/1530</u>
- AMAP, 2017. Adaptation Actions for a Changing Arctic: Perspectives from the Barents Area (applicable to the region mentioned) <u>https://www.amap.no/documents/doc/adaptation-actions-for-a-chang-ing-arctic-perspectives-from-the-barents-area/1604</u>
- AMAP, 2017. Adaptation Actions for a Changing Arctic (AACA) Barents Area Overview report. <u>https://www.amap.no/documents/doc/adaptation-actions-for-a-changing-arctic-aaca-bar-ents-area-overview-report/1529</u>
- AMAP Assessment 2018: Biological Effects of Contaminants on Arctic Wildlife and Fish https://www.amap.no/documents/doc/amap-assessment-2018-biological-effects-of-contam-inants-on-arctic-wildlife-and-fish/1663
- AMAP, 2019. AMAP Assessment 2018: Biological Effects of Contaminants on Arctic Wildlife and Fish. Summary for Policy-Makers <u>https://www.amap.no/documents/doc/biological-effects-of-contaminants-on-arctic-wildlife-and-fish.-summary-for-policy-makers/1763</u>
- AMAP. 2016. Influence of Climate Change on Transport, Levels, and Effects of Contaminants in Northern Areas – Part 2 <u>https://www.amap.no/documents/doc/influence-of-climate-change-on-transport-levels-and-effects-of-contaminants-in-northern-areas-part-2/1561</u>
- Socio-Economic Drivers of Change in the Arctic <u>https://www.amap.no/documents/doc/socio-eco-nomic-drivers-of-change-in-the-arctic/1115</u>

The Influence of Global Change on Contaminant Pathways to, within, and from the Arctic: <u>https://www.amap.no/documents/doc/amap-assessment-2002-the-influence-of-global-</u> <u>change-on-contaminant-pathways-to-within-and-from-the-arctic/94</u>
6 ToR e. Review and report on new studies on fishes in the CAO ecosystem

Contributing: Hauke Flores, Harald Gjøsæter, Kevin Hedges, Pauline Snoeijs-Leijonmalm, Ingeborg Mulder, Barbara Niehoff, Hyoung Chul Shin

6.1 Introduction

Over the years, several – partly overlapping – initiatives have been undertaken to describe, review, map, and monitor fish populations living in the CAO. It is worth noting that some experts are active in several of these initiatives.

Various working groups under the Arctic Council, such as AMAP, CAFF, and PAME have produced comprehensive reports dealing with Arctic biota, including fishes. The ten Signatories to the "Agreement to Prevent Unregulated High Seas Fisheries in the Central Arctic Ocean" established a group of scientific experts, including experts on fishes (the Scientific Experts on Fish Stocks in the Central Arctic Ocean group; FiSCAO), who met five times and produced reports summarizing the current knowledge about fishes in the CAO (FiSCAO 2017¹, 2018²). In 2019, when most Signatories had ratified the Agreement, the group of experts was re-established as the "Provisional Scientific Coordinating Group (PSCG)" to the Agreement and tasked to develop a monitoring plan for the CAO under the Agreement. The EU, one of the signatories to the Agreement mentioned above, published a report in spring 2020 titled "Review of the research knowledge and gaps on fish populations, fisheries and linked ecosystems in the Central Arctic Ocean (CAO)" (Snoeijs-Leijonmalm et al. 2020). The WGICA, established in 2016, was tasked to report on the current knowledge about all ecosystem components and the physical environment of the CAO ecosystem, to develop advice on IEA and to develop an Ecosystem Overview for the CAO. The two latter tasks distinguish the WGICA from the other groups. To accomplish its tasks, it is vital that the WGICA utilizes the knowledge already collected about fishes in the CAO, to the extent possible, including knowledge that has been summarized and reviewed by FiSCAO, the PSCG and AMAP, CAFF and PAME working groups.

In this chapter we summarize some of the fish research that has been carried out in recent years (2017-2020) in or adjacent to the CAO. We also compiled a list of peer-reviewed articles and reports covering the period 2017-2020, and new knowledge that has been gained through recent and ongoing research activities that have been reported in peer-reviewed journals as well as in cruise reports and project reports.

¹ FiSCAO, 2017. Final Report of the Fourth Meeting of Scientific Experts on Fish Stocks in the Central Arctic Ocean. Tromsø, Norway, September 26-28, 2016, 82 pp. <u>https://www.afsc.noaa.gov/Arctic fish stocks fourth meeting/default.htm</u>.

² FiSCAO, 2018. Final Report of the Fifth Meeting of Scientific Experts on Fish Stocks in the Central Arctic Ocean. Ottawa, Canada, October 24-26, 2017, 45 pp. <u>https://www.afsc.noaa.gov/Arc-tic fish stocks fifth meeting/pdfs/Final report of the 5th FiSCAO meeting.pdf</u>.

L

6.2 An analysis of literature published in the past three years (2017-2020)

The Fifth FiSCAO Report (FiSCAO 2018) provides a summary of literature on marine fishes in the CAO up to 2017. To provide an overview of literature on fishes published during 2017-2020, Alf Hakon Hoel and Pauline Snoeijs Leijonmalm prepared a list of relevant literature prior to the WGICA Meeting on 27-29 April 2020 and attendees updated the list during the meeting . The final list was analyzed by geographical area (shelf areas, gateways, CAO), number of species investigated (one species, multiple-species), and research topic (e.g. abundance, diet, genetics, experimental study). In this analysis, the Beaufort Sea is characterized as outside the CAO (as defined by PAME in 2016), even though parts of the Beaufort Sea are deeper than most shelf areas, and in that respect those parts are more similar to the CAO than to shelf areas. For consistency, the common name "polar cod" is used for the species *Boreogadus saida*, even though this species is also known as "Arctic cod" in some areas. Altogether, 109 publications were included in the analysis.

6.2.1 Shelf area

Most of the studies published during 2017-2020 focus on the shelves and slopes surrounding the CAO where mainly Arctic and Arctic-boreal fish species occur (*Figure 6-1*). The circumpolar genetic variation of polar cod has been assessed (Gordeeva & Mishin 2019, Nelson et al, 2020). Trawl surveys have been carried out in the Siberian shelf LMEs (Kara Sea, Laptev Sea, and East Siberian Sea) to study the role of polar cod in these ecosystems (Chernova 2018), and to assess the size composition and distribution of larval and juvenile polar cod (Mishin et al. 2018). Other studies focused on specific areas within the Siberian shelf such as the Kara Sea (Figure 1) (Antonov et al. 2017; Borodavkina et al. 2019; Dolgov & Benzik 2017; Prokopchuk 2017). Research included diet studies of polar cod (Prokopchuk 2017) and Greenland halibut (*Reinhardtius hippoglossoides*; Dolgov & Benzik 2017), visual observations of Greenland shark (*Somniosus microcephalus*; Borodavkina et al. 2019), and the abundance of polar cod and its fishery potential in the Kara Sea (Antonov et al. 2017).



Figure 6-1. Number of studies on fishes in the CAO and adjacent shelf seas by geographical area.

Bouchard & Fortier (2020) studied the diet composition and feeding success of young polar cod in the Kara Sea and extended their research into other Arctic seas: the Laptev Sea, Beaufort Sea and parts of the Canadian High Arctic. Understanding the polar cod's feeding ecology is important for predicting its response to the ongoing borealization of the Arctic seas. In the Canadian High Arctic, new research on polar cod in 2017-2020 includes genomics, thermal limits, movement and distribution (Drost 2017, Kessel et al. 2017, Wilson et al. 2017, Nelson et al. 2020). Other studies focused on boreal species that are currently expanding their range into the Canadian Arctic Archipelago (e.g. sand lance, *Ammodytes hexapterus*, Falardeau et al. 2017) and the continental slope of northeast Greenland (e.g. northern wolffish, *Anarhichas denticulatus*, Karamushko et al. 2017).

A recent study by Steiner et al. (2019) illustrates how combining physical, biological and socioeconomical models in a common framework can help assess the impacts of climate change on an ecological keystone species, and resulting effects on subsistence fisheries by local communities. Steiner et al. conducted a case study focusing on polar cod, applying a comprehensive framework of linked Climate, Ecosystem and Economic (CEE) models. The study applied two scenarios for future climate projections, the so-called "business as usual scenario" (Representative Concentration Pathway, RCP, 8.5), and a scenario representing the Paris Agreement target limiting global temperature rise to 1.5°C (RCP 2.6). Regional models show declines in sea-ice concentration and sea-ice thickness throughout the whole Canadian High Arctic region over the past three decades. Associated with these changes, phytoplankton primary production increased. Ice-algae primary production also shows a somewhat positive trend, albeit with large inter-annual variability. A pronounced increase in sea-ice velocity indicates decreasing accessibility of hunting grounds for local coastal communities.



Figure 6-2. Projected changes in abundance for polar cod relative to the 2001–2010 average, as simulated with a Dynamic Bioclimatic Envelope Model (DBEM). Thin lines indicate the individual DBEM results with forcing input from three earth system models. Thick lines represent the averaged results from the respective three runs. Dashed lines indicate DBEM results that include impacts of ocean acidification. Red indicates DBEM results using the high CO2 emission scenario (RCP8.5), and blue indicates results for the low emission scenario (RCP2.6). Modified from AMAP (2018).

Future projections show continued rise of ocean temperature, and declines in oxygen concentration, pH and sea-ice concentration until the end of this century. The magnitude of these changes varies according to the scenario applied (RCP 8.5 or RCP 2.6). Model projections for the future (2100) development of polar cod abundance show an average 17% decrease under the RCP 8.5 emission scenario, and ~ 5% in the RCP 2.6 scenario, but the model uncertainty is high (*Figure 6-*2). Steiner et al. (2019) hypothesize that the demise of polar cod will negatively impact specialised predators, such as ringed seals and beluga whales, and therefore will also affect subsistence fisheries. In conclusion, Steiner et al. (2019) recommend that future decisions regarding commercial fisheries need to be precautionary and adaptive in light of the many uncertainties. Furthermore, inclusion of indigenous knowledge and the interests of local communities are essential.

6.2.2 Gateways

Increasing water temperature and loss of sea-ice trigger shifts in species distributions causing community-wide reorganizations and subsequent changes in ecosystem functioning. Ecosystems prone to fauna shifts are located on Arctic inflow shelves (major gateways into the Arctic), such as the Northeast Greenland shelf (Andrews et al. 2019), Barents Sea, and the northern Bering Sea and Chukchi Sea. To characterize potential future variations and changes in ecosystems, baseline information is needed. Several studies have reported such baseline information, including current demersal fish community structure and distributions (Bergstad et al. 2018, De Robertis et al. 2017, Vestfals et al. 2019), seasonal variations in biomass, species assemblages and trophic relationships (Geoffroy et al. 2019, Kortsch et al. 2019, Sigler et al. 2017), and the structure of water masses and plankton in the Atlantic Water Boundary Current (Hop et al. 2019). Other studies focused on the impacts of environmental variables (e.g. temperature, ice-cover), the behavior of individual fish species, and the functioning of fish communities (Frainer et al. 2017, Huserbraten et al. 2019, Marsh et al. 2019, Mueter et al. 2017, Schmidt et al. 2017). Recently recognized stressors for polar cod include plastic litter. Kühn et al. (2018) showed a 2.8 % frequency of occurrence of ingested non-fibrous microplastic particles among 72 polar cod specimens, which is similar to levels in previous studies. In combination with ocean warming and sea-ice decline, plastic ingestion might act as an increasing stressor on polar cod (Kühn et al. 2018).

The deep scattering layer (DSL, >200 m) south of the CAO was described by Gjøsæter et al. (2017) and Knutsen et al. (2017), as documented from most other oceans. Species (zooplankton, small fish) occurring in these layers play a key role in carbon sequestration (Knutsen et al. 2017). Gjøsæter et al. (2017) showed that these layers display diel vertical migration with a clear ascending movement during nighttime and a descending movement during daytime. Berge et al. (2020) showed that artificial light can influence the vertical distribution of fishes and zooplankton.

6.2.3 CAO

Some studies published on the CAO in the last three years used previous collected publicly available data to make model predictions about near future and longer-term (decreased) development of sea-ice cover (Screen & Williamson 2017, Serreze & Meier 2019) and northward movement of sub-Arctic and boreal fishes and shellfish with climate-induced environmental change (Hollowed et al. 2017).

Other studies combined and summarized previous datasets to understand general structures or movement patterns of zooplankton and small fishes (Hobbs et al. 2018) or compiled an organic carbon budget for the deep Arctic Ocean (Wiedmann et al. 2020). Others collected new expedition data in the CAO. Kohlbach et al. (2017) examined the contribution of ice algae-produced carbon to the carbon budget of polar cod and Flores et al. (2018) reported on the biodiversity of sea-ice associated habitats, with emphasis on polar cod and its ice-associated and pelagic prey species, using data from the Polarstern expedition "IceArc" in 2012. Melnikov (2018) used data obtained during Pan-Arctic ice camp expeditions (PAICEX) in 2007-2011 and 2015 to characterize the biodiversity of ice-associated flora and fauna. Most samples in the above-mentioned papers were collected at the outer edges of the Amerasian and Eurasian Basins (Hobbs et al. 2018, Kühn et al. 2018), a few were collected near the Lomonosov Ridge (Hobbs et al. 2018), and some were collected around 89 oN (Kohlback et al. 2017, Hobbs et al. 2018, Melnikov 2018).

6.2.4 Experimental studies

Several experimental studies have been published on Arctic fish species and species that are showing a northward shift in their distribution, such as Atlantic cod (*Gadus morhua*). Most of these studies have looked at the impacts of temperature on lipid allocation (Copeman et al. 2017), habitat loss (Dahlke et al. 2018), condition, survival and growth (Koenker et al. 2018a, Koenker et al. 2018b, Laurel et al. 2017, Laurel et al. 2018), metabolism and swimming performance (Kunz et al. 2018), and behaviour (Schmidt et al. 2017). Most of these studies focused on polar cod, but some also examined saffron cod (*Eleginus gracilis*) (e.g. Copeman et al. 2017) and walleye pollock (*Gadus chalcogrammus*) (e.g. Koenker et al. 2018a). Another well-studied topic is the effect of oil exposure on the long-term survival, growth, lipid allocation, lipid metabolism and biotransformation, and reproductive development of polar cod (Bender et al. 2018, Laurel et al. 2019, Vieweg et al. 2017).

6.2.5 Policy

The opening of the Arctic Ocean provides many opportunities for both Arctic and non-Arctic states. Therefore, several policy papers have examined the challenge of researching sustainable management of living resources (Crépin et al. 2017, IPCC 2018, Landriault 2018, Vestergaard et al. 2018). As climate change is expected to transform the CAO ecosystem, research has focused on how these changes might impact geophysical variables and as a result directly or indirectly impact socio-economic activities like transportation, marine sea food production and resource exploitation (Crépin et al. 2018, IPCC 2018). Other publications focused mainly on fisheries in the CAO, including gaps in fisheries research (Papastavridis 2018, Van Pelt et al. 2017), science coordination (Van Pelt 2017), effective tools for fisheries enforcement (Papastavridis 2018), the role of law (Rayfuse 2019), ocean governance for species on the move (Pinsky et al. 2018), determination of total allowable catch given certain restrictions (Richter et al. 2018), and how social-ecological cross-scale interactions affect the development of Arctic fisheries (Niiranen et al. 2018).

6.2.6 Research topics

The literature on fishes in the Arctic Ocean published in the last three years (2017-2020) covers a wide variety of topics, from life history to food web interactions and policy (*Figure 6-3*). Most studies report the abundance and distribution of certain species or fish assemblages in particular areas – in some cases to assess fishery potential. The number of studies that investigated a single species (N=38) or multiple-species (N = 32) are relatively similar. Several studies focus on the effects of climate change (warming ocean, retreating ice-cover and acidification) in the Arctic region both from an ecological and policy perspective (Figure 2). Topics such as growth, condition, life history, genetics and conservation were little studied in the last three years.



Figure 6-3: Number of studies on fishes in the CAO and adjacent shelves by research topic.

References

For new studies on polar cod see also:

https://link.springer.com/journal/300/43/8?utm_source=toc&utm_medium=email&utm_campai gn=toc_300_43_8&utm_content=etoc_springer_20200818

- Andrews AJ, et al. (2019) Boreal marine fauna from the Barents Sea disperse to Arctic Northeast Greenland. Scientific Reports 9:5799 [http://doi.org/10.1038/s41598-019-42097-x]
- Antonov NP, et al. (2017) Ecology of Arctic cod *Boreogadus saida* (Gadiformes, Gadidae) and its fishery potential in Kara Sea. Journal of Ichthyology 57:721–729 [http://doi.org/10.1134/S0032945217050022]
- Bender, M. L., Frantzen, M., Camus, L., Le Floch, S., Palerud, J., and Nahrgang, J. 2018. Effects of acute exposure to dispersed oil and burned oil residue on long-term survival, growth, and reproductive development in polar cod (*Boreogadus saida*). Marine Environmental Research, 140: 468-477. [http://doi.org/10.1016/j.marenvres.2018.09.005]
- Berge J, Geoffroy M, Daase M, Cottier F, Priou P, Cohen JH, Johnsen G, McKee D, Kostakis I, Renaud PE, Vogedes D, Anderson P, Last KS, Gauthier S (2020) Artificial light during the polar night disrupts Arctic fish and zooplankton behaviour down to 200 m depth. Communications Biology 3: 102 [http://doi.org/10.1038/s42003-020-0807-6]
- Berger, Chloe & Bougas, Berenice & Turgeon, Samuel & Ferchiou, Sophia & Menard, Nadia & Bernatchez, Louis. (2020). Groundtruthing of pelagic forage fish detected by hydroacoustics in a whale feeding area using environmental DNA. Environmental DNA. [http://doi.org/10.1002/edn3.73].
- Bergstad, O. A., Johannesen, E., Høines, Å., Ellingsen, K. E., Lien, V. S., Byrkjedal, I., Yoccoz, N. G., et al. 2017. Demersal fish assemblages in the boreo-Arctic shelf waters around Svalbard during the warm period 2007–2014. Polar Biology, 41: 125-142. [https://doi.org/10.1007/s00300-017-2176-2]
- Bertelsen RG (2018) The International Political Systemic Context of Arctic Marine Resource Governance. Arctic Marine Resource Governance and Development 3:17 [Abstract]
- Borodavkina, M. V., Chernova, N. V., and Chekmeneva, N. A. 2019. About New Findings of the Greenland Shark *Somniosus microcephalus* in the Kara Sea. Journal of Ichthyology, 59: 623-627.
- Bouchard C, et al. (2017) Climate warming enhances polar cod recruitment, at least transiently. Progress in Oceanography 156:121–129 [http://doi.org/10.1016/j.pocean.2017.06.008]

- Bouchard, C. 2020. Ichthyoplankton and pelagic fish assemblages in the Greenland Sea in 2017. Technical report no. 110, Greenland Institute of Natural Resources, Greenland. ISBN 87-91214-88-2 24 pp.
- Bouchard, C., and Fortier, L. 2020. The importance of *Calanus glacialis* for the feeding success of young polar cod: a circumpolar synthesis. Polar Biology. [https://doi.org/10.1007/s00300-020-02643-0]
- Breines, Ragna & Ursvik, Anita & Nymark, Marianne & Johansen, Steinar & Coucheron, Dag. (2008). Complete mitochondrial genome sequences of the Arctic Ocean codfishes *Arctogadus glacialis* and *Boreogadus saida* reveal oriL and tRNA gene duplications. Polar Biology. 31. 1245-1252. [https://doi.org/10.1007/s00300-008-0463-7].
- CAFF (2017) State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland. 195 pp. [https://www.caff.is/marine/marine-monitoringpublications/state-of-the-arctic-marine-biodiversity-report]
- Castellani G., Schaafsma F.L., Arndt s., Lange B.A., Peeken I, Ehrlich J, David CL, Ricker R, Krumpen T, Hendricks S, Schwegmann S, Massicotte P, Flores H (in press) Large Scale Biophysical Characterization of Arctic and Antarctic Under-Ice Environments. Frontiers in Marine Science
- Chernova, N. V. 2017. Catching of Greenland halibut *Reinhardtius hippoglossoides* (Pleuronectidae) on the shelf edge of the Laptev and East Siberian Seas. Journal of Ichthyology, 57: 219-227.
- Chernova N (2018) Arctic cod in the Russian Arctic: new data, with notes on intraspecific forms. Journal of Aquaculture & Marine Biology 7(1):00180 [https://medcraveonline.com/JAMB/JAMB-07-00180.pdf]
- Cheung WWL, Jones MC, Reygondeau G, Stock CA, Lam VWY, Frolicher TL (2016) Structural uncertainty in projecting global fisheries catches under climate change. Ecol Model 325: 57-66 [https://doi.org/10.1016/j.ecolmodel.2015.12.018]
- Christiansen JS (2017) No future for Euro-Arctic ocean fishes? Marine Ecology Progress Series 575:217–227 [http://doi.org/10.3354/meps12192]
- Copeman, L. A., Laurel, B. J., Spencer, M., and Sremba, A. 2017. Temperature impacts on lipid allocation among juvenile gadid species at the Pacific Arctic-Boreal interface: an experimental laboratory approach. Marine Ecology Progress Series, 566: 183-198.
- Crépin AS, et al. (2017a) Operationalising a social-ecological system perspective on the Arctic Ocean. Ambio 46:475–485 [http://doi.org/10.1007/s13280-017-0960-4]
- Crépin AS, et al. (2017b) Arctic Climate Change, Economy and Society (ACCESS): Integrated perspectives. Ambio 46:341–354 [http://doi.org/10.1007/s13280-017-0953-3]
- Dahlke FT, et al. (2018) Northern cod species face spawning habitat losses if global warming exceeds 1.5°C. Science Advances 4(11):eaas8821 [http://doi.org/10.1126/sciadv.aas8821]
- David C, Lange B, Rabe B, Flores H (2015) Community structure of under-ice fauna in the Eurasian central Arctic Ocean in relation to environmental properties of sea-ice habitats. Mar Ecol Prog Ser 522: 15-32 doi 10.3354/meps11156 [http://doi.org/10.3354/meps12040]
- David C, Lange B, Krumpen T, Schaafsma F, van Franeker JA, Flores H (2016) Under-ice distribution of polar cod *Boreogadus saida* in the central Arctic Ocean and their association with sea-ice habitat properties. Polar Biology 39: 981-994 [http://doi.org/10.1007/s00300-015-1774-0]
- Dalpadado, P., B. Bogstad, I. Prokopchuk, A. Gordeeva, A. Dolgov, A. Rey and H. Gjøsæter (2019). Diet and growth of capelin and polar cod in the Barents Sea. Influence of Ecosystem Changes on Harvestable Resources at High Latitudes. The Proceedings of the 18th Russian-Norwegian Symposium, Murmansk, Russia, 5–7 June 2018. E. Shamray, G. Huse, A. Trofimov et al. Murmansk, IMR/PINRO Joint Report Series. 1-2019: 149-150.
- De Robertis A, et al. (2017) Abundance and distribution of Arctic cod (*Boreogadus saida*) and other pelagic fishes over the U.S. Continental Shelf of the Northern Bering and Chukchi Seas. Deep Sea Research II 135:51-65 [http://doi.org/10.1016/j.dsr2.2016.03.002] 2
- Devine BM, et al. (2018) First estimates of Greenland shark (*Somniosus microcephalus*) local abundances in Arctic waters. Scientific reports 8:974 [http://doi.org/10.1038/s41598-017-19115-x]

- Dias Bernardes I, et al. (2020) Study of the Arctic mesopelagic layer with vessel and profiling multifrequency acoustics. Progress in Oceanography 182(102260) [http://doi.org/10.1016/j.pocean.2019.102260]
- Dolgov, A. V., and Benzik, A. N. 2017. Feeding of Greenland halibut *Reinhardtius hippoglossoides* (Pleuronectidae) in the Kara Sea. Journal of Ichthyology, 57: 402-409. [http://doi.org/10.1134/S0032945217030043]
- Drost, H. 2017. Upper thermal limits and acclimation potential of Arctic cod (*Boreogadus saida*): a key food web species in the Arctic Ocean. University of British Columbia, Retrieved from https://open.library.ubc.ca/collections/ubctheses/24/items/1.0343462.
- Dupont N, Durant JM, Langangen Ø, Gjøsæter H, Stige LC (2020). Sea ice, temperature, and prey effects on annual variations in mean lengths of a key Arctic fish, *Boreogadus saida*, in the Barents Sea. ICES Journal of Marine Science. [http://doi.org/10.1093/icesjms/fsaa040]
- Ehrlich J., Schaafsma F.L., Bluhm B.A, Peeken I., Castellani G., Brandt a., Flores H. (2020) Sympagic fauna in- and under Arctic pack ice in the annual sea-ice system of the new Arctic. Frontiers in Marine Science (in press)
- Eriksen E, et al. (2019) Polar cod eggs and larval drift pattern in the Svalbard archipelago. Polar Biology [http://doi.org/10.1007/s00300-019-02549-6]
- Falardeau M, et al. (2017) First records of Pacific sand lance (*Ammodytes hexapterus*) in the Canadian Arctic Archipelago. Polar Biology 40:2291–2296 [http://doi.org/10.1007/s00300-017-2141-0]
- Fey, D. P., and Węsławski, J. M. 2017. Age, growth rate, and otolith growth of polar cod (*Boreogadus saida*) in two fjords of Svalbard, Kongsfjorden and Rijpfjorden. Oceanologia, 59: 576-584. [http://dx.doi.org/10.1016/j.oceano.2017.03.011]
- FiSCAO (2017) Final Report of the Fourth Meeting of Scientific Experts on Fish Stocks in the Central Arctic Ocean. 82 pp. [Link Fourth FiSCAO Report] 3
- FiSCAO (2018) Final Report of the Fifth Meeting of Scientific Experts on Fish Stocks in the Central Arctic Ocean. 45 pp. [Link Fifth FiSCAO Report]
- Flores H, van Franeker JA, Siegel V, Haraldsson M, Strass VH, Meesters EHWG, Bathmann U, Wolff WJ (2012) The association of Antarctic krill *Euphausia superba* with the under-ice habitat. PLoS One 7: e31775 [http://dx.doi.org/10.1371/journal.pone.0031775]
- Flores H, et al. (2018) Under-ice fauna, zooplankton and endotherms. In: Macke A, Flores H (Eds) The Expeditions PS106/1 and 2 of the Research Vessel POLARSTERN to the Arctic Ocean in 2017. Alfred Wegener Institute for Polar and Marine Research, Bremerhaven. pp 123–133 [http://doi.org/10.2312/BzPM_0719_2018]
- Flores H, David C, Ehrlich J, Hardge K, Kohlbach D, Lange BA, Niehoff B, Nothig EM, Peeken I, Metfies K (2019) Sea-ice properties and nutrient concentration as drivers of the taxonomic and trophic structure of high-Arctic protist and metazoan communities. Polar Biology 42: 1377-1395 [http://dx.doi.org/10.1007/s00300-019-02526-z]
- Flores H., Mueter F., ten Boer R., van Dorssen M., Edenfield L., Klasmeier A., Kunz K.L., Maes S., Pinchuk A., Weems J., Zakharova N. (2020) Go West: Sea-ice association of polar cod and its prey in the western Arctic Ocean. Cruise report. Available at: epic.awi.de
- Frainer, A., Primicerio, R., Kortsch, S., Aune, M., Dolgov, A. V., Fossheim, M., and Aschan, M. M. 2017. Climate-driven changes in functional biogeography of Arctic marine fish communities. Proc Natl Acad Sci U S A, 114: 12202-12207.
- Geoffroy, M., Daase, M., Cusa, M., Darnis, G., Graeve, M., Santana Hernández, N., Berge, J., et al. 2019. Mesopelagic sound scattering layers of the High Arctic: Seasonal variations in biomass, Species assemblage, and trophic relationships. Frontiers in Marine Science, 6.
- Gjøsæter H, et al. (2017) Evidence of diel vertical migration of mesopelagic sound-scattering organisms in the Arctic. Frontiers of Marine Science 4:332 [http://doi.org/10.3389/fmars.2017.00332]

- Gjøsæter, H., Huserbråten, M., Vikebø, F., and Eriksen, E. 2020. Key processes regulating the early life history of Barents Sea polar cod. Polar Biology. [http://doi.org/10.1007/s00300-020-02656-9]
- Gjøsæter, H., Ingvaldsen, R., Christiansen, J.S., 2020. Acoustic scattering layers reveal a faunal connection across the Fram Strait. Progress in Oceanography 185:102348. [https://doi.org/10.1016/j.pocean.2020.102348]
- Gradinger RR, Bluhm BA (2004) In-situ observations on the distribution and behavior of amphipods and Arctic cod (*Boreogadus saida*) under the sea ice of the High Arctic Canada Basin. Polar Biol 27: 595-603 [https://doi.org/10.1007/s00300-004-0630-4]
- Gordeeva, N. V., and Mishin, A. V. 2019. Population Genetic Diversity of Arctic Cod (*Boreogadus saida*) of Russian Arctic Seas. Journal of Ichthyology, 59: 246-254. [https://doi.org/10.1134/S0032945219020073]
- Grebmeier JM, et al. (2018) Trends in benthic macrofaunal populations, seasonal sea ice persistence, and bottom water temperatures in the Bering Strait region. Oceanography 31:136–151 [http://doi.org/10.5670/oceanog.2018.224]
- Harris P, et al. (2018) Arctic marine conservation is not prepared for the coming melt. ICES Journal of Marine Science 75:61–71 [http://doi.org/10.1093/icesjms/fsx153]
- Haug, T., Bogstad, B., Chierici, M., Gjøsæter, H., Hallfredsson, E. H., Høines, Å. S., Hoel, A. H., et al. 2017.
 Future harvest of living resources in the Arctic Ocean north of the Nordic and Barents Seas: A review of possibilities and constraints. Fisheries Research, 188: 38-57.
 [http://dx.doi.org/10.1016/j.fishres.2016.12.002]
- Hoag H (2017) Nations put science before fishing in the Arctic historic fishing ban gives scientists time to probe ecology as northern waters warm. Science 358(6368):1235 [http://doi.org/10.1126/science.358.6368.1235]
- Hobbs, L., Cottier, F. R., Last, K. S., and Berge, J. 2018. Pan-Arctic diel vertical migration during the polar night. Marine Ecology Progress Series, 605: 61-72. [https://doi.org/10.3354/meps12753]
- Hollowed, A. B., Cheng, W., Loeng, H., Logerwell, E., and Reist, J. 2017. Regional Assessment of Climate Change Impacts on Arctic Marine Ecosystems. In Climate Change Impacts on Fisheries and Aquaculture, pp. 703-728. Ed. by B. F. Phillips, and M. Pérez-Ramirez.
- Hop, H., Assmy, P., Wold, A., Sundfjord, A., Daase, M., Duarte, P., Kwasniewski, S., et al. 2019. Pelagic ecosystem characteristics across the Atlantic water boundary current from Rijpfjorden, Svalbard, to the Arctic Ocean During Summer (2010–2014). Frontiers in Marine Science, 6. [http://doi.org/10.3389/fmars.2019.00181]
- Hop H, Poltermann M, Lonne OJ, Falk-Petersen S, Korsnes R, Budgell WP (2000) Ice amphipod distribution relative to ice density and under-ice topography in the northern Barents Sea. Polar Biol 23: 357-367 doi [http://doi.org/10.1007/s003000050456]
- Huserbråten MBO, et al. (2019) Polar cod in jeopardy under the retreating Arctic sea ice. Communications Biology 2:407 [http://doi.org/10.1038/s42003-019-0649-2]
- ICES (2017) Interim Report of the ICES/PICES/PAME Working Group on Integrated Ecosystem Assessment (IEA) for the Central Arctic Ocean (WGICA). WGICA 2017 REPORT 19-21 April 2017. Seattle, USA. ICES CM 2017/SSSGIEA:11. 47 pp. [https://www.ices.dk/community/groups/Pages/WGICA.aspx]
- ICES (2018) Report of the Arctic Fisheries Working Group (AFWG), 18–24 April 2018, Ispra, Italy. ICES CM 2018/ACOM:06. 865 pp. [http://www.ices.dk/sites/pub/Publication%20Reports/Expert%20Group%20Report/acom/2018/AFWG/00-AFWG%202018%20Report.pdf]
- Ingvaldsen RB, et al. (2017) Atlantic cod (Gadus morhua) feeding over deep water in the high Arctic. Polar Biology 40:2105–2111 [http://doi.org/10.1007/s00300-017-2115-2]
- Ingvaldsen, R. B., Gjøsæter, H., Hallfredsson, E., Haug, T., Hosia, A., Jørgensen, L. L., Knutsen, T., et al. 2017. Cruise report SI_ARCTIC/Arctic Ecosystem Survey R/V Helmer Hanssen, 2-16 September 2016. Toktrapport/Havforskningsinstituttet.

- Ingvaldsen, R. B., Gjøsæter, H., Haug, T., Jørgensen, L. L., Knutsen, T., Lødemel, H. H., Menze, S., et al. 2017. Cruise report SI_ARCTIC/Arctic Ecosystem Survey R/V Helmer Hanssen, 21 August-7 September 2017. Toktrapport/Havforskningsinstituttet, Nr. 5-2017.
- IPCC (2018) Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. Masson-Delmotte V et al. (Eds) in Press. [https://www.ipcc.ch/sr15]
- Johannesen, E., Mørk, H. L., Korsbrekke, K., Wienerroither, R., Eriksen, E., Fossheim, M., Wenneck, T. d. L., et al. 2017. Arctic fishes in the Barents Sea 2004-2015: Changes in abundance and distribution. IMR/PINRO Joint Report Series, 2017-1.
- Kaartvedt, S., Titelman, J., and Browman, H. 2018. Planktivorous fish in a future Arctic Ocean of changing ice and unchanged photoperiod. ICES Journal of Marine Science, 75: 2312-2318. [http://doi.org/10.1093/icesjms/fsx248]
- Karamushko, O. V., Birkjedal, I., Christiansen, J. S., and Lynghammar, A. 2017. First record of juvenile northern wolffish Anarhichas denticulatus Krøyer, 1845, on the continental slope of Northeast Greenland. Journal of Ichthyology, 57: 633-638. [http://doi.org/10.1134/S0032945217040051]
- Kessel ST, et al. (2017) First documented large-scale horizontal movements of individual Arctic cod (*Boreogadus saida*). Canadian Journal of Fisheries and Aquatic Sciences 74:292–296 [http://doi.org/10.1139/cjfas-2016-0196]
- Knutsen T, et al. (2017) High latitude epipelagic and mesopelagic scattering layers A reference for future Arctic ecosystem change. Frontiers in Marine Science 4:334 [http://doi.org/10.3389/fmars.2017.00334]
- Koenker BL, Copeman LA, Laurel BJ (2018) Impacts of temperature and food availability on the condition of larval Arctic cod (*Boreogadus saida*) and walleye pollock (*Gadus chalcogrammus*). ICES Journal of Marine Science 75: 2370-2385 [http://doi.org/10.1093/icesjms/fsy052]
- Koenker, B. L., Laurel, B. J., Copeman, L. A., and Ciannelli, L. 2018. Effects of temperature and food availability on the survival and growth of larval Arctic cod (*Boreogadus saida*) and walleye pollock (Gadus chalcogrammus). ICES Journal of Marine Science. [http://doi.org/10.1093/icesjms/fsy062]
- Kohlbach D, Graeve M, Lange BA, David C, Peeken I, Flores H (2016) The importance of ice algae-produced carbon in the central Arctic Ocean ecosystem: Food web relationships revealed by lipid and stable isotope analyses. Limnol Oceanogr 61: 2027-2044 [http://doi.org/10.1002/lno.10351]
- Kohlbach D, Schaafsma FL, Graeve M, Lebreton B, Lange BA, David C, Vortkamp M, Flores H (2017) Strong linkage of polar cod (*Boreogadus sa*ida) to sea ice algae-produced carbon: Evidence from stomach content, fatty acid and stable isotope analyses. Progress in Oceanography 152: 62-74 [http://doi.org/10.1016/j.pocean.2017.02.003]
- Kortsch S, et al. (2018) Food-web structure varies along environmental gradients in a high-latitude marine ecosystem. Ecography 42:295–308 [http://doi.org/10.1111/ecog.03443]
- Kunz, K. L., Claireaux, G., Pörtner, H.-O., Knust, R., and Mark, F. C. 2018. Aerobic capacities and swimming performance of polar cod (*Boreogadus saida*) under ocean acidification and warming conditions. UNSP jeb184473. [http://doi.org/10.1242/jeb.184473].
- Kunz KL, Frickenhaus S, Hardenberg S, Johansen T, Leo E, Portner HO, Schmidt M, Windisch HS, Knust R, Mark FC (2016) New encounters in Arctic waters: a comparison of metabolism and performance of polar cod (*Boreogadus saida*) and Atlantic cod (*Gadus morhua*) under ocean acidification and warming. Polar Biology 39: 1137-1153 [http://doi.org/10.1007/s00300-016-1932-z]
- Kühn S, Schaafsma FL, van Werven B, Flores H, Bergmann M, Egelkraut-Holtus M, Tekman MB, van Franeker JA (2018) Plastic ingestion by juvenile polar cod (*Boreogadus saida*) in the Arctic Ocean. Polar Biology 41: 1269-1278 [http://doi.org/10.1007/s00300-018-2283-8]
- Kvile KO, et al. (2018) Pushing the limit: Resilience of an Arctic copepod to environmental fluctuations. Global Change Biology 24:5426–5439 [http://doi.org/10.1111/gcb.14419]

- Landriault M (2018) Opening a new ocean: Arctic Ocean fisheries regime as a (potential) turning point for Canada's Arctic policy. International Journal 73:158–165 [http://doi.org/10.1177/0020702018764753]
- Lange BA, Katlein C, Nicolaus M, Peeken I, Flores H (2016) Sea ice algae chlorophyll a concentrations derived from under-ice spectral radiation profiling platforms. Journal of Geophysical Research: Oceans 121: 8511-8534 [http://doi.org/10.1002/2016JC011991]
- Lange BA, Katlein C, Castellani G, Fernandez-Mendez M, Nicolaus M, Peeken I, Flores H (2017) Characterizing Spatial Variability of Ice Algal Chlorophyll a and Net Primary Production between Sea Ice Habitats Using Horizontal Profiling Platforms. Frontiers in Marine Science 4 [https://doi.org/10.3389/fmars.2017.00349]
- Laurel, B. J., Copeman, L. A., Iseri, P., Spencer, M. L., Hutchinson, G., Nordtug, T., Donald, C. E., et al. 2019. Embryonic crude oil exposure impairs growth and lipid allocation in a keystone Arctic forage fish. iScience, 19: 1101-1113. [https://doi.org/10.1016/j.isci.2019.08.051]
- Laurel, B. J., Copeman, L. A., Spencer, M., and Iseri, P. 2017. Temperature-dependent growth as a function of size and age in juvenile Arctic cod (*Boreogadus saida*). ICES Journal of Marine Science. [https://doi.org/10.1093/icesjms/fsx028]
- Laurel, B. J., Copeman, L. A., Spencer, M., and Iseri, P. 2018. Comparative effects of temperature on rates of development and survival of eggs and yolk-sac larvae of Arctic cod (*Boreogadus saida*) and walleye pollock (*Gadus chalcogrammus*). ICES Journal of Marine Science. [https://doi.org/10.1093/icesjms/fsy042]
- Manero SA (2018) The Arctic Environmental Protection and the Agenda 2030. Actualidad Jurídica Ambiental, ISSN-e 1989-5666, No. 77, pp. 4-34 5
- Marsh, J. M., Mueter, F. J., and Quinn, T. J. 2019. Environmental and biological influences on the distribution and population dynamics of polar cod (*Boreogadus saida*) in the US Chukchi Sea. Polar Biology. [https://doi.org/10.1007/s00300-019-02561-w]
- Massicotte P, Peeken I, Katlein C, Flores H, Huot Y, Castellani G, Arndt S, Lange BA, Tremblay J-É, Babin M Sensitivity of phytoplankton primary production estimates to available irradiance under heterogeneous sea ice conditions. Journal of Geophysical Research: Oceans 0 [https://doi.org/10.1029/2019jc015007]
- McNicholl DG, et al. (2018) Life history variation across latitudes: observations between capelin (*Mallotus villosus*) from Newfoundland and the eastern Canadian Arctic. Polar Biology 41:643–651 [http://doi.org/10.1007/s00300-017-2225-x]
- Mecklenburg CW, et al. (2018) Marine Fishes of the Arctic Region, Volume I. CAFF Monitoring Series Report 28:1–454 Conservation of Arctic Flora and Fauna, Akureyri, Iceland [https://www.caff.is/monitoring-series/451-marine-fishes-of-the-arctic-region-vol-1]
- Mecklenburg, C. W., Lynghammar, A., Johannesen, E., Byrkjedal, I., Christiansen, J. S., Dolgov, A. V., Karamushko, O. V., et al. 2018. Marine Fishes of the Arctic Region vol 2. CAFF Monitoring Series Report, 28: 455-731.
- Meyer Ottesen, C. A., Hop, H., Falk-Petersen, S., and Christiansen, J. S. 2018. Reproduction and sexual dimorphism of daubed shanny (Teleostei: *Leptoclinus maculatus*) in Svalbard waters. Polar Biology. [https://doi.org/10.1007/s00300-018-2328-z]
- Melnikov IA (2018) Characterization of the biodiversity of modern sea ice in the North Pole Region. Doklady Earth Sciences 480:792–795 [http://doi.org/10.1134/S1028334x18060132]
- Melnikov IA, Chernova NV (2013) Characteristics of under-ice swarming of polar cod *Boreogadus saida* (Gadidae) in the Central Arctic Ocean. Journal of Ichthyology 53: 7-15 [https://doi.org/10.1134/s0032945213010086]
- Mishin, A. V., Evseenko, S. A., Bol'shakov, D. V., and Bol'shakova, Y. Y. 2018. Ichthyoplankton of Russian Arctic Seas: 1. Polar cod *Boreogadus saida*. Journal of Ichthyology, 58: 710-716. [https://doi.org/10.1134/S0032945218050156]
- Moore SE, (2019) Baleen whale ecology in arctic and subarctic seas in an era of rapid habitat alteration. Progress in Oceanography 176 [http://doi.org/10.1016/j.pocean.2019.05.010]

- Mueter FJ, Weems J, Farley EV, Sigler MF (2017) Arctic Ecosystem Integrated Survey (Arctic Eis): Marine ecosystem dynamics in the rapidly changing Pacific Arctic Gateway. Deep-Sea Research Part Ii-Topical Studies in Oceanography 135: 1-6 [http://doi.org/10.1016/j.dsr2.2016.11.005]
- Nelson, R. & Bouchard, Caroline & Madsen, Matias & Præbel, Kim & Rondeau, Eric & Schalburg, Kristian & Leong, Jong & Jantzen, Stuart & Sandwith, Zoe & Puckett, Stephanie & Messmer, Amber & Fevolden, Svein-Eik & Koop, Ben. (2013). Microsatellite loci for genetic analysis of the arctic gadids *Boreogadus saida* and *Arctogadus* glacialis. Conservation Genetics Resources. 5. 445-448. [http://doi.org/10.1007/s12686-012-9824-1].
- Nelson RJ, Bouchard C, Fortier L, Majewski AR, Reist JD, Præbel K, Madsen ML, Rose GA, Kessel ST, Divoky GJ (2020) Circumpolar genetic population structure of polar cod, *Boreogadus saida*. Polar Biology [http://doi.org/10.1007/s00300-020-02660-z]
- Niemi, A., Ferguson, S., Hedges, K., Melling, H., Michel, C., Ayles, B., Azetsu-Scott, K., Coupel, P., Deslauriers, D., Devred, E., Doniol-Valcroze, T., Dunmall, K., Eert, J., Galbraith, P., Geoffroy, M., Gilchrist, G., Hennin, H., Howland, K., Kendall, M., Kohlbach, D., Lea, E., Loseto, L., Majewski, A., Marcoux, M., Matthews, C., McNicholl, D., Mosnier, A., Mundy, C.J., Ogloff, W., Perrie, W., Richards, C., Richardson, E., Reist, R., Roy, V., Sawatzky, C., Scharffenberg, K., Tallman, R., Tremblay, J-É., Tufts, T., Watt, C., Williams, W., Worden, E., Yurkowski, D., Zimmerman, S. 2019. State of Canada's Arctic Seas. Can. Tech. Rep. Fish. Aquat. Sci. 3344: xv + 189 p.
- Niiranen S, et al. (2018) Global connectivity and cross-scale interactions create uncertainty for Blue Growth of Arctic fisheries. Marine Policy 87:321–330 [http://doi.org/10.1016/j.marpol.2017.10.024]
- Nordam T, et al. (2017) Impact of climate change and seasonal trends on the fate of Arctic oil spills. Ambio 46:442–452 [http://doi.org/10.1007/s13280-017-0961-3]
- Norris AJ, McKinley P (2017) The Central Arctic Ocean preventing another tragedy of the commons. Polar Record 53:43–51 [http://doi.org/10.1017/S003224741600067X]
- Papastavridis E (2018) Fisheries enforcement on the High Seas of the Arctic Ocean: gaps, solutions and the potential contribution of the European Union and Its Member States. International Journal of Marine and Coastal Law 33:324–360 [http://doi.org/10.1163/15718085-13320002]
- Pinsky ML, et al. (2018) Preparing ocean governance for species on the move. Science 360(6394):1189–1191 [http://doi.org/10.1126/science.aat2360]
- Prokopchuk, I. P. 2017. Feeding of polar cod *Boreogadus saida* in the Kara Sea. Journal of Ichthyology, 57: 591-602. [http://doi.org/10.1134/S0032945217040129]
- Raskhozheva, E. V., and Karamushko, O. V. 2018. Dynamics of production ccharacteristics of polar cod Boreogadus saida (Gadidae) in the Barents Sea. Journal of Ichthyology, 58: 732-740. [http://doi.org/10.1134/S0032945218050181]
- Rayfuse R (2018) Regulating fisheries in the Central Arctic Ocean: Much ado about nothing? In: Vestergaard N, et al. (Eds) Arctic Marine Resource Governance and Development, Springer Nature. pp. 35–51 [http://doi.org/10.1007/978-3-319-67365-3_3]
- Rayfuse, R. 2019. The role of law in the regulation of fishing activities in the Central Arctic Ocean. Marine Policy, 110. [https://doi.org/10.1016/j.marpol.2019.103562]
- Rees, Helen & Maddison, Ben & Middleditch, David & Patmore, James & Gough, Kevin. (2014). The detection of aquatic animal species using environmental DNA a review of eDNA as a survey tool in ecology. Journal of Applied Ecology. 51. [https://doi.org/10.1111/1365-2664.12306].
- Richter A, et al. (2018) Optimal management under institutional constraints: determining a total allowable catch for different fleet segments in the Northeast Arctic cod fishery. Environmental and Resource Economics 69:811–835 [http://doi.org/10.1007/s10640-016-0106-3]
- Salter, Ian & Joensen, Mourits & Kristiansen, Regin & Steingrund, Petur & Vestergaard, Poul. (2019). Environmental DNA concentrations are correlated with regional biomass of Atlantic cod in oceanic waters. Communications Biology. 2. 461. [https://doi.org/10.1038/s42003-019-0696-8].

- Screen JA, Williamson D (2017) Ice-free Arctic at 1.5 °C? Nature Climate Change 7:230–231 [https://www.nature.com/articles/nclimate3248] 6
- Schmidt M, Gerlach G, Leo E, Kunz KL, Swoboda S, Portner HO, Bock C, Storch D (2017) Impact of ocean warming and acidification on the behaviour of two co-occurring gadid species, *Boreogadus saida* and *Gadus morhua*, from Svalbard. Mar Ecol Prog Ser 571: 183-191 [https://doi.org/10.3354/meps12130]
- Serreze C, Meier WN (2019) The Arctic's sea ice cover: trends, variability, predictability, and comparisons to the Antarctic. Annals of the New York Academy of Sciences 1436:36–53 [http://doi.org/10.1111/nyas.13856]
- Shu, Lu & Ludwig, Arne & Peng, Zuogang. (2020). Standards for Methods Utilizing Environmental DNA for Detection of Fish Species. Genes. 11. 296. [https://doi.org/10.3390/genes11030296].
- Sigler MF, Mueter FJ, Bluhm BA, Busby MS, Cokelet ED, Danielson SL, Robertis AD, Eisner LB, Farley EV, Iken K, Kuletz KJ, Lauth RR, Logerwell EA, Pinchuk AI (2017) Late summer zoogeography of the northern Bering and Chukchi seas. Deep-Sea Research Part II: Topical Studies in Oceanography 135: 168-189 [https://doi.org/10.1016/j.dsr2.2016.03.005]
- Smirnova, E. V., Karamushko, O. V., and Chernova, N. V. 2019. Species Composition, Distribution, and Some Biological Aspects for Fishes of the Genus Lycodes (Zoarcidae) in the Laptev Sea in 2014. Journal of Ichthyology, 59: 488-498. [https://doi.org/10.1134/S0032945219040167]
- Snoeijs-Leijonmalm P, Flores H, Volckaert F, Niehoff B, Schaafsma F, Hjelm J, Hentati-Sundberg J, Niiranen S, Crépin A-S, Österblom H (2020) Review of the research knowledge and gaps on fish populations, fisheries and linked ecosystems in the Central Arctic Ocean (CAO). In: Commission E (ed), Luxemburg. [https://doi.org/10.2826/387890]
- Steiner NS, Cheung WWL, Cisneros-Montemayor AM, Drost H, Hayashida H, Hoover C, Lam J, Sou T, Sumaila UR, Suprenand P, Tai TC and VanderZwaag DL (2019) Impacts of the Changing Ocean-Sea Ice System on the Key Forage Fish Arctic Cod (*Boreogadus saida*) and Subsistence Fisheries in the Western Canadian Arctic—Evaluating Linked Climate, Ecosystem and Economic (CEE) Models. Front. Mar. Sci. 6:179. [https://doi.org/10.3389/fmars.2019.00179]
- Strand, K. O., Sundby, S., Albretsen, J., and Vikebø, F. B. 2017. The Northeast Greenland Shelf as a Potential Habitat for the Northeast Arctic Cod. Frontiers in Marine Science, 4. [https://doi.org/10.3389/fmars.2017.00304]
- Thomsen, Philip, Kielgast, Jos & Iversen, Lars & Moller, Peter & Rasmussen, Morten & Willerslev, Eske. (2012). Detection of a Diverse Marine Fish Fauna Using Environmental DNA from Seawater Samples. PloS one, 7: e41732. [https://doi.org/10.1371/journal.pone.0041732].
- Thomsen, Philip & Moller, Peter & Sigsgaard, Eva & Knudsen, Steen & Jørgensen, Ole & Willerslev, Eske. (2016). Environmental DNA from Seawater Samples Correlate with Trawl Catches of Subarctic, Deepwater Fishes. PLOS ONE. 11. e0165252. [https://doi.org/10.1371/journal.pone.0165252].
- Thorsson JT, et al. (2019) Comparison of near-bottom fish densities show rapid community and population shifts in Bering and Barents Seas. Arctic Report Card 2019 [Link to Arctic Report Cards]
- Troell M, et al. (2017) Seafood from a changing Arctic. Ambio 46, Supplement 3:368–386 [http://doi.org/10.1007/s13280-017-0954-2]
- Troell, M., Eide, A., Isaksen, J., Hermansen, O., and Crepin, A. S. 2017. Seafood from a changing Arctic. Ambio, 46: 368-386.
- Tsuji, Satsuki & Takahara, Teruhiko & Doi, Hideyuki & Shibata, Naoki & Yamanaka, Hiroki. (2019). The detection of aquatic macroorganisms using environmental DNA analysis—A review of methods for collection, extraction, and detection. Environmental DNA. [https://doi.org/10.1002/edn3.21].
- van Franeker JA, Flores H, Van Dorssen M (2009) The Surface and Under Ice Trawl (SUIT). In: Flores H (ed) Frozen Desert Alive - The role of sea ice for pelagic macrofauna and its predators. PhD thesis. University of Groningen, Groningen, pp 181-188
- Van Pelt TI, et al. (2017) The missing middle: Central Arctic Ocean gaps in fishery research and science coordination. Marine Policy 85:79–86 [http://doi.org/10.1016/j.marpol.2017.08.008]

- Vestfals CD, et al. (2019) Spatio-temporal distribution of polar cod (*Boreogadus saida*) and saffron cod (Eleginus gracilis) early life stages in the Pacific Arctic. Polar Biology 42:969–990 [http://doi.org/10.1007/s00300-019-02494-4]
- Vedishcheva, E. V., Maznikova, O. A., and Orlov, A. M. 2019. New data on the age and growth of Greenland halibut, *Reinhardtius hippoglossoides* (Pleuronectidae), from the Laptev Sea. Journal of Ichthyology, 58: 845-850. [http://doi.org/10.1134/S0032945218060164]
- Vieweg, I., Bilbao, E., Meador, J. P., Cancio, I., Bender, M. L., Cajaraville, M. P., and Nahrgang, J. 2018. Effects of dietary crude oil exposure on molecular and physiological parameters related to lipid homeostasis in polar cod (*Boreogadus saida*). Comparative Biochemistry and Physiology Part C: Toxicology & Pharmacology, 206-207: 54-64. [https://doi.org/10.1016/j.cbpc.2018.03.003]
- Westgaard JI, et al. (2017) Genetic population structure in Greenland halibut (Reinhardtius hippoglossoides) and its relevance to fishery management. Canadian Journal of Fisheries and Aquatic Sciences 74:475–485 [http://doi.org/10.1139/cjfas-2015-0430]
- Weydmann A, et al. (2018) Warming of Subarctic waters accelerates development of a key marine zooplankton Calanus finmarchicus. Global Change Biology 24:172–183 [http://doi.org/10.1111/gcb.13864]
- Wiedmann I, Ershova E, Bluhm BA, Nöthig EM, Gradinger RR., Kosobokova K, Boetius A (2020) What Feeds the Benthos in the Arctic Basins? Assembling a Carbon Budget for the Deep Arctic Ocean. Frontiers in Marine Science, 7, [http://doi.org/10.3389/fmars.2020.00224]
- Wilkinson J, et al. (2017) Oil spill response capabilities and technologies for ice-covered Arctic marine waters: A review of recent developments and established practices. Ambio 46:423–441 [http://doi.org/10.1007/s13280-017-0958-y]
- Wilson RE, et al. (2017) Genomics of Arctic Cod. Anchorage, AK: US Dept. of the Interior, Bureau of Ocean Energy Management, Alaska OCS Region. OCS Study BOEM 2017-066. 92 pp. [https://www.boem.gov/BOEM-2017-066-Genomics-of-Arctic-Cod]
- Xavier, J. C., Cherel, Y., Allcock, L., Rosa, R., Sabirov, R. M., Blicher, M. E., and Golikov, A. V. 2018. A review on the biodiversity, distribution and trophic role of cephalopods in the Arctic and Antarctic marine ecosystems under a changing ocean. Marine Biology, 165. [https://doi.org/10.1007/s00227-018-3352-9]
- Xuan BB, Armstrong CW (2017) Marine reserve creation and interactions between fisheries and capturebased aquaculture: A bio-economic model analysis. Natural Resource Modelling 30(2):e12122 [http://doi.org/10.1111/nrm.12122] 7
- Zou LL, Huntington HP (2018) Implications of the convention on the conservation and management of pollock resources in the central Bering Sea for the management of fisheries in the Central Arctic Ocean. Marine Policy 88:132–138 [http://doi.org/10.1016/j.marpol.2017.11.019]

7 ToR f. Continue to identify priority research needs and monitor how identifies knowledge gaps are being addressed and filled

Leads: Jacqueline Grebmeier¹ and Randi Ingvaldsen² - with text and additions from chapter 3, 4, 5, and 6 (see authors of each of the chapters).

¹Chesapeake Biological Laboratory, University of Maryland Center for Environmental Science, Solomons, Maryland, USA; ²Institute of Marine Research, Norway

7.1 Introduction

The goal of ToR f is to identify priority research needs and questions to understand the impact of ongoing environmental change and human activities on the Arctic ecosystem, and to assess whether ongoing and planned activities will address the identified knowledge gaps.

The report is organized as follows; first we describe important case studies from ongoing research, mapping and monitoring (7.2), as well as some new tools and methods necessary for conducting the needed research including recent experience with sampling gear (7.3). Thereafter we list the identified research questions and monitoring needs for each of the ToRs b-e (7.4), before we summarize the overall conclusions and discuss them in relation to earlier assessments as the SAMBR report (CAFF, 2017) and the reports from the FISCAO process (FISCAO 2016 and 2018, PSCG 2020) (7.5).

7.2 Examples of Case studies form ongoing research, mapping and monitoring in the CAO

n inventory of monitoring programs can also be found in PSCG 2020.

7.2.1 Distributed Biological Observatory (DBO)

Variations in upper-ocean hydrography, light penetration, lower and upper trophic levels, pelagic-benthic coupling and carbon cycling are being evaluated through the Distributed Biological Observatory (DBO), which was initiated in 2010 in the Pacific Arctic (Moore and Grebmeier 2018). The DBO sampling approach emphasizes annual standardized sampling by an international suite of ships occupying agreed-to transect lines in order to measure the status and developing trends in key measurements for the marine ecosystem. Continuous data are also obtained through mooring and satellite observations. The first decade of DBO sampling has revealed seasonal and interannual hydrographic changes are driving shifts in biological species composition and abundance, northward range expansions for some temperate species and negative impacts for some ice dependent species. This model of change detection is being expanded to other Arctic regions beyond the initial implementation in the Pacific Arctic. An Atlantic DBO is in development through coordination of ongoing international field activities in the Eurasian Arctic, and planning has started for an effort in Davis Strait/Baffin Bay. The FISCAO process identified the use of standard measurements and section lines in the shelf to outer slope in the DBO network as a potential model for coordinated international activities for the Central Arctic Ocean (CAO), including the developing Synoptic Arctic Survey (SAS) to include a suite of Essential Ocean Variables (EOVs) to be collected by multiple ships sampling in the CAO in a pan-Arctic effort, including associated data exchange (PSCG 2020, Paasche et al. 2019). Information on the DBO can be found at <u>https://dbo.cbl.umces.edu/</u>.



Figure 7.1 The eight sampling regions of the Distributed Biological Observatory (DBO) extending from the northern Bering Sea into the Beaufort Sea in the Pacific Arctic Region that are focused on 'hotspots' of biological productivity and biodiversity. Maximum and minimum median ice extent based on SMMR, SSM/I, and SSMIS satellite-derived sea-ice concentrations (1979–2018) are also shown on this figure (from Grebmeier et al. 2019).

7.2.2 Pacific Arctic Climate Ecosystem Observatory (PACEO)

The Pacific Arctic Climate Ecosystem Observatory (PACEO) is a developing long-term monitoring Pacific Arctic Gateway (PAG) activity in the high Pacific Arctic and CAO (Figure 7.1). PA-CEO includes a set of climate-ecosystem lines extending from west of the Chukchi Borderlands into the Canada Basin in the CAO. This developing international initiative is coordinating multiship operations in the Pacific Arctic Ocean that are including standard physical, biochemical and biological measurements, similar to the DBO. The PACEO sites are occupied by international entities in the CAO with a shared data plan. Key fisheries measurements are also planned to understand the Pacific CAO's marine ecology and the potential impacts of climate change before fishing becomes widespread and to prevent unregulated high seas fisheries in the CAO. Figure 7.2a provides a schematic of the PACEO, multi-ship effort for sampling in the Pacific CAO. Figure 7.2b shows the 10-yr time series stations undertaken as part of the Korean Polar Research Institute (KOPRI) efforts over the Chukchi Borderland into the CAO as part of the PACEO effort.



Proposed international Pacific Arctic

Black vectors: average sea ice motion vectors for Nov. 2007- Apr. 2008 (Sea Ice Beaufort Gyre)

Figure 7.2a Developing international Pacific Arctic climate monitoring sections as part of the Pacific Arctic Climate Ecosystem Observatory (PACEO).



Figure 7.2b. Time series stations occupied on the RV ARAON as part of the PACEO program.

7.2.3 Joint Ocean Ice Studies (JOIS)

Canada has conducted a variety of studies in the Beaufort Sea and adjacent waters since the early 2000s (Figure 7.3). The Joint Ocean Ice Studies (JOIS) program is an international collaboration among Canada, the United States and Japan; the program provides ongoing assessment and monitoring of oceanographic and sea ice conditions through a combination of ice buoys, CTD casts, XCTDs, bongo net tows and closing nets. JOIS program cruises collect data for the Arctic Observing Network - Beaufort Gyre Observing System (AON-BGOS) and more recently represent an important part of Canada's contribution to SAS (Paasche et al. 2019). Information about the 2018 JOIS cruise can be found at: https://www.dfo-mpo.gc.ca/science/atsea-enmer/missions/2018/jois-eng.html.

Symbols: Mooring array in 2012-2013 (TUMSAT/KOPRI/NIPR & WHOI)



Figure 7.3. Sampling and observation locations from the 2019 Joint Ocean Ice Studies - Arctic Observing Network - Beaufort Gyre Observing System– Synoptic Arctic Survey cruise in the Canadian Beaufort Sea and adjacent waters.

7.2.4 Canadian Beaufort Sea Marine Ecosystem Assessment (CBS-MEA)

Surveys of phytoplankton, zooplankton, marine invertebrates, fishes and mammals have been less regular in the Canadian Beaufort Sea, but significant data have been collected through short term (typically 1-3) year programs. The Canadian Beaufort Sea Marine Ecosystem Assessment (CBS-MEA) is a multi-year program that represents a resumption or continuation of the Beaufort Regional Ecosystem Assessment–Marine Fishes Project that was conducted during 2012-2014. CBS-MEA is collecting baseline data on phytoplankton, zooplankton, marine invertebrates and fishes throughout the Canadian Beaufort Sea (Figure 7.4) and supports long-term monitoring. CBS-MEA links to the JOIS program as well as coastal, community-based research programs to integrate data throughout the Beaufort Sea ecosystem and across trophic levels. Sample collection includes rosette and CTD casts, box coring, bongo/multi-net collections, benthic and midwater trawling and ship-based hydroacoustics (EK80). Information about the 2019 CBS-MEA cruise can be found at:

https://www.dfo-mpo.gc.ca/science/atsea-enmer/missions/2019/beaufort-eng.html.



Figure 7.4. 2017 Canadian Beaufort Sea Marine Ecosystem Assessment sampling stations and transects.

7.2.5 Canadian Beaufort Sea Marine Mammal research

Community-based programs focused on marine fishes and mammals provide coastal data collections in the Canadian Beaufort Sea (Figure 7.5a and b). In particular, marine mammal research largely involves community-based programs supplemented with occasional aerial surveys. Community-based marine mammal programs annually collect data on animal health, species distributions/habitat use (through direct observations, moored hydrophones or deployment of pop-off archival satellite tags) and record any unusual occurrences with marine mammals or other ecosystem components. Aerial surveys support population assessments to support harvest management. In combination, these programs provide ongoing data on the presence, habitat use, movement patterns and health of marine mammals.



Figure 7.5a. Positions of eight whales recorded between mid-July to mid-November in 2018 and 2019; majority of records from late-August to mid-October.



Figure 7.5b. Transects flown during an aerial survey on marine mammals in the Beaufort Sea in 2019. Field programs, both vessel (JOIS and CBS-MEA) and community-based (marine mammal health assessments and tagging), are planned for summer-fall 2020, but their status is currently uncertain because of the COVID-19 pandemic and resulting travel restrictions.

7.2.6 Nansen Legacy

The Norwegian research program Nansen Legacy is a 6-year project (2018-2023), mainly using the new Norwegian ice-going research vessel *Kronprins Haakon*. The core activities of the program include research expeditions into ice-covered waters of the northern Barents Sea and adjacent Arctic Basin. This is an integrated ecosystem research program spanning meteorology, ocean physics and chemistry, primary production and higher trophic levels, including marine mammals and birds. The first Nansen Legacy cruise was conducted in fall 2018, and some cruises were conducted in 2019, but not all planned cruises were completed due to required maintenance and improvements to the new vessel during the warranty period. By the end of 2023 the program is expected to complete 370 days at sea. Unfortunately, most cruises planned for the first half of

2020 have been postponed/cancelled due to the COVID-19 pandemic. Information about plans and cruises can be found at: <u>https://arvenetternansen.com/</u>.

7.2.7 Multidiciplinary Drifting Observatory for the Study of Arctic Change (MOSAiC) Expedition

The MOSAiC expedition started on 20 September 2019 using the German research icebreaker *Polarstern*. The goal of the expedition is to spend a full year drifting with the sea ice in the Arctic Ocean and conduct a continuous year-round multidisciplinary study of the coupled atmosphereice-ocean-biosphere system while following the Transpolar Drift (Figure 7.6). Throughout the expedition, scientists record data on sea ice and snow, oceanography, atmosphere, nutrient availability and fluxes, primary productivity, zooplankton, marine invertebrates, and fishes. In addition to observations of ambient conditions and biota, experiments are conducted on the vessel to understand ecosystem processes and linkages, in order to improve physical and ecosystem models. An accurate representation of these processes in models has been lacking so far but is essential for examining how climate change and anthropogenic stressors may affect the Arctic marine ecosystem and biota. Information about the background and progress of the MOSAiC Expedition can be found at: <u>mosaic-expedition.org</u> and <u>https://follow.mosaic-expedition.org</u>.



Figure 7.6. Cruise track of RV Polarstern during the MOSAiC Expedition from 20 September 2019 to 02 June 2020. In May 2020 Polarstern temporarily left the MOSAiC ice floe to exchange crew and cargo near Svalbard.

7.2.8 Ecosystem monitoring in the Atlantic gateway

Monitoring of the Atlantic gateway to the Arctic Ocean is undertaken through repeat section crossings of the shelf-break north of Svalbard (the Hinlopen section at about 20°E). The section is sampled annually in late summer/early autumn (since 2014). The current sampling routine includes physical and chemical oceanography, zooplankton (meso and macro) biomass and species composition, juvenile fish in the upper layers (trawling), pelagic fishes in the mesopelagic layer (trawling) and demersal fishes and benthos (trawling). Underway acoustic measurements

L

(EK80) and visual sightings of marine mammals and sea birds are also conducted. Year-round monitoring of physical, chemical and acoustic properties is conducted with moorings slightly further to the east (at about 30°E as part of the ATWAIN-SIOS project). The repeated section and the moorings contribute to long-term monitoring detecting changes in physical, chemical and biological properties at the Atlantic gateway, and can be used to document changes in the deep scattering layer and the pelagic ecosystem at this entrance to the Arctic Ocean.

7.2.9 Synoptic Arctic Survey (SAS)

The CAO remains profoundly understudied, particularly carbon cycling, ecosystem alteration, and associated changes in atmosphere, ice and ocean physics that influence those biological and biogeochemical systems (Paasche et al. 2019). The region is expected to experience marked changes over the next decades, driven by ongoing climate warming, yet our understanding of key process is limited for this area. SAS seeks to quantify the present states of the physical, biological, and biogeochemical systems. Multiple countries have both confirmed and pending cruises as part of the 2020/2021 SAS networked activities, although many 2020 cruises have been postponed to 2021 due to the COVID-19 pandemic and some cruises are now planned for 2022. Key goals of the SAS are to establish the present state of the Arctic system, to document temporal changes where possible through comparison with historical data, and to quantify linkages between the adjacent shelves, slopes, and deep basins, objectives that are shared with the broader Pan-Arctic effort of the composite SAS. The SAS consists of regional shelf-to-basin shipbased surveys in 2020 and 2021 (and now into 2022) to obtain a Pan-Arctic understanding of essential ocean variables (EOVs) on a quasi-synoptic, spatially distributed basis in which no single nation bears the full burden of collecting the requisite data. The SAS will also provide an important legacy of field activities to future, quasi-decadal assessments of rapid and evolving Arctic Ocean system change.

7.2.10 North Pole 2020 expedition

Members of the Russian Academy of Sciences (RAS) plan a North Pole 2020 expedition involving 20 RAS institutions (<u>https://arctic.ru/news/20200311/932193.html</u>). The icebreaker *Kapitan Dranitsyn* will be frozen in the ice to become the base for the expedition which will also include satellite ice camps. It is expected to start drifting in the area of the New Siberian Islands (about 85°N), after which it will go to the Amundsen Basin, move along the Lomonosov Ridge and turn left, crossing the Gakkel Ridge. Major investigations involved in the expedition (with emphasis to climate change) include hydrometeorology, oceanology and ice studies, air pollution and aerosols, pollution (POPs, microplastic, trace metals and radionuclides), biodiversity (benthos, ice microbiota and plankton, pelagic plankton, marine mammals and seabirds), geology and seismology. *The status of the expedition is currently uncertain because of the COVID-19 pandemic and resulting travel restrictions*.

7.3 New tools, methods, and sampling gear for studies of the CAO

Sampling in the CAO is challenging due to the heavy sea ice conditions, in particular the impact on sampling for fish. Below we have given a brief overview of ongoing investigations and recent progress on this matter.

7.3.1 Sampling of fish in the ice-covered CAO (EU project EFICA)

Addressing the urgent need to assess fish stocks of the CAO in the framework of the Agreement to prevent unregulated high seas fisheries in the CAO, the EU-funded European Fish Inventory of the Central Arctic Ocean (EFICA) Consortium aims to map the fish stocks present in the CAO for the first time. Sampling fish with trawled nets is impossible as long as the CAO is covered by perennial sea ice. Therefore, new methods are being applied to maximize the information on fish without using trawled nets. To this end, EFICA is participating in the ongoing MOSAiC expedition on the RV *Polarstern* and planned to sample on the SAS expedition with RV *Oden* that was originally scheduled for August-September 2020 but which, due to the COVID-19 crisis, now will take place in 2021. On MOSAiC, EFICA uses a combination of hydroacoustics, an *in situ* deep-sea camera system, longlines, gillnets, fishing rods and eDNA sampling. These approaches will be extended with large ring nets that can sample the water column vertically during the SAS-*Oden* 2021 Expedition.

7.3.2 The SUIT in the Arctic Ocean – Recent sampling of polar cod under sea ice in the Chukchi-Beaufort region

The ice-water interface constitutes a unique habitat of the CAO, providing the only surface substrate in the euphotic zone. At this interface, sympagic and cryo-pelagic organisms interact, and carbon is transferred from internal sea-ice biota to the pelagic food web. This important habitat, however, is inaccessible with conventional sampling gear. Based on small-scale sampling with divers and dip nets it became evident that the ice-water interface can host significant abundances of under-ice fauna and constitutes an important nursery habitat for young polar cod Boreogadus saida (Hop et al. 2000, Gradinger and Bluhm 2004, Melnikov and Chernova 2013). During the past eight years, a novel sampling device was used to study the large-scale distribution of underice invertebrates and polar cod in the Eurasian Basin of the CAO, the Surface and Under-Ice Trawl (SUIT, van Franeker et al. 2009, Flores et al. 2012). Studies using SUITs equipped with environmental sensor arrays demonstrated that the under-ice community structure varies with changes in sea-ice properties and oceanographic conditions (David et al. 2015, Flores et al. 2019, Ehrlich et al. 2020), and investigated the spatial variability of physical properties, ice-algae biomass and primary production in ice-covered waters (Lange et al. 2016, 2017, Massicotte et al. 2019, Castellani et al. in press). In particular, SUIT sampling enabled the first basin-scale study of the distribution of polar cod in the ice-water interface layer and highlighted the potential of sea ice as a vector of trans-Arctic migrations in the Eurasian Basin (David et al. 2016). Further SUIT-based studies confirmed a high dependency of young polar cod and their prey on ice algaeproduced carbon (Kohlbach et al. 2016, 2017) and investigated microplastic ingestion of underice polar cod (Kühn et al. 2018).

7.3.3 The SUIT in the Arctic Ocean – Results from the Go-West project

Based on their observations in the Eurasian part of the CAO, David et al. (2016) had proposed that young polar cod associate with sea ice on the Siberian shelf during autumn and then get advected across the CAO during their first and second year of life. Building on these investigations, the University of Alaska Fairbanks and AWI speer-headed a first field survey with SUIT in the American Arctic in November/December 2019. This survey was realized within the project "Go-West", using the US vessel RV Siquliak. The overall goal of Go-West was to test the hypothesis that entrainment of young polar cod into the under-ice habitat during autumn is significant also in the Chukchi and Beaufort Seas and adjacent CAO. The specific objectives of Go-West

were to: 1) Identify sea ice habitats favourable for polar cod in terms of physical properties, ice algal biomass and prey composition; 2) test for the presence of deep scattering layers in the underlying water column; 3) quantify the carbon flux between sea ice, plankton, and polar cod and assess the pre-winter condition of polar cod; and 4) study the connectivity between central Arctic under-ice populations and shelf-based spawning populations.

Go-West was in large parts funded by the Arctic Research Icebreaker Consortium (ARICE). The project represented 7 institutes from 5 nations, and the team consisted of 11 members from 6 countries. Go-West was conducted during RV Sikuliaq expedition SKQ201923S (06 November – 02 December 2019), together with the physical-oceanographic project Coastal Ocean Dynamics in the Arctic (CODA). The study region covered the continental slope between the shelf break and the Canada Basin between the Chukchi Sea and the Beaufort Sea off Alaska's North Slope (Figure 7.7). Go-West sampled Polar cod and its prey in the ice-water interface layer with a Surface and Under-Ice Trawl (SUIT), along with high resolution profiles of sea-ice and surface water properties. In addition, backscatter of fish and zooplankton in the water column was recorded with an EK80 echosounder, and pelagic communities were sampled with midwater trawls and vertical zooplankton nets. A CTD probe equipped with a rosette water sampler sampled vertical profiles of water column properties and was used to collect water samples for biogeochemical parameters. During four ice stations, sea ice was sampled for biogeochemical and optical properties.



Figure 7.7. Working area and track chart of R/V Sikuliaq, Cruise SKQ201923S with fishing stations and ship CTD station numbers indicated (from: Flores, Mueter et al., 2020).

Altogether, Go-West completed 11 SUIT stations (1 in open water, 10 under ice), 4 ice stations and 3 midwater trawls. On the continental shelf, under-ice macrofauna abundance was low. The highest densities of under-ice macrofauna were found at the deep-water stations, including several neritic taxa. Polar cod were caught at all SUIT stations (Figure 7.7). Most fish appeared to be first-year juveniles between 6 and 8 cm in size, pending age determinations. The trophic relationships between ice algae, zooplankton and sea-ice fauna and Polar cod will be analyzed in detail based on diet and trophic biomarker samples. Investigations of otolith microchemistry and population genetics studies on each sampled fish will help elucidate their origins and migration patterns.

The preliminary results of this expedition support the hypothesis that juvenile Polar cod associate with sea ice in autumn and show that prey is available to sustain them at the onset of winter. In conclusion, SUIT can be a useful tool to study polar cod and ice-associated fauna at large spatial scales, and to link spatial distribution with sea-ice properties and life cycle studies.

7.3.4 A new, non-invasive tool, environmental DNA

More detailed information on fish communities in the CAO is in dire need for ecosystem assessment and to support management. Despite the number of current and planned campaigns in the CAO, usual field protocols do not always include typical fish sampling elements using large nets and trawls. Non-invasive techniques that can be employed without affecting the vessel's planned trajectory can offer a great deal of opportunities to collect information on the fish fauna. These would include hydroacoustics and environmental DNA.

Environmental DNA (eDNA) means DNA genetic substances contained in a state of free-floating DNA, fragments of cells, feces, saliva, urine and skin cells (Tsuji et al. 2019). The constant loss and shedding of genetic material from macrorganisms impart a molecular footprint in environmental samples that can be analyzed to determine either the presence of specific target species or to characterize biodiversity without the need for capture or visual surveys. Target organisms can also be detected at any life stage, including in the egg, larval, and juvenile forms from eDNA samples. Therefore, emerging eDNA technologies are non-invasive, sensitive, and cost-efficient compared to traditional survey approaches (Rees et al. 2014). Environmental DNA approaches are used to detect specific fish species and determine fish community diversity (Thomsen et al. 2012, Shu et al. 2020). Although the technique has become widely available only recently, there are now numerous accounts of its application. Further utility of eDNA for fisheries management is the potential ability of eDNA quantities to reflect fish biomass in the ocean. Positive relationships between eDNA quantities and fish biomass and abundance are now being reported, for example from Atlantic cod, Gadus morhua (Salter et al 201). Application in the very central Arctic has yet to be made but some related sub-Arctic water fauna has been subject to such attempts, especially on the identical species with some success. Thomsen et al (2016) reported eDNA metabarcoding of seawater samples from continental slope depths in Southwest Greenland and compared seawater eDNA to catch data from trawling. Environmental DNA sequence reads from the fish assemblages correlated with biomass and abundance data obtained from trawling. Environmental DNA reads can be used as a qualitative and quantitative proxy for marine fish assemblages and this relates to monitoring effects of ongoing climate change on marine biodiversity in Arctic waters.

It is encouraging for such application that there is a fair amount of information regarding the genetics of some key species, for example Arctogadus glacialis and Boreogadus saida. Breines et al. (2008) determined the complete mitochondrial genome sequences of both species. Nelson et al. (2013) reported sets of 19 and 16 microsatellite loci for the examination of the population genetics of B. saida and A. glacialis, respectively. Six of these loci were developed from a collection of 9,497 expressed sequences from B. saida while the remaining loci were found in the literature and optimized for use in B. saida and A. glacialis. Later on, Nelson et al. (2020) provided good knowledge of B. saida population genetic structure that can provide insight into how the species may respond to environmental change, and allow for establishment of meaningful management units by analysing nine microsatellite DNA loci in 2269 fish collected at 19 locations across the species' range. The population genetic structure detected is likely to be important for the response of B. saida to environmental change and should be considered in management of human activities that may impact this foundational species.

For future surveys in the CAO, two non-invasive tools could be used in combination, such as hydroacoustics and eDNA, to record the occurrence of fish species and to estimate their abundance. Hydroacoustics can discriminate the classification of echoes from fish, while eDNA might be used to validate the acoustics data and to improve species identification. Berger et al. (2020) used eDNA to groundtruth the hydroacoustics they employed to detect pelagic fish in a whale feeding area in Canadian waters. The eDNA analyses helped determine the dominant species and the variability in their abundance. This kind of study will contribute toward collection of the

badly wanted data for science-based conservation in the CAO and in particular, will provide extra capacity for the field programs with no direct fishing plan and enable full use of the vessels' path and ship time.

7.4 Priority research and monitoring needs per ToR

Priority research, monitoring needs and recommendations for each of the expert groups ToR bd are given below.

7.4.1 (ToR b): On-going and recent changes and events in the CAO

- Priority research is needed in ice edge areas (marginal ice zones) because the ice edge is lacking data, but is expected to have unique conditions of atmosphere and ocean interaction, resulting in potential biological hotspot locations.
- Long-term monitoring is needed to detect changes in ocean properties and sea ice in the Arctic gateways, as changes in the gateways could be used as initial conditions for assessment of changes in the CAO.
- Monitoring of the Arctic slopes is necessary as these are the regions that will experience the largest changes in the near future.
- Winter observations are of critical importance since the number of such observations is very small, while detailed understanding of winter processes is essential for correct interpretation of the entire pattern of ongoing changes.

7.4.2 (ToR c): Continue to examine effects of climate change on the ecosystem and assess projected future changes

7.4.2.1 Phytoplankton and algae

- For the CAO, there is a lack of regular monitoring of phytoplankton and algae (CAFF 2017).
- It is important to have comparable, supporting data regarding the physical and chemical environment for evaluating this system.
- There is a general lack of baseline data in the CAO, especially throughout the year, and it may therefore be challenging to distinguish between natural variations and changes of primary producer communities due to anthropogenic modification (von Quillfeldt and Smith 2016).
- Monitoring should be standardized regarding gear, collections, timing, sample preservation and processing, storage and data management.
- Available monitoring protocols need to be more widely available for standardized monitoring parameters.
- Species information is required for monitoring biotic community changes that are driven by climate. However, comparable data from different laboratories requires taxonomic expertise, which is becoming rare. An alternative or complement to microscopy identification is the use of marker genes to identify species, although this approach requires infrastructure and access to sequencing facilities.
- Understanding how nutrient limitation enhances heterotrophy in ecosystems and quantifying the potential impact this limitation has on ecosystem function is fundamental for the development of future scenarios of the changing Arctic ecosystems (Flores et al. 2019)

7.4.2.2 Sea-ice biota

- Monitoring should be conducted with consistent methodology after coordinated plans (gear, collections, timing, sample preservation and processing, storage, and data management) for sampling at set locations, with consistent sampling and analyses to ensure comparability (Gill et al. 2011).
- Available protocols need to be more widely implemented for monitoring.
- A central repository for data as well as long-term funding for monitoring should be considered.
- Data should be deposited in existing databases and made available to researchers and beyond.
- Proper monitoring requires seasonal and annual field campaigns as well as sufficient, long-term financial support.
- As minimum, the bottom 10 cm of ice cores (which can be split into the lowermost 3-4 cm bottom ice and the next 6-7 cm of the core) should be sampled, because the bottom 10 cm often includes the majority of the biomass and abundance of ice biota (~65 % of ice meiofauna, up to 95% for algae based on the here compiled data sets).
- It is recommended that the 10-30 cm section also to be included (separately) where feasible, as an average 12% of the meiofauna have been observed in this ice section.
- Sterile procedures should be used to avoid contaminating the ice cores (e.g., Collins et al. 2010).
- The need for sampling the entire ice column will depend on project goals, which may also include modelling aspects of biota in sea ice (Duarte et al. 2015).
- Ice biota has rarely been studied as a whole at a given location and time period. *Gammarus wilkitzkii* and *Eusirus holmi*, both amphipod species, should therefore be monitored, since changes in their relative abundance may reflect changes in sea ice conditions.
- The most quantitative samples appear to be based on SCUBA-operated sampling of squares with electrical suction pump (Hop et al. 2000) and the under-ice trawl net SUIT (van Franeker et al. 2009), as well as imagery with sufficient resolution.
- Variability in contemporary and historic methods of sampling as well as temporal variability of sampling obscure potential temporal trends. Thus, actual changes in ice biota diversity and community structure need to be substantial enough to rise beyond the existing variability in order to be detected.

7.4.2.3 Zooplankton

Ship-borne, ocean-going field programs in the CAO are growing, yet regular monitoring program are not well-established and still sporadic. Information toward the present status and potential future prospects is insufficient and scarce. Coordinated field efforts and standard protocols are essential and warranted. Below are examples of research needs:

- Consolidated information on the distribution and abundance of major zooplankton species in the Central Arctic Ocean proper in different years and in different seasons (non-summer seasons in particular) is needed.
- Community change governed by the exchange of water masses from the Atlantic and the Pacific, and the nature of the population interactions between sub-Arctic and the Central Arctic, as well as between shelf and basins need to be studied.
- Differential life history response of major copepod species with regards to changing primary and secondary productivity, which is determined by the magnitude of melt (the position of sea ice edge, and sea ice thickness) and the timing of ice break-up, need investigations in different sectors of the CAO.
- Modelling efforts are needed to predict the degree and nature of borealization aided by new observational data, especially for the key biological input parameters.

- Laboratory and field-based research to evaluate zooplankton adaptation to future human activity such as shipping, pollution and energy exploration, should be developed. Topics of plastic ingestion, oil spill impacts, and ocean acidification are important for future studies.
- Applications of new tools and methods in the CAO should include, but not be limited to, environmental DNA, automated image collection and processing.

7.4.2.4 Marine mammals

- *Continued and expanded monitoring of marine mammal occurrence in the CAO:* Monitoring of geographic and seasonal occurrence patterns is of fundamental importance for the understanding of highly mobile species in a changing environment. Historically, monitoring of marine mammals in the CAO has mainly relied on opportunistically collected data from ice breakers, aerial ice monitoring programs and various research and tourist expeditions (e.g. Belikov and Boltunov 2002). Collection, collation and analyses of this type of data is highly valuable and should continue and expand with the increase of human activities in the region. Efforts should be made to optimize the quality, synthesis and dissemination of opportunistic observations. In addition, passive acoustic monitoring stations within the CAO would valuable tools for the majority of the arctic marine mammals. Based on prior indications and judgement of ecological habitat suitabilities, reconnaissance surveys should be carried out in selected areas to evaluate the need and feasibility of specialized abundance surveys.
- *Continued and expanded dedicated abundance surveys in the CAO:* Continued and geographically expanded dedicated abundance surveys for ice dependent species like narwhals, polar bears, seals and bowheads around the CAO ice edge are needed for responsible management of human activities. In particular, a more complete survey of narwhals in the Nansen basin should be a near future priority as this population component has shown high affinity to conditions prevailing in the CAO. It is furthermore likely to constitute a significant proportion of the total abundance of a distinct grouping of narwhals to the East of Greenland.
- Telemetry based studies of marine mammal movements in the CAO: Satellite telemetry, preferably with oceanographic tags, provide highly valuable information on movement patterns, habitat selection, diving behavior, potential diets and population affiliation. Currently available telemetry data for marine mammals in the CAO are generally based on animals originally tagged outside the area, which may be biased towards short term visitors. Tagging animals already within the CAO would be more likely to reveal information on potential long-term residents of the CAO. Elucidating the population affiliation of Nansen basin narwhals is of particularly high importance as these animals are likely to represent a significant proportion of the total abundance of narwhals to the East of Greenland.
- *Genetic studies:* Genetic assignment of population affiliation of marine mammals observed in the CAO is a high priority for all species. In addition, genetic samples may provide information on sex and some physiological parameters through quantitative analyses of gene expression (Sigsgaard et al. 2019). High quality DNA from individual biopsies provide the most straightforward technical basis for these analyses, but faecal dna or environmental DNA from targeted areas may provide similar information with extended protocols (Sigsgaard et al. 2019). In addition, eDNA may provide community data on suites of species.
- Studies of life history, health and ecology based on biological material: For marine mammals observed in the CAO access to biological material from live animals for assessment of health, age and reproductive status are generally limited to biopsies, faeces, cells in exhaled air, and occasional live sampling of blood samples, epithelial cells, hair and

whiskers. These samples can provide useful data on reproductive status (sex hormones), health (stress hormones, contaminants, pathogens and antibodies), trophic information (stable isotope and fatty acid analyses). With the exception of data collected from live-captured animals for tagging, many of these techniques are, however, highly opportunistic and time consuming and unlikely to be widely operational for realistic levels of field time in the CAO. For abundant species like ringed seals, lethal sampling may be defendable for more detailed analyses of health and reproductive status, contaminant levels, age and diet. Similar data derived from indigenous and commercial hunting have previously proven highly useful for arctic populations (e.g. Nguyen et al. 2017, Beatty et al. 2019). The relevance of available hunted samples for CAO related studies, should be assessed.

- Analyses of soundscapes relevant for marine mammals within the CAO: In addition to data on occurrence patterns of vocal marine mammals, general analyses of CAO soundscapes are important for the overall understanding of marine mammal acoustic ecology in the area. This is because levels of both anthropogenic noise and sounds from natural sources like wind and wave action are likely to change in response to changes in ice cover (Stafford et al. 2018). Soundscape data can be obtained from passive acoustic monitoring devices. Baseline studies for the pacific and arctic gateways are available (Moore et al. 2012, Ahonen et al. 2017, Stafford et al. 2018).
- Studies of arctic mammal sensitivity to low frequency anthropogenic noise: Although several studies are available on the sensitivity of bowhead and beluga whales to seismic airguns and vessel noise, there are still considerable gaps in the knowledge base required for vulnerability assessments of marine mammals in the CAO. Very few studies of behavioural responses to noise exist for other arctic marine mammals and no studies appear to exist on effects of military sonars on arctic marine mammals (Gomez et al. 2016). With expected increases in industrial and military activity in the CAO and other parts of the Arctic, these gaps need to be filled.
- Since marine mammals in the CAO are most likely closely connected with larger populations outside the core area of the CAO, continued monitoring of these more central populations are important in order to put the information from the CAO into a population and species-wide context. Generally agreed circumpolar research and monitoring recommendations for arctic mammals are summarized in CAFF (2017).

7.4.2.5 Benthos

- There is a clear need for a better understanding of the quantity, phenology, and the regionality of carbon supply and demand in the deep Arctic basins, which will allow us to evaluate how the ecosystem may change in the future (Wiedmann et al. 2020).
- The high heterogeneity of the CAO sea-floor have implications for sampling, monitoring and potential response of use of the deep sea in the Arctic such as: (1) larger and more stratified sampling strategies are required to cover different benthic habitat types present than previously appreciated, (2) advanced autonomous technology would be required to capture short-term events and small-scale areas of high densities/biomass/flux, (3) recovery after disturbance of the seafloor is likely extremely slow, and (4) our poor understanding of biotic communities makes is challenging at best to distinguish any potential response to change from new findings filling an earlier knowledge gap.
- Further temperature monitoring is needed to evaluate the recent warming trend in the CAO and its impact on carbon export to the deep sea.

7.4.2.6 Fish

- Consolidated information is needed on the distribution and abundance of major fish species (*Boreogadus* and *Arctogadus* in particular) in the CAO in different years, and across different habitats (water column and sea ice associated areas).
- Evaluating the diversity of fish species across different habitats (upper water column, mid-water, demersal, sea ice associated) should occur.
- The life history response of by polar cod species with regards to changing biological productivity and abundance of prey organisms, which is determined by the nature of sea ice melt, should be studied, especially in a pan Arctic perspective.
- Evaluation of the pan-Arctic population structure of polar cod species and the nature of the interaction of populations between low latitude Arctic and the high latitude Arctic, as well as between shelf and basin, is needed.
- Studies of the degree of borealization of fish communities as determined by field data and specimens is needed, along with developing predictions by modelling efforts.
- Investigations into the damage to and adaptation of major fish populations (especially to the early life history stage) by future human activity through both laboratory and field-based research. Such topics as plastic ingestion, oil spill impacts, ocean acidification impacts, and disturbance to the sound scape need to be studied.
- Evaluating the interactions of human activities with apex predators and predicting the changing availability of prey to the top of the food web is needed.
- Applications of new tools and methods in the CAO to include, but not be limited to, more sophisticated hydroacoustics, environmental DNA, automated image collection and processing.

7.4.3 (ToR d): Assess the consequences of climatic and oceanographic changes on transport pathways and potential effects of contaminants

- There is a need to standardize and harmonize methods to quantify microplastics in abiotic and biotic matrices in order to establish monitoring data series in the CAO.
- There should be monitoring of contaminants (including litter and microplastics) in all parts of the CAO (Note: this was also identified as current Arctic research priorities in the '2020: State of the Arctic science' report of the International Arctic Science Committee).
- Monitoring long-distance pollution transport in the Arctic (metals, NOx, SOx, organic pollutants, etc.) is needed. (Note: this was also identified as current Arctic research priorities in the '2020: State of the Arctic science' report of the International Arctic Science Committee).
- Monitoring human activities, such as fisheries, shipping, cruise, scientific exploration (of which physical contact or operation is expected), both within and in the surrounding areas of the CAO, should be implemented.

7.4.4 (ToR e): Review and report on new studies on fish as well as other biological components

• The largest knowledge gap concerns the presence and size of fish stocks inside the CAO. New research programs are needed to remedy this gap, and they should include the development of methodologies to sample fish in an ice-covered sea. Data collected should focus on spatial and seasonal distributions of fish in the CAO, and potential migration L

L

patterns of fish stocks in the CAO, and use advanced technologies, such as ice-moored autonomous hydroacoustic buoys and eDNA samplers.

- The significance of the under-ice habitat for the key forage fishes polar cod *Boreogadus saida* and ice cod *Arctogadus glacialis* is not fully understood. A detailed understanding of the role of sea ice for the survival and distribution of these species is urgently needed to predict the impact of climate change on these species and their ecosystems, and to develop sustainable management schemes.
- Assessments of the current and future carrying capacity of the CAO for fish populations must combine detailed knowledge on the changing habitats, distribution of potential prey stocks, predators, and competitors in relation to the physiological envelope in which Arctic fishes thrive.
- Potential range expansions of commercial fish stocks into the CAO must be assessed in combination with economic expansion potential of the fishing industry and emerging management schemes.
- New management tools mediating between local communities, industry and societies must be developed that can safeguard sustainable development of ecosystems and human stakeholders in the face of a rapidly changing environment.
- Model studies must be better constrained by the factors mentioned above.

7.5 Summary and Conclusions

Physics, contaminants, primary and secondary producers and sea ice biota dominated the identified key points for monitoring in the CAO. Important issues included standardization of protocols/sampling and obtaining data at times when data historically are rare (i.e. during winter).

For benthos and fishes, mapping and baseline studies were identified as priorities, including the development/use of new technology. Regarding fish and possible future fisheries in the region, priority research includes the development of new management tools mediating between local communities, industry and societies that can safeguard sustainable development of ecosystems and human stakeholders in the face of a rapidly changing environment.

Regarding marine mammals, monitoring and mapping of distribution and abundance were identified as key points, but also studies on life history, health and ecology were highlighted. In addition, studies directly related to increased human activity, like analyses of soundscapes relevant for marine mammals, and studies of arctic mammal sensitivity to low frequency anthropogenic noise, were identified as priority research.

A synthesis of the state of knowledge about biodiversity in Arctic marine ecosystems, detectable changes, and important gaps in our ability to assess state and trends in biodiversity across sea ice biota, plankton, benthos, marine fishes, seabirds and marine mammals were conducted in the SAMBR report (CAFF, 2017), and the key priority research, monitoring needs and recommendations identified in section 7.4 fit well with those. The monitoring needs and recommendations for fish (7.4) are consistent with the outcome from the FISCAO process (PSCG 2020), i.e. the high-lighted concern about the current knowledge gap on the presence and size of fish stocks inside the CAO as well as the necessity of development of methodologies to sample fish in an ice-covered sea.

A coordinated array of regional transect lines are recommended to be developed with key parameters to evaluate the regional to pan-Arctic scales, something the international DBO network and SAS are working towards. With more coordinated observations, we can then identify focal areas for monitoring on an interannual and decadal basis, knowing the limitations of required funding and the need for international agreements. The 5th FISCAO report developed monitoring recommendations by leveraging ongoing observing activities, which is only as viable as long

as those time series of observations are maintained. There is also a need for an international, coordinated effort (involving among others CAFF and its Circumpolar Seabird Expert Group (Cbird); the IUCN Polar Bear Specialist Group, and other relevant working groups) for these investigations to eventually develop monitoring capacity at various trophic levels, with associated environmental drivers.

References

Additional references can be found in the subchapters from ToR b-e.

- CAFF. 2017. State of the Arctic Marine Biodiversity Report. Conservation of Arctic Flora and Fauna International Secretariat, Akureyri, Iceland. 978-9935-431-63-9.
- FISCAO 2016. Final Report of the Fourth Meeting of Scientific Experts on Fish Stocks in the Central Arctic Ocean. 26-28 September, Tromsø, Norway (<u>https://www.fisheries.noaa.gov/event/fourth-meeting-scientific-experts-fish-stocks-central-arctic-ocean</u>).
- FISCAO 2018. Final Report of the Fifth Meeting of Scientific Experts on Fish Stocks in the Central Arctic Ocean. 24-26 October, Ottawa, Canada (<u>https://archive.afsc.noaa.gov/Arc-tic_fish_stocks_fifth_meeting/508_Documents/508_Final_report_of_the_5th_FiSCAO_meet-ing.pdf</u>)
- Grebmeier JM, Moore SE, Cooper LW, and Frey KE, 2019. The Distributed Biological Observatory: A change detection array in the Pacific Arctic - An Introduction. Deep-Sea Research Part II 162, 1–7; <u>https://doi.org/10.1016/j.dsr2.2019.05.005</u>.
- Moore SE, and Grebmeier JM, 2018. The Distributed Biological Observatory: Linking Physics to Biology in the Pacific Arctic Region. Arctic 71, Suppl. 1: 1-7. <u>http://doi.org/10.14430/arctic4606.</u>
- Paasche Ø, Olsen A, Årthun M, Anderson LG, Wängberg S-Å, Ashjian CJ, Grebmeier JM, Kikuchi T, Nishino S, Yasunaka S, Kang S-H, Cho K-Ho, Azetsu-Scott K, Williams WJ, Carmack E, Torres-Valdés S, Tyrrell T, Edelvang K, He J, and Kassens H, 2019. Addressing Arctic challenges requires a synoptic ocean survey, Eos, 100, https://doi.org/10.1029/2019EO136200. Published on 01 November 2019.
- PSCG 2020. Report of the 1st meeting of the Provisional Scientific Coordinating Group (PSCG) of the Agreement to Prevent Unregulated High Seas Fisheries in the Central Arctic Ocean. 11-13 February 2020, Ispra, Italy (<u>https://www.fisheries.noaa.gov/event/fifth-meeting-scientific-experts-fish-stocks-central-arctic-ocean</u>)
- Wiedmann T, Lenzen M, Keyßer LT et al., 2020. Scientists' warning on affluence. Nat Commun **11**, 3107 (2020). <u>https://doi.org/10.1038/s41467-020-16941-v</u>.

8 ToR g. Prepare an Ecosystem Overview for the CAO ecosystem

Contributors: Martine van den Heuvel-Greve, Inigo Martinez, Lis Lindal Jørgensen

Working group: all WGICA members

8.1 Introduction

An Ecosystem Overview (EO) provides a description of an ecosystem, identifies the main human pressures, and explains how these affect key ecosystem components. The audience for EOs includes client commissions and stakeholders as well as the ICES community and networks. The EO for the CAO will be completed by an advice drafting group and approved by the ICES Advisory Committee, ACOM.

The purpose of the EO is to:

- a) describe the location, scale, and management and assessment boundaries of the ecoregion;
- b) describe the distribution of human activities and resultant pressures (in space and time) on the environment and ecosystem;
- c) describe the state of the ecosystem (in space and time) and comment on pressures accounting for changes in state.

The EO consists of the following parts:

- 1. **Ecoregion description** Maps and text showing the boundaries of the ecosystem and depth contours, relevant management and assessment regions, human usage, catchment areas, and designated areas (i.e. Natura 2000 areas). Appropriate subregions should be described.
- 2. **Key signals within the environment and ecosystem** This focuses on changes on time and space scales that have consequences for advice. Major trends and shifts that are required to provide advice should be described.
- 3. **Activities and pressures** Identify regional priorities, listing the predominant pressures in the ecoregion, with an indicative list of activities. The goal is to assess cumulative pressures from multiple activities.
- 4. **State** Short concise descriptions of the main state of the ecosystem components within each ecoregion, linking the selected pressures to the state of the ecosystem.

WGICA will be preparing an EO for the CAO in 2020-2021. The focus in 2020 will be on conducting the activities and pressures assessment and developing draft text for the EO. In 2020, the EO will focus on the current situation where ice-cover is largely restricting human activities that occur inside the CAO. Most "current" pressures are therefore of a "global nature" and involve long distance transport of contaminants, litter and new species.

In 2021 WGICA will explore methods to merge both current and future pressures into the conceptual model of the EO. This "future" aspect is added because rapid changes are occurring in the CAO and the "precautionary principal" could be applied by evaluating the consequences of pressures on the ecosystem as early as possible. The CAO is defined as the High Seas ice-cover, water surface, and water column in addition to the seabed under national jurisdictions in the deep basins, including the slopes but excluding the bordering shelf seas.

8.2 Pressure assessment

The first step in the selection of the main pressures involved the assessment of all pressures currently affecting the CAO. This assessment was conducted according to the ICES guidelines.

The following 16 WGICA experts were involved in this assessment: Henry Huntington, Natsuhiko Otsuka, Kathy Kuletz, Thomas Van Pelt, Lis Lindal Jørgensen, Karen Edelvang, Bodil Bluhm, Jackie Grebmeier, Fujio Ohnishi, Harald Gjøsæter, Hauke Flores, Pauline Snoeijs-Leijonmalm, Barbara Niehoff, Hyoung Chul Shin, Martine van den Heuvel-Greve, Bjørn Einar Grøsvik.

The following five main pressures, for the current situation, were identified for the CAO (see also figure 8.1):

- 1. Sea ice loss;
- 2. Introduction of non-indigenous species;
- 3. Introduction of contaminating compounds;
- 4. Marine litter;
- 5. Noise.



Figure 8-1. Results of the pressure assessment for the CAO for the current situation, based on scores from 16 WGICA experts. The higher the score the more important the pressure for the ecosystem. The five main pressures are presented in green, the other pressures in blue.

8.3 Linking main pressures with human activities and ecosystem components

Using the five main pressures for the CAO as a starting point, two further assessments were made to identify current human activities that contribute to these pressures and direct links that exist between these pressures and all ecosystem components. The ICES guidelines were followed for these assessments.

These assessments were conducted by the following 9 experts of WGICA: Kevin Hedges, Hauke Flores, Lis Jorgensen, Henry Huntington, Harald Gjøsæter, Martine van den Heuvel-Greve, Barbara Niehoff, Thomas van Pelt, Anders Mosbech.

L

8.3.1 Linking main pressures with human activities

The draft results for this assessment were just compiled. Further discussion on the outcome will follow later this year. The draft results identified a few main human activities that are linked to the current main pressures for the CAO (Table 8.1).

Table 8.1. Current human activities that are linked to the current main pressures in the CAO, based on scores of 9 WGICA experts. Scores ranged between 0-3; the higher the score the more important the link.

Pressure	Score≥2.5	Score ≥ 2.0	Score ≥ 1.5
Sea-ice		Global sources	
Non-indigenous spe- cies*		Shipping	Tourism & recreation, research
Contaminating com- pounds		Shipping, global sources	Military, tourism & recreation
Marine litter			Tourism & recreation, shipping, global sources, military
Noise	Military, ship- ping		Tourism & recreation

*Global sources may be added as a contributing human activity to this pressure

8.3.2 Linking main pressures with ecosystem components

For this assessment the draft results were just compiled. The outcomes will be discussed further later this year. The draft results identified the following links between ecosystem components and the five main pressures for the CAO (Table 8.2).

Table 8.2. Direct links between ecosystem components and the current main pressures in the CAO, based on scores of 9 WGICA experts. Scores ranged between 0-3; the higher the score the more important the link.

Ecosystem compo- nent	Score = 3.0	Score ≥ 2.5	Score ≥ 2.0	Score ≥ 1.5
Habitats	Sea-ice loss			
Food webs		Sea-ice loss	Non-indigenous species	Contaminating compounds
Productivity (algae)	Sea-ice loss			Non-indigenous species
Zooplankton		Sea-ice loss		Non-indigenous species, contaminat- ing compounds
Benthos				Sea-ice loss, non-indigenous species
Fish			Sea-ice loss, non-indigenous species	Contaminating compounds
Sea birds		Sea-ice loss	Contaminating compounds	Marine litter
Marine mammals		Sea-ice loss	Contaminating compounds, noise	Non-indigenous species, marine lit- ter

9 Outline of Report 2

Human activities, climate- and vulnerability assessment of the Central Arctic Ocean (CAO)

It was discussed that Report 2, as a natural continuation of Report 1, could describe:

<u>Part 1</u>

Human activities and related management bodies:

- ongoing, as well as future, human activities (global impact as well as local) and management bodies responsible for managing these Human activities.
- scenarios of a future where the ice cover is reduced, allowing a variety of different human activities, and how this will evolve under different management regimes.

Part 2

Climate assessment:

• the ongoing and future climate impact on the physical, chemical and biological CAO

Vulnerability assessment:

- the ongoing and future pressures/impacts from human activities and their effects on all the ecosystem components, including the fauna associated with the sea ice, the water column, and the seabed:
 - Sensitive and vulnerable areas
 - Fishable and harvestable concentrations of vertebrates and invertebrates and the potential production
 - Effects of pollution (including plastics), contaminants, noise and other
 - Risk of introduction and spread of invasive species
 - Future risk in case of exploration (seabed mining) and other activities

Originally the Report 2 (part 1 and 2) was given a deadline delivery of November 2021. The group considers that to be not enough time, since the report has not been started.
We suggest a revised timeline, as follows:

- Fall 2020: Finalize the Ecosystem Overview (EO) for the November 2020 delivery
- Spring summer 2021: Write Part 1 (Human activities and management bodies) by following the structure of the EO
- Fall 2021: to have a physical meeting in ICES HQ to discuss the way forward to finalize Part 1, and to draft the ToRs for 2022-2024 that will support the writing of Report 2 part 2.

2022-2024: write part 2 (Climate Assessment and Vulnerability Assessment).

Annex 1: List of participants

[Text]

Name	Institute	Coun-	Email	Function
		try		
Lis Lindal Jørgensen	Institute of Marine Research Tromsø	Norway	lis.lindal.joergensen@hi .no	Chair (ICES, PAME)
Sei-ichi Saitoh	Hokkaido University	Japan	ssaitoh@salmon.fish.ho kudai.ac.jp	Chair (PICES)
Alf Håkon Hoel	Institute of Marine Research Tromsø	Norway	alf.haakon.hoel@hi.no	Invited
Anders Mosbech	Department of Arctic Environment	Denmark	amo@bios.au.dk	Invited
Fugio Ohnishi	Arctic Research Center	Japan	fujio.ohnishi@arc.hoku dai.ac.jp	Invited
Grøsvik, Bjørn Einar	Institute of Marine Reseach	Norway	<u><bjorn.grosvik@hi.no></bjorn.grosvik@hi.no></u>	Invited
Hauke Flores	Alfred-Wegener-Inst. Foun. for Polar and Mar. Research	Germany	Hauke.Flores@awi.de	Invited
Hein Rune Skjoldal	Institute of Marine Research Tromsø	Norway	<u>hein.rune.skjoldal@hi.n</u> <u>0</u>	Invited
lgor Melnikov	P.P. Shirshov Institute of Oceanology	Russia	migor39@yandex.ru	Invited
Jacqueline Grebmeier	Center for Environmental Science	USA	jgrebmei@umces.edu	Invited
Jan René Larsen	Arctic Monitoring and Assessment Pro- gramme	Norway	jan.rene.larsen@amap. no	Invited
Kathy Kuletz	US Fish and Wildlife Service - Alaska Re- gion	USA	Kathy_Kuletz@fws.gov	Invited
Kevin Hedges	DFO Winnipeg	Canada	Kevin.Hedges@dfo- mpo.gc.ca	Invited
Lisa Speer	Natural Resources Defense Council	USA	lspeer@nrdc.org	Invited
Maria Gavrilo		Russia	m_gavrilo@mail.ru	Invited
Naomi Harada	Research Institute for Global Change	Japan	haradan@jamstec.go.jp	Invited
Natsuhiko Otsuka	Arctic Research Center	Japan	natsuhiko.otsuka@arc.h okudai.ac.jp	Invited
Pauline Snoejis Leonmalm	Stockholm University	Sweden	pauline.snoeijs- leijonmalm@su.se	Invited
Senjie Yang	Hokkaido University	Japan	76008563@qu.com	Invited
Shigeto Nishino	Research Institute for Global Change	Japan	nishinos@jamstec.go.jp	Invited
Taka Hirata	Hokkaido University	Japan	tahi@arc.hokudai.ac.jp	Invited

Thomas Van Pelt	University of Washington	USA	<u>tvanpelt@transboundar</u> <u>y.net</u>	Invited
Tom Christensen	Department of Arctic Environment	Denmark	toch@bios.au.dk	Invited
Toru Hirawake	Hokkaido University	Japan	Hirawake@salmon.fish. hokudai.ac.jp	Invited
Vladimir Ivanov	Arctic and Antarctic Research Institute	Russia	<u>vladimir.ivanov@aari.r</u> <u>u</u>	Invited
Yasushi Fukamachi	Hokkaido University	Japan	yasuf@arc.hokudai.ac.j p	Invited
Anne Kirstine Frie	Institute of Marine Research Tromsø	Norway	anne.kirstine@hi.no	Member
Bodil Bluhm	Department of Arctic Biology	Norway	bodil.bluhm@uit.no	Member
Cecilie von Quillfeldt	Norwegian Polar Institute	Norway	cecilie.von.quillfeldt@n polar.no	Member
Harald Gjøsæter	Institute of Marine Research	Norway	Harald.Gjoesaeter@hi.n o	Member
John Bengtson	Alaska Fisheries Science Center	USA	john.bengtson@noaa.go v	Member
Karen Edelvang	DTU Aqua, National Institute of Aquatic Resources	Denmark	kaede@aqua.dtu.dk	Member
Martine van den Heuvel-Greve	Wageningen Marine Research	Netherland s	Martine.vandenHeuvel- Greve@wur.nl	Member
Randi Ingvaldsen	Institute of Marine Research	Norway	randi.ingvaldsen@hi.no	Member
Hyoung Chul Shin	Korea Polar Research Institute	Korea	hcshin@kopri.re.kr	Member (PICES)
Barbara Niehoff	Alfred-Wegener-Inst. Foun. for Polar and Mar. Research	Germany	barbara.niehoff@awi.de	Stepping in
Ingeborg Mulder	Wageningen Marine Research	Germany	<u>ingeborg.mulder@wur.</u> nl	Stepping in

Annex 2: Resolutions

WGICA - ICES/PICES/PAME Working Group on Integrated Ecosystem Assessment (IEA) for the Central Arctic Ocean

2018/MA2/IEASG06 A Joint ICES/PICES/PAME Working Group on Integrated Ecosystem Assessment of the Central Arctic Ocean (WGICA), chaired by John Bengtson, USA, Sei-Ichi Saitoh, Japan, and Lindal Jørgensen, Norway, will work on ToRs and generate deliverables as listed in the Table below.

	MEETING DATES	VENUE	REPORTING DETAILS	Comments (change in Chair, etc.)
Year 2019	8-10 May 2019	Sapporo, Japan	ICES Scientific Report by 1 September 2019	
Year 2020	27-29 April	by correspondence	ICES Scientific Report by 1 September 2020	Hein Rune Skjoldal, Norway and John Bengtson, USA as outgoing Chairs. Lis Lindal Jørgensen, Norway as incoming Chair
Year 2021	To be decided	To be decided	Final ICES Scientific Report by 31 December 2021	Martine van den Heuvel- Greve, Netherlands, as incoming Chair

ToR descriptors

			Science Plan		Expected Deliver-
ToR	Description	Background	<u>codes</u>	Duration	ables
a	Review and consider ap- proaches and methodol- ogies for conducting an IEA of the CAO ecosys- tem including Human Activities from the view- point of Climate and Vulnerability Assess- ments.	WGICA has produced a first version IEA report for the CAO. Before producing an updated and extended version, the basic approach and methodologies should again be considered.	2.2, 6.1, 6.5	Year 1	Report outcome in the 2019 interim re- port.
b	Review and report on ongoing and recent changes and events in the CAO associated with changes in sea ice, oceanographic circula- tion, and hydrographic properties	There is a need to follow developments in the CAO resulting from the predicted further loss of sea ice and other physi- cal changes associated with global climate change.	1.1, 2.2, 6.5	Years 1-3	New information will be reported in interim reports in 2019 and 2020. A more full account will be given as part of a second version IEA report for the CAO in 2021.

c	Continue to examine ef- fects of climate change on the CAO ecosystem by compiling and re- viewing information on changes in response to the ongoing 'Great melt', and assess likely conse- quences to the CAO eco- system of projected fu- ture changes associated with further loss of sea ice and other climate-re- lated changes (i.e. a cli- mate impact assess- ment).	This activity was started in the first 3-year pe- riod, and some infor- mation is included in the 2018 IEA report. There is a need to con- tinue and carry out a more detailed assess- ment of the documented and/or inferred bologi- cal and ecological changes associated with the large physical changes that have al- ready taken place (e.g. loss of half the area and ¾ of volume of summer sea ice).	1.1, 1.3, 6.1, 6.5	Years 1-3	Progress will be reported in interim reports in 2019 and 2020. A more full account will be given as part of the new version of the IEA report for the CAO in 2021.
d	Assess the potential ef- fects on the CAO ecosys- tem of recent, ongoing and future climatic and oceanographic changes on Human activities (shipping, tourism, pos- sible future fisheries, seabed exploitation of minerals and security) and recent on-going pol- lution (contaminant, gar- bage, and micro plastics)	This is a new activity which relates to assess- ment of pollution in the CAO. Pollution can be expected to be one of the more serious threat to the CAO ecosystem and should be included in an IEA.	2.1, 2.5, 6.1	Years 2, 3	Progress will be re- ported in interim report in 2020. As- pects of pollution wil be included in the new IEA report for the CAO in 2021.
e	Review and report on new studies on fish of the CAO ecosystem (the High Seas).	The information on many parts of the CAO ecosystem is still lim- ited. New information is expected to come over the next few years as re- search ice-breakers pay more attention and use scientific ecchosounders and other observation techniques to record fish and other organ- isms in the water col- umn and at the seafloor.	5.2, 6.1, 6.5, 6.6	Years 1-3	Progress will be re- ported in interim reports in 2019 and 2020. A more full account will be given as part of the new version of the IEA report for the CAO in 2021.

e	Continue to identify pri- ority research needs and monitor how identified knowledge gaps (needed to improve IEA and management effec- tiveness) are being ad- dressed and filled.	A by-product of doing the first version IEA of the CAO is a priority list of research needs. It is necessary to monitor how knowledge gaps are filled that will im- prove new versions of IEA.	1.3, 2.2, 3.1, 6.1, 6.5	Years 2, 3	Progress will be re- ported in the in- terim report in 2020 and outcome re- ported in 2021.
f	Prepare an Ecosystem Overview for the CAO ecosystem	This will be an addition to the series of Ecosys- tem Overviews pre- pared by ICES.	6.5, 6.6	Years 2, 3	Draft version will be reported in the interim report in 2020 and final ver- sion reported in 2021.

Summary of the Work Plan

Year 1	Review IEA methodologies for IEA of the CAO. Review and report new information and
	changes in the CAO ecosystem.
Year 2	Review and report new information and changes in the CAO ecosystem. Address path- ways and effects of contaminants, make an initial list of research needs, and prepare draft
	Ecosystem Overview.
Year 3	Prepare a second version IEA eport for the CAO with information on status and trends, in-
	cluding impacts of climate change, pollution, and other relevant human pressures. Report
	on research needs and prepare final draft of Ecosystem Overview.

Supporting information

Priority	WGICA is one of several groups in ICES that do integrated ecosystem assessments, which is one of the priority action areas for ICES. Being a WG for the central Arctic Ocean, WGICA also contributes to the Arctic research action area. Jointly sponsored by PICES and the PAME working group of the Arctic Council, WGICA represents a collaborative effort that links ICES work in the wider Arctic Mediterranean Sea (the Nordic Seas and the central Arctic Ocean) with expertise on the Pacific Arctic through PICES.
	The work planned in WGICA will directly address ICES science priority area 6 De- veloping tools, knowledge and evidence of effective conservation and management and some elements of priority area 2 (Understanding ecosystems) and 3 (Impacts of human activities).
Scientific justification	ICES IEA EGs provide science based assessments of ecosystem status, trends and vulnerabilities to support implementation of the ecosystem approach to management.
	ToR a – The CAO is a data-deficient system where much of the data and knowledge comes from research activities, while monitoring is a more limited source of information. Based on the first version IEA report for the CAO, as well as experiences from the other IEA WGs in ICES, the approach and methods for IEA for the CAO will be considered prior to producing a second version IEA report in 2021.
	ToR b – The CAO is on a trajectory of reduction of sea ice with considerable inter- annual variablity. Trends and events will be reported to draw attention to the ongo- ing changes in the CAO.
	ToR c – The purpose and aim of this item is to provide a careful evaluation and summary of what we can say about the biological and ecological effects of climate change over the recent decades up to present. This can in turn be used for projec- tions of likely effects of continued warming and loss of sea ice over next decades.

Linkages to other or- ganizations	This is a joint ICES, PICES, and PAME WG.
Linkages to other com- mittees or groups	Within ICES links across all ICES IEA working groups and to HAPISG EGs on hu- man pressures on marine ecosystems, such as pollution.
Linkages to ACOM and groups under ACOM	Link to ACOM through the development of Ecosystem Overviews and advice.
Financial	No financial implications for ICES.
Secretariat facilities	Support for meetings at ICES HQ, when appropriate.
Participants	Experts from ICES, PICES, and PAME
Resource requirements	No major resourcing.
	ToR e – This will add to the suit of Ecosystem Overviews prepared and published by ICES.
	ToR d – This is an item meant to provide guidance to the research community at large on priority research issues to improve the knowledge base for continued IEA work.
	ToR e – It is expected that new information will be forthcoming on occurrence of fish and other biota in the CAO from planned research activies. There is for instance increased awareness that scientific echosounders on research ice-breakers can provide valuable information. We will report on developments and include new information in the next IEA report.
	ToR d – This item addresses pollution with focus on contaminant pathways (physi- cal and biological) and potential effects in foodwebs of the CAO. The scale of activ- ity will depend on the expertice available in the WG.